Licenses of third-party software used by AC2 (Windows)

Overview	4
CET	6
docopt	10
EISPACK	11
FDE	12
finterp license	15
h5py	16
HDF5, 2016	20
HDF5, 2006	21
IDL, EULA	23
IDL, legal notices	28
Intel® Fortran Redist	81
Intel® C++ Redist list	100
Intel® Fortran Redist list	109
Intel® MPI Library	111
Intel® MPI Library 3rd-party	113
Intel® oneMKL	125
Intel® oneMKL 3rd-party	127

Intel® oneMKL 3rd-party Benchmarks14	41
Intel® oneMKL 3rd-party IPP	43
Intel® oneMKL 3rd-party OpenMP14	16
Intel® oneMKL 3rd-party Safe String	55
JSON-Fortran	56
libaec	58
libsz	59
METIS	60
Apache License Version 2.0	31
Microsoft® Visual C++ 2019 Runtime	3 5
MINPACK	71
MMA	72
MUMPS	73
CeCILL-C 1.0	74
NumPy	33
OpenBLAS	34
pathlib2	35
PETSc	36
pyqtgraph	37

pyside6 1	88
Python1	91
Qt2	06
Licensed code in Qt	09
ruamel.yaml	12
virids	13
zlib	15
zlib	17

Overview

Software	Version	License Type	Link
CET		CC-BY	Link
docopt	0.6.2	MIT	Link
EISPACK		BSD-style License	Link
FDE	2.8.2	LGPL v3 [1]	Link
finterp	1.2.3	Permissive free software license	Link
h5py	3.7.0	BSD 3-Clause	Link
HDF5	1.10.6	HDF5, CNLT, 2016 [2]	Link
IDL	8.8.1	IDL® EULA [3]	Link
Intel® Fortran Redist	2023.1.0	Intel End User License Agreement [4]	Link
Intel® MPI Library	2021.9.0	Intel Simplified Software License [5]	Link
Intel® oneMKL	2023.1.0	Intel Simplified Software License [6]	Link
JSON-Fortran	7.1.0	BSD 3-clause License	Link
libaec	1.1.3	BSD-2-Clause	Link
libsz	1.1.3	BSD-2-Clause	Link
METIS	5.1.0	Apache License Version 2.0	Link
Microsoft® Visual C++ 2019 Runtime	14.29.30133	MICROSOFT SOFTWARE LICENSE TERMS [7]	Link
MINPACK		BSD-style License	Link
MMA	1.0	BSD 2-Clause	Link
MUMPS	5.5.1	CeCILL-C 1.0	Link
NumPy	1.26.4	BSD 3-Clause	Link
OpenBLAS	v0.3.23	BSD 3-Clause	Link
pathlib2	2.3.6	MIT	Link
PETSc	3.19.4	BSD 2-Clause	Link
pyqtgraph	0.13.7	MIT	Link
PySide6_Essentials	6.7.0	LGPL-3.0-only	Link
Python	3.11.8	PSF 2 [8]	Link
Qt	5.12.11	GNU LESSER GENERAL PUBLIC LICENSE 3.0	Link
ruamel.yaml	0.17.21	MIT	Link
virids		CC0	[Link]
zlib	v1.2.11	free use	Link
zlib	1.2.13	MIT	Link

- [1] GNU LESSER GENERAL PUBLIC LICENSE Version 3
- [2] Copyright Notice and License Terms for HDF5 (Hierarchical Data Format 5) Software Library and Utilities
- [3] HARRIS GEOSPATIAL SOLUTIONS, INC. END USER LICENSE AGREEMENT FOR IDL® 8.8.1 AND ENVI® 5.6.1 AND L3HARRIS GEOSPATIAL LICENSE SERVER 3.0 AND ENVI® PHOTOGRAMMETRY MODULE 5.6.1 AND IDL® DICOM TOOLKIT 8.8.1

- [4] Intel End User License Agreement for Developer Tools (Version October 2021)
- [5] Intel Simplified Software License (Version October 2022)
- [6] Intel Simplified Software License (Version October 2022)
- [7] MICROSOFT SOFTWARE LICENSE TERMS MICROSOFT VISUAL STUDIO ENTERPRISE 2019, VISUAL STUDIO PROFESSIONAL 2019, VISUAL STUDIO TEST PROFESSIONAL 2019 AND TRIAL EDITION
- [8] Python Software Foundation License Version 2
- In compliance with CeCILL-C 1.0 access to the source code of the software MUMPS is provided on request.

Contact

For enquiries please send an e-mail to AC2@grs.de.

Creative Commons Attribution 4.0 International Public License (CC-BY)

By exercising the Licensed Rights (defined below), You accept and agree to be bound by the terms and conditions of this Creative Commons Attribution 4.0 International Public License ("Public License"). To the extent this Public License may be interpreted as a contract, You are granted the Licensed Rights in consideration of Your acceptance of these terms and conditions, and the Licensor grants You such rights in consideration of benefits the Licensor receives from making the Licensed Material available under these terms and conditions.

Section 1 - Definitions.

- a. Adapted Material means material subject to Copyright and Similar Rights that is derived from or based upon the Licensed Material and in which the Licensed Material is translated, altered, arranged, transformed, or otherwise modified in a manner requiring permission under the Copyright and Similar Rights held by the Licensor. For purposes of this Public License, where the Licensed Material is a musical work, performance, or sound recording, Adapted Material is always produced where the Licensed Material is synched in timed relation with a moving image.
- b. Adapter's License means the license You apply to Your Copyright and Similar Rights in Your contributions to Adapted Material in accordance with the terms and conditions of this Public License.
- c. Copyright and Similar Rights means copyright and/or similar rights closely related to copyright including, without limitation, performance, broadcast, sound recording, and Sui Generis Database Rights, without regard to how the rights are labeled or categorized. For purposes of this Public License, the rights specified in Section [5]2(b)(1)-(2) are not Copyright and Similar Rights.
- d. Effective Technological Measures means those measures that, in the absence of proper authority, may not be circumvented under laws fulfilling obligations under Article 11 of the WIPO Copyright Treaty adopted on December 20, 1996, and/or similar international agreements.
- e. Exceptions and Limitations means fair use, fair dealing, and/or any other exception or limitation to Copyright and Similar Rights that applies to Your use of the Licensed Material.
- f. Licensed Material means the artistic or literary work, database, or other material to which the Licensor applied this Public License.
- g. Licensed Rights means the rights granted to You subject to the terms and conditions of this Public License, which are limited to all Copyright and Similar Rights that apply to Your use of the Licensed Material and that the Licensor has authority to license.
- h. Licensor means the individual(s) or entity(ies) granting rights under this Public License.
- i. Share means to provide material to the public by any means or process that requires permission under the Licensed Rights, such as reproduction, public display, public performance, distribution, dissemination, communication, or importation, and to make material available to the public including in ways that members of the public may access the material from a place and at a time individually chosen by them.
- j. Sui Generis Database Rights means rights other than copyright resulting from Directive 96/9/EC of the European Parliament and of the Council of 11 March 1996 on the legal protection of databases, as amended and/or succeeded, as well as other essentially equivalent rights anywhere in the world.
- k. You means the individual or entity exercising the Licensed Rights under this Public License. Your has a corresponding meaning.

Section 2 - Scope.

a. License grant.

- 1. Subject to the terms and conditions of this Public License, the Licensor hereby grants You a worldwide, royalty-free, non-sublicensable, non-exclusive, irrevocable license to exercise the Licensed Rights in the Licensed Material to:
 - A. reproduce and Share the Licensed Material, in whole or in part; and
 - B. produce, reproduce, and Share Adapted Material.
- 2. Exceptions and Limitations. For the avoidance of doubt, where Exceptions and Limitations apply to Your use, this Public License does not apply, and You do not need to comply with its terms and conditions.
- 3. Term. The term of this Public License is specified in Section [6]6(a).
- 4. Media and formats; technical modifications allowed. The Licensor authorizes You to exercise the Licensed Rights in all media and formats whether now known or hereafter created, and to make technical modifications necessary to do so. The Licensor waives and/or agrees not to assert any right or authority to forbid You from making technical modifications necessary to exercise the Licensed Rights, including technical modifications necessary to circumvent Effective Technological Measures. For purposes of this Public License, simply making modifications authorized by this Section [7]2(a)(4) never produces Adapted Material.
- 5. Downstream recipients.
 - A. Offer from the Licensor Licensed Material. Every recipient of the Licensed Material automatically receives an offer from the Licensor to exercise the Licensed Rights under the terms and conditions of this Public License.
 - B. No downstream restrictions. You may not offer or impose any additional or different terms or conditions on, or apply any Effective Technological Measures to, the Licensed Material if doing so restricts exercise of the Licensed Rights by any recipient of the Licensed Material.
- 6. No endorsement. Nothing in this Public License constitutes or may be construed as permission to assert or imply that You are, or that Your use of the Licensed Material is, connected with, or sponsored, endorsed, or granted official status by, the Licensor or others designated to receive attribution as provided in Section [8]3(a)(1)(A)(i).
- b. Other rights.
 - 1. Moral rights, such as the right of integrity, are not licensed under this Public License, nor are publicity, privacy, and/or other similar personality rights; however, to the extent possible, the Licensor waives and/or agrees not to assert any such rights held by the Licensor to the limited extent necessary to allow You to exercise the Licensed Rights, but not otherwise.
 - 2. Patent and trademark rights are not licensed under this Public License.
 - 3. To the extent possible, the Licensor waives any right to collect royalties from You for the exercise of the Licensed Rights, whether directly or through a collecting society under any voluntary or waivable statutory or compulsory licensing scheme. In all other cases the Licensor expressly reserves any right to collect such royalties.

Section 3 - License Conditions.

Your exercise of the Licensed Rights is expressly made subject to the following conditions.

a. Attribution.

- 1. If You Share the Licensed Material (including in modified form), You must:
 - A. retain the following if it is supplied by the Licensor with the Licensed Material:
 - i. identification of the creator(s) of the Licensed Material and any others designated to receive attribution, in any reasonable manner requested by the Licensor (including by pseudonym if designated);
 - ii. a copyright notice;
 - iii. a notice that refers to this Public License;
 - iv. a notice that refers to the disclaimer of
 warranties;
 - v. a URI or hyperlink to the Licensed Material to the extent reasonably practicable;
 - B. indicate if You modified the Licensed Material and retain an indication of any previous modifications; and
 - C. indicate the Licensed Material is licensed under this Public License, and include the text of, or the URI or hyperlink to, this Public License.
- 2. You may satisfy the conditions in Section [9]3(a)(1) in any reasonable manner based on the medium, means, and context in which You Share the Licensed Material. For example, it may be reasonable to satisfy the conditions by providing a URI or hyperlink to a resource that includes the required information.
- 3. If requested by the Licensor, You must remove any of the information required by Section [10]3(a)(1)(A) to the extent reasonably practicable.
- 4. If You Share Adapted Material You produce, the Adapter's License You apply must not prevent recipients of the Adapted Material from complying with this Public License.

Section 4 - Sui Generis Database Rights.

Where the Licensed Rights include Sui Generis Database Rights that apply to Your use of the Licensed Material:

- a. for the avoidance of doubt, Section [11]2(a)(1) grants You the right to extract, reuse, reproduce, and Share all or a substantial portion of the contents of the database;
- b. if You include all or a substantial portion of the database contents in a database in which You have Sui Generis Database Rights, then the database in which You have Sui Generis Database Rights (but not its individual contents) is Adapted Material; and
- c. You must comply with the conditions in Section [12]3(a) if You Share all or a substantial portion of the contents of the database.

For the avoidance of doubt, this Section [13]4 supplements and does not replace Your obligations under this Public License where the Licensed Rights include other Copyright and Similar Rights.

Section 5 - Disclaimer of Warranties and Limitation of Liability.

- a. Unless otherwise separately undertaken by the Licensor, to the extent possible, the Licensor offers the Licensed Material as-is and as-available, and makes no representations or warranties of any kind concerning the Licensed Material, whether express, implied, statutory, or other. This includes, without limitation, warranties of title, merchantability, fitness for a particular purpose, non-infringement, absence of latent or other defects, accuracy, or the presence or absence of errors, whether or not known or discoverable. Where disclaimers of warranties are not allowed in full or in part, this disclaimer may not apply to You.
- b. To the extent possible, in no event will the Licensor be liable to You on any legal theory (including, without limitation, negligence) or otherwise for any direct, special, indirect, incidental,

consequential, punitive, exemplary, or other losses, costs, expenses, or damages arising out of this Public License or use of the Licensed Material, even if the Licensor has been advised of the possibility of such losses, costs, expenses, or damages. Where a limitation of liability is not allowed in full or in part, this limitation may not apply to You.

c. The disclaimer of warranties and limitation of liability provided above shall be interpreted in a manner that, to the extent possible, most closely approximates an absolute disclaimer and waiver of all liability.

Section 6 - Term and Termination.

- a. This Public License applies for the term of the Copyright and Similar Rights licensed here. However, if You fail to comply with this Public License, then Your rights under this Public License terminate automatically.
- b. Where Your right to use the Licensed Material has terminated under Section [14]6(a), it reinstates:
 - automatically as of the date the violation is cured, provided it is cured within 30 days of Your discovery of the violation; or
 - 2. upon express reinstatement by the Licensor. For the avoidance of doubt, this Section [15]6(b) does not affect any right the Licensor may have to seek remedies for Your violations of this Public License.
- c. For the avoidance of doubt, the Licensor may also offer the Licensed Material under separate terms or conditions or stop distributing the Licensed Material at any time; however, doing so will not terminate this Public License.
- d. Sections [16]1, [17]5, [18]6, [19]7, and [20]8 survive termination of this Public License.

Section 7 - Other Terms and Conditions.

- a. The Licensor shall not be bound by any additional or different terms or conditions communicated by You unless expressly agreed.
- b. Any arrangements, understandings, or agreements regarding the Licensed Material not stated herein are separate from and independent of the terms and conditions of this Public License.

Section 8 - Interpretation.

- a. For the avoidance of doubt, this Public License does not, and shall not be interpreted to, reduce, limit, restrict, or impose conditions on any use of the Licensed Material that could lawfully be made without permission under this Public License.
- b. To the extent possible, if any provision of this Public License is deemed unenforceable, it shall be automatically reformed to the minimum extent necessary to make it enforceable. If the provision cannot be reformed, it shall be severed from this Public License without affecting the enforceability of the remaining terms and conditions.
- c. No term or condition of this Public License will be waived and no failure to comply consented to unless expressly agreed to by the Licensor.
- d. Nothing in this Public License constitutes or may be interpreted as a limitation upon, or waiver of, any privileges and immunities that apply to the Licensor or You, including from the legal processes of any jurisdiction or authority.

Copyright (c) 2012 Vladimir Keleshev, <vladimir@keleshev.com>

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Copyright (c) 1992-2013 The University of Tennessee and The University of Tennessee Research Foundation. All rights reserved.

Copyright (c) 2000-2013 The University of California Berkeley. All rights reserved.

Copyright (c) 2006-2013 The University of Colorado Denver. All rights reserved. Additional copyrights may follow

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.

Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer listed in this license in the documentation and/or other materials provided with the distribution.

Neither the name of the copyright holders nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

The copyright holders provide no reassurances that the source code provided does not infringe any patent, copyright, or any other intellectual property rights of third parties. The copyright holders disclaim any liability to any recipient for claims brought against recipient by any third party for infringement of that parties intellectual property rights.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

GNU LESSER GENERAL PUBLIC LICENSE Version 3, 29 June 2007

Copyright (C) 2007 Free Software Foundation, Inc. http://fsf.org/ Everyone is permitted to copy and distribute verbatim copies of this license document, but changing it is not allowed.

This version of the GNU Lesser General Public License incorporates the terms and conditions of version 3 of the GNU General Public License, supplemented by the additional permissions listed below.

O. Additional Definitions.

As used herein, "this License" refers to version 3 of the GNU Lesser General Public License, and the "GNU GPL" refers to version 3 of the GNU General Public License.

"The Library" refers to a covered work governed by this License, other than an Application or a Combined Work as defined below.

An "Application" is any work that makes use of an interface provided by the Library, but which is not otherwise based on the Library. Defining a subclass of a class defined by the Library is deemed a mode of using an interface provided by the Library.

A "Combined Work" is a work produced by combining or linking an Application with the Library. The particular version of the Library with which the Combined Work was made is also called the "Linked Version".

The "Minimal Corresponding Source" for a Combined Work means the Corresponding Source for the Combined Work, excluding any source code for portions of the Combined Work that, considered in isolation, are based on the Application, and not on the Linked Version.

The "Corresponding Application Code" for a Combined Work means the object code and/or source code for the Application, including any data and utility programs needed for reproducing the Combined Work from the Application, but excluding the System Libraries of the Combined Work.

1. Exception to Section 3 of the GNU GPL.

You may convey a covered work under sections 3 and 4 of this License without being bound by section 3 of the GNU GPL.

2. Conveying Modified Versions.

If you modify a copy of the Library, and, in your modifications, a facility refers to a function or data to be supplied by an Application that uses the facility (other than as an argument passed when the facility is invoked), then you may convey a copy of the modified version:

- a) under this License, provided that you make a good faith effort to ensure that, in the event an Application does not supply the function or data, the facility still operates, and performs whatever part of its purpose remains meaningful, or
- b) under the GNU GPL, with none of the additional permissions of this License applicable to that copy.
- 3. Object Code Incorporating Material from Library Header Files.

The object code form of an Application may incorporate material from a header file that is part of the Library. You may convey such object code under terms of your choice, provided that, if the incorporated material is not limited to numerical parameters, data structure layouts and accessors, or small macros, inline functions and templates (ten or fewer lines in length), you do both of the following:

- a) Give prominent notice with each copy of the object code that the Library is used in it and that the Library and its use are covered by this License.
- b) Accompany the object code with a copy of the GNU GPL and this license document.

4. Combined Works.

You may convey a Combined Work under terms of your choice that, taken together, effectively do not restrict modification of the portions of the Library contained in the Combined Work and reverse engineering for debugging such modifications, if you also do each of the following:

- a) Give prominent notice with each copy of the Combined Work that the Library is used in it and that the Library and its use are covered by this License.
- b) Accompany the Combined Work with a copy of the GNU GPL and this license document.
- c) For a Combined Work that displays copyright notices during execution, include the copyright notice for the Library among these notices, as well as a reference directing the user to the copies of the GNU GPL and this license document.
- d) Do one of the following:
 - 0) Convey the Minimal Corresponding Source under the terms of this License, and the Corresponding Application Code in a form suitable for, and under terms that permit, the user to recombine or relink the Application with a modified version of the Linked Version to produce a modified Combined Work, in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.
 - 1) Use a suitable shared library mechanism for linking with the Library. A suitable mechanism is one that (a) uses at run time a copy of the Library already present on the user's computer system, and (b) will operate properly with a modified version of the Library that is interface-compatible with the Linked Version.
- e) Provide Installation Information, but only if you would otherwise be required to provide such information under section 6 of the GNU GPL, and only to the extent that such information is necessary to install and execute a modified version of the Combined Work produced by recombining or relinking the Application with a modified version of the Linked Version. (If you use option 4d0, the Installation Information must accompany the Minimal Corresponding Source and Corresponding Application Code. If you use option 4d1, you must provide the Installation Information in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.)
- 5. Combined Libraries.

You may place library facilities that are a work based on the Library side by side in a single library together with other library facilities that are not Applications and are not covered by this License, and convey such a combined library under terms of your choice, if you do both of the following:

- a) Accompany the combined library with a copy of the same work based on the Library, uncombined with any other library facilities, conveyed under the terms of this License.
- b) Give prominent notice with the combined library that part of it is a work based on the Library, and explaining where to find the accompanying uncombined form of the same work.
- 6. Revised Versions of the GNU Lesser General Public License.

The Free Software Foundation may publish revised and/or new versions of the GNU Lesser General Public License from time to time. Such new versions will be similar in spirit to the present version, but may differ in detail to address new problems or concerns.

Each version is given a distinguishing version number. If the Library as you received it specifies that a certain numbered version of the GNU Lesser General Public License "or any later version" applies to it, you have the option of following the terms and conditions either of that published version or of any later version published by the Free Software Foundation. If the Library as you received it does not specify a version number of the GNU Lesser General Public License, you may choose any version of the GNU Lesser General Public License ever published by the Free Software Foundation.

If the Library as you received it specifies that a proxy can decide whether future versions of the GNU Lesser General Public License shall apply, that proxy's public statement of acceptance of any version is permanent authorization for you to choose that version for the Library.

finterp: Modern Fortran Multidimensional Linear Interpolation https://github.com/jacobwilliams/finterp

Copyright (c) 2016-2021, Jacob Williams All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * The names of its contributors may not be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Licenses and legal info

Copyright Notice and Statement for the h5py Project

::

Copyright (c) 2008 Andrew Collette and contributors All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. Neither the name of the copyright holder nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

HDF5 Copyright Statement

::

HDF5 (Hierarchical Data Format 5) Software Library and Utilities Copyright 2006-2007 by The HDF Group (THG).

NCSA HDF5 (Hierarchical Data Format 5) Software Library and Utilities Copyright 1998-2006 by the Board of Trustees of the University of Illinois.

All rights reserved.

Contributors: National Center for Supercomputing Applications (NCSA) at the University of Illinois, Fortner Software, Unidata Program Center (netCDF), The Independent JPEG Group (JPEG), Jean-loup Gailly and Mark Adler (gzip), and Digital Equipment Corporation (DEC).

Redistribution and use in source and binary forms, with or without modification, are permitted for any purpose (including commercial purposes) provided that the following conditions are met:

1. Redistributions of source code must retain the above copyright notice, this list of conditions, and the following disclaimer.

- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions, and the following disclaimer in the documentation and/or materials provided with the distribution.
- 3. In addition, redistributions of modified forms of the source or binary code must carry prominent notices stating that the original code was changed and the date of the change.
- 4. All publications or advertising materials mentioning features or use of this software are asked, but not required, to acknowledge that it was developed by The HDF Group and by the National Center for Supercomputing Applications at the University of Illinois at Urbana-Champaign and credit the contributors.
- 5. Neither the name of The HDF Group, the name of the University, nor the name of any Contributor may be used to endorse or promote products derived from this software without specific prior written permission from THG, the University, or the Contributor, respectively.

DISCLAIMER: THIS SOFTWARE IS PROVIDED BY THE HDF GROUP (THG) AND THE CONTRIBUTORS "AS IS" WITH NO WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED. In no event shall THG or the Contributors be liable for any damages suffered by the users arising out of the use of this software, even if advised of the possibility of such damage.

Portions of HDF5 were developed with support from the University of California, Lawrence Livermore National Laboratory (UC LLNL). The following statement applies to those portions of the product and must be retained in any redistribution of source code, binaries, documentation, and/or accompanying materials:

This work was partially produced at the University of California, Lawrence Livermore National Laboratory (UC LLNL) under contract no. W-7405-ENG-48 (Contract 48) between the U.S. Department of Energy (DOE) and The Regents of the University of California (University) for the operation of UC LLNL.

DISCLAIMER: This work was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor the University of California nor any of their employees, makes any warranty, express or implied, or assumes any liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately-owned rights. Reference herein to any specific commercial products, process, or service by trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or the University of California, and shall not be used for advertising or product endorsement purposes.

PyTables Copyright Statement

::

Copyright Notice and Statement for PyTables Software Library and Utilities:

Copyright (c) 2002, 2003, 2004 Francesc Altet Copyright (c) 2005, 2006, 2007 Carabos Coop. V. All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are

met:

- a. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- b. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- c. Neither the name of the Carabos Coop. V. nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

S	t	d	i	n	t		h		(W	i	n	d	o	W	S		V	е	r	s	i	0	n)		L	i	С	e	n	s	e
-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

::

Copyright (c) 2006-2008 Alexander Chemeris

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. The name of the author may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE AUTHOR ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE AUTHOR BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Python license

#. This LICENSE AGREEMENT is between the Python Software Foundation ("PSF"), and the Individual or Organization ("Licensee") accessing and otherwise using Python Python 2.7.5 software in source or binary form and its associated documentation.

#. Subject to the terms and conditions of this License Agreement, PSF hereby grants Licensee a nonexclusive, royalty-free, world-wide license to reproduce,

analyze, test, perform and/or display publicly, prepare derivative works, distribute, and otherwise use Python Python 2.7.5 alone or in any derivative version, provided, however, that PSF's License Agreement and PSF's notice of copyright, i.e., "Copyright 2001-2013 Python Software Foundation; All Rights Reserved" are retained in Python Python 2.7.5 alone or in any derivative version

prepared by Licensee.

#. In the event Licensee prepares a derivative work that is based on or incorporates Python Python 2.7.5 or any part thereof, and wants to make the derivative work available to others as provided herein, then Licensee hereby agrees to include in any such work a brief summary of the changes made to Python

Python 2.7.5.

- #. PSF is making Python Python 2.7.5 available to Licensee on an "AS IS" basis. PSF MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED. BY WAY OF EXAMPLE, BUT NOT LIMITATION, PSF MAKES NO AND DISCLAIMS ANY REPRESENTATION OR WARRANTY OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE OR THAT THE USE OF PYTHON Python 2.7.5 WILL NOT INFRINGE ANY THIRD PARTY RIGHTS.
- #. PSF SHALL NOT BE LIABLE TO LICENSEE OR ANY OTHER USERS OF PYTHON Python 2.7.5 FOR ANY INCIDENTAL, SPECIAL, OR CONSEQUENTIAL DAMAGES OR LOSS AS A RESULT OF MODIFYING, DISTRIBUTING, OR OTHERWISE USING PYTHON Python 2.7.5, OR ANY DERIVATIVE

THEREOF, EVEN IF ADVISED OF THE POSSIBILITY THEREOF.

- #. This License Agreement will automatically terminate upon a material breach of its terms and conditions.
- #. Nothing in this License Agreement shall be deemed to create any relationship of agency, partnership, or joint venture between PSF and Licensee. This License

Agreement does not grant permission to use PSF trademarks or trade name in a trademark sense to endorse or promote products or services of Licensee, or any

third party.

#. By copying, installing or otherwise using Python Python 2.7.5, Licensee agrees

to be bound by the terms and conditions of this License Agreement.

Copyright Notice and License Terms for HDF5 (Hierarchical Data Format 5) Software Library and Utilities

HDF5 (Hierarchical Data Format 5) Copyright (c) 2016, The Regents of the University of California, through Lawrence Berkeley National Laboratory (subject to receipt of any required approvals from the U.S. Dept. of Energy).

All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions, and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions, and the following disclaimer in the documentation and/or materials provided with the distribution.
- 3. Neither the name of the University of California, Lawrence Berkeley National Laboratory, U.S. Dept. of Energy nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

You are under no obligation whatsoever to provide any bug fixes, patches, or upgrades to the features, functionality or performance of the source code ("Enhancements") to anyone; however, if you choose to make your Enhancements available either publicly, or directly to Lawrence Berkeley National Laboratory, without imposing a separate written license agreement for such Enhancements, then you hereby grant the following license: a non-exclusive, royalty-free perpetual license to install, use, modify, prepare derivative works, incorporate into other computer software, distribute, and sublicense such enhancements or derivative works thereof, in binary and source code form.

Copyri	ght Notice a	and License	Terms fo	r			
HDF5 (Hierarchica ⁻	l Data Form	at 5) Sof	tware Li	brary and	Utilities	

HDF5 (Hierarchical Data Format 5) Software Library and Utilities Copyright 2006 by The HDF Group.

NCSA HDF5 (Hierarchical Data Format 5) Software Library and Utilities Copyright 1998-2006 by The Board of Trustees of the University of Illinois.

All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted for any purpose (including commercial purposes) provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions, and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions, and the following disclaimer in the documentation and/or materials provided with the distribution.
- 3. Neither the name of The HDF Group, the name of the University, nor the name of any Contributor may be used to endorse or promote products derived from this software without specific prior written permission from The HDF Group, the University, or the Contributor, respectively.

DISCLAIMER:

THIS SOFTWARE IS PROVIDED BY THE HDF GROUP AND THE CONTRIBUTORS "AS IS" WITH NO WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED. IN NO EVENT SHALL THE HDF GROUP OR THE CONTRIBUTORS BE LIABLE FOR ANY DAMAGES SUFFERED BY THE USERS ARISING OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

You are under no obligation whatsoever to provide any bug fixes, patches, or upgrades to the features, functionality or performance of the source code ("Enhancements") to anyone; however, if you choose to make your Enhancements available either publicly, or directly to The HDF Group, without imposing a separate written license agreement for such Enhancements, then you hereby grant the following license: a non-exclusive, royalty-free perpetual license to install, use, modify, prepare derivative works, incorporate into other computer software, distribute, and sublicense such enhancements or derivative works thereof, in binary and source code form.

works thereof, in binary and source code form.

Limited portions of HDF5 were developed by Lawrence Berkeley National Laboratory (LBNL). LBNL's Copyright Notice and Licensing Terms can be found here: COPYING_LBNL_HDF5 file in this directory or at http://support.hdfgroup.org/ftp/HDF5/releases/COPYING_LBNL_HDF5.

Contributors: National Center for Supercomputing Applications (NCSA) at the University of Illinois, Fortner Software, Unidata Program Center (netCDF), The Independent JPEG Group (JPEG), Jean-loup Gailly and Mark Adler (gzip), and Digital Equipment Corporation (DEC).

Portions of HDF5 were developed with support from the Lawrence Berkeley

National Laboratory (LBNL) and the United States Department of Energy under Prime Contract No. DE-AC02-05CH11231.

Portions of HDF5 were developed with support from the University of California, Lawrence Livermore National Laboratory (UC LLNL). The following statement applies to those portions of the product and must be retained in any redistribution of source code, binaries, documentation, and/or accompanying materials:

This work was partially produced at the University of California, Lawrence Livermore National Laboratory (UC LLNL) under contract no. W-7405-ENG-48 (Contract 48) between the U.S. Department of Energy (DOE) and The Regents of the University of California (University) for the operation of UC LLNL.

DISCLAIMER:

THIS WORK WAS PREPARED AS AN ACCOUNT OF WORK SPONSORED BY AN AGENCY OF THE UNITED STATES GOVERNMENT. NEITHER THE UNITED STATES GOVERNMENT NOR THE UNIVERSITY OF CALIFORNIA NOR ANY OF THEIR EMPLOYEES, MAKES ANY WARRANTY, EXPRESS OR IMPLIED, OR ASSUMES ANY LIABILITY OR RESPONSIBILITY FOR THE ACCURACY, COMPLETENESS, OR USEFULNESS OF ANY INFORMATION, APPARATUS, PRODUCT, OR PROCESS DISCLOSED, OR REPRESENTS THAT ITS USE WOULD NOT INFRINGE PRIVATELY- OWNED RIGHTS. REFERENCE HEREIN TO ANY SPECIFIC COMMERCIAL PRODUCTS, PROCESS, OR SERVICE BY TRADE NAME, TRADEMARK, MANUFACTURER, OR OTHERWISE, DOES NOT NECESSARILY CONSTITUTE OR IMPLY ITS ENDORSEMENT, RECOMMENDATION, OR FAVORING BY THE UNITED STATES GOVERNMENT OR THE UNIVERSITY OF CALIFORNIA. THE VIEWS AND OPINIONS OF AUTHORS EXPRESSED HEREIN DO NOT NECESSARILY STATE OR REFLECT THOSE OF THE UNITED STATES GOVERNMENT OR THE UNIVERSITY OF CALIFORNIA, AND SHALL NOT BE USED FOR ADVERTISING OR PRODUCT ENDORSEMENT PURPOSES.

HDF5 is available with the SZIP compression library but SZIP is not part of HDF5 and has separate copyright and license terms. See SZIP Compression in HDF Products (www.hdfgroup.org/doc_resource/SZIP/) for further details.

HARRIS GEOSPATIAL SOLUTIONS, INC. END USER LICENSE AGREEMENT FOR IDL® 8.8.1 AND ENVI® 5.6.1 AND L3HARRIS GEOSPATIAL LICENSE SERVER 3.0 AND ENVI® PHOTOGRAMMETRY MODULE 5.6.1 AND IDL® DICOM TOOLKIT 8.8.1

IMPORTANT: PLEASE READ THIS END USER LICENSE AGREEMENT CAREFULLY. DOWNLOADING, INSTALLING OR USING THE SOFTWARE CONSTITUTES ACCEPTANCE OF THIS AGREEMENT.

HARRIS GEOSPATIAL SOLUTIONS, INC. ("HGS"), OR ITS AUTHORIZED LICENSOR, IS WILLING TO LICENSE THE SOFTWARE TO YOU OR THE BUSINESS ENTITY YOU REPRESENT ("LICENSEE") UPON THE CONDITION THAT LICENSEE ACCEPTS ALL OF THE TERMS CONTAINED IN THIS END USER LICENSE AGREEMENT ("AGREEMENT"). BY DOWNLOADING, INSTALLING, OR USING THE SOFTWARE, YOU ARE BINDING YOURSELF AND THE BUSINESS ENTITY THAT YOU REPRESENT TO THE TERMS OF THIS AGREEMENT.

IF LICENSEE DOES NOT AGREE TO ALL OF THE TERMS OF THE AGREEMENT, THEN HGS IS UNWILLING TO LICENSE THE SOFTWARE TO LICENSEE AND (A) LICENSEE MAY NOT DOWNLOAD, INSTALL OR USE THE SOFTWARE, AND (B) LICENSEE MAY RETURN THE SOFTWARE (INCLUDING ANY UNOPENED CD/DVD PACKAGE AND ANY WRITTEN MATERIALS) TO THE PLACE OF PURCHASE FOR A FULL REFUND, OR, IF THE SOFTWARE AND WRITTEN MATERIALS ARE SUPPLIED AS PART OF ANOTHER PRODUCT, LICENSEE MAY RETURN THE ENTIRE PRODUCT TO THE PLACE OF PURCHASE FOR A FULL REFUND. LICENSEE'S RIGHT TO RETURN AND REFUND EXPIRES 60 DAYS AFTER PURCHASE FROM LICENSOR, AND APPLIES ONLY IF LICENSEE IS THE ORIGINAL END USER PURCHASER.

1. DEFINITIONS.

- a. "Authorization Code" means any key, activation code or other mechanism required for use of the Software.
- b. "Documentation" means written information (whether contained in user or technical manuals, training manuals, specifications or otherwise) pertaining to the Software and made available by HGS in any manner (including CD-ROM or on-line).
- c. "Ordering Documents" means a sales quotation, purchase order, invoice, or other document identifying the Software that Licensee procures.
- d. "Software" means the computer programs provided to Licensee by an approved source, and any upgrades, updates, bug fixes or modified versions thereto (collectively "Upgrades").
- **2. LICENSE.** Conditioned upon compliance with the terms and conditions of the Agreement, HGS grants to Licensee a nonexclusive license to use for Licensee's internal use and specific business needs ("Scope"), the Software and Documentation for which Licensee has paid the required license fees as set forth on the applicable Ordering Documents from HGS or an authorized HGS licensor. In order to use the Software, Licensee may be required to provide a maintenance ID number or installation ID number to obtain the necessary Authorization Code.
- **3. GENERAL LIMITATIONS.** Licensee may install and use the Software on a single computer, install and store the Software on a storage device, such as a network server, install the Software on Licensee's other computers over an internal network, or install on a third party internet-based network server for use within the Scope, provided Licensee has purchased the appropriate license and adheres to the terms and conditions herein. Other than as specifically set forth herein, you may not make or distribute copies of the Software, use the Software for commercial network services, time sharing or remote processing services, or provide unlicensed end users with access to the Software functionality. Licensee may make one copy of the Software in machine-readable form solely for backup purposes. Licensee must reproduce on any such copy all copyright notices and any other proprietary legends on the original copy of the Software.

This is a license, not a transfer of title to the Software and Documentation, and HGS and/or its licensors retain ownership of all copies of the Software and Documentation. Licensee acknowledges that the Software and Documentation contain trade secrets, proprietary, or confidential information of HGS or its suppliers or licensors, which includes but is not limited to the specific internal design and structure of individual programs and associated interface information. Except as otherwise expressly provided under the Agreement, Licensee shall have no right, and specifically agrees not to:

(i) rent, lease, loan, sell, transfer, assign or sublicense its license rights to any other person or entity (other than in compliance with any HGS relicensing/transfer policy then in force), and Licensee acknowledges that any attempted transfer, assignment, sublicense or use shall be

void;

- (ii) make error corrections to or otherwise modify, alter or adapt the Software or create derivative works based upon the Software, or permit third parties to do the same;
- (iii) translate, decipher, reverse engineer or decompile, decrypt, disassemble or otherwise reduce the Software to human-readable form, except to the extent otherwise expressly permitted under applicable law notwithstanding this restriction;
- (iv) use or permit the Software to be used to perform services for third parties, whether on a service bureau or time sharing basis or otherwise, without the express written authorization of HGS:
- (v) disclose, provide, or otherwise make available trade secrets, proprietary or confidential information contained within the Software and Documentation in any form to any third party without the prior written consent of HGS unless otherwise required by applicable law or court order. Licensee shall implement reasonable security measures to protect such trade secrets, proprietary or confidential information.
- (vi) use more Software licenses at any given time than the total quantity authorized by Licensees.
- **4. SOFTWARE, UPGRADES AND ADDITIONAL COPIES.** NOTWITHSTANDING ANY OTHER PROVISION OF THE AGREEMENT: (1) LICENSEE HAS NO LICENSE OR RIGHT TO MAKE OR USE ANY ADDITIONAL COPIES OR UPGRADES UNLESS LICENSEE, AT THE TIME OF MAKING OR ACQUIRING SUCH COPY OR UPGRADE, ALREADY HOLDS A VALID LICENSE TO THE ORIGINAL SOFTWARE AND HAS PAID THE APPLICABLE FEE FOR THE UPGRADE OR ADDITIONAL COPIES; AND (2) THE MAKING AND USE OF ADDITIONAL COPIES IS LIMITED TO NECESSARY BACKUP PURPOSES ONLY.
- **5. CONCURRENT PROCESS LICENSE.** If Licensee has purchased a concurrent process license, HGS grants Licensee the right to use the Software on a single computer or a computer network (a) based upon the specific configuration set forth in the Ordering Documents, and (b) not to exceed the allowable number of simultaneous processes supported by the number of licenses acquired.
- **6. EDUCATIONAL LAB, MINI-LAB, or DEPARTMENT LICENSE.** If Licensee has purchased an educational lab, mini-lab, or department license, the Software is for use at an accredited, degree-granting academic institution. The Software shall be installed on a single network on computer systems owned, operated and located on the campus of the academic institution. The Software shall be used for teaching and academic-related research purposes only. Licensee may have as many copies of the Software in simultaneous use on the network as authorized by the license fees paid and in accordance with the HGS Academic Use Policy. Specifically excluded is use by or for any government, hospital, non-academic research, or commercial use.
- **7. HASP LICENSE.** If Licensee has purchased a Hasp hardware key, Licensee may use the Software on any machine with the provided hardware key attached.
- **8. STUDENT LICENSE.** If Licensee has purchased a student license, HGS grants Licensee a license to use the Software on a single computer system owned by the Licensee. Licensee must be a student pursuing a degree at qualifying college or university. Professional, commercial, government, and other usage are strictly prohibited. The Software may not be installed or run on computer systems owned or operated by a College or University, or on other computer systems not owned by the student. The license shall be terminated after 12 months or when Licensee is no longer a student pursuing a degree at a qualifying college or university, whichever occurs first, at which time Licensee agrees to cease using the Software and agrees to destroy all copies of the Software.
- 9. IDL VIRTUAL MACHINE LICENSE. If Licensee has acquired an IDL Virtual Machine license, Licensee is granted a non-exclusive license to use the IDL Virtual Machine Software, to modify the IDL Virtual Machine Software, or combine it with other software. Licensee may distribute the IDL Virtual Machine Software with other software subject to the requirement that the derived software is subject to the terms of this Agreement. Licensee may not reverse engineer, decompile or disassemble the IDL Virtual Machine Software. Licensee may not alter or modify the installation procedure, suppress the End User License Agreement, alter the software operating environment in any way that prevents the IDL Virtual Machine splash screen from appearing at start-up, or automatically dismiss the IDL Virtual Machine splash screen. Licensee may not use the IDL Virtual Machine if it has been altered to automatically dismiss the IDL Virtual Machine splash screen from appearing at start-up. Licensee may not modify the IDL Virtual Machine in a way that exposes the IDL Development Environment (IDLDE), or in any manner that replicates or mimics the functionality of the IDL Development Environment or the IDL Command Line. Licensee may reproduce Documentation, but only for Licensee's use of the IDL Virtual Machine Software in accordance with the IDL Virtual Machine license terms described herein. All such copies of Software or Documentation must

contain all proprietary, copyright notices, and trademarks contained as part of the original Software or Documentation.

- 10. ENGINE LICENSE. If Licensee has purchased an Engine license, HGS grants Licensee the right to use the Software as a processing engine only based upon the configuration and usage type set forth in the Ordering Documents, and not to exceed the allowable number of simultaneous processes supported by the number of licenses acquired. An Engine license includes the ability to deploy on both a production and non-production system, however non-production licenses are strictly limited to testing, development and integration purposes. Engine licenses whose usage type is identified as 'external' in the Ordering Documents grand Licensee the right to provide Licensee's end users access to Software functionality through Licensee's production platform.
- 11. U.S. GOVERNMENT END USER PURCHASERS. The Software and Documentation qualify as "commercial items," as that term is defined at Federal Acquisition Regulation ("FAR") (48 C.F.R.) 2.101, consisting of "commercial computer software" and "commercial computer software documentation" as such terms are used in FAR 12.212. Consistent with FAR 12.212 and DoD FAR Supp. 227.7202-1 through 227.7202-4, and notwithstanding any other FAR or other contractual clause to the contrary in any contract into which this Agreement may be incorporated, Government end user will acquire the Software and Documentation with only those rights set forth in this Agreement. Use of either the Software or Documentation or both constitutes agreement by the Government that the Software and Documentation are "commercial computer software" and "commercial computer software documentation," and constitutes acceptance of the rights and restrictions herein.
- 12. LIMITED WARRANTY. Subject to the limitations and conditions set forth herein, HGS warrants that commencing from the date of shipment to Licensee and for a period no longer than sixty (60) days: (a) the media on which the Software is furnished will be free of defects in materials and workmanship under normal use; and (b) the Software shall perform in substantial conformity with the Documentation. Except for the foregoing, the Software is provided "AS IS." This limited warranty extends only to the Licensee who is the original licensee. Licensee's sole and exclusive remedy and the entire liability of HGS and its distributors or suppliers under this limited warranty will be (i) replacement of defective media; and/or (ii) at HGS' option, repair, replacement, or refund of the purchase price of the Software license. In both cases, the remedy is subject to the condition that any error or defect constituting a breach of this limited warranty is reported to HGS or the party supplying the Software to Licensee if different than HGS, within the 60-day warranty period. HGS or the party supplying the Software to Licensee may, at its option, require return of the Software and/or Documentation as a condition to the remedy. In no event does HGS warrant that the Software is error free or that Licensee will be able to operate the Software without problems or interruptions.
- **13. WARRANTY RESTRICTIONS.** This warranty does not apply if the Software, (a) has been altered, except by HGS; (b) has not been installed, operated, repaired, or maintained in accordance with instructions supplied by HGS; (c) has been subjected to abnormal physical or electrical stress, abnormal environmental conditions, misuse, negligence, or accident; (d) is licensed for beta, evaluation, testing or demonstration purposes; or (e) any Software for which HGS does not receive a license fee.
- 14. THIRD PARTY AND OPEN SOURCE SOFTWARE. PORTIONS OF THE SOFTWARE MAY INCLUDE COPYRIGHTED THIRD PARTY SOFTWARE PROVIDED UNDER LICENSE. Such third party software may include "free" or "open source" software licensed under separate terms from this EULA ("Open Source Software"). Such Open Source Software is distributed WITHOUT ANY WARRANTY, without even the implied warranty of MERCHANTABILITY or FITNESS FOR A PARTICULAR PURPOSE. HGS is not obligated to provide any warranty, maintenance, technical or other support for the Open Source Software or its use in the Software. Attributions and license terms for third party and Open Source Software are available in the source folder of the software installation, the "Help/Legal" directory located within the software, and at http://www.harrisgeospatial.com/portals/0/pdfs/legal_notices.pdf. Where applicable, source code for Open Source software is also provided in the source folder of the software installation and/or our web site.
- 15. DISCLAIMER OF WARRANTY. EXCEPT AS SPECIFIED IN THIS WARRANTY SECTION, ALL EXPRESS OR IMPLIED CONDITIONS, REPRESENTATIONS, AND WARRANTIES INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTY OR CONDITION OF TITLE, MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT, SATISFACTORY QUALITY, NON-INTERFERENCE, ACCURACY OF INFORMATIONAL CONTENT, OR ARISING FROM A COURSE OF DEALING, LAW, USAGE, OR TRADE PRACTICE, ARE HEREBY EXCLUDED TO THE EXTENT ALLOWED BY APPLICABLE LAW AND ARE EXPRESSLY DISCLAIMED BY HGS, ITS SUPPLIERS AND LICENSORS. TO THE EXTENT AN IMPLIED WARRANTY CANNOT BE EXCLUDED, SUCH WARRANTY IS LIMITED IN DURATION TO THE EXPRESS WARRANTY PERIOD. BECAUSE SOME STATES OR JURISDICTIONS DO NOT ALLOW LIMITATIONS ON HOW LONG AN IMPLIED WARRANTY LASTS, THE ABOVE LIMITATION MAY NOT APPLY. THIS WARRANTY GIVES LICENSEE SPECIFIC LEGAL RIGHTS, AND LICENSEE

MAY ALSO HAVE OTHER RIGHTS WHICH VARY FROM JURISDICTION TO JURISDICTION. THIS DISCLAIMER AND EXCLUSION SHALL APPLY EVEN IF THE EXPRESS WARRANTY SET FORTH ABOVE FAILS ITS ESSENTIAL PURPOSE.

- 16. IDL VIRTUAL MACHINE DISCLAIMER OF WARRANTY. THE IDL VIRTUAL MACHINE SOFTWARE IS PROVIDED ON AN "AS IS" BASIS. HGS MAY CHANGE OR ALTER IDL VIRTUAL MACHINE SOFTWARE AT ANY TIME, AND WITHOUT PRIOR NOTICE. HGS AND ITS LICENSORS SPECIFICALLY DISCLAIM ALL EXPRESS STATUTORY OR IMPLIED WARRANTIES RELATING TO IDL VIRTUAL MACHINE SOFTWARE, INCLUDING BUT NOT LIMITED TO MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR NON-INFRINGEMENT OF ANY THIRD PARTY RIGHTS.
- 17. DISCLAIMER OF LIABILITIES LIMITATION OF LIABILITY. NOTWITHSTANDING ANYTHING ELSE IN THE AGREEMENT TO THE CONTRARY, ALL LIABILITY OF HGS, ITS AFFILIATES, OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, SUPPLIERS AND LICENSORS COLLECTIVELY, TO LICENSEE, WHETHER IN CONTRACT, TORT (INCLUDING NEGLIGENCE), BREACH OF WARRANTY OR OTHERWISE, SHALL NOT EXCEED THE PRICE PAID BY LICENSEE TO HGS OR ITS AUTHORIZED RESELLER FOR THE SOFTWARE THAT GAVE RISE TO THE CLAIM. THIS LIMITATION OF LIABILITY FOR SOFTWARE IS CUMULATIVE AND NOT PER INCIDENT (I.E., THE EXISTENCE OF TWO OR MORE CLAIMS WILL NOT ENLARGE THIS LIMIT). IN NO EVENT WILL HGS, ITS AFFILIATES, OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, SUPPLIERS AND LICENSORS, BE LIABLE FOR ANY LOST REVENUE, LOST PROFIT, OR LOST OR DAMAGED DATA, BUSINESS INTERRUPTION, LOSS OF CAPITAL, COST TO COVER OR FOR SPECIAL, INDIRECT, CONSEQUENTIAL, INCIDENTAL, OR PUNITIVE DAMAGES, HOWSOEVER ARISING, INCLUDING, WITHOUT LIMITATION, IN CONTRACT, TORT (INCLUDING NEGLIGENCE) OR WHETHER ARISING OUT OF THE USE OF OR INABILITY TO USE THE SOFTWARE, EVEN IF, IN EACH CASE, HGS, ITS AFFILIATES, OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, SUPPLIERS AND LICENSORS, HAVE BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES. BECAUSE SOME STATES OR JURISDICTIONS DO NOT ALLOW LIMITATION OR EXCLUSION OF CONSEQUENTIAL OR INCIDENTAL DAMAGES, THE ABOVE LIMITATION MAY NOT FULLY APPLY TO LICENSEE.
- **18. INDEMNITY.** HGS shall defend or settle at its expense a claim or suit against Licensee arising out of or in connection with an assertion that the Software infringes any U.S. copyright or U.S. registered patent. HGS shall indemnify and hold Licensee harmless from and against the damages, costs and expenses (including, without limitation, reasonable legal and expert witness fees), if any, finally awarded in such suit or the amount of the settlement thereof, provided that HGS is notified in writing of the existence of such claim by Licensee within five (5) business days of Licensee's first learning of the same, and provided that HGS is given full authority to control the defense, cost and settlement of the claim. HGS will not be obligated to defend or otherwise indemnify Licensee in any lawsuit or as to any claim which arises from or relates to: (1) any combination of the Software with another product not supplied by HGS; (2) if such a claim is based upon use of the Software for purposes for which it was not designed; or (3) if the Software has been modified by any party other than HGS. In lieu of the foregoing indemnification obligations, HGS shall have the option, at its expense, either to procure for Licensee the right to continue using the Software or to replace or modify the Software so that it becomes non-infringing, or to refund to Licensee the amount actually paid by the Licensee for the Software.
- 19. EXPORT RESTRICTIONS. The Software is subject to export controls under the laws and regulations of the United States of America, including but not limited to the U.S. Export Administration Regulations and U.S. Department of the Treasury, Office of Foreign Assets Control, and may be subject to additional export and import regulations of the country in which the software is obtained. You agree that you will comply with all applicable laws and regulations governing the export, import, reexport, transfer and use, and that you have the responsibility to obtain any required authorization from the appropriate authority. If HGS receives notice that You are or You become identified as a sanctioned or restricted party under applicable law, then HGS will not be obligated to perform any of its obligations under this Agreement if such performance would result in violation of the sanctions or restrictions.
- **20. PROPRIETARY NOTICES.** Licensee agrees to maintain and reproduce all copyright and other proprietary notices on all copies, in any form, of the Software and Documentation in the same form and manner that such copyright and other proprietary notices are included on the Software and Documentation. Except as expressly authorized in the Agreement, Licensee shall not make any copies or duplicates of any Software without the prior written permission of HGS.
- **21. WAIVER.** The failure of HGS to insist on the performance of any of the terms or conditions of this Agreement or to exercise any right hereunder shall not be a waiver of such terms, conditions, or rights in the future, nor shall it be deemed to be a waiver of any other term, condition, or right under this Agreement.
- **22. MODIFICATION OF TERMS AND CONDITIONS.** No terms and conditions other than those stated herein, and no modification of these terms or conditions, shall be binding on HGS without HGS'

written consent.

- 23. TERM AND TERMINATION. The Agreement and the license granted herein shall remain effective until terminated. Licensee may terminate the Agreement and the license at any time by destroying all copies of Software and any Documentation. Licensee's rights under the Agreement will terminate immediately without notice from HGS if Licensee fails to comply with any provision of the Agreement. Upon termination, Licensee shall destroy all copies of Software and Documentation in its possession or control. All confidentiality obligations of Licensee and all limitations of liability and disclaimers and restrictions of warranty shall survive termination of this Agreement.
- **24. LICENSEE RECORDS.** Licensee grants to HGS and its independent accountants the right to examine Licensee's books, records and accounts during Licensee's normal business hours to verify compliance of with this Agreement as it relates to the Software. In the event such audit discloses noncompliance with this Agreement, Licensee shall promptly pay to HGS the appropriate license fees, plus the reasonable cost of conducting the audit.
- **25. GOVERNING LAW.** The Agreement shall be governed as follows:
 - a. For commercial entities, the laws of the State of Colorado without regard to its choice of law rules.
 - b. For U.S. state and local governments and/or higher education schools governed by state laws, contracts shall be governed by the laws of the state in which they are located without reference to conflict of laws principles.
 - c. For the U.S. Government, contracts shall be governed by U.S. federal laws.
 - d. Contracts will not be governed by the United Nations Convention on Contracts for International Sale of Goods; this application is expressly excluded.
- **26. GENERAL PROVISIONS.** If any part of the Agreement is found void and unenforceable, it will not affect the validity of the balance of the Agreement, which will remain valid and enforceable according to its terms. This Agreement may only be modified by a writing signed by an authorized officer of HGS. The English version of this agreement will be the version used when interpreting or construing this Agreement.
- **27. NOTICE.** Any notice relating to the Agreement should be sent by personal delivery or U.S. certified mail (return receipt requested) to the address provided below and will be effective upon receipt: Harris Geospatial Solutions, Inc., ATTN: Contracts Department, 385 Interlocken Crescent, Suite 300, Broomfield, CO, 80021, USA.

Legal and Copyright Notices

The IDL® and ENVI® software programs and the accompanying procedures, functions, and documentation described herein are sold under license agreements. Their use, duplication, and disclosure are subject to the restrictions stated in the license agreement. Harris Geospatial Solutions, Inc. reserves the right to make changes to this document at any time and without notice.

Limitation of Warranty

HARRIS GEOSPATIAL SOLUTIONS, INC. MAKES NO WARRANTIES, EITHER EXPRESS OR IMPLIED, AS TO ANY MATTER NOT EXPRESSLY SET FORTH IN THE LICENSE AGREEMENT, INCLUDING WITHOUT LIMITATION THE CONDITION OF THE SOFTWARE, MERCHANTABILITY, OR FITNESS FOR ANY PARTICULAR PURPOSE. HARRIS GEOSPATIAL SOLUTIONS, INC. SHALL NOT BE LIABLE FOR ANY DIRECT, CONSEQUENTIAL, OR ANY OTHER DAMAGES, SUFFERED BY THE LICENSED USER OR ANY OTHER USERS RESULTING FROM THE USE OF THE SOFTWARE PACKAGES OR THE SOFTWARE DOCUMENTATION.

Permission to Reproduce Manuals

If you are a licensed user of Harris Geospatial Solutions, Inc. software, Harris Geospatial Solutions, Inc. grants you a limited, nontransferable license to reproduce its software's manuals provided such copies are for your use only and are not sold or distributed to third parties. All such copies must contain the title page and Harris Geospatial Solutions, Inc. copyright notice.

Export Control Information

Harris Geospatial Solutions, Inc. Software and its associated technology are subject to U.S. export controls including the United States Export Administration Regulations. The licensed user is responsible for ensuring compliance with all applicable U.S. export control laws and regulations. These laws include restrictions on destinations, end users and end use.

Copyright and Trademark Notices

IDL® and ENVI® are registered trademarks of L3Harris Technologies, Inc.

© 1988-2021 Harris Geospatial Solutions, Inc.

Esri®, ArcGIS®, ArcView®, and ArcInfo® are registered trademarks of Esri.

Adobe Illustrator® and Adobe PDF® Print Engine are either registered trademarks or trademarks of Adobe Systems Incorporated in the United States and/or other countries.

Macintosh® is a registered trademark of Apple Inc., registered in the U.S. and other countries.

 $PowerPoint @, PowerPoint icon \ and \ Windows @ \ are \ registered \ trademarks \ of \ Microsoft \ Corporation \ in \ the \ United \ States \ and/or \ other \ countries.$

UNIX® is a registered trademark of The Open Group.

FLAASH® and QUAC® are registered trademarks of Spectral Sciences, Inc.

Other trademarks and registered trademarks are the property of the respective trademark holders.

IDL® Software Components

IDL® software contains components that are copyrighted and require the following acknowledgments:

3DLabs, Inc. Ltd. - © 2002-2005 All Rights Reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

Redistributions of source code must retain the above copyright notice this list of conditions and the following disclaimer.

Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

Neither the name of 3Dlabs Inc. Ltd. nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDERS OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION: HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

AdoptOpenJDK JDK

AdoptOpen JRE

See GPL LICENSE for details.

-- Class Path Exception--

Linking this library statically or dynamically with other modules is making a combined work based on this library. Thus, the terms and conditions of the GNU General Public License cover the whole combination.

As a special exception, the copyright holders of this library give you permission to link this library with independent modules to produce an executable, regardless of the license terms of these independent modules, and to copy and distribute the resulting executable under terms of your choice, provided that you also meet, for each linked independent module, the terms and conditions of the license of that module. An independent module is a module which is not derived from or based on this library. If you modify this library, you may extend this exception to your version of the library, but you are not obligated to do so. If you do not wish to do so, delete this exception statement from your version.

-GPL LICENSE-

GNU General Public License, version 2, with the Class Path Exception

Version 2, June 1991

Copyright (C) 1989, 1991 Free Software Foundation, Inc.

59 Temple Place, Suite 330, Boston, MA 02111-1307 USA

Everyone is permitted to copy and distribute verbatim copies of this license document, but changing it is not allowed.

Preamble

The licenses for most software are designed to take away your freedom to share and change it. By contrast, the GNU General Public License is intended to guarantee your freedom to share and change free software—to make sure the software is free for all its users. This General Public License applies to most of the Free Software Foundation's software and to any other program whose authors commit to using it. (Some other Free Software Foundation software is covered by the GNU Library General Public License instead.) You can apply it to your programs, too.

When we speak of free software, we are referring to freedom, not price. Our General Public Licenses are designed to make sure that you have the freedom to distribute copies of free software (and charge for this service if you wish), that you receive source code or can get it if you want it, that you can change the software or use pieces of it in new free programs; and that you know you can do these things.

To protect your rights, we need to make restrictions that forbid anyone to deny you these rights or to ask you to surrender the rights. These restrictions translate to certain responsibilities for you if you distribute copies of the software, or if you modify it.

For example, if you distribute copies of such a program, whether gratis or for a fee, you must give the recipients all the rights that you have. You must make sure that they, too, receive or can get the source code. And you must show them these terms so they know their rights.

We protect your rights with two steps: (1) copyright the software, and (2) offer you this license which gives you legal permission to copy, distribute and/or modify the software.

Also, for each author's protection and ours, we want to make certain that everyone understands that there is no warranty for this free software. If the software is modified by someone else and passed on, we want its recipients to know that what they have is not the original, so that any problems introduced by others will not reflect on the original authors' reputations.

Finally, any free program is threatened constantly by software patents. We wish to avoid the danger that redistributors of a free program will individually obtain patent licenses, in effect making the program proprietary. To prevent this, we have made it clear that any patent must be licensed for everyone's free use or not licensed at all.

The precise terms and conditions for copying, distribution and modification follow.

TERMS AND CONDITIONS FOR COPYING, DISTRIBUTION AND MODIFICATION

0. This License applies to any program or other work which contains a notice placed by the copyright holder saying it may be distributed under the terms of this General Public License. The "Program", below, refers to any such program or work, and a "work based on the Program" means either the Program or any derivative work under copyright law: that is to say, a work containing the Program or a portion of it, either verbatim or with modifications and/or translated into another language. (Hereinafter, translation is included without limitation in the term "modification".) Each licensee is addressed as "you".

Activities other than copying, distribution and modification are not covered by this License; they are outside its scope. The act of running the Program is not restricted, and the output from the Program is covered only if its contents constitute a work based on the Program (independent of having been made by running the Program). Whether that is true depends on what the Program does.

1. You may copy and distribute verbatim copies of the Program's source code as you receive it, in any medium, provided that you conspicuously and appropriately publish on each copy an appropriate copyright notice and disclaimer of warranty; keep intact all the notices that refer to this License and to the absence of any warranty; and give any other recipients of the Program a copy of this License along with the Program.

You may charge a fee for the physical act of transferring a copy, and you may at your option offer warranty protection in exchange for a fee.

- 2. You may modify your copy or copies of the Program or any portion of it, thus forming a work based on the Program, and copy and distribute such modifications or work under the terms of Section 1 above, provided that you also meet all of these conditions:
 - a. You must cause the modified files to carry prominent notices stating that you changed the files and the date of any change.
 - b. You must cause any work that you distribute or publish, that in whole or in part contains or is derived from the Program or any part thereof, to be licensed as a whole at no charge to all third parties under the terms of this License.
 - c. If the modified program normally reads commands interactively when run, you must cause it, when started running for

such interactive use in the most ordinary way, to print or display an announcement including an appropriate copyright notice and a notice that there is no warranty (or else, saying that you provide a warranty) and that users may redistribute the program under these conditions, and telling the user how to view a copy of this License. (Exception: if the Program itself is interactive but does not normally print such an announcement, your work based on the Program is not required to print an announcement.)

These requirements apply to the modified work as a whole. If identifiable sections of that work are not derived from the Program, and can be reasonably considered independent and separate works in themselves, then this License, and its terms, do not apply to those sections when you distribute them as separate works. But when you distribute the same sections as part of a whole which is a work based on the Program, the distribution of the whole must be on the terms of this License, whose permissions for other licensees extend to the entire whole, and thus to each and every part regardless of who wrote it.

Thus, it is not the intent of this section to claim rights or contest your rights to work written entirely by you; rather, the intent is to exercise the right to control the distribution of derivative or collective works based on the Program.

In addition, mere aggregation of another work not based on the Program with the Program (or with a work based on the Program) on a volume of a storage or distribution medium does not bring the other work under the scope of this License.

- 3. You may copy and distribute the Program (or a work based on it, under Section 2) in object code or executable form under the terms of Sections 1 and 2 above provided that you also do one of the following:
 - a. Accompany it with the complete corresponding machine-readable source code, which must be distributed under the terms of Sections 1 and 2 above on a medium customarily used for software interchange; or,
 - b. Accompany it with a written offer, valid for at least three years, to give any third party, for a charge no more than your cost of physically performing source distribution, a complete machine-readable copy of the corresponding source code, to be distributed under the terms of Sections 1 and 2 above on a medium customarily used for software interchange; or.
 - c. Accompany it with the information you received as to the offer to distribute corresponding source code. (This alternative is allowed only for noncommercial distribution and only if you received the program in object code or executable form with such an offer, in accord with Subsection b above.)

The source code for a work means the preferred form of the work for making modifications to it. For an executable work, complete source code means all the source code for all modules it contains, plus any associated interface definition files, plus the scripts used to control compilation and installation of the executable. However, as a special exception, the source code distributed need not include anything that is normally distributed (in either source or binary form) with the major components (compiler, kernel, and so on) of the operating system on which the executable runs, unless that component itself accompanies the executable.

If distribution of executable or object code is made by offering access to copy from a designated place, then offering equivalent access to copy the source code from the same place counts as distribution of the source code, even though third parties are not compelled to copy the source along with the object code.

- 4. You may not copy, modify, sublicense, or distribute the Program except as expressly provided under this License. Any attempt otherwise to copy, modify, sublicense or distribute the Program is void, and will automatically terminate your rights under this License. However, parties who have received copies, or rights, from you under this License will not have their licenses terminated so long as such parties remain in full compliance.
- 5. You are not required to accept this License, since you have not signed it. However, nothing else grants you permission to modify or distribute the Program or its derivative works. These actions are prohibited by law if you do not accept this License. Therefore, by modifying or distributing the Program (or any work based on the Program), you indicate your acceptance of this License to do so, and all its terms and conditions for copying, distributing or modifying the Program or works based on it.
- 6. Each time you redistribute the Program (or any work based on the Program), the recipient automatically receives a license from the original licensor to copy, distribute or modify the Program subject to these terms and conditions. You may not impose any further restrictions on the recipients' exercise of the rights granted herein. You are not responsible for enforcing compliance by third parties to this License.
- 7. If, as a consequence of a court judgment or allegation of patent infringement or for any other reason (not limited to patent issues), conditions are imposed on you (whether by court order, agreement or otherwise) that contradict the conditions of this License, they do not excuse you from the conditions of this License. If you cannot distribute so as to satisfy simultaneously your obligations under this License and any other pertinent obligations, then as a consequence you may not distribute the Program at all. For example, if a patent license would not permit royalty-free redistribution of the Program by all those who receive copies directly or indirectly through you, then the only way you could satisfy both it and this License would be to refrain entirely from distribution of the Program.

If any portion of this section is held invalid or unenforceable under any particular circumstance, the balance of the section is intended to apply and the section as a whole is intended to apply in other circumstances.

It is not the purpose of this section to induce you to infringe any patents or other property right claims or to contest validity of any such claims; this section has the sole purpose of protecting the integrity of the free software distribution system, which is implemented by public license practices. Many people have made generous contributions to the wide range of software distributed through that system in reliance on consistent application of that system; it is up to the author/donor to decide if he or she is willing to distribute software through any other system and a licensee cannot impose that choice.

This section is intended to make thoroughly clear what is believed to be a consequence of the rest of this License.

- 8. If the distribution and/or use of the Program is restricted in certain countries either by patents or by copyrighted interfaces, the original copyright holder who places the Program under this License may add an explicit geographical distribution limitation excluding those countries, so that distribution is permitted only in or among countries not thus excluded. In such case, this License incorporates the limitation as if written in the body of this License.
- 9. The Free Software Foundation may publish revised and/or new versions of the General Public License from time to time. Such new versions will be similar in spirit to the present version, but may differ in detail to address new problems or concerns. Each version is given a distinguishing version number. If the Program specifies a version number of this License which applies to it and "any later version", you have the option of following the terms and conditions either of that version or of any later version published by the Free Software Foundation. If the Program does not specify a version number of this License, you may choose any version ever published by the Free Software Foundation.

10. If you wish to incorporate parts of the Program into other free programs whose distribution conditions are different, write to the author to ask for permission. For software which is copyrighted by the Free Software Foundation, write to the Free Software Foundation; we sometimes make exceptions for this. Our decision will be guided by the two goals of preserving the free status of all derivatives of our free software and of promoting the sharing and reuse of software generally.

NO WARRANTY

- 11. BECAUSE THE PROGRAM IS LICENSED FREE OF CHARGE, THERE IS NO WARRANTY FOR THE PROGRAM, TO THE EXTENT PERMITTED BY APPLICABLE LAW. EXCEPT WHEN OTHERWISE STATED IN WRITING THE COPYRIGHT HOLDERS AND/OR OTHER PARTIES PROVIDE THE PROGRAM "AS IS" WITHOUT WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE. THE ENTIRE RISK AS TO THE QUALITY AND PERFORMANCE OF THE PROGRAM IS WITH YOU. SHOULD THE PROGRAM PROVE DEFECTIVE, YOU ASSUME THE COST OF ALL NECESSARY SERVICING, REPAIR OR CORRECTION.
- 12. IN NO EVENT UNLESS REQUIRED BY APPLICABLE LAW OR AGREED TO IN WRITING WILL ANY COPYRIGHT HOLDER, OR ANY OTHER PARTY WHO MAY MODIFY AND/OR REDISTRIBUTE THE PROGRAM AS PERMITTED ABOVE, BE LIABLE TO YOU FOR DAMAGES, INCLUDING ANY GENERAL, SPECIAL, INCIDENTAL OR CONSEQUENTIAL DAMAGES ARISING OUT OF THE USE OR INABILITY TO USE THE PROGRAM (INCLUDING BUT NOT LIMITED TO LOSS OF DATA OR DATA BEING RENDERED INACCURATE OR LOSSES SUSTAINED BY YOU OR THIRD PARTIES OR A FAILURE OF THE PROGRAM TO OPERATE WITH ANY OTHER PROGRAMS), EVEN IF SUCH HOLDER OR OTHER PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

END OF TERMS AND CONDITIONS

How to Apply These Terms to Your New Programs

If you develop a new program, and you want it to be of the greatest possible use to the public, the best way to achieve this is to make it free software which everyone can redistribute and change under these terms.

To do so, attach the following notices to the program. It is safest to attach them to the start of each source file to most effectively convey the exclusion of warranty; and each file should have at least the "copyright" line and a pointer to where the full notice is found.

One line to give the program's name and a brief idea of what it does.

Copyright (C) < year > < name of author >

This program is free software; you can redistribute it and/or modify it under the terms of the GNU General Public License as published by the Free Software Foundation; either version 2 of the License, or (at your option) any later version. This program is distributed in the hope that it will be useful, but WITHOUT ANY WARRANTY; without even the implied warranty of MERCHANTABILITY or FITNESS FOR A PARTICULAR PURPOSE. See the GNU General Public License for more details. You should have received a copy of the GNU General Public License along with this program; if not, write to the Free Software Foundation, Inc., 59 Temple Place, Suite 330, Boston, MA 02111-1307 USA

Also add information on how to contact you by electronic and paper mail.

If the program is interactive, make it output a short notice like this when it starts in an interactive mode:

Gnomovision version 69, Copyright (C) year name of author Gnomovision comes with ABSOLUTELY NO WARRANTY; for details type 'show w'. This is free software, and you are welcome to redistribute it under certain conditions; type 'show c' for details.

The hypothetical commands 'show w' and 'show c' should show the appropriate parts of the General Public License. Of course, the commands you use may be called something other than 'show w' and 'show c'; they could even be mouse-clicks or menu items--whatever suits your program.

You should also get your employer (if you work as a programmer) or your school, if any, to sign a "copyright disclaimer" for the program, if necessary. Here is a sample; alter the names:

Yoyodyne, Inc., hereby disclaims all copyright interest in the program

'Gnomovision' (which makes passes at compilers) written by James Hacker.

signature of Ty Coon, 1 April 1989

Ty Coon, President of Vice

This General Public License does not permit incorporating your program into proprietary programs. If your program is a subroutine library, you may consider it more useful to permit linking proprietary applications with the library. If this is what you want to do, use the GNU Library General Public License instead of this License.

-END GPL LICENSE-

Aha-Soft Free Icon - www.aha-soft.com

Licensed under the Creative Commons Attribution-Share Alike 3.0 License found at:

http:\\creativecommons.org/licenses/by/3.0/us

ANTLR - © 1989-2004 Developed by Terence Parr. Partially supported by University of San Francisco & ¡Guru.com

ANTLR 2 License

We reserve no legal rights to the ANTLR--it is fully in the public domain. An individual or company may do whatever they wish with source code distributed with ANTLR or the code generated by ANTLR, including the incorporation of ANTLR, or its output, into commercial software.

We encourage users to develop software with ANTLR. However, we do ask that credit is given to us for developing ANTLR. By

"credit", we mean that if you use ANTLR or incorporate any source code into one of your programs (commercial product, research project, or otherwise) that you acknowledge this fact somewhere in the documentation, research report, etc... If you like ANTLR and have developed a nice tool with the output, please mention that you developed it using ANTLR. In addition, we ask that the headers remain intact in our source code. As long as these guidelines are kept, we expect to continue enhancing this system and expect to make other tools available as they are completed.

Apache commons-logging. © **2001-2013 The Apache Software Foundation. All Rights Reserved.** Licensed under the Apache License Version 2.0. Apache Commons, Apache Commons Logging, Apache, the Apache feather logo and the Apache Commons project logos are trademarks of The Apache Software Foundation.

Apache FontBox - Licensed under the Apache License Version 2.0.

Apache JSON simple - Licensed under the Apache License Version 2.0

ApachePDF Box - © 2014. The Apache Software Foundation. Licensed under the Apache License Version 2.0. Apache PDFBox, PDFBox, Apache, the Apache feather logo and the Apache PDFBox project logos are trademarks of The Apache Software Foundation.

Arimo, Cousine, Tinos TrueType fonts - designed by Steve Matteson, and are © 2010-2012 by Google Corporation, and released under the Apache License, Version 2.0.

Cousine-Symbol and Tinos-Symbol TrueType fonts - The non-symbol characters are based upon Cousine and Tinos (see above). A few of the symbol characters are from DejaVu Sans, by Tavmjong Bah (see license within this document). The remaining symbol characters are © 2017 Harris Geospatial Solutions, Inc..

ecCodes - © 2005-2019 ECMWF. Licensed under the Apache License, Version 2.0.

fontbox - © 2010 The Apache Software Foundation. Licensed under the Apache License, Version 2.0.

Jackson JSON Processor - © 2007- Tatu Saloranta. Licensed under the Apache License, Version 2.0.

json-schema-validator - © 2014, Francis GAliegue (fgaliegue@gmail.com) . Licensed under the Apache License Version 2.0

json-simple - © 2016 Clifton Labs. Licensed under the Apache License Version 2.0

xerces-c - © 1999-2017 The Apache Software Foundation. All Rights Reserved.

Openmp - © 1997-2019 Intel Corporation. Licensed under the Apache License Version 2.0, with LLVM exceptions.

---- LLVM Exceptions to the Apache 2.0 License ----

As an exception, if, as a result of your compiling your source code, portions of this Software are embedded into an Object form of such source code, you may redistribute such embedded portions in such Object form without complying with the conditions of Sections 4(a), 4(b) and 4(d) of the License.

In addition, if you combine or link compiled forms of this Software with software that is licensed under the GPLv2 ("Combined Software") and if a court of competent jurisdiction determines that the patent provision (Section 3), the indemnity provision (Section 9) or other Section of the License conflicts with the conditions of the GPLv2, you may retroactively and prospectively choose to deem waived or otherwise exclude such Section(s) of the License, but only in their entirety and only with respect to the Combined Software.

Software from third parties included in the LLVM Project:

The LLVM Project contains third party software which is under different license terms. All such code will be identified clearly using at least one of two mechanisms:

1) It will be in a separate directory tree with its own `LICENSE.txt` or `LICENSE` file at the top containing the specific license and restrictions which apply to that software, or

2) It will contain specific license and restriction terms at the top of every file.

 $Legacy\ LLVM\ License\ (https://llvm.org/docs/DeveloperPolicy.html \# legacy):$

The software contained in this directory tree is dual licensed under both the University of Illinois "BSD-Like" license and the MIT license. As a user of this code you may choose to use it under either license. As a contributor, you agree to allow your code to be used under both. The full text of the relevant licenses is included below.

In addition, a license agreement from the copyright/patent holders of the software contained in this directory tree is included below.

University of Illinois/NCSA

Open Source License

Copyright © 1997-2019 Intel Corporation

All rights reserved.

Developed by:

OpenMP Runtime Team

Intel Corporation

http://www.openmprtl.org

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal with the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimers.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimers in the documentation and/or other materials provided with the distribution.
- * Neither the names of Intel Corporation OpenMP Runtime Team nor the names of its contributors may be used to endorse or promote products derived from this Software without specific prior written permission.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE CONTRIBUTORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS WITH THE SOFTWARE.

Copyright © 1997-2019 Intel Corporation

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Intel Corporation

Software Grant License Agreement ("Agreement")

Except for the license granted herein to you, Intel Corporation ("Intel") reserves all right, title, and interest in and to the Software (defined below).

Definition

- "Software" means the code and documentation as well as any original work of authorship, including any modifications or additions to an existing work, that is intentionally submitted by Intel to llvm.org (http://llvm.org) ("LLVM") for inclusion in, or documentation of, any of the products owned or managed by LLVM (the "Work"). For the purposes of this definition, "submitted" means any form of electronic, verbal, or written communication sent to LLVM or its representatives, including but not limited to communication on electronic mailing lists, source code control systems, and issue tracking systems that are managed by, or on behalf of, LLVM for the purpose of discussing and improving the Work, but excluding communication that is conspicuously marked otherwise.
- 1. Grant of Copyright License. Subject to the terms and conditions of this Agreement, Intel hereby grants to you and to recipients of the Software distributed by LLVM a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare derivative works of, publicly display, publicly perform, sublicense, and distribute the Software and such derivative works.
- 2. Grant of Patent License. Subject to the terms and conditions of this Agreement, Intel hereby grants you and to recipients of the Software distributed by LLVM a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable (except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by Intel that are necessarily infringed by Intel's Software alone or by combination of the Software with the Work to which such Software was submitted. If any entity institutes patent litigation against Intel or any other entity (including a cross-claim or counterclaim in a lawsuit) alleging that Intel's Software, or the Work to which Intel has contributed constitutes direct or contributory patent infringement, then any patent licenses granted to that entity under this Agreement for the Software or Work shall terminate as of the date such litigation is filed.

Unless required by applicable law or agreed to in writing, the software is provided on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE.

-APACHE LICENSE-

Apache License Version 2.0, January 2004 http://www.apache.org/licenses/

TERMS AND CONDITIONS FOR USE, REPRODUCTION, AND DISTRIBUTION

1. Definitions.

"License" shall mean the terms and conditions for use, reproduction, and distribution as defined by Sections 1 through 9 of this document.

"Licensor" shall mean the copyright owner or entity authorized by the copyright owner that is granting the License.

"Legal Entity" shall mean the union of the acting entity and all other entities that control, are controlled by, or are under common control with that entity. For the purposes of this definition, "control" means (i) the power, direct or indirect, to cause the direction or management of such entity, whether by contract or otherwise, or (ii) ownership of fifty percent (50%) or more of the outstanding shares, or (iii) beneficial ownership of such entity.

"You" (or "Your") shall mean an individual or Legal Entity exercising permissions granted by this License.

"Source" form shall mean the preferred form for making modifications, including but not limited to software source code, documentation source, and configuration files.

"Object" form shall mean any form resulting from mechanical transformation or translation of a Source form, including but not limited to compiled object code, generated documentation, and conversions to other media types.

"Work" shall mean the work of authorship, whether in Source or Object form, made available under the License, as indicated by a copyright notice that is included in or attached to the work (an example is provided in the Appendix below).

"Derivative Works" shall mean any work, whether in Source or Object form, that is based on (or derived from) the Work and for which the editorial revisions, annotations, elaborations, or other modifications represent, as a whole, an original work of authorship. For the purposes of this License, Derivative Works shall not include works that remain separable from, or merely link (or bind by name) to the interfaces of, the Work and Derivative Works thereof.

"Contribution" shall mean any work of authorship, including the original version of the Work and any modifications or additions to that Work or Derivative Works thereof, that is intentionally submitted to Licensor for inclusion in the Work by the copyright owner or by an individual or Legal Entity authorized to submit on behalf of the copyright owner. For the purposes of this definition, "submitted" means any form of electronic, verbal, or written communication sent to the Licensor or its representatives, including but not limited to communication on electronic mailing lists, source code control systems, and issue tracking systems that are managed by, or on behalf of, the Licensor for the purpose of discussing and improving the Work, but excluding communication that is conspicuously marked or otherwise designated in writing by the copyright owner as "Not a Contribution."

"Contributor" shall mean Licensor and any individual or Legal Entity on behalf of whom a Contribution has been received by Licensor and subsequently incorporated within the Work.

2. Grant of Copyright License.

Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare Derivative Works of, publicly display, publicly perform, sublicense, and distribute the Work and such Derivative Works in Source or Object form.

3. Grant of Patent License.

Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable (except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by such Contributor that are necessarily infringed by their Contribution(s) alone or by combination of their Contribution(s) with the Work to which such Contribution(s) was submitted. If You institute patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Work or a Contribution incorporated within the Work constitutes direct or contributory patent infringement, then any patent licenses granted to You under this License for that Work shall terminate as of the date such litigation is filed.

4. Redistribution.

You may reproduce and distribute copies of the Work or Derivative Works thereof in any medium, with or without modifications, and in Source or Object form, provided that You meet the following conditions:

- a. You must give any other recipients of the Work or Derivative Works a copy of this License; and
- b. You must cause any modified files to carry prominent notices stating that You changed the files; and
- c. You must retain, in the Source form of any Derivative Works that You distribute, all copyright, patent, trademark, and attribution notices from the Source form of the Work, excluding those notices that do not pertain to any part of the Derivative Works: and
- d. If the Work includes a "NOTICE" text file as part of its distribution, then any Derivative Works that You distribute must include a readable copy of the attribution notices contained within such NOTICE file, excluding those notices that do not pertain to any part of the Derivative Works, in at least on of the following places: within a NOTICE text file distributed as part of the Derivative Works; within the Source form or documentation, if provided along with the Derivative Works; or, within a display generated by the Derivative Works, if and wherever such third-party notices normally appear. The contents of the NOTICE file are for informational purposes only and do not modify the License. You may add Your own attribution notices within Derivative Works that You distribute, alongside or as an addendum to the NOTICE text from the Work, provided that such additional attribution notices cannot be construed as modifying the License.

You may add Your own copyright statement to Your modifications and may provide additional or different license terms and conditions for use, reproduction, or distribution of Your modifications, or for any such Derivative Works as a whole, provided Your use, reproduction, and distribution of the Work otherwise complies with the conditions stated in this License.

5. Submission of Contributions.

Unless You explicitly state otherwise, any Contribution intentionally submitted for inclusion in the Work by You to the Licensor shall be under the terms and conditions of this License, without any additional terms or conditions. Notwithstanding the above, nothing herein shall supersede or modify the terms of any separate license agreement you may

have executed with Licensor regarding such Contributions.

Trademarks.

This License does not grant permission to use the trade names, trademarks, service marks, or product names of the Licensor, except as required for reasonable and customary use in describing the origin of the Work and reproducing the content of the NOTICE file.

7. Disclaimer of Warranty.

Unless required by applicable law or agreed to in writing, Licensor provides the Work (and each Contributor provides its Contributions) on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE. You are solely responsible for determining the appropriateness of using or redistributing the Work and assume any risks associated with Your exercise of permissions under this License.

8. Limitation of Liability.

In no event and under no legal theory, whether in tort (including negligence), contract, or otherwise, unless required by applicable law (such as deliberate and grossly negligent acts) or agreed to in writing, shall any Contributor be liable to You for damages, including any direct, indirect, special, incidental, or consequential damages of any character arising as a result of this License or out of the use or inability to use the Work (including but not limited to damages for loss of goodwill, work stoppage, computer failure or malfunction, or any and all other commercial damages or losses), even if such Contributor has been advised of the possibility of such damages.

9. Accepting Warranty or Additional Liability.

While redistributing the Work or Derivative Works thereof, You may choose to offer, and charge a fee for, acceptance of support, warranty, indemnity, or other liability obligations and/or rights consistent with this License. However, in accepting such obligations, You may act only on Your own behalf and on Your sole responsibility, not on behalf of any other Contributor, and only if You agree to indemnify, defend, and hold each Contributor harmless for any liability incurred by, or claims asserted against, such Contributor by reason of your accepting any such warranty or additional liability.

END OF TERMS AND CONDITIONS

APPENDIX: How to apply the Apache License to your work.

To apply the Apache License to your work, attach the following boilerplate notice, with the fields enclosed by brackets "[]" replaced with your own identifying information. (Don't include the brackets!) The text should be enclosed in the appropriate comment syntax for the file format. We also recommend that a file or class name and description of purpose be included on the same "printed page" as the copyright notice for easier identification within third-party archives.

Copyright [yyyy] [name of copyright owner]

Licensed under the Apache License, Version 2.0 (the "License"); you may not use this file except in compliance with the License.

You may obtain a copy of the License at http://www.apache.org/licenses/LICENSE-2.0

Unless required by applicable law or agreed to in writing, software distributed under the License is distributed on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied. See the License for the specific language governing permissions and limitations under the License.

-END APACHE LICENSE-

Arev Fonts - © 2020 Indigo Rose Software

Permission is hereby granted, free of charge, to any person obtaining a copy of the fonts accompanying this license ("Fonts") and associated documentation files (the "Font Software"), to reproduce and distribute the modifications to the Bitstream Vera Font Software, including without limitation the rights to use, copy, merge, publish, distribute, and/or sell copies of the Font Software, and to permit persons to whom the Font Software is furnished to do so, subject to the following conditions:

The above copyright and trademark notices and this permission notice shall be included in all copies of one or more of the Font Software typefaces.

The Font Software may be modified, altered, or added to, and in particular the designs of glyphs or characters in the Fonts may be modified and additional glyphs or characters may be added to the Fonts, only if the fonts are renamed to names not containing either the words "Tavmjong Bah" or the word "Arev".

This License becomes null and void to the extent applicable to Fonts or Font Software that has been modified and is distributed under the "Tavmjong Bah Arev" names.

The Font Software may be sold as part of a larger software package but no copy of one or more of the Font Software typefaces may be sold by itself.

THE FONT SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO ANY WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT OF COPYRIGHT, PATENT, TRADEMARK, OR OTHER RIGHT. IN NO EVENT SHALL TAVMJONG BAH BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, INCLUDING ANY GENERAL, SPECIAL, INDIRECT, INCIDENTAL, OR CONSEQUENTIAL DAMAGES, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF THE USE OR INABILITY TO USE THE FONT SOFTWARE OR FROM OTHER DEALINGS IN THE FONT SOFTWARE.

Except as contained in this notice, the name of Tavmjong Bah shall not be used in advertising or otherwise to promote the sale, use or other dealings in this Font Software without prior written authorization from Tavmjong Bah. For further information, contact: tavmjong @ free . fr.

AutoPlay Media Studio - © 2020 Indigo Rose Software

AVC - Portions of the IDL software are under a license agreement with MPEGLA.

DejaVu Sans Fonts - © 2003 by Bitstream, Inc. All Rights Reserved. Bitstream Vera is a trademark of Bitstream, Inc.

Permission is hereby granted, free of charge, to any person obtaining a copy of the fonts accompanying this license ("Fonts") and associated documentation files (the "Font Software"), to reproduce and distribute the Font Software, including without limitation the rights to use, copy, merge, publish, distribute, and/or sell copies of the Font Software, and to permit persons to whom the Font Software is furnished to do so, subject to the following conditions:

The above copyright and trademark notices and this permission notice shall be included in all copies of one or more of the Font Software typefaces.

The Font Software may be modified, altered, or added to, and in particular the designs of glyphs or characters in the Fonts may be modified and additional glyphs or characters may be added to the Fonts, only if the fonts are renamed to names not containing either the words "Bitstream" or the word "Vera".

This License becomes null and void to the extent applicable to Fonts or Font Software that has been modified and is distributed under the "Bitstream Vera" names.

The Font Software may be sold as part of a larger software package but no copy of one or more of the Font Software typefaces may be sold by itself.

THE FONT SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO ANY WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT OF COPYRIGHT, PATENT, TRADEMARK, OR OTHER RIGHT. IN NO EVENT SHALL BITSTREAM OR THE GNOME FOUNDATION BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, INCLUDING ANY GENERAL, SPECIAL, INDIRECT, INCIDENTAL, OR CONSEQUENTIAL DAMAGES, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF THE USE OR INABILITY TO USE THE FONT SOFTWARE OR FROM OTHER DEALINGS IN THE FONT SOFTWARE.

Except as contained in this notice, the names of Gnome, the Gnome Foundation, and Bitstream Inc., shall not be used in advertising or otherwise to promote the sale, use or other dealings in this Font Software without prior written authorization from the Gnome Foundation or Bitstream Inc., respectively. For further information, contact: fonts at gnome dot org.

BSD - IDL Software includes software developed by the University of California, Berkeley and its contributors.

© 1989, 1993, 1994 The Regents of the University of California. All Rights Reserved.

This code is derived from software contributed to Berkeley by Guido van Rossum.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with distributions.
- 3. All advertising materials mentioning features or use of this software must display the following acknowledgement: this product includes software developed by the University of California, Berkeley and its contributors.
- 4. Neither the name of the University nor the names of its contributors maybe used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDERS OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION: HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

-BSD 3 CLAUSE LICENSES-

- Chromium Embedded Framework © 2000-2005 Chih-Chung Chang and Chih-Jen Lin
- $\bullet\,$ libsvm © 2000-2005 Chih-Chung Chang and Chih-Jen Lin
- libvpx © 2010, The WebM Project authors. All rights reserved.
- NetCDF-C © 2018 Unidata
- NumPy © 2005-2020, NumPy Developers.
- strlcopy © 1998 Todd C. Miller < Todd.Miller@courtesan.com >. All Rights Reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- Neither the name of Google Inc. nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

-END BSD 3 CLAUSE LICENSES-

CDF - © 1999 National Space Science Data Center, NASA/Goddard Space Flight Center.

This software may be copied or redistributed as long as it is not sold for profit, but it can be incorporated into any other substantive product with or without modifications for profit or non-profit. If the software is modified, it must include the following notices:

- The software is not the original (for protection of the original author's reputations from any problems introduced by others).
- · Change history (e.g. date, functionality, etc.).

This copyright notice must be reproduced on each copy made. This software is provided as is without any express or implied warranties whatsoever.

Harris Geospatial Solutions, Inc. Addendum

 $The \ software \ is \ not \ the \ original \ (for \ protection \ of \ the \ original \ author's \ reputations \ from \ any \ problems \ introduced \ by \ others)$

Change history:

2017 Apr 05 - A fix was made in the CDF_TT2000_from_UTC_string() function in cdftt2000.c.

The change was given back to the owners and is included in CDF 3.7.1.

Central Test Node - © 1993, 1994, RSNA and Washington University

The software and supporting documentation for the Radiological Society of North America (RSNA) 1993, 1994 Digital Imaging and Communications in Medicine (DICOM) Demonstration were developed at the

Electronic Radiology Laboratory

Mallinckrodt Institute of Radiology

Washington University School of Medicine

510 S. Kingshighway Blvd.

St. Louis, MO 63110

as part of the 1993, 1994 DICOM Central Test Node project for, and under contract with, the Radiological Society of North America.

THIS SOFTWARE IS MADE AVAILABLE, AS IS, AND NEITHER RSNA NOR WASHINGTON UNIVERSITY MAKE ANY WARRANTY ABOUT THE SOFTWARE, ITS PERFORMANCE, ITS MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR USE, FREEDOM FROM ANY COMPUTER DISEASES OR ITS CONFORMITY TO ANY SPECIFICATION. THE ENTIRE RISK AS TO QUALITY AND PERFORMANCE OF THE SOFTWARE IS WITH THE USER.

Copyright of the software and supporting documentation is jointly owned by RSNA and Washington University, and free access is hereby granted as a license to use this software, copy this software and prepare derivative works based upon this software.

However, any distribution of this software source code or supporting documentation or derivative works (source code and supporting documentation) must include the three paragraphs of the copyright notice.

cURL - © 1996 - 2019, Daniel Stenberg, daniel@haxx.se, and many contributors, see the THANKS file. All rights reserved

Permission to use, copy, modify, and distribute this software for any purpose with or without fee is hereby granted, provided that the above copyright notice and this permission notice appear in all copies.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT OF THIRD PARTY RIGHTS. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Except as contained in this notice, the name of a copyright holder shall not be used in advertising or otherwise to promote the sale, use or other dealings in this Software without prior written authorization of the copyright holder.

DXF - © 1988-1998 Visio Corporation. All rights reserved.

Eclipse 1.0

eclipse-subversive-plugin - © 2005, 2014 Polarion Software and other. All rights reserved.

Eclipse Public License - v 1.0

THE ACCOMPANYING PROGRAM IS PROVIDED UNDER THE TERMS OF THIS ECLIPSE PUBLIC LICENSE ("AGREEMENT"). ANY USE, REPRODUCTION OR DISTRIBUTION OF THE PROGRAM CONSTITUTES RECIPIENT'S ACCEPTANCE OF THIS AGREEMENT.

1. DEFINITIONS

"Contribution" means:

- a) in the case of the initial Contributor, the initial code and documentation distributed under this Agreement, and
- b) in the case of each subsequent Contributor:
- i) changes to the Program, and
- ii) additions to the Program;

where such changes and/or additions to the Program originate from and are distributed by that particular Contributor. A Contribution 'originates' from a Contributor if it was added to the Program by such Contributor itself or anyone acting on such Contributor's behalf. Contributions do not include additions to the Program which: (i) are separate modules of software distributed in conjunction with the Program under their own license agreement, and (ii) are not derivative works of the Program.

"Contributor" means any person or entity that distributes the Program.

"Licensed Patents" mean patent claims licensable by a Contributor which are necessarily infringed by the use or sale of its Contribution alone or when combined with the Program.

"Program" means the Contributions distributed in accordance with this Agreement.

"Recipient" means anyone who receives the Program under this Agreement, including all Contributors.

2. GRANT OF RIGHTS

- a) Subject to the terms of this Agreement, each Contributor hereby grants Recipient a non-exclusive, worldwide, royalty-free copyright license to reproduce, prepare derivative works of, publicly display, publicly perform, distribute and sublicense the Contribution of such Contributor, if any, and such derivative works, in source code and object code form.
- b) Subject to the terms of this Agreement, each Contributor hereby grants Recipient a non-exclusive, worldwide, royalty-free patent license under Licensed Patents to make, use, sell, offer to sell, import and otherwise transfer the Contribution of such Contributor, if any, in source code and object code form. This patent license shall apply to the combination of the Contribution and the Program if, at the time the Contribution is added by the Contributor, such addition of the Contribution causes such combination to be covered by the Licensed Patents. The patent license shall not apply to any other combinations which include the Contribution. No hardware per se is licensed hereunder.
- c) Recipient understands that although each Contributor grants the licenses to its Contributions set forth herein, no assurances are provided by any Contributor that the Program does not infringe the patent or other intellectual property rights of any other entity. Each Contributor disclaims any liability to Recipient for claims brought by any other entity based on infringement of intellectual property rights or otherwise. As a condition to exercising the rights and licenses granted hereunder, each Recipient hereby assumes sole responsibility to secure any other intellectual property rights needed, if any. For example, if a third party patent license is required to allow Recipient to distribute the Program, it is Recipient's responsibility to acquire that license before distributing the Program.
- d) Each Contributor represents that to its knowledge it has sufficient copyright rights in its Contribution, if any, to grant the copyright license set forth in this Agreement.

3. REQUIREMENTS

A Contributor may choose to distribute the Program in object code form under its own license agreement, provided that:

- a) it complies with the terms and conditions of this Agreement; and
- b) its license agreement:
- i) effectively disclaims on behalf of all Contributors all warranties and conditions, express and implied, including warranties or conditions of title and non-infringement, and implied warranties or conditions of merchantability and fitness for a particular purpose;
- ii) effectively excludes on behalf of all Contributors all liability for damages, including direct, indirect, special, incidental and consequential damages, such as lost profits;
- iii) states that any provisions which differ from this Agreement are offered by that Contributor alone and not by any other party; and
- iv) states that source code for the Program is available from such Contributor, and informs licensees how to obtain it in a reasonable manner on or through a medium customarily used for software exchange.

When the Program is made available in source code form:

- a) it must be made available under this Agreement; and
- b) a copy of this Agreement must be included with each copy of the Program.

Contributors may not remove or alter any copyright notices contained within the Program.

Each Contributor must identify itself as the originator of its Contribution, if any, in a manner that reasonably allows subsequent Recipients to identify the originator of the Contribution.

4. COMMERCIAL DISTRIBUTION

Commercial distributors of software may accept certain responsibilities with respect to end users, business partners and the like. While this license is intended to facilitate the commercial use of the Program, the Contributor who includes the Program in a commercial product offering should do so in a manner which does not create potential liability for other Contributors. Therefore, if a Contributor includes the Program in a commercial product offering, such Contributor ("Commercial Contributor") hereby agrees to defend and indemnify every other Contributor ("Indemnified Contributor") against any losses, damages and costs (collectively "Losses") arising from claims, lawsuits and other legal actions brought by a third party against the Indemnified Contributor to the extent caused by the acts or omissions of such Commercial Contributor in connection with its

distribution of the Program in a commercial product offering. The obligations in this section do not apply to any claims or Losses relating to any actual or alleged intellectual property infringement. In order to qualify, an Indemnified Contributor must: a) promptly notify the Commercial Contributor in writing of such claim, and b) allow the Commercial Contributor to control, and cooperate with the Commercial Contributor in, the defense and any related settlement negotiations. The Indemnified Contributor may participate in any such claim at its own expense.

For example, a Contributor might include the Program in a commercial product offering, Product X. That Contributor is then a Commercial Contributor. If that Commercial Contributor then makes performance claims, or offers warranties related to Product X, those performance claims and warranties are such Commercial Contributor's responsibility alone. Under this section, the Commercial Contributor would have to defend claims against the other Contributors related to those performance claims and warranties, and if a court requires any other Contributor to pay any damages as a result, the Commercial Contributor must pay those damages.

5. NO WARRANTY

EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, THE PROGRAM IS PROVIDED ON AN "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, EITHER EXPRESS OR IMPLIED INCLUDING, WITHOUT LIMITATION, ANY WARRANTIES OR CONDITIONS OF TITLE, NON-INFRINGEMENT, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE. Each Recipient is solely responsible for determining the appropriateness of using and distributing the Program and assumes all risks associated with its exercise of rights under this Agreement, including but not limited to the risks and costs of program errors, compliance with applicable laws, damage to or loss of data, programs or equipment, and unavailability or interruption of operations.

6. DISCLAIMER OF LIABILITY

EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, NEITHER RECIPIENT NOR ANY CONTRIBUTORS SHALL HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING WITHOUT LIMITATION LOST PROFITS), HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OR DISTRIBUTION OF THE PROGRAM OR THE EXERCISE OF ANY RIGHTS GRANTED HEREUNDER, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

7. GENERAL

If any provision of this Agreement is invalid or unenforceable under applicable law, it shall not affect the validity or enforceability of the remainder of the terms of this Agreement, and without further action by the parties hereto, such provision shall be reformed to the minimum extent necessary to make such provision valid and enforceable.

If Recipient institutes patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Program itself (excluding combinations of the Program with other software or hardware) infringes such Recipient's patent(s), then such Recipient's rights granted under Section 2(b) shall terminate as of the date such litigation is filed.

All Recipient's rights under this Agreement shall terminate if it fails to comply with any of the material terms or conditions of this Agreement and does not cure such failure in a reasonable period of time after becoming aware of such noncompliance. If all Recipient's rights under this Agreement terminate, Recipient agrees to cease use and distribution of the Program as soon as reasonably practicable. However, Recipient's obligations under this Agreement and any licenses granted by Recipient relating to the Program shall continue and survive.

Everyone is permitted to copy and distribute copies of this Agreement, but in order to avoid inconsistency the Agreement is copyrighted and may only be modified in the following manner. The Agreement Steward reserves the right to publish new versions (including revisions) of this Agreement from time to time. No one other than the Agreement Steward has the right to modify this Agreement. The Eclipse Foundation is the initial Agreement Steward. The Eclipse Foundation may assign the responsibility to serve as the Agreement Steward to a suitable separate entity. Each new version of the Agreement will be given a distinguishing version number. The Program (including Contributions) may always be distributed subject to the version of the Agreement under which it was received. In addition, after a new version of the Agreement is published, Contributor may elect to distribute the Program (including its Contributions) under the new version. Except as expressly stated in Sections 2(a) and 2(b) above, Recipient receives no rights or licenses to the intellectual property of any Contributor under this Agreement, whether expressly, by implication, estoppel or otherwise. All rights in the Program not expressly granted under this Agreement are reserved.

This Agreement is governed by the laws of the State of New York and the intellectual property laws of the United States of America. No party to this Agreement will bring a legal action under this Agreement more than one year after the cause of action arose. Each party waives its rights to a jury trial in any resulting litigation.

Eclipse 2.0

- \bullet eclipse-cvs-client-plugin \circledcirc Eclipse contributors and others 2000, 2019. All rights reserved.
- $\bullet\,$ eclipse-egit-plugin © 2005, 2009 EGit project committers and others. All rights reserved.
- eclipse-platform Eclipse contributors and others 2000, 2019. All rights reserved.

Eclipse Public License - v 2.0

THE ACCOMPANYING PROGRAM IS PROVIDED UNDER THE TERMS OF THIS ECLIPSE PUBLIC LICENSE ("AGREEMENT"). ANY USE, REPRODUCTION OR DISTRIBUTION OF THE PROGRAM CONSTITUTES RECIPIENT'S ACCEPTANCE OF THIS AGREEMENT.

1. DEFINITIONS

"Contribution" means:

- a) in the case of the initial Contributor, the initial content Distributed under this Agreement, and
- b) in the case of each subsequent Contributor:
- i) changes to the Program, and

ii) additions to the Program;

where such changes and/or additions to the Program originate from and are Distributed by that particular Contributor. A Contribution "originates" from a Contributor if it was added to the Program by such Contributor itself or anyone acting on such Contributor's behalf. Contributions do not include changes or additions to the Program that are not Modified Works.

- "Contributor" means any person or entity that Distributes the Program.
- "Licensed Patents" mean patent claims licensable by a Contributor which are necessarily infringed by the use or sale of its Contribution alone or when combined with the Program.
- "Program" means the Contributions Distributed in accordance with this Agreement.
- "Recipient" means anyone who receives the Program under this Agreement or any Secondary License (as applicable), including Contributors.
- "Derivative Works" shall mean any work, whether in Source Code or other form, that is based on (or derived from) the Program and for which the editorial revisions, annotations, elaborations, or other modifications represent, as a whole, an original work of authorship.
- "Modified Works" shall mean any work in Source Code or other form that results from an addition to, deletion from, or modification of the contents of the Program, including, for purposes of clarity any new file in Source Code form that contains any contents of the Program. Modified Works shall not include works that contain only declarations, interfaces, types, classes, structures, or files of the Program solely in each case in order to link to, bind by name, or subclass the Program or Modified Works thereof.
- "Distribute" means the acts of a) distributing or b) making available in any manner that enables the transfer of a copy.
- "Source Code" means the form of a Program preferred for making modifications, including but not limited to software source code, documentation source, and configuration files.
- "Secondary License" means either the GNU General Public License, Version 2.0, or any later versions of that license, including any exceptions or additional permissions as identified by the initial Contributor.

2. GRANT OF RIGHTS

- a) Subject to the terms of this Agreement, each Contributor hereby grants Recipient a non-exclusive, worldwide, royalty-free copyright license to reproduce, prepare Derivative Works of, publicly display, publicly perform, Distribute and sublicense the Contribution of such Contributor, if any, and such Derivative Works.
- b) Subject to the terms of this Agreement, each Contributor hereby grants Recipient a non-exclusive, worldwide, royalty-free patent license under Licensed Patents to make, use, sell, offer to sell, import and otherwise transfer the Contribution of such Contributor, if any, in Source Code or other form. This patent license shall apply to the combination of the Contribution and the Program if, at the time the Contribution is added by the Contributor, such addition of the Contribution causes such combination to be covered by the Licensed Patents. The patent license shall not apply to any other combinations which include the Contribution. No hardware per se is licensed hereunder.
- c) Recipient understands that although each Contributor grants the licenses to its Contributions set forth herein, no assurances are provided by any Contributor that the Program does not infringe the patent or other intellectual property rights of any other entity. Each Contributor disclaims any liability to Recipient for claims brought by any other entity based on infringement of intellectual property rights or otherwise. As a condition to exercising the rights and licenses granted hereunder, each Recipient hereby assumes sole responsibility to secure any other intellectual property rights needed, if any. For example, if a third party patent license is required to allow Recipient to Distribute the Program, it is Recipient's responsibility to acquire that license before distributing the Program.
- d) Each Contributor represents that to its knowledge it has sufficient copyright rights in its Contribution, if any, to grant the copyright license set forth in this Agreement.
- e) Notwithstanding the terms of any Secondary License, no Contributor makes additional grants to any Recipient (other than those set forth in this Agreement) as a result of such Recipient's receipt of the Program under the terms of a Secondary License (if permitted under the terms of Section 3).

3. REQUIREMENTS

- 3.1 If a Contributor Distributes the Program in any form, then:
- a) the Program must also be made available as Source Code, in accordance with section 3.2, and the Contributor must accompany the Program with a statement that the Source Code for the Program is available under this Agreement, and informs Recipients how to obtain it in a reasonable manner on or through a medium customarily used for software exchange; and
- b) the Contributor may Distribute the Program under a license different than this Agreement, provided that such license:
- i) effectively disclaims on behalf of all other Contributors all warranties and conditions, express and implied, including warranties or conditions of title and non-infringement, and implied warranties or conditions of merchantability and fitness for a particular purpose;
- ii) effectively excludes on behalf of all other Contributors all liability for damages, including direct, indirect, special, incidental and consequential damages, such as lost profits;
- iii) does not attempt to limit or alter the recipients' rights in the Source Code under section 3.2; and
- iv) requires any subsequent distribution of the Program by any party to be under a license that satisfies the requirements of this section 3.
- $3.2\ When the Program is Distributed as Source Code:$
- a) it must be made available under this Agreement, or if the Program (i) is combined with other material in a separate file or

files made available under a Secondary License, and (ii) the initial Contributor attached to the Source Code the notice described in Exhibit A of this Agreement, then the Program may be made available under the terms of such Secondary Licenses, and

b) a copy of this Agreement must be included with each copy of the Program.

3.3 Contributors may not remove or alter any copyright, patent, trademark, attribution notices, disclaimers of warranty, or limitations of liability ("notices") contained within the Program from any copy of the Program which they Distribute, provided that Contributors may add their own appropriate notices.

4. COMMERCIAL DISTRIBUTION

Commercial distributors of software may accept certain responsibilities with respect to end users, business partners and the like. While this license is intended to facilitate the commercial use of the Program, the Contributor who includes the Program in a commercial product offering should do so in a manner which does not create potential liability for other Contributors. Therefore, if a Contributor includes the Program in a commercial product offering, such Contributor ("Commercial Contributor") hereby agrees to defend and indemnify every other Contributor ("Indemnified Contributor") against any losses, damages and costs (collectively "Losses") arising from claims, lawsuits and other legal actions brought by a third party against the Indemnified Contributor to the extent caused by the acts or omissions of such Commercial Contributor in connection with its distribution of the Program in a commercial product offering. The obligations in this section do not apply to any claims or Losses relating to any actual or alleged intellectual property infringement. In order to qualify, an Indemnified Contributor must: a) promptly notify the Commercial Contributor in writing of such claim, and b) allow the Commercial Contributor to control, and cooperate with the Commercial Contributor in, the defense and any related settlement negotiations. The Indemnified Contributor may participate in any such claim at its own expense.

For example, a Contributor might include the Program in a commercial product offering, Product X. That Contributor is then a Commercial Contributor. If that Commercial Contributor then makes performance claims, or offers warranties related to Product X, those performance claims and warranties are such Commercial Contributor's responsibility alone. Under this section, the Commercial Contributor would have to defend claims against the other Contributors related to those performance claims and warranties, and if a court requires any other Contributor to pay any damages as a result, the Commercial Contributor must pay those damages.

5. NO WARRANTY

EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, AND TO THE EXTENT PERMITTED BY APPLICABLE LAW, THE PROGRAM IS PROVIDED ON AN "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, EITHER EXPRESS OR IMPLIED INCLUDING, WITHOUT LIMITATION, ANY WARRANTIES OR CONDITIONS OF TITLE, NON-INFRINGEMENT, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE. Each Recipient is solely responsible for determining the appropriateness of using and distributing the Program and assumes all risks associated with its exercise of rights under this Agreement, including but not limited to the risks and costs of program errors, compliance with applicable laws, damage to or loss of data, programs or equipment, and unavailability or interruption of operations.

6. DISCLAIMER OF LIABILITY

EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, AND TO THE EXTENT PERMITTED BY APPLICABLE LAW, NEITHER RECIPIENT NOR ANY CONTRIBUTORS SHALL HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING WITHOUT LIMITATION LOST PROFITS), HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OR DISTRIBUTION OF THE PROGRAM OR THE EXERCISE OF ANY RIGHTS GRANTED HEREUNDER, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

7. GENERAL

If any provision of this Agreement is invalid or unenforceable under applicable law, it shall not affect the validity or enforceability of the remainder of the terms of this Agreement, and without further action by the parties hereto, such provision shall be reformed to the minimum extent necessary to make such provision valid and enforceable.

If Recipient institutes patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Program itself (excluding combinations of the Program with other software or hardware) infringes such Recipient's patent(s), then such Recipient's rights granted under Section 2(b) shall terminate as of the date such litigation is filed.

All Recipient's rights under this Agreement shall terminate if it fails to comply with any of the material terms or conditions of this Agreement and does not cure such failure in a reasonable period of time after becoming aware of such noncompliance. If all Recipient's rights under this Agreement terminate, Recipient agrees to cease use and distribution of the Program as soon as reasonably practicable. However, Recipient's obligations under this Agreement and any licenses granted by Recipient relating to the Program shall continue and survive.

Everyone is permitted to copy and distribute copies of this Agreement, but in order to avoid inconsistency the Agreement is copyrighted and may only be modified in the following manner. The Agreement Steward reserves the right to publish new versions (including revisions) of this Agreement from time to time. No one other than the Agreement Steward has the right to modify this Agreement. The Eclipse Foundation is the initial Agreement Steward. The Eclipse Foundation may assign the responsibility to serve as the Agreement Steward to a suitable separate entity. Each new version of the Agreement will be given a distinguishing version number. The Program (including Contributions) may always be Distributed subject to the version of the Agreement under which it was received. In addition, after a new version of the Agreement is published, Contributor may elect to Distribute the Program (including its Contributions) under the new version.

Except as expressly stated in Sections 2(a) and 2(b) above, Recipient receives no rights or licenses to the intellectual property of any Contributor under this Agreement, whether expressly, by implication, estoppel or otherwise. All rights in the Program not expressly granted under this Agreement are reserved. Nothing in this Agreement is intended to be enforceable by any entity that is not a Contributor or Recipient. No third-party beneficiary rights are created under this Agreement.

ECW - This application supports the ECW data format and ECWP compression protocols. Portions of this computer program are Copyright 2007-2015 Intergraph Corporation. All rights reserved. Creating compressed files using ECW technology is protected by one or more of U.S. Patent No. 6,201,897, No. 6,442,298 and No. 6,633,688.

-LGPL LICENSES-

GNU LESSER GENERAL PUBLIC LICENSE

Version 2.1, February 1999

© 1991, 1999 Free Software Foundation, Inc.

51 Franklin Street, Fifth Floor, Boston, MA 02110-1301 USA

Everyone is permitted to copy and distribute verbatim copies of this license document, but changing it is not allowed.

[This is the first released version of the Lesser GPL. It also counts as the successor of the GNU Library Public License, version 2, hence the version number 2.1.]

Preamble

The licenses for most software are designed to take away your freedom to share and change it. By contrast, the GNU General Public Licenses are intended to guarantee your freedom to share and change free software--to make sure the software is free for all its users.

This license, the Lesser General Public License, applies to some specially designated software packages--typically libraries--of the Free Software Foundation and other authors who decide to use it. You can use it too, but we suggest you first think carefully about whether this license or the ordinary General Public License is the better strategy to use in any particular case, based on the explanations below.

When we speak of free software, we are referring to freedom of use, not price. Our General Public Licenses are designed to make sure that you have the freedom to distribute copies of free software (and charge for this service if you wish); that you receive source code or can get it if you want it; that you can change the software and use pieces of it in new free programs; and that you are informed that you can do these things.

To protect your rights, we need to make restrictions that forbid distributors to deny you these rights or to ask you to surrender these rights. These restrictions translate to certain responsibilities for you if you distribute copies of the library or if you modify it.

For example, if you distribute copies of the library, whether gratis or for a fee, you must give the recipients all the rights that we gave you. You must make sure that they, too, receive or can get the source code. If you link other code with the library, you must provide complete object files to the recipients, so that they can relink them with the library after making changes to the library and recompiling it. And you must show them these terms so they know their rights.

We protect your rights with a two-step method: (1) we copyright the library, and (2) we offer you this license, which gives you legal permission to copy, distribute and/or modify the library.

To protect each distributor, we want to make it very clear that there is no warranty for the free library. Also, if the library is modified by someone else and passed on, the recipients should know that what they have is not the original version, so that the original author's reputation will not be affected by problems that might be introduced by others.

Finally, software patents pose a constant threat to the existence of any free program. We wish to make sure that a company cannot effectively restrict the users of a free program by obtaining a restrictive license from a patent holder. Therefore, we insist that any patent license obtained for a version of the library must be consistent with the full freedom of use specified in this license.

Most GNU software, including some libraries, is covered by the ordinary GNU General Public License. This license, the GNU Lesser General Public License, applies to certain designated libraries, and is quite different from the ordinary General Public License. We use this license for certain libraries in order to permit linking those libraries into non-free programs.

When a program is linked with a library, whether statically or using a shared library, the combination of the two is legally speaking a combined work, a derivative of the original library. The ordinary General Public License therefore permits such linking only if the entire combination fits its criteria of freedom. The Lesser General Public License permits more lax criteria for linking other code with the library.

We call this license the "Lesser" General Public License because it does Less to protect the user's freedom than the ordinary General Public License. It also provides other free software developers Less of an advantage over competing non-free programs. These disadvantages are the reason we use the ordinary General Public License for many libraries. However, the Lesser license provides advantages in certain special circumstances.

For example, on rare occasions, there may be a special need to encourage the widest possible use of a certain library, so that it becomes a de-facto standard. To achieve this, non-free programs must be allowed to use the library. A more frequent case is that a free library does the same job as widely used non-free libraries. In this case, there is little to gain by limiting the free library to free software only, so we use the Lesser General Public License. In other cases, permission to use a particular library in non-free programs enables a greater number of people to use a large body of free software. For example, permission to use the GNU C Library in non-free programs enables many more people to use the whole GNU operating system, as well as its variant, the GNU/Linux operating system.

Although the Lesser General Public License is Less protective of the users' freedom, it does ensure that the user of a program that is linked with the Library has the freedom and the wherewithal to run that program using a modified version of the Library.

The precise terms and conditions for copying, distribution and modification follow. Pay close attention to the difference between a "work based on the library" and a "work that uses the library." The former contains code derived from the library, whereas the latter must be combined with the library in order to run.

GNU LESSER GENERAL PUBLIC LICENSE

TERMS AND CONDITIONS FOR COPYING, DISTRIBUTION AND MODIFICATION

0. This License Agreement applies to any software library or other program which contains a notice placed by the copyright holder or other authorized party saying it may be distributed under the terms of this Lesser General Public License (also called "this License"). Each licensee is addressed as "you."

A "library" means a collection of software functions and/or data prepared so as to be conveniently linked with application programs (which use some of those functions and data) to form executables.

The "Library", below, refers to any such software library or work which has been distributed under these terms. A "work based on the Library" means either the Library or any derivative work under copyright law: that is to say, a work containing the Library or a portion of it, either verbatim or with modifications and/or translated straightforwardly into another language. (Hereinafter, translation is included without limitation in the term "modification.")

"Source code" for a work means the preferred form of the work for making modifications to it. For a library, complete source code means all the source code for all modules it contains, plus any associated interface definition files, plus the scripts used to control compilation and installation of the library.

Activities other than copying, distribution and modification are not covered by this License; they are outside its scope. The act of running a program using the Library is not restricted, and output from such a program is covered only if its contents constitute a work based on the Library (independent of the use of the Library in a tool for writing it). Whether that is true depends on what the Library does and what the program that uses the Library does.

1. You may copy and distribute verbatim copies of the Library's complete source code as you receive it, in any medium, provided that you conspicuously and appropriately publish on each copy an appropriate copyright notice and disclaimer of warranty; keep intact all the notices that refer to this License and to the absence of any warranty; and distribute a copy of this License along with the Library.

You may charge a fee for the physical act of transferring a copy, and you may at your option offer warranty protection in exchange for a fee.

- 2. You may modify your copy or copies of the Library or any portion of it, thus forming a work based on the Library, and copy and distribute such modifications or work under the terms of Section 1 above, provided that you also meet all of these conditions:
 - a. The modified work must itself be a software library.
 - b. You must cause the files modified to carry prominent notices stating that you changed the files and the date of any change.
 - c. You must cause the whole of the work to be licensed at no charge to all third parties under the terms of this License.
 - d. If a facility in the modified Library refers to a function or a table of data to be supplied by an application program that uses the facility, other than as an argument passed when the facility is invoked, then you must make a good faith effort to ensure that, in the event an application does not supply such function or table, the facility still operates, and performs whatever part of its purpose remains meaningful.

(For example, a function in a library to compute square roots has a purpose that is entirely well-defined independent of the application. Therefore, Subsection 2d requires that any application-supplied function or table used by this function must be optional: if the application does not supply it, the square root function must still compute square roots.)

These requirements apply to the modified work as a whole. If identifiable sections of that work are not derived from the Library, and can be reasonably considered independent and separate works in themselves, then this License, and its terms, do not apply to those sections when you distribute them as separate works. But when you distribute the same sections as part of a whole which is a work based on the Library, the distribution of the whole must be on the terms of this License, whose permissions for other licensees extend to the entire whole, and thus to each and every part regardless of who wrote it.

Thus, it is not the intent of this section to claim rights or contest your rights to work written entirely by you; rather, the intent is to exercise the right to control the distribution of derivative or collective works based on the Library.

In addition, mere aggregation of another work not based on the Library with the Library (or with a work based on the Library) on a volume of a storage or distribution medium does not bring the other work under the scope of this License.

3. You may opt to apply the terms of the ordinary GNU General Public License instead of this License to a given copy of the Library. To do this, you must alter all the notices that refer to this License, so that they refer to the ordinary GNU General Public License, version 2, instead of to this License. (If a newer version than version 2 of the ordinary GNU General Public License has appeared, then you can specify that version instead if you wish.) Do not make any other change in these notices.

Once this change is made in a given copy, it is irreversible for that copy, so the ordinary GNU General Public License applies to all subsequent copies and derivative works made from that copy.

This option is useful when you wish to copy part of the code of the Library into a program that is not a library.

4. You may copy and distribute the Library (or a portion or derivative of it, under Section 2) in object code or executable form under the terms of Sections 1 and 2 above provided that you accompany it with the complete corresponding machine-readable source code, which must be distributed under the terms of Sections 1 and 2 above on a medium customarily used for software interchange.

If distribution of object code is made by offering access to copy from a designated place, then offering equivalent access to copy the source code from the same place satisfies the requirement to distribute the source code, even though third parties are not compelled to copy the source along with the object code.

5. A program that contains no derivative of any portion of the Library, but is designed to work with the Library by being compiled or linked with it, is called a "work that uses the Library." Such a work, in isolation, is not a derivative work of the Library, and therefore falls outside the scope of this License.

However, linking a "work that uses the Library" with the Library creates an executable that is a derivative of the Library (because it contains portions of the Library), rather than a "work that uses the library." The executable is therefore covered by this License. Section 6 states terms for distribution of such executables.

When a "work that uses the Library" uses material from a header file that is part of the Library, the object code for the work may be a derivative work of the Library even though the source code is not. Whether this is true is especially

significant if the work can be linked without the Library, or if the work is itself a library. The threshold for this to be true is not precisely defined by law.

If such an object file uses only numerical parameters, data structure layouts and accessors, and small macros and small inline functions (ten lines or less in length), then the use of the object file is unrestricted, regardless of whether it is legally a derivative work. (Executables containing this object code plus portions of the Library will still fall under Section 6.)

Otherwise, if the work is a derivative of the Library, you may distribute the object code for the work under the terms of Section 6. Any executables containing that work also fall under Section 6, whether or not they are linked directly with the Library itself.

6. As an exception to the Sections above, you may also combine or link a "work that uses the Library" with the Library to produce a work containing portions of the Library, and distribute that work under terms of your choice, provided that the terms permit modification of the work for the customer's own use and reverse engineering for debugging such modifications.

You must give prominent notice with each copy of the work that the Library is used in it and that the Library and its use are covered by this License. You must supply a copy of this License. If the work during execution displays copyright notices, you must include the copyright notice for the Library among them, as well as a reference directing the user to the copy of this License. Also, you must do one of these things:

- a. Accompany the work with the complete corresponding machine-readable source code for the Library including whatever changes were used in the work (which must be distributed under Sections 1 and 2 above); and, if the work is an executable linked with the Library, with the complete machine-readable "work that uses the Library", as object code and/or source code, so that the user can modify the Library and then relink to produce a modified executable containing the modified Library. (It is understood that the user who changes the contents of definitions files in the Library will not necessarily be able to recompile the application to use the modified definitions.)
- b. Use a suitable shared library mechanism for linking with the Library. A suitable mechanism is one that (1) uses at run time a copy of the library already present on the user's computer system, rather than copying library functions into the executable, and (2) will operate properly with a modified version of the library, if the user installs one, as long as the modified version is interface-compatible with the version that the work was made with.
- c. Accompany the work with a written offer, valid for at least three years, to give the same user the materials specified in Subsection 6a, above, for a charge no more than the cost of performing this distribution.
- d. If distribution of the work is made by offering access to copy from a designated place, offer equivalent access to copy the above specified materials from the same place.
- e. Verify that the user has already received a copy of these materials or that you have already sent this user a copy.

For an executable, the required form of the "work that uses the Library" must include any data and utility programs needed for reproducing the executable from it. However, as a special exception, the materials to be distributed need not include anything that is normally distributed (in either source or binary form) with the major components (compiler, kernel, and so on) of the operating system on which the executable runs, unless that component itself accompanies the executable.

It may happen that this requirement contradicts the license restrictions of other proprietary libraries that do not normally accompany the operating system. Such a contradiction means you cannot use both them and the Library together in an executable that you distribute.

- 7. You may place library facilities that are a work based on the Library side-by-side in a single library together with other library facilities not covered by this License, and distribute such a combined library, provided that the separate distribution of the work based on the Library and of the other library facilities is otherwise permitted, and provided that you do these two things:
 - a. Accompany the combined library with a copy of the same work based on the Library, uncombined with any other library facilities. This must be distributed under the terms of the Sections above.
 - b. Give prominent notice with the combined library of the fact that part of it is a work based on the Library, and explaining where to find the accompanying uncombined form of the same work.
- 8. You may not copy, modify, sublicense, link with, or distribute the Library except as expressly provided under this License. Any attempt otherwise to copy, modify, sublicense, link with, or distribute the Library is void, and will automatically terminate your rights under this License. However, parties who have received copies, or rights, from you under this License will not have their licenses terminated so long as such parties remain in full compliance.
- 9. You are not required to accept this License, since you have not signed it. However, nothing else grants you permission to modify or distribute the Library or its derivative works. These actions are prohibited by law if you do not accept this License. Therefore, by modifying or distributing the Library (or any work based on the Library), you indicate your acceptance of this License to do so, and all its terms and conditions for copying, distributing or modifying the Library or works based on it.
- 10. Each time you redistribute the Library (or any work based on the Library), the recipient automatically receives a license from the original licensor to copy, distribute, link with or modify the Library subject to these terms and conditions. You may not impose any further restrictions on the recipients' exercise of the rights granted herein. You are not responsible for enforcing compliance by third parties with this License.
- 11. If, as a consequence of a court judgment or allegation of patent infringement or for any other reason (not limited to patent issues), conditions are imposed on you (whether by court order, agreement or otherwise) that contradict the conditions of this License, they do not excuse you from the conditions of this License. If you cannot distribute so as to satisfy simultaneously your obligations under this License and any other pertinent obligations, then as a consequence you may not distribute the Library at all. For example, if a patent license would not permit royalty-free redistribution of the Library by all those who receive copies directly or indirectly through you, then the only way you could satisfy both it and this License would be to refrain entirely from distribution of the Library.

If any portion of this section is held invalid or unenforceable under any particular circumstance, the balance of the section is intended to apply, and the section as a whole is intended to apply in other circumstances.

It is not the purpose of this section to induce you to infringe any patents or other property right claims or to contest validity of any such claims; this section has the sole purpose of protecting the integrity of the free software distribution system which is implemented by public license practices. Many people have made generous contributions to the wide range of software distributed through that system in reliance on consistent application of that system; it is up to the author/donor to decide if he or she is willing to distribute software through any other system and a licensee cannot impose that choice.

This section is intended to make thoroughly clear what is believed to be a consequence of the rest of this License.

- 12. If the distribution and/or use of the Library is restricted in certain countries either by patents or by copyrighted interfaces, the original copyright holder who places the Library under this License may add an explicit geographical distribution limitation excluding those countries, so that distribution is permitted only in or among countries not thus excluded. In such case, this License incorporates the limitation as if written in the body of this License.
- 13. The Free Software Foundation may publish revised and/or new versions of the Lesser General Public License from time to time. Such new versions will be similar in spirit to the present version, but may differ in detail to address new problems or concerns.

Each version is given a distinguishing version number. If the Library specifies a version number of this License which applies to it and "any later version", you have the option of following the terms and conditions either of that version or of any later version published by the Free Software Foundation. If the Library does not specify a license version number, you may choose any version ever published by the Free Software Foundation.

14. If you wish to incorporate parts of the Library into other free programs whose distribution conditions are incompatible with these, write to the author to ask for permission. For software which is copyrighted by the Free Software Foundation, write to the Free Software Foundation; we sometimes make exceptions for this. Our decision will be guided by the two goals of preserving the free status of all derivatives of our free software and of promoting the sharing and reuse of software generally.

NO WARRANTY

- 15. BECAUSE THE LIBRARY IS LICENSED FREE OF CHARGE, THERE IS NO WARRANTY FOR THE LIBRARY, TO THE EXTENT PERMITTED BY APPLICABLE LAW. EXCEPT WHEN OTHERWISE STATED IN WRITING THE COPYRIGHT HOLDERS AND/OR OTHER PARTIES PROVIDE THE LIBRARY "AS IS" WITHOUT WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE. THE ENTIRE RISK AS TO THE QUALITY AND PERFORMANCE OF THE LIBRARY IS WITH YOU. SHOULD THE LIBRARY PROVE DEFECTIVE, YOU ASSUME THE COST OF ALL NECESSARY SERVICING, REPAIR OR CORRECTION.
- 16. IN NO EVENT UNLESS REQUIRED BY APPLICABLE LAW OR AGREED TO IN WRITING WILL ANY COPYRIGHT HOLDER, OR ANY OTHER PARTY WHO MAY MODIFY AND/OR REDISTRIBUTE THE LIBRARY AS PERMITTED ABOVE, BE LIABLE TO YOU FOR DAMAGES, INCLUDING ANY GENERAL, SPECIAL, INCIDENTAL OR CONSEQUENTIAL DAMAGES ARISING OUT OF THE USE OR INABILITY TO USE THE LIBRARY (INCLUDING BUT NOT LIMITED TO LOSS OF DATA OR DATA BEING RENDERED INACCURATE OR LOSSES SUSTAINED BY YOU OR THIRD PARTIES OR A FAILURE OF THE LIBRARY TO OPERATE WITH ANY OTHER SOFTWARE), EVEN IF SUCH HOLDER OR OTHER PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

END OF TERMS AND CONDITIONS

How to Apply These Terms to Your New Libraries

If you develop a new library, and you want it to be of the greatest possible use to the public, we recommend making it free software that everyone can redistribute and change. You can do so by permitting redistribution under these terms (or, alternatively, under the terms of the ordinary General Public License).

To apply these terms, attach the following notices to the library. It is safest to attach them to the start of each source file to most effectively convey the exclusion of warranty; and each file should have at least the "copyright" line and a pointer to where the full notice is found.

<one line to give the library's name and a brief idea of what it does.>

Copyright (C) < year > < name of author >

This library is free software; you can redistribute it and/or modify it under the terms of the GNU Lesser General Public License as published by the Free Software Foundation; either version 2.1 of the License, or (at your option) any later version.

This library is distributed in the hope that it will be useful, but WITHOUT ANY WARRANTY; without even the implied warranty of MERCHANTABILITY or FITNESS FOR A PARTICULAR PURPOSE. See the GNU Lesser General Public License for more details.

You should have received a copy of the GNU Lesser General Public License along with this library; if not, write to the Free Software Foundation, Inc., 51 Franklin Street, Fifth Floor, Boston, MA 02110-1301 USA

Also add information on how to contact you by electronic and paper mail.

You should also get your employer (if you work as a programmer) or your school, if any, to sign a "copyright disclaimer" for the library, if necessary. Here is a sample; alter the names:

Yoyodyne, Inc., hereby disclaims all copyright interest in the library `Frob' (a library for tweaking knobs) written by James Random Hacker.

<signature of Ty Coon>, 1 April 1990

Ty Coon, President of Vice

That's all there is to it!

GNU LESSER GENERAL PUBLIC LICENSE

Version 3, 29 June 2007

Copyright (C) 2007 Free Software Foundation, Inc. http://fsf.org/

Everyone is permitted to copy and distribute verbatim copies of this license document, but changing it is not allowed.

This version of the GNU Lesser General Public License incorporates the terms and conditions of version 3 of the GNU General Public License, supplemented by the additional permissions listed below.

0. Additional Definitions.

As used herein, "this License" refers to version 3 of the GNU Lesser General Public License, and the "GNU GPL" refers to version 3 of the GNU General Public License.

"The Library" refers to a covered work governed by this License, other than an Application or a Combined Work as defined below.

An "Application" is any work that makes use of an interface provided by the Library, but which is not otherwise based on the Library.

Defining a subclass of a class defined by the Library is deemed a mode of using an interface provided by the Library.

A "Combined Work" is a work produced by combining or linking an Application with the Library. The particular version of the Library with which the Combined Work was made is also called the "Linked Version."

The "Minimal Corresponding Source" for a Combined Work means the Corresponding Source for the Combined Work, excluding any source code for portions of the Combined Work that, considered in isolation, are based on the Application, and not on the Linked Version.

The "Corresponding Application Code" for a Combined Work means the object code and/or source code for the Application, including any data and utility programs needed for reproducing the Combined Work from the Application, but excluding the System Libraries of the Combined Work.

1. Exception to Section 3 of the GNU GPL.

You may convey a covered work under sections 3 and 4 of this License without being bound by section 3 of the GNU GPL.

2. Conveying Modified Versions.

If you modify a copy of the Library, and, in your modifications, a facility refers to a function or data to be supplied by an Application that uses the facility (other than as an argument passed when the facility is invoked), then you may convey a copy of the modified version:

- a. under this License, provided that you make a good faith effort to ensure that, in the event an Application does not supply the function or data, the facility still operates, and performs whatever part of its purpose remains meaningful, or
- b. under the GNU GPL, with none of the additional permissions of this License applicable to that copy.

3. Object Code Incorporating Material from Library Header Files.

The object code form of an Application may incorporate material from a header file that is part of the Library. You may convey such object code under terms of your choice, provided that, if the incorporated material is not limited to numerical parameters, data structure layouts and accessors, or small macros, inline functions and templates (ten or fewer lines in length), you do both of the following:

- a. Give prominent notice with each copy of the object code that the Library is used in it and that the Library and its use are covered by this License.
- b. Accompany the object code with a copy of the GNU GPL and this license document.

4. Combined Works.

You may convey a Combined Work under terms of your choice that, taken together, effectively do not restrict modification of the portions of the Library contained in the Combined Work and reverse engineering for debugging such modifications, if you also do each of the following:

- a. Give prominent notice with each copy of the Combined Work that the Library is used in it and that the Library and its use are covered by this License.
- b. Accompany the Combined Work with a copy of the GNU GPL and this license document.
- c. For a Combined Work that displays copyright notices during execution, include the copyright notice for the Library among these notices, as well as a reference directing the user to the copies of the GNU GPL and this license document.

d. Do one of the following:

- O. Convey the Minimal Corresponding Source under the terms of this License, and the Corresponding Application Code in a form suitable for, and under terms that permit, the user to recombine or relink the Application with a modified version of the Linked Version to produce a modified Combined Work, in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.
- 1. Use a suitable shared library mechanism for linking with the Library. A suitable mechanism is one that (a) uses at run time a copy of the Library already present on the user's computer system, and (b) will operate properly with a modified version of the Library that is interface-compatible with the Linked Version.
- e. Provide Installation Information, but only if you would otherwise be required to provide such information under section 6 of the GNU GPL, and only to the extent that such information is necessary to install and execute a modified version of the Combined Work produced by recombining or relinking the Application with a modified version of the Linked Version. (If you use option 4d0, the Installation Information must accompany the Minimal Corresponding Source and Corresponding Application Code. If you use option 4d1, you must provide the Installation Information in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.)

5. Combined Libraries.

You may place library facilities that are a work based on the Library side by side in a single library together with other library facilities that are not Applications and are not covered by this License, and convey such a combined library under terms of your choice, if you do both of the following:

- a. Accompany the combined library with a copy of the same work based on the Library, uncombined with any other library facilities, conveyed under the terms of this License.
- b. Give prominent notice with the combined library that part of it is a work based on the Library, and explaining where to find the accompanying uncombined form of the same work.
- 6. Revised Versions of the GNU Lesser General Public License.

The Free Software Foundation may publish revised and/or new versions of the GNU Lesser General Public License from time to time. Such new versions will be similar in spirit to the present version, but may differ in detail to address new problems or concerns.

Each version is given a distinguishing version number. If the Library as you received it specifies that a certain numbered version of the GNU Lesser General Public License "or any later version" applies to it, you have the option of following the terms and conditions either of that published version or of any later version published by the Free Software Foundation. If the Library as you received it does not specify a version number of the GNU Lesser General Public License, you may choose any version of the GNU Lesser General Public License ever published by the Free Software Foundation.

If the Library as you received it specifies that a proxy can decide whether future versions of the GNU Lesser General Public License shall apply, that proxy's public statement of acceptance of any version is permanent authorization for you to choose that version for the Library.

IDL software uses libraries from the FFmpeg project under the LGPLv2.1. FFmpeg is a trademark of Fabrice Bellard, originator of the FFmpeg project.

GSHHG Database developed and maintained by Paul Wessel, SOEST, University of Hawaii, Honolulu, HI and Walter H. F. Smith, NOAA Geosciences Lab, National Ocean Service, Silver Spring, MD. This dataset is licensed under the GNU Lesser General Public License version 3 (http://www.ecmwf.int/products/data/software/lgpl-3.0.txt) which incorporates the terms and conditions of version 3 of the GNU General Public License (http://www.ecmwf.int/products/data/software/gpl-3.0.txt).

JTS - © 2001 Vivid Solutions, Inc.

JTS 1.13 is open-source and is licensed under the LGPL version 2.1 as published by the Free Software Foundation at http://www.gnu.org/.

Motif - © 1987-2012, The Open Group. All rights reserved.

-END LGPL LICENSES-

Esri - Portions of IDL and ENVI software and/or documentation included intellectual property of Esri and are used herein by permission. © 1999-2018 Environmental Systems Research Institute, Inc. All Rights Reserved.

FFT - © 2000, 2006 Harris Geospatial Solutions, Inc.

Adapted from FFTN algorithm January 2000

FTTN routine itself is copyright © 1995, 1997 Mark Olsen, Queen's University at Kingston.

Permission to use, copy, modify and distribute this software for any purpose without fee is hereby granted, provided that this entire notice is included in all copies of any software which is or includes a copy or modification of this software and in all copies of the supporting documentation for such software.

THIS SOFTWARE IS BEING PROVIDED "AS IS", WITHOUT ANY EXPRESS OR IMPLIED WARRANTY. IN PARTICULAR, NEITHER THE AUTHOR NOR QUEEN'S UNIVERSITY AT KINGSTON MAKES ANY REPRESENTATION OR WARRANTY OF ANY KIND CONCERNING THE MERCHANTABILITY OF THIS SOFTWARE OR ITS FITNESS FOR ANY PARTICULAR PURPOSE.

All of which is to say that you can do what you like with this source code provided you don't try to sell it as your own and you include an unaltered copy of this message (including the copyright_.

It is also implicitly understood that bug fixes and improvements should make their way back to the general Internet community so that everyone benefits.

Flexera - © 2019 Flexera.

Copyright Notice

Copyright © 2019 Flexera.

This software contains proprietary and confidential information and creative works owned by Flexera Software LLC and its licensors, if any. Any use, copying, publication, distribution, display, modification, or transmission of this software in whole or in part in any form or by any means without the prior express written permission of Flexera Software LLC is strictly prohibited. Except where expressly provided by Flexera Software LLC in writing, possession of this software shall not be construed to confer any license or rights under any Flexera Software LLC intellectual property rights, whether by estoppel, implication, or otherwise.

All copies of the technology and related information, if allowed by Flexera Software LLC, must display this notice of copyright and ownership in full.

Intellectual Property

For a list of trademarks and patents that are owned by Flexera Software, see http://www.flexerasoftware.com/intellectual-property. All other brand and product names mentioned in Flexera Software products, product documentation, and marketing materials are the trademarks and registered trademarks of their respective owners.

Restricted Rights Legend

The Software is commercial computer software. If the user or licensee of the Software is an agency, department, or other entity of the United States Government, the use, duplication, reproduction, release, modification, disclosure, or transfer of the

Software, or any related documentation of any kind, including technical data and manuals, is restricted by a license agreement or by the terms of this Agreement in accordance with Federal Acquisition Regulation 12.212 for civilian purposes and Defense Federal Acquisition Regulation Supplement 227.7202 for military purposes. The Software was developed fully at private expense. All other use is prohibited.

FreeType - Portions of IDL software are © 1996-2002, 2006. The FreeType Project (www.freetype.org). All rights reserved.

Introduction

=========

The FreeType Project is distributed in several archive packages; some of them may contain, in addition to the FreeType font engine, various tools and contributions which rely on, or relate to, the FreeType Project.

This license applies to all files found in such packages, and which do not fall under their own explicit license. The license affects thus the FreeType font engine, the test programs, documentation and makefiles, at the very least.

This license was inspired by the BSD, Artistic, and IJG (Independent JPEG Group) licenses, which all encourage inclusion and use of free software in commercial and freeware products alike. As a consequence, its main points are that:

- We don't promise that this software works. However, we will be interested in any kind of bug reports. (`as is' distribution)
- You can use this software for whatever you want, in parts or full form, without having to pay us. ('royalty-free' usage)
- You may not pretend that you wrote this software. If you use it, or only parts of it, in a program, you must acknowledge somewhere in your documentation that you have used the FreeType code. (`credits')

We specifically permit and encourage the inclusion of this software, with or without modifications, in commercial products. We disclaim all warranties covering The FreeType Project and assume no liability related to The FreeType Project.

Finally, many people asked us for a preferred form for a credit/disclaimer to use in compliance with this license. We thus encourage you to use the following text:

....

Portions of this software are copyright © <year> The FreeType Project (www.freetype.org). All rights reserved.

.....

Please replace <year> with the value from the FreeType version you actually use.

Legal Terms

========

0. Definitions

Throughout this license, the terms `package', `FreeType Project', and `FreeType archive' refer to the set of files originally distributed by the authors (David Turner, Robert Wilhelm, and Werner Lemberg) as the `FreeType Project', be they named as alpha, beta or final release.

`You' refers to the licensee, or person using the project, where `using' is a generic term including compiling the project's source code as well as linking it to form a program' or `executable'. This program is referred to as `a program using the FreeType engine'.

This license applies to all files distributed in the original FreeType Project, including all source code, binaries and documentation, unless otherwise stated in the file in its original, unmodified form as distributed in the original archive. If you are unsure whether or not a particular file is covered by this license, you must contact us to verify this.

The FreeType Project is copyright (C) 1996-2000 by David Turner, Robert Wilhelm, and Werner Lemberg. All rights reserved except as specified below.

1. No Warranty

THE FREETYPE PROJECT IS PROVIDED `AS IS' WITHOUT WARRANTY OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE. IN NO EVENT WILL ANY OF THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY DAMAGES CAUSED BY THE USE OR THE INABILITY TO USE, OF THE FREETYPE PROJECT.

2. Redistribution

This license grants a worldwide, royalty-free, perpetual and irrevocable right and license to use, execute, perform, compile, display, copy, create derivative works of, distribute and sublicense the FreeType Project (in both source and object code forms) and derivative works thereof for any purpose; and to authorize others to exercise some or all of the rights granted herein, subject to the following conditions:

- Redistribution of source code must retain this license file (`FTL.TXT') unaltered; any additions, deletions or changes to the original files must be clearly indicated in accompanying documentation. The copyright notices of the unaltered, original files must be preserved in all copies of source files.
- Redistribution in binary form must provide a disclaimer that states that the software is based in part of the work of the
 FreeType Team, in the distribution documentation. We also encourage you to put an URL to the FreeType web page in your
 documentation, though this isn't mandatory.

These conditions apply to any software derived from or based on the FreeType Project, not just the unmodified files. If you use our work, you must acknowledge us. However, no fee need be paid to us.

3. Advertising

Neither the FreeType authors and contributors nor you shall use the name of the other for commercial, advertising, or promotional purposes without specific prior written permission.

We suggest, but do not require, that you use one or more of the following phrases to refer to this software in your documentation or advertising materials: `FreeType Project', `FreeType Engine', `FreeType library', or `FreeType Distribution'.

As you have not signed this license, you are not required to accept it. However, as the FreeType Project is copyrighted material, only this license, or another one contracted with the authors, grants you the right to use, distribute, and modify it. Therefore, by using, distributing, or modifying the FreeType Project, you indicate that you understand and accept all the terms of this license.

4. Contacts

There are two mailing lists related to FreeType:

· freetype@nongnu.org

Discusses general use and applications of FreeType, as well as future and wanted additions to the library and distribution. If you are looking for support, start in this list if you haven't found anything to help you in the documentation.

• freetype-devel@nongnu.org

Discusses bugs, as well as engine internals, design issues, specific licenses, porting, etc.

Our home page can be found at

https://www.freetype.org

GDAL - © 2000, Frank Warmerdam. GDAL/OGR is distributed under the terms of the X11/MIT License.

MIT License

Copyright © <year> <copyright holders>

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice (including the next paragraph) shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

GIF Decode - © 1987, by Steven A. Bennet. GIF and "Graphics Interchange Format" are trademarks of Compuserv, Incorporated, an H&R Block Company.

Permission is given by the author to freely distribute and include this code in any program as long as this credit is given where due.

GPC - © Advanced Interfaces Group, University of Manchester.

GRG2™ © 1979-1997 by Windward Technologies, Inc. and Optimal Methods, Inc.

Copyright © 1979-1997 by Windward Technologies, Inc. and Optimal Methods, Inc.

Windward Technologies, Inc. makes no warranty of any kind with regard to this material, including, but not limited to, the implied warranties of merchantability and fitness for a particular purpose.

Neither Optimal Methods nor Windward Technologies shall be liable for errors contained herein or for incidental, consequential, or other indirect damages in connection with the furnishing, performance, or use of this material.

All rights reserved. No part of code contained on this disk may be copied or reproduced without the prior written consent of Windward Technologies, Inc.

Restricted Rights Legend

Use, duplication, or disclosure by the US Government is subject to restrictions as set forth in FAR 52.227-19, subparagraph ©(i) (ii) of DOD FAR SUPP 252.227-7013, or equivalent government clause for other agencies.

Restricted Rights Notice: The version of the product described in this document is sold under a per user license agreement. Its use, duplication, and disclosure are subject to the restrictions in the license agreement.

Hasp Sentinel-LDK_SDK - © Aladdin Knowledge Systems Ltd. 1985-2011. All Rights Reserved.

HDF - Hierarchical Data Format (HDF) Software Library and Utilities Copyright 2006-2015 by The HDF Group. NCSA Hierarchical Data Format (HDF) Software Library and Utilities Copyright 1988-2006 by the Board of Trustees of the University of Illinois. All rights reserved.

Copyright Notice and License Terms for

Hierarchical Data Format (HDF) Software Library and Utilities

Hierarchical Data Format (HDF) Software Library and Utilities Copyright 2006-2015 by The HDF Group.

NCSA Hierarchical Data Format (HDF) Software Library and Utilities Copyright 1988-2006 by the Board of Trustees of the University of Illinois.

All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted for any purpose (including commercial purposes) provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions, and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions, and the following disclaimer in the documentation and/or materials provided with the distribution.
- 3. In addition, redistributions of modified forms of the source or binary code must carry prominent notices stating that the original code was changed and the date of the change.
- 4. All publications or advertising materials mentioning features or use of this software are asked, but not required, to acknowledge that it was developed by The HDF Group and by the National Center for Supercomputing Applications at the University of Illinois at Urbana-Champaign and credit the contributors.
- 5. Neither the name of The HDF Group, the name of the University, nor the name of any Contributor may be used to endorse or promote products derived from this software without specific prior written permission from The HDF Group, the University, or the Contributor, respectively.

DISCLAIMER:

THIS SOFTWARE IS PROVIDED BY THE HDF GROUP AND THE CONTRIBUTORS "AS IS" WITH NO WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED. In no event shall The HDF Group or the Contributors be liable for any damages suffered by the users arising out of the use of this software, even if advised of the possibility of such damage.

Contributors: National Center for Supercomputing Applications (NCSA) at the University of Illinois, Fortner Software, Unidata Program Center (netCDF), The Independent JPEG Group (JPEG), Jean-loup Gailly and Mark Adler (gzip), and Digital Equipment Corporation (DEC).

HDF EOS Library - @ 1996 Hughes and Applied Research Corporation.

Permission to use, modify, and distribute this software and its documentation for any purpose without fee is hereby granted, provided that the above copyright notice appear in all copies and that both that copyright and this permission notice appear in supporting documentation.

HDF5 - HDF5 (Hierarchical Data Format 5) Software Library and Utilities © 2006-2018, The HDF Group. NCSA HDF5 (Hierarchical Data Format 5) Software Library and Utilities © 1998-2006, The Board of Trustees of the University of Illinois. All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted for any purpose (including commercial purposes)

provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions, and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions, and the following disclaimer in the documentation and/or materials provided with the distribution.
- 3. Neither the name of The HDF Group, the name of the University, nor the name of any Contributor may be used to endorse or promote products derived from this software without specific prior written permission from The HDF Group, the University, or the Contributor, respectively.

DISCLAIMER:

THIS SOFTWARE IS PROVIDED BY THE HDF GROUP AND THE CONTRIBUTORS "AS IS" WITH NO WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED. IN NO EVENT SHALL THE HDF GROUP OR THE CONTRIBUTORS BE LIABLE FOR ANY DAMAGES SUFFERED BY THE USERS ARISING OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

You are under no obligation whatsoever to provide any bug fixes, patches, or upgrades to the features, functionality or performance of the source code ("Enhancements") to anyone; however, if you choose to make your Enhancements available

derivative works, incorporate into other computer software, distribute, and sublicense such enhancements or derivative works thereof, in binary and source code form.
Limited portions of HDF5 were developed by Lawrence Berkeley National Laboratory (LBNL). LBNL's Copyright Notice and Licensing Terms can be found here: COPYING_LBNL_HDF5 file in this directory or at http://support.hdfgroup.org/ftp/HDF5/releases/COPYING_LBNL_HDF5.
Contributors: National Center for Supercomputing Applications (NCSA) at the University of Illinois, Fortner Software, Unidata Program Center (netCDF), The Independent JPEG Group (JPEG), Jean-loup Gailly and Mark Adler (gzip), and Digital Equipment Corporation (DEC).
Portions of HDF5 were developed with support from the Lawrence Berkeley National Laboratory (LBNL) and the United States Department of Energy under Prime Contract No. DE-AC02-05CH11231.

either publicly, or directly to The HDF Group, without imposing a separate written license agreement for such Enhancements,

Portions of HDF5 were developed with support from the University of California, Lawrence Livermore National Laboratory (UC LLNL). The following statement applies to those portions of the product and must be retained in any redistribution of source code, binaries, documentation, and/or accompanying materials:

This work was partially produced at the University of California, Lawrence Livermore National Laboratory (UC LLNL) under contract no. W-7405-ENG-48 (Contract 48) between the U.S. Department of Energy (DOE) and The Regents of the University of California (University) for the operation of UC LLNL.

DISCLAIMER:

This work was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor the University of California nor any of their employees, makes any warranty, express or implied, or assumes any liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately- owned rights. Reference herein to any specific commercial products, process, or service by trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or the University of California, and shall not be used for advertising or product endorsement purposes.

HDF5 is available with the SZIP compression library but SZIP is not part of HDF5 and has separate copyright and license terms. See SZIP Compression in HDF Products (www.hdfgroup.org/doc_resource/SZIP/) for further details.

Heap Management- © Biit Inc. 1996-1998. All Rights Reserved.

Permission to use, copy, modify, and distribute this software, in both binary and source forms, and together with its documentation, for any purpose and without fee is hereby granted, provided that the above copyright notice appear in all copes and that both the copyright notice and this permission notice appear in support documentation, and that the name of Bit not be used in advertising or publicity pertaining to distribution of this software without specific, written permission.

BIIT DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS. IN NO EVENT SHALL BIIT BE LIABLE FOR ANY SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTUOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

IDL Wavelet Toolkit Copyright © 2002, Christopher Torrence.

IMSL - © 2001-2015, Rogue Wave Software, Inc. All Rights Reserved. Protected by copyright and licenses restricting use, copying, distribution and decompilation.

Rogue Wave and IMSL are registered trademarks of Rogue Wave Software, Inc. in the United States and other countries.

Inno Setup

© 1997-2019 Jordan Russell. All rights reserved.

Portions Copyright © 2000-2016 Martijn Laan. All rights reserved.

This software is provided "as-is," without any express or implied warranty. In no event shall the author be held liable for any damages arising from the use of this software.

Permission is granted to anyone to use this software for any purpose, including commercial applications, and to alter and redistribute it, provided that the following conditions are met:

- All redistributions of source code files must retain all copyright notices that are currently in place, and this list of conditions without modification.
- 2. All redistributions in binary form must retain all occurrences of the above copyright notice and web site addresses that are

- currently in place (for example, in the About boxes).
- 3. The origin of this software must not be misrepresented; you must not claim that you wrote the original software. If you use this software to distribute a product, an acknowledgment in the product documentation would be appreciated but is not required.
- 4. Modified versions in source or binary form must be plainly marked as such, and must not be misrepresented as being the original software.

Jordan Russell

jr-2010 AT jrsoftware.org

http://www.jrsoftware.org/

Intel® Math Kernel Library

Intel Simplified Software License (Version April 2018)

© 2018 Intel Corporation.

Use and Redistribution. You may use and redistribute the software (the "Software"), without modification, provided the following conditions are met:

- Redistributions must reproduce the above copyright notice and the following terms of use in the Software and in the documentation and/or other materials provided with the distribution.
- Neither the name of Intel nor the names of its suppliers may be used to endorse or promote products derived from this Software without specific prior written permission.
- No reverse engineering, decompilation, or disassembly of this Software is permitted.

Limited patent license. Intel grants you a world-wide, royalty-free, non-exclusive license under patents it now or hereafter owns or controls to make, have made, use, import, offer to sell and sell ("Utilize") this Software, but solely to the extent that any such patent is necessary to Utilize the Software alone. The patent license shall not apply to any combinations which include this software. No hardware per se is licensed hereunder.

Third party and other Intel programs. "Third Party Programs" are the files listed in the "third-party-programs.txt" text file that is included with the Software and may include Intel programs under separate license terms. Third Party Programs, even if included with the distribution of the Materials, are governed by separate license terms and those license terms solely govern your use of those programs.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE DISCLAIMED. THIS SOFTWARE IS NOT INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE PERSONAL INJURY OR DEATH AND YOU AGREE THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS, COSTS, DAMAGES, EXPENSES, AND ATTORNEYS' FEES ARISING OUT OF ANY SUCH USE, EVEN IF ANY CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE MATERIALS.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE. YOU AGREE TO INDEMNIFY AND HOLD INTEL HARMLESS AGAINST ANY CLAIMS AND EXPENSES RESULTING FROM YOUR USE OR UNAUTHORIZED USE OF THE SOFTWARE.

No support. Intel may make changes to the Software, at any time without notice, and is not obligated to support, update or provide training for the Software.

Termination. Intel may terminate your right to use the Software in the event of your breach of this Agreement and you fail to cure the breach within a reasonable period of time.

Feedback. Should you provide Intel with comments, modifications, corrections, enhancements or other input ("Feedback") related to the Software Intel will be free to use, disclose, reproduce, license or otherwise distribute or exploit the Feedback in its sole discretion without any obligations or restrictions of any kind, including without limitation, intellectual property rights or licensing obligations.

Compliance with laws. You agree to comply with all relevant laws and regulations governing your use, transfer, import or export (or prohibition thereof) of the Software.

Governing law. All disputes will be governed by the laws of the United States of America and the State of Delaware without reference to conflict of law principles and subject to the exclusive jurisdiction of the state or federal courts sitting in the State of Delaware, and each party agrees that it submits to the personal jurisdiction and venue of those courts and waives any objections. The United Nations Convention on Contracts for the International Sale of Goods (1980) is specifically excluded and will not apply to the Software.

*Other names and brands may be claimed as the property of others.

Jas Per - © 2001-2006 Michael David Adams, © 1999-2000 Image Power, Inc., © 1999-2000 University of British Columbia.

Copyright © 2001-2016 Michael David Adams

Copyright © 1999-2000 Image Power, Inc.

Copyright © 1999-2000 The University of British Columbia

All rights reserved.

Permission is hereby granted, free of charge, to any person (the "User") obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

- 1. The above copyright notices and this permission notice (which includes the disclaimer below) shall be included in all copies or substantial portions of the Software.
- 2. The name of a copyright holder shall not be used to endorse or promote products derived from the Software without specific prior written permission.

THIS DISCLAIMER OF WARRANTY CONSTITUTES AN ESSENTIAL PART OF THIS LICENSE. NO USE OF THE SOFTWARE IS AUTHORIZED HEREUNDER EXCEPT UNDER THIS DISCLAIMER. THE SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT OF THIRD PARTY RIGHTS. IN NO EVENT SHALL THE COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, OR ANY SPECIAL INDIRECT OR CONSEQUENTIAL DAMAGES, OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE. NO ASSURANCES ARE PROVIDED BY THE COPYRIGHT HOLDERS THAT THE SOFTWARE DOES NOT INFRINGE THE PATENT OR OTHER INTELLECTUAL PROPERTY RIGHTS OF ANY OTHER ENTITY. EACH COPYRIGHT HOLDER DISCLAIMS ANY LIABILITY TO THE USER FOR CLAIMS BROUGHT BY ANY OTHER ENTITY BASED ON INFRINGEMENT OF INTELLECTUAL PROPERTY RIGHTS OR OTHERWISE. AS A CONDITION TO EXERCISING THE RIGHTS GRANTED HEREUNDER, EACH USER HEREBY ASSUMES SOLE RESPONSIBILITY TO SECURE ANY OTHER INTELLECTUAL PROPERTY RIGHTS NEEDED, IF ANY. THE SOFTWARE IS NOT FAULT-TOLERANT AND IS NOT INTENDED FOR USE IN MISSION-CRITICAL SYSTEMS, SUCH AS THOSE USED IN THE OPERATION OF NUCLEAR FACILITIES, AIRCRAFT NAVIGATION OR COMMUNICATION SYSTEMS, AIR TRAFFIC CONTROL SYSTEMS, DIRECT LIFE SUPPORT MACHINES, OR WEAPONS SYSTEMS, IN WHICH THE FAILURE OF THE SOFTWARE OR SYSTEM COULD LEAD DIRECTLY TO DEATH, PERSONAL INJURY, OR SEVERE PHYSICAL OR ENVIRONMENTAL DAMAGE ("HIGH RISK ACTIVITIES"). THE COPYRIGHT HOLDERS SPECIFICALLY DISCLAIM ANY EXPRESS OR IMPLIED WARRANTY OF FITNESS FOR HIGH RISK ACTIVITIES.

jaxb-api

COMMON DEVELOPMENT AND DISTRIBUTION LICENSE (CDDL)

Version 1.1

- 1. Definitions.
- 1.1. "Contributor" means each individual or entity that creates or contributes to the creation of Modifications.
- 1.2. "Contributor Version" means the combination of the Original Software, prior Modifications used by a Contributor (if any), and the Modifications made by that particular Contributor.
- 1.3. "Covered Software" means (a) the Original Software, or (b) Modifications, or © the combination of files containing Original Software with files containing Modifications, in each case including portions thereof.
- $1.4. \ \hbox{\tt "Executable" means the Covered Software in any form other than Source Code.}$
- 1.5. "Initial Developer" means the individual or entity that first makes Original Software available under this License.
- 1.6. "Larger Work" means a work which combines Covered Software or portions thereof with code not governed by the terms of this License.
- 1.7. "License" means this document.
- 1.8. "Licensable" means having the right to grant, to the maximum extent possible, whether at the time of the initial grant or subsequently acquired, any and all of the rights conveyed herein.
- 1.9. "Modifications" means the Source Code and Executable form of any of the following:
- A. Any file that results from an addition to, deletion from or modification of the contents of a file containing Original Software or previous Modifications;
- B. Any new file that contains any part of the Original Software or previous Modification; or
- C. Any new file that is contributed or otherwise made available under the terms of this License.
- 1.10. "Original Software" means the Source Code and Executable form of computer software code that is originally released under this License.
- 1.11. "Patent Claims" means any patent claim(s), now owned or hereafter acquired, including without limitation, method, process, and apparatus claims, in any patent Licensable by grantor.
- 1.12. "Source Code" means (a) the common form of computer software code in which modifications are made and (b) associated documentation included in or with such code.
- 1.13. "You" (or "Your") means an individual or a legal entity exercising rights under, and complying with all of the terms of, this License. For legal entities, "You" includes any entity which controls, is controlled by, or is under common control with You. For purposes of this definition, "control" means (a) the power, direct or indirect, to cause the direction or management of such entity, whether by contract or otherwise, or (b) ownership of more than fifty percent (50%) of the outstanding shares or beneficial ownership of such entity.
- 2. License Grants.
- 2.1. The Initial Developer Grant.

Conditioned upon Your compliance with Section 3.1 below and subject to third party intellectual property claims, the Initial Developer hereby grants You a world-wide, royalty-free, non-exclusive license:

- (a) under intellectual property rights (other than patent or trademark) Licensable by Initial Developer, to use, reproduce, modify, display, perform, sublicense and distribute the Original Software (or portions thereof), with or without Modifications, and/or as part of a Larger Work; and
- (b) under Patent Claims infringed by the making, using or selling of Original Software, to make, have made, use, practice, sell, and offer for sale, and/or otherwise dispose of the Original Software (or portions thereof).
- © The licenses granted in Sections 2.1(a) and (b) are effective on the date Initial Developer first distributes or otherwise makes the Original Software available to a third party under the terms of this License.
- (d) Notwithstanding Section 2.1(b) above, no patent license is granted: (1) for code that You delete from the Original Software, or (2) for infringements caused by: (i) the modification of the Original Software, or (ii) the combination of the Original Software with other software or devices.

2.2. Contributor Grant.

Conditioned upon Your compliance with Section 3.1 below and subject to third party intellectual property claims, each Contributor hereby grants You a world-wide, royalty-free, non-exclusive license:

- (a) under intellectual property rights (other than patent or trademark) Licensable by Contributor to use, reproduce, modify, display, perform, sublicense and distribute the Modifications created by such Contributor (or portions thereof), either on an unmodified basis, with other Modifications, as Covered Software and/or as part of a Larger Work; and
- (b) under Patent Claims infringed by the making, using, or selling of Modifications made by that Contributor either alone and/or in combination with its Contributor Version (or portions of such combination), to make, use, sell, offer for sale, have made, and/or otherwise dispose of: (1) Modifications made by that Contributor (or portions thereof); and (2) the combination of Modifications made by that Contributor with its Contributor Version (or portions of such combination).
- © The licenses granted in Sections 2.2(a) and 2.2(b) are effective on the date Contributor first distributes or otherwise makes the Modifications available to a third party.
- (d) Notwithstanding Section 2.2(b) above, no patent license is granted: (1) for any code that Contributor has deleted from the Contributor Version; (2) for infringements caused by: (i) third party modifications of Contributor Version, or (ii) the combination of Modifications made by that Contributor with other software (except as part of the Contributor Version) or other devices; or (3) under Patent Claims infringed by Covered Software in the absence of Modifications made by that Contributor.
- 3. Distribution Obligations.
- 3.1. Availability of Source Code.

Any Covered Software that You distribute or otherwise make available in Executable form must also be made available in Source Code form and that Source Code form must be distributed only under the terms of this License. You must include a copy of this License with every copy of the Source Code form of the Covered Software You distribute or otherwise make available. You must inform recipients of any such Covered Software in Executable form as to how they can obtain such Covered Software in Source Code form in a reasonable manner on or through a medium customarily used for software exchange.

3.2. Modifications

The Modifications that You create or to which You contribute are governed by the terms of this License. You represent that You believe Your Modifications are Your original creation(s) and/or You have sufficient rights to grant the rights conveyed by this License.

3.3. Required Notices.

You must include a notice in each of Your Modifications that identifies You as the Contributor of the Modification. You may not remove or alter any copyright, patent or trademark notices contained within the Covered Software, or any notices of licensing or any descriptive text giving attribution to any Contributor or the Initial Developer.

3.4. Application of Additional Terms.

You may not offer or impose any terms on any Covered Software in Source Code form that alters or restricts the applicable version of this License or the recipients' rights hereunder. You may choose to offer, and to charge a fee for, warranty, support, indemnity or liability obligations to one or more recipients of Covered Software. However, you may do so only on Your own behalf, and not on behalf of the Initial Developer or any Contributor. You must make it absolutely clear that any such warranty, support, indemnity or liability obligation is offered by You alone, and You hereby agree to indemnify the Initial Developer and every Contributor for any liability incurred by the Initial Developer or such Contributor as a result of warranty, support, indemnity or liability terms You offer.

3.5. Distribution of Executable Versions.

You may distribute the Executable form of the Covered Software under the terms of this License or under the terms of a license of Your choice, which may contain terms different from this License, provided that You are in compliance with the terms of this License and that the license for the Executable form does not attempt to limit or alter the recipient's rights in the Source Code form from the rights set forth in this License. If You distribute the Covered Software in Executable form under a different license, You must make it absolutely clear that any terms which differ from this License are offered by You alone, not by the Initial Developer or Contributor. You hereby agree to indemnify the Initial Developer and every Contributor for any liability incurred by the Initial Developer or such Contributor as a result of any such terms You offer.

3.6. Larger Works.

You may create a Larger Work by combining Covered Software with other code not governed by the terms of this License and distribute the Larger Work as a single product. In such a case, You must make sure the requirements of this License are fulfilled for the Covered Software.

4. Versions of the License.

4.1. New Versions.

Oracle is the initial license steward and may publish revised and/or new versions of this License from time to time. Each version will be given a distinguishing version number. Except as provided in Section 4.3, no one other than the license steward has the right to modify this License.

4.2. Effect of New Versions.

You may always continue to use, distribute or otherwise make the Covered Software available under the terms of the version of the License under which You originally received the Covered Software. If the Initial Developer includes a notice in the Original Software prohibiting it from being distributed or otherwise made available under any subsequent version of the License, You must distribute and make the Covered Software available under the terms of the version of the License under which You originally received the Covered Software. Otherwise, You may also choose to use, distribute or otherwise make the Covered Software available under the terms of any subsequent version of the License published by the license steward.

4.3 Modified Versions

When You are an Initial Developer and You want to create a new license for Your Original Software, You may create and use a modified version of this License if You: (a) rename the license and remove any references to the name of the license steward (except to note that the license differs from this License); and (b) otherwise make it clear that the license contains terms which differ from this License.

5. DISCLAIMER OF WARRANTY.

COVERED SOFTWARE IS PROVIDED UNDER THIS LICENSE ON AN "AS IS" BASIS, WITHOUT WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED, INCLUDING, WITHOUT LIMITATION, WARRANTIES THAT THE COVERED SOFTWARE IS FREE OF DEFECTS, MERCHANTABLE, FIT FOR A PARTICULAR PURPOSE OR NON-INFRINGING. THE ENTIRE RISK AS TO THE QUALITY AND PERFORMANCE OF THE COVERED SOFTWARE IS WITH YOU. SHOULD ANY COVERED SOFTWARE PROVE DEFECTIVE IN ANY RESPECT, YOU (NOT THE INITIAL DEVELOPER OR ANY OTHER CONTRIBUTOR) ASSUME THE COST OF ANY NECESSARY SERVICING, REPAIR OR CORRECTION. THIS DISCLAIMER OF WARRANTY CONSTITUTES AN ESSENTIAL PART OF THIS LICENSE. NO USE OF ANY COVERED SOFTWARE IS AUTHORIZED HEREUNDER EXCEPT UNDER THIS DISCLAIMER.

6. TERMINATION.

- 6.1. This License and the rights granted hereunder will terminate automatically if You fail to comply with terms herein and fail to cure such breach within 30 days of becoming aware of the breach. Provisions which, by their nature, must remain in effect beyond the termination of this License shall survive.
- 6.2. If You assert a patent infringement claim (excluding declaratory judgment actions) against Initial Developer or a Contributor (the Initial Developer or Contributor against whom You assert such claim is referred to as "Participant") alleging that the Participant Software (meaning the Contributor Version where the Participant is a Contributor or the Original Software where the Participant is the Initial Developer) directly or indirectly infringes any patent, then any and all rights granted directly or indirectly to You by such Participant, the Initial Developer (if the Initial Developer is not the Participant) and all Contributors under Sections 2.1 and/or 2.2 of this License shall, upon 60 days notice from Participant terminate prospectively and automatically at the expiration of such 60 day notice period, unless if within such 60 day period You withdraw Your claim with respect to the Participant Software against such Participant either unilaterally or pursuant to a written agreement with Participant.
- 6.3. If You assert a patent infringement claim against Participant alleging that the Participant Software directly or indirectly infringes any patent where such claim is resolved (such as by license or settlement) prior to the initiation of patent infringement litigation, then the reasonable value of the licenses granted by such Participant under Sections 2.1 or 2.2 shall be taken into account in determining the amount or value of any payment or license.
- 6.4. In the event of termination under Sections 6.1 or 6.2 above, all end user licenses that have been validly granted by You or any distributor hereunder prior to termination (excluding licenses granted to You by any distributor) shall survive termination.

7. LIMITATION OF LIABILITY.

UNDER NO CIRCUMSTANCES AND UNDER NO LEGAL THEORY, WHETHER TORT (INCLUDING NEGLIGENCE), CONTRACT, OR OTHERWISE, SHALL YOU, THE INITIAL DEVELOPER, ANY OTHER CONTRIBUTOR, OR ANY DISTRIBUTOR OF COVERED SOFTWARE, OR ANY SUPPLIER OF ANY OF SUCH PARTIES, BE LIABLE TO ANY PERSON FOR ANY INDIRECT, SPECIAL, INCIDENTAL, OR CONSEQUENTIAL DAMAGES OF ANY CHARACTER INCLUDING, WITHOUT LIMITATION, DAMAGES FOR LOSS OF GOODWILL, WORK STOPPAGE, COMPUTER FAILURE OR MALFUNCTION, OR ANY AND ALL OTHER COMMERCIAL DAMAGES OR LOSSES, EVEN IF SUCH PARTY SHALL HAVE BEEN INFORMED OF THE POSSIBILITY OF SUCH DAMAGES. THIS LIMITATION OF LIABILITY SHALL NOT APPLY TO LIABILITY FOR DEATH OR PERSONAL INJURY RESULTING FROM SUCH PARTY'S NEGLIGENCE TO THE EXTENT APPLICABLE LAW PROHIBITS SUCH LIMITATION. SOME JURISDICTIONS DO NOT ALLOW THE EXCLUSION OR LIMITATION OF INCIDENTAL OR CONSEQUENTIAL DAMAGES, SO THIS EXCLUSION AND LIMITATION MAY NOT APPLY TO YOU.

8. U.S. GOVERNMENT END USERS.

The Covered Software is a "commercial item," as that term is defined in 48 C.F.R. 2.101 (Oct. 1995), consisting of "commercial computer software" (as that term is defined at 48 C.F.R. § 252.227-7014(a)(1)) and "commercial computer software documentation" as such terms are used in 48 C.F.R. 12.212 (Sept. 1995). Consistent with 48 C.F.R. 12.212 and 48 C.F.R. 227.7202-1 through 227.7202-4 (June 1995), all U.S. Government End Users acquire Covered Software with only those rights set forth herein. This U.S. Government Rights clause is in lieu of, and supersedes, any other FAR, DFAR, or other clause or provision that addresses Government rights in computer software under this License.

9. MISCELLANEOUS.

This License represents the complete agreement concerning subject matter hereof. If any provision of this License is held to be unenforceable, such provision shall be reformed only to the extent necessary to make it enforceable. This License shall be governed by the law of the jurisdiction specified in a notice contained within the Original Software (except to the extent

applicable law, if any, provides otherwise), excluding such jurisdiction's conflict-of-law provisions. Any litigation relating to this License shall be subject to the jurisdiction of the courts located in the jurisdiction and venue specified in a notice contained within the Original Software, with the losing party responsible for costs, including, without limitation, court costs and reasonable attorneys' fees and expenses. The application of the United Nations Convention on Contracts for the International Sale of Goods is expressly excluded. Any law or regulation which provides that the language of a contract shall be construed against the drafter shall not apply to this License. You agree that You alone are responsible for compliance with the United States export administration regulations (and the export control laws and regulation of any other countries) when You use, distribute or otherwise make available any Covered Software.

10. RESPONSIBILITY FOR CLAIMS.

As between Initial Developer and the Contributors, each party is responsible for claims and damages arising, directly or indirectly, out of its utilization of rights under this License and You agree to work with Initial Developer and Contributors to distribute such responsibility on an equitable basis. Nothing herein is intended or shall be deemed to constitute any admission of liability.

NOTICE PURSUANT TO SECTION 9 OF THE COMMON DEVELOPMENT AND DISTRIBUTION LICENSE (CDDL)

The code released under the CDDL shall be governed by the laws of the State of California (excluding conflict-of-law provisions). Any litigation relating to this License shall be subject to the jurisdiction of the Federal Courts of the Northern District of California and the state courts of the State of California, with venue lying in Santa Clara County, California.

JP2K - Portions of the IDL software were developed using Unisearch's Kakadu software, for which Harris Geospatial Solutions, Inc. has a commercial license. Kakadu Software. © 2001. The University of New South Wales, UNSW, Sydney NSW 2052, Australia, and Unisearch Ltd., Australia.

libgeotiff - © 1999 Frank Warmerdam

Licensed under the MIT License, which is available in this document here.

libjpeg-turbo - ©2009-2019 D. R. Commander. All Rights Reserved. ©2015 Viktor Szathmáry. All Rights Reserved.

libjpeq-turbo Licenses

libjpeg-turbo is covered by three compatible BSD-style open source licenses:

- The IJG (Independent JPEG Group) License, which is listed in README.ijg

This license applies to the libjpeg API library and associated programs (any code inherited from libjpeg, and any modifications to that code.)

- The Modified (3-clause) BSD License, which is listed below This license covers the TurboJPEG API library and associated programs, as well as the build system.
- The [zlib License](https://opensource.org/licenses/Zlib) This license is a subset of the other two, and it covers the libjpeg-turbo SIMD extensions.

Complying with the libjpeg-turbo Licenses

This section provides a roll-up of the libjpeg-turbo licensing terms, to the best of our understanding.

If you are distributing a modified version of the libjpeg-turbo source, then:

- 1. You cannot alter or remove any existing copyright or license notices from the source.
- **Origin**
- Clause 1 of the IJG License
- Clause 1 of the Modified BSD License
- Clauses 1 and 3 of the zlib License
- 2. You must add your own copyright notice to the header of each source file you modified, so others can tell that you modified that file (if there is not an existing copyright header in that file, then you can simply add a notice stating that you modified the file.)
- **Origin**
- Clause 1 of the IJG License
- Clause 2 of the zlib License
- 3. You must include the IJG README file, and you must not alter any of the copyright or license text in that file.
- **Origin**
- Clause 1 of the IJG License
- 2. If you are distributing only libjpeg-turbo binaries without the source, or if you are distributing an application that statically links with

libjpeg-turbo, then:

1. Your product documentation must include a message stating:

This software is based in part on the work of the Independent JPEG Group.

- **Origin**
- Clause 2 of the IJG license
- 2. If your binary distribution includes or uses the TurboJPEG API, then your product documentation must include the text of the Modified BSD License (see below.)
- **Origin**
- Clause 2 of the Modified BSD License
- 3. You cannot use the name of the IJG or The libjpeq-turbo Project or the contributors thereof in advertising, publicity, etc.
- **Origin**
- IJG License
- Clause 3 of the Modified BSD License
- 4. The IJG and The libjpeg-turbo Project do not warrant libjpeg-turbo to be free of defects, nor do we accept any liability for undesirable consequences resulting from your use of the software.
- **Origin**
- IJG License
- Modified BSD License
- zlib License

The Modified (3-clause) BSD License

Copyright © 2009-2019 D. R. Commander. All Rights Reserved.

Copyright ©2015 Viktor Szathmáry. All Rights Reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- Neither the name of the libjpeg-turbo Project nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS", AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDERS OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Why Three Licenses?

The zlib License could have been used instead of the Modified (3-clause) BSD License, and since the IJG License effectively subsumes the distribution conditions of the zlib License, this would have effectively placed libjpeg-turbo binary distributions under the IJG License. However, the IJG License specifically refers to the Independent JPEG Group and does not extend attribution and endorsement protections to other entities. Thus, it was desirable to choose a license that granted us the same protections for new code that were granted to the IJG for code derived from their software.

libxml2 - © 1998-2012 Daniel Veillard. All Rights Reserved.

Licensed under the MIT License, which is available in this document here.

libtiff - © 1988-1997 Sam Leffler. © 1991-1997 Silicon Graphics, Inc.

Permission to use, copy, modify, distribute, and sell this software and its documentation for any purpose is hereby granted without fee, provided that (i) the above copyright notices and this permission notice appear in all copies of the software and related documentation, and (ii) the names of Sam Leffler and Silicon Graphics may not be used in any advertising or publicity relating to the software without the specific, prior written permission of Sam Leffler and Silicon Graphics.

THE SOFTWARE IS PROVIDED "AS-IS" AND WITHOUT WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR OTHERWISE, INCLUDING WITHOUT LIMITATION, ANY WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE.

IN NO EVENT SHALL SAM LEFFLER OR SILICON GRAPHICS BE LIABLE FOR ANY SPECIAL, INCIDENTAL, INDIRECT OR CONSEQUENTIAL DAMAGES OF ANY KIND, OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR

PROFITS, WHETHER OR NOT ADVISED OF THE POSSIBILITY OF DAMAGE, AND ON ANY THEORY OF LIABILITY, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

libpng - © 1995-2019 The PNG Reference Library Authors. © 2018-2019 Cosmin Truta. © 2000-2002, 2004, 2006-2018 Glenn Randers-Pehrson. © 1996-1997 Andreas Dilger. © 1995-1996 Guy Eric Schalnat, Group 42, Inc.

This copy of the libpng notices is provided for your convenience. In case of any discrepancy between this copy and the notices in the file png.h that is included in the libpng distribution, the latter shall prevail.

COPYRIGHT NOTICE, DISCLAIMER, and LICENSE:

If you modify libping you may insert additional notices immediately following this sentence.

This code is released under the libpng license.

libpng versions 1.2.6, August 15, 2004, through 1.4.5, December 9, 2010, are Copyright © 2004, 2006-2010 Glenn Randers-Pehrson, and are distributed according to the same disclaimer and license as libpng-1.2.5 with the following individual added to the list of Contributing Authors

Cosmin Truta

libpng versions 1.0.7, July 1, 2000, through 1.2.5 - October 3, 2002, are Copyright © 2000-2002 Glenn Randers-Pehrson, and are distributed according to the same disclaimer and license as libpng-1.0.6 with the following individuals added to the list of Contributing Authors

Simon-Pierre Cadieux

Eric S. Raymond

Gilles Vollant

and with the following additions to the disclaimer:

There is no warranty against interference with your enjoyment of the library or against infringement. There is no warranty that our efforts or the library will fulfill any of your particular purposes or needs. This library is provided with all faults, and the entire risk of satisfactory quality, performance, accuracy, and effort is with the user.

libpng versions 0.97, January 1998, through 1.0.6, March 20, 2000, are

Copyright © 1998, 1999 Glenn Randers-Pehrson, and are distributed according to the same disclaimer and license as libpng-0.96, with the following individuals added to the list of Contributing Authors:

Tom Lane

Glenn Randers-Pehrson

Willem van Schaik

libpng versions 0.89, June 1996, through 0.96, May 1997, are

Copyright © 1996, 1997 Andreas Digger

Distributed according to the same disclaimer and license as libpng-0.88, with the following individuals added to the list of Contributing Authors:

John Bowler

Kevin Bracey

Sam Bushell

Magnus Holmgren

Greg Roelofs

Tom Tanner

libpng versions 0.5, May 1995, through 0.88, January 1996, are

Copyright © 1995, 1996 Guy Eric Schalnat, Group 42, Inc.

 $For the purposes of this copyright and \ license, \ "Contributing Authors" \ is \ defined \ as \ the \ following \ set \ of \ individuals:$

Andreas Dilger

Dave Martindale

Guy Eric Schalnat

Paul Schmidt

Tim Wegner

The PNG Reference Library is supplied "AS IS". The Contributing Authors and Group 42, Inc. disclaim all warranties, expressed or implied, including, without limitation, the warranties of merchantability and of fitness for any purpose. The Contributing Authors and Group 42, Inc. assume no liability for direct, indirect, incidental, special, exemplary, or consequential damages, which may result from the use of the PNG Reference Library, even if advised of the possibility of such damage.

Permission is hereby granted to use, copy, modify, and distribute this source code, or portions hereof, for any purpose, without

fee, subject to the following restrictions:

- 1. The origin of this source code must not be misrepresented.
- 2. Altered versions must be plainly marked as such and must not be misrepresented as being the original source.
- 3. This Copyright notice may not be removed or altered from any source or altered source distribution.

The Contributing Authors and Group 42, Inc. specifically permit, without fee, and encourage the use of this source code as a component to supporting the PNG file format in commercial products. If you use this source code in a product, acknowledgment is not required but would be appreciated.

A "png get copyright" function is available, for convenient use in "about" boxes and the like:

printf("%s",png get copyright(NULL));

Also, the PNG logo (in PNG format, of course) is supplied in the files "pngbar.png" and "pngbar.jpg (88x31) and "pngnow.png" (98x31).

Libpng is OSI Certified Open Source Software. OSI Certified Open Source is a certification mark of the Open Source Initiative.

Glenn Randers-Pehrson

glennrp at users.sourceforge.net

December 9, 2010

libXScrnSaver - © © 1992 X Consortium Copyright © 2003 The XFree86 Project, Inc. All Rights Reserved.

Licensed under the MIT License, which is available in this document here.

Merge DICOM Toolkit - © 2019 Merge Healthcare Incorporated, an IBM Company. Unauthorized use, reproduction, or disclosure is prohibited.

Merge DICOM Toolkit (TM) is used under a commercial software license for use in Harris Geospatial Solutions, Inc. products and services only.

Unauthorized use, reproduction, or disclosure is prohibited.

Mersenne Twister - © 1997-2002, Makoto Matsumoto and Takuji Nishimura, All Rights Reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. The names of its contributors may not be used to endorse or promote the products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OR MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OR SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Mesa - © 1999-2007 Brian Paul. All Rights Reserved.

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE Mesa Component Licenses:

Component Location License	

Main Mesa code src/mesa/ MIT

Device drivers src/mesa/drivers/* MIT, generally

Gallium code src/gallium/ MIT

Ext headers include/GL/glext.h Khronos include/GL/glxext.h

GLX client code src/glx/ SGI Free Software License B

C11 thread include/c11/threads*.h Boost (permissive) emulation

Mesa GLU © 1991-2000 Silicon Graphics, Inc. All Rights Reserved.

SGI FREE SOFTWARE LICENSE B

(Version 2.0, Sept. 18, 2008)

Copyright © [dates of first publication] Silicon Graphics, Inc. All Rights Reserved.

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice including the dates of first publication and either this permission notice or a reference to http://oss.sgi.com/projects/FreeB/ shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL SILICON GRAPHICS, INC. BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Except as contained in this notice, the name of Silicon Graphics, Inc. shall not be used in advertising or otherwise to promote the sale, use or other dealings in this Software without prior written authorization from Silicon Graphics, Inc.

MGRS

© 2016 Howard Butler <howard@hobu.co>

Licensed under the MIT License, which is available in this document here.

Microline - © Copyright 1994, 1995, 1996 Microline Software, Inc. All Rights Reserved.

THIS SOFTWARE IS FURNISHED UNDER A LICENSE AND MAY BE COPIED AND USED ONLY IN ACCORDANCE WITH THE TERMS OF THAT LICENSE AND WITH THE INCLUSION OF THE ABOVE COPYRIGHT NOTICE. THIS SOFTWARE AND DOCUMENTATION, AND ITS COPYRIGHTS ARE OWNED BY MICROLINE SOFTWARE AND ARE PROTECTED BY UNITED STATES COPYRIGHT LAWS AND INTERNATIONAL TREATY PROVISIONS.

THE INFORMATION IN THIS SOFTWARE IS SUBJECT TO CHANGE WITHOUT NOTICE AND SHOULD NOT BE CONSTRUED AS A COMMITMENT BY MICROLINE SOFTWARE. THIS SOFTWARE AND REFERENCE MATERIALS ARE PROVIDED "AS IS" WITHOUT WARRANTY AS TO THEIR PERFORMANCE, MERCHANTABILITY, FITNESS FOR ANY PARTICULAR PURPOSE, OR AGAINST INFRINGEMENT. MICROLINE SOFTWARE ASSUMES NO RESPONSIBILITY FOR THE USE OR INABILITY TO USE THIS SOFTWARE.

MICROLINE SOFTWARE SHALL NOT BE LIABLE FOR INDIRECT, SPECIAL OR CONSEQUENTIAL DAMAGES RESULTING FROM THE USE OF THIS PRODUCT. SOME STATES DO NOT ALLOW THE EXCLUSION OR LIMITATION OF INCIDENTAL OR CONSEQUENTIAL DAMAGES, SO THE ABOVE LIMITATIONS MIGHT NOT APPLY TO YOU.

MICROLINE SOFTWARE SHALL HAVE NO LIABILITY OR RESPONSIBILITY FOR SOFTWARE ALTERED, MODIFIED, OR CONVERTED BY YOU OR A THIRD PARTY, DAMAGES RESULTING FROM ACCIDENT, ABUSE OR MISAPPLICATION, OR FOR PROBLEMS DUE TO THE MALFUNCTION OF YOUR EQUIPMENT OR SOFTWARE NOT SUPPLIED BY MICROLINE SOFTWARE.

U.S. GOVERNMENT RESTRICTED RIGHTS

This Software and documentation are provided with RESTRICTED RIGHTS. Use, duplication or disclosure by the Government is subject to restrictions as set forth in subparagraph ©(1) of the Rights in Technical Data and Computer Software Clause at DFARS 252.227-7013 or subparagraphs ©(1)(ii) and (2) of Commercial Computer Software - Restricted Rights at 48 CFR 52.227-19, as applicable, supplier is Microline Software, 41 Sutter St Suite 1374, San Francisco, CA 94104.

microsoft-vcredist - © Microsoft Corporation. All rights reserved.

MICROSOFT SOFTWARE LICENSE TERMS

MICROSOFT VISUAL C++ 2019 RUNTIME

These license terms are an agreement between Microsoft Corporation (or based on where you live, one of its affiliates) and you. They apply to the software named above. The terms also apply to any Microsoft services or updates for the software, except to the extent those have different terms.

IF YOU COMPLY WITH THESE LICENSE TERMS, YOU HAVE THE RIGHTS BELOW.

INSTALLATION AND USE RIGHTS.

* You may install and use any number of copies of the software.

TERMS FOR SPECIFIC COMPONENTS.

- * Microsoft Platforms. The software may include components from Microsoft Windows; Microsoft Windows Server; Microsoft SQL Server; Microsoft Exchange; Microsoft Office; and Microsoft SharePoint. These components are governed by separate agreements and their own product support policies, as described in the Microsoft "Licenses" folder accompanying the software, except that, if license terms for those components are also included in the associated installation directory, those license terms control.
- * Third Party Components. The software may include third party components with separate legal notices or governed by other

agreements, as may be described in the ThirdPartyNotices file(s) accompanying the software.

SCOPE OF LICENSE.

The software is licensed, not sold. This agreement only gives you some rights to use the software. Microsoft reserves all other rights. Unless applicable law gives you more rights despite this limitation, you may use the software only as expressly permitted in this agreement. In doing so, you must comply with any technical limitations in the software that only allow you to use it in certain ways. You may not

- * work around any technical limitations in the software;
- * reverse engineer, decompile or disassemble the software, or otherwise attempt to derive the source code for the software except, and only to the extent required by third party licensing terms governing the use of certain open source components that may be included in the software;
- * remove, minimize, block or modify any notices of Microsoft or its suppliers in the software;
- * use the software in any way that is against the law; or
- * share, publish, rent or lease the software, or provide the software as a stand-alone offering for others to use, or transfer the software or this agreement to any third party.

EXPORT RESTRICTIONS.

You must comply with all domestic and international export laws and regulations that apply to the software, which include restrictions on destinations, end users, and end use. For further information on export restrictions, visit www.microsoft.com/exporting http://www.microsoft.com/exporting>.

SUPPORT SERVICES.

Because this software is "as is," we may not provide support services for it.

ENTIRE AGREEMENT.

This agreement, and the terms for supplements, updates, Internet-based services and support services that you use, are the entire agreement for the software and support services.

APPLICABLE LAW.

If you acquired the software in the United States, Washington law applies to interpretation of and claims for breach of this agreement, and the laws of the state where you live apply to all other claims. If you acquired the software in any other country, its laws apply.

CONSUMER RIGHTS; REGIONAL VARIATIONS.

This agreement describes certain legal rights. You may have other rights, including consumer rights, under the laws of your state or country. Separate and apart from your relationship with Microsoft, you may also have rights with respect to the party from which you acquired the software. This agreement does not change those other rights if the laws of your state or country do not permit it to do so. For example, if you acquired the software in one of the below regions, or mandatory country law applies, then the following provisions apply to you:

- * Australia. You have statutory guarantees under the Australian Consumer Law and nothing in this agreement is intended to affect those rights.
- * Canada. If you acquired this software in Canada, you may stop receiving updates by turning off the automatic update feature, disconnecting your device from the Internet (if and when you re-connect to the Internet, however, the software will resume checking for and installing updates), or uninstalling the software. The product documentation, if any, may also specify how to turn off updates for your specific device or software.
- * Germany and Austria.
- (i) Warranty. The properly licensed software will perform substantially as described in any Microsoft materials that accompany the software. However, Microsoft gives no contractual guarantee in relation to the licensed software.
- (ii) Limitation of Liability. In case of intentional conduct, gross negligence, claims based on the Product Liability Act, as well as, in case of death or personal or physical injury, Microsoft is liable according to the statutory law.
- * Subject to the foregoing clause (ii), Microsoft will only be liable for slight negligence if Microsoft is in breach of such material contractual obligations, the fulfillment of which facilitate the due performance of this agreement, the breach of which would endanger the purpose of this agreement and the compliance with which a party may constantly trust in (so-called "cardinal obligations"). In other cases of slight negligence, Microsoft will not be liable for slight negligence.

DISCLAIMER OF WARRANTY.

THE SOFTWARE IS LICENSED "AS-IS." YOU BEAR THE RISK OF USING IT. MICROSOFT GIVES NO EXPRESS WARRANTIES, GUARANTEES OR CONDITIONS. TO THE EXTENT PERMITTED UNDER YOUR LOCAL LAWS, MICROSOFT EXCLUDES THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NON-INFRINGEMENT.

LIMITATION ON AND EXCLUSION OF DAMAGES.

YOU CAN RECOVER FROM MICROSOFT AND ITS SUPPLIERS ONLY DIRECT DAMAGES UP TO U.S. \$5.00. YOU CANNOT RECOVER ANY OTHER DAMAGES, INCLUDING CONSEQUENTIAL, LOST PROFITS, SPECIAL, INDIRECT OR INCIDENTAL DAMAGES.

This limitation applies to (a) anything related to the software, services, content (including code) on third party Internet sites, or third party applications; and (b) claims for breach of contract, breach of warranty, guarantee or condition, strict liability, negligence, or other tort to the extent permitted by applicable law.

It also applies even if Microsoft knew or should have known about the possibility of the damages. The above limitation or exclusion may not apply to you because your country may not allow the exclusion or limitation of incidental, consequential or other damages.

MPEG-2 - © 1996 MPEG Software Simulation Group.

Copyright © 1996, MPEG Software Simulation Group. All Rights Reserved.

Disclaimer of Warranty

These software programs are available to the user without any license fee or royalty on an "as is" basis. The MPEG Software Simulation Group disclaims any and all warranties, whether express, implied, or statuary, including any implied warranties or merchantability or of fitness for a particular purpose. In no event shall the copyright-holder be liable for any incidental, punitive, or consequential damages of any kind whatsoever arising from the use of these programs.

This disclaimer of warranty extends to the user of these programs and user's customers, employees, agents, transferees, successors, and assigns.

The MPEG Software Simulation Group does not represent or warrant that the programs furnished hereunder are free of infringement of any third-party patents.

Commercial implementations of MPEG-1 and MPEG-2 video, including shareware, are subject to royalty fees to patent holders. Many of these patents are general enough such that they are unavoidable regardless of implementation design.

NCOM - Portions of the IDL software use NCOM procedures and functions. Harris Geospatial Solutions, Inc..

NetPBM - © 1992 by David W. Sanderson. © 1990 by Mark W. Snitily. © 1989 by Jeff Pskanzer.

Permission to use, copy, modify, and distribute this software and its documentation for any purpose and without fee is hereby granted, provided that the above copyright notice appears in all copies and that both the copyright notice and this permission notice appear in supporting documentation.

This software is provided 'as is' without express or implied warranty.

NewtonSoft.Json - © 2007 James Newton-King

Licensed under the MIT License, which is available in this document here.

James Newton-King Url: https://www.nuget.org/packages/Newtonsoft.Json/9.0.1

Numerical Recipes™ is a trademark of Numerical Recipes Software. © Numerical Recipes Software 1988-2002. Numerical Recipes routines are used by permission. Numerical Recipes™ in C: The Art of Scientific Computing. Cambridge University Press. www.nr.com.

OpenSSL - © 1998-2018 The OpenSSL Project. All Rights Reserved. © 1995-1998 Eric Young (eay@cryptsoft.com). All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. All advertising materials mentioning features or use of this software must display the following acknowledgment:
 - "This product includes software developed by the OpenSSL Project for use in the OpenSSL Toolkit. (http://www.openssl.org/)"
- 4. The names "OpenSSL Toolkit" and "OpenSSL Project" must not be used to endorse or promote products derived from this software without prior written permission. For written permission, please contact openssl-core@openssl.org.
- 5. Products derived from this software may not be called "OpenSSL" nor may "OpenSSL" appear in their names without prior written permission of the OpenSSL Project.
- $6. \ \ Redistributions \ of \ any \ form \ \ what so ever \ must \ retain \ the \ following \ acknowledgment:$

"This product includes software developed by the OpenSSL Project for use in the OpenSSL Toolkit (http://www.openssl.org/)"

THIS SOFTWARE IS PROVIDED BY THE OPENSL PROJECT ``AS IS" AND ANY EXPRESSED OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE OPENSL PROJECT OR ITS CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

This product includes cryptographic software written by Eric Young (eay@cryptsoft.com). This product includes software written by Tim Hudson (tjh@cryptsoft.com).

Original SSLeay Licen	ise
Copyright (C) 1995-19	998 Eric Young (eay@cryptsoft.com)

0 : : 1001

All rights reserved.

This package is an SSL implementation written by Eric Young (eay@cryptsoft.com).

The implementation was written so as to conform with Netscapes SSL.

This library is free for commercial and non-commercial use as long as the following conditions are adhered to. The following conditions apply to all code found in this distribution, be it the RC4, RSA, lhash, DES, etc., code; not just the SSL code. The SSL documentation included with this distribution is covered by the same copyright terms except that the holder is Tim Hudson (tjh@cryptsoft.com).

Copyright remains Eric Young's, and as such any Copyright notices in the code are not to be removed.

If this package is used in a product, Eric Young should be given attribution as the author of the parts of the library used.

This can be in the form of a textual message at program startup or in documentation (online or textual) provided with the package.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. All advertising materials mentioning features or use of this software must display the following acknowledgment:

"This product includes cryptographic software written by Eric Young (eay@cryptsoft.com)"

The word 'cryptographic' can be left out if the routines from the library being used are not cryptographic related :-).

4. If you include any Windows specific code (or a derivative thereof) from the apps directory (application code) you must include an acknowledgment:

"This product includes software written by Tim Hudson (tjh@cryptsoft.com)"

THIS SOFTWARE IS PROVIDED BY ERIC YOUNG "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE AUTHOR OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

The license and distribution terms for any publicly available version or derivative of this code cannot be changed. i.e. this code cannot simply be copied and put under another distribution license (including the GNU Public License).

Perforce SCM Plug-in - © Perforce Software 2003, 2016. All rights reserved.

Progress DataDirect

- Progress DataDirect Connect/Connect64 for ODBC Copyright © 2020 Progress Software Corporation and/or one of its subsidiaries or affiliates. All rights reserved.
- Progress DataDirect Connect/Connect64 for ODBC Doc Copyright © 2020 Progress Software Corporation and/or one of its subsidiaries or affiliates. All rights reserved.

READ THIS END USER LICENSE AGREEMENT ("EULA") BEFORE INSTALLING OR USING THE PRODUCT TO WHICH THIS EULA APPLIES. BY ACCEPTING THIS EULA, COMPLETING THE REGISTRATION PROCESS, AND/OR INSTALLING OR USING THE PRODUCT, YOU AGREE ON BEHALF OF YOURSELF AND YOUR COMPANY (IF APPLICABLE) TO THE TERMS BELOW. IF YOU DO NOT AGREE WITH THESE TERMS, OR DO NOT HAVE THE AUTHORITY TO BIND YOUR COMPANY, DO NOT INSTALL, REGISTER FOR OR USE THE PRODUCT, AND DESTROY OR RETURN ALL COPIES OF THE PRODUCT. ONCE YOU HAVE DONE THIS, YOU MAY REQUEST FROM THE POINT OF PURCHASE A FULL REFUND OF THE LICENSE FEES, IF ANY, PAID FOR THE PRODUCT (OR, IF THE PRODUCT IS PROVIDED TO YOU AS A HOSTED SERVICE, A REFUND OF THE PREPAID SERVICE FEES FOR THE REMAINDER OF THE SUBSCRIPTION PERIOD OF THE PRODUCT. SUCH REQUEST MUST BE COMPLETED WITHIN THIRTY (30) DAYS OF DELIVERY OF THE PRODUCT TO YOU. UNLESS OTHERWISE SPECIFIED IN THIS EULA, PROGRESS SOFTWARE CORPORATION IS THE LICENSOR OF THE PRODUCT. THE LICENSOR MAY BE REFERRED TO HEREIN AS "Licensor", "we", "us", or "our". IF YOU ARE AGREEING TO THIS EULA ON BEHALF OF YOURSELF IN YOUR INDIVIDUAL CAPACITY, THEN YOU ARE THE LICENSEE AND YOU MAY BE REFERRED TO HEREIN AS "Licensee", "you", or "your". IF YOU ARE AGREEING TO THIS EULA ON BEHALF OF YOURSELF IN YOUR COMPANY, THEN YOUR

This EULA includes the following sections:

- 1. GENERAL TERMS AND CONDITIONS these terms apply to all Products;
- 2.A. TERMS FOR ON-PREMISE PRODUCTS these terms apply to Products that you or Permitted Third Parties install on computers;
- 2.B. TERMS FOR HOSTED SERVICES these terms apply to Products that we host;
- 3. PRODUCT FAMILY SPECIFIC TERMS these terms apply to all Products that are part of the family of Products referenced in this section; and
- 4. PRODUCT SPECIFIC TERMS these terms apply to specific Products referenced in this section.

1. GENERAL TERMS AND CONDITIONS

1.1. Definitions.

- 1.1.1. "Affiliate" means any legal entity that directly or indirectly controls, is controlled by, or is under common control with you or us. For the purposes of this definition, "control" means ownership, directly or indirectly, of more than fifty percent (50%) of the voting shares or other equity interest in an entity.
- 1.1.2. "Applicable Laws" means national, federal, state, and local laws, rules, and regulations including, without limitation, those laws and regulations relating to data privacy and security in each applicable jurisdiction.
- 1.1.3. "Authorized Reseller" means a third party who is not our Affiliate and who is authorized by us or our Affiliate to resell the Product.
- 1.1.4. "Authorized User" means you, your employee or a third-party consultant or agent that you authorize to use the Product for your benefit in accordance with section 1.2.3 (Third Party Use).
- 1.1.5. "Documentation" means any technical instructions or materials describing the operation of the Product made available to you (electronically or otherwise) by us for use with the Product, expressly excluding any user blogs, reviews or forums.
- 1.1.6. "Hosted Services" means computer software program(s), content and related services provided by us on a software-as-aservice basis through computers we or our Affiliates or our respective contractors (including cloud infrastructure suppliers) control.
- 1.1.7. "Intellectual Property Rights" means any and all current and future (a) rights associated with works of authorship, including copyrights, mask work rights, and moral rights; (b) trademark or service mark rights; © trade secret rights; (d) patents, patent rights, and industrial property rights; (e) layout design rights, design rights, and other proprietary rights of every kind and nature other than trademarks, service marks, trade dress, and similar rights; and (f) registrations, applications, renewals, extensions, or reissues of any of (a) to (e), in each case, in any jurisdiction throughout the world.
- 1.1.8. "On-Premise Product(s)" means computer software program(s) provided to you to download, install and use on computer(s) controlled directly or indirectly by you.
- 1.1.9. "Order" means a written or electronic order document entered into between you and us (or our Affiliate or an Authorized Reseller) for the Product. Unless an Order says something different, each Order will be governed by the terms of this EULA and include the name of the Product being licensed and any usage limitations, applicable fees, and any other details related to the transaction.
- 1.1.10. "Our Technology" means any software, code, tools, libraries, scripts, application programming interfaces, templates, algorithms, data science recipes (including any source code for data science recipes and any modifications to such source code), data science workflows, user interfaces, links, proprietary methods and systems, know-how, trade secrets, techniques, designs, inventions, and other tangible or intangible technical material, information and works of authorship underlying or otherwise used to make available the Product, including, without limitation, all Intellectual Property Rights therein and thereto.
- 1.1.11. "Permitted Third Party" has the meaning given in section 1.2.3 (Third Party Use).
- 1.1.12. "Product" means the On-Premise Product(s) or Hosted Services, as applicable, identified in an Order, and any Updates.
- 1.1.13. "Update" means any update, enhancement, error correction, modification or new release to the Product that we make available to you.
- 1.2. General License Terms, Restrictions and Order of Precedence.
- 1.2.1. General License Terms. The Product is licensed, not sold, to you by us under the terms of this EULA and the Order. The scope of license granted by us to you for the Product is set out in section 3 (Product Family Specific Terms) and section 4 (Product Specific Terms).
- 1.2.2. Authorized Users. Anything your Authorized Users do or fail to do will be considered your act or omission, and you accept full responsibility for any such act or omission to the extent you would be liable if it were your act or omission.
- 1.2.3. Third Party Use. You may allow your agents, contractors and outsourcing service providers (each a "Permitted Third Party") to use the Product(s) licensed to you hereunder solely for your benefit in accordance with the terms of this EULA and you are responsible for any such Permitted Third Party's compliance with this EULA in such use. Any breach by any Permitted Third Party of the terms of this EULA will be considered your breach.
- 1.2.4. Restrictions. Except as otherwise expressly permitted in this EULA, you will not (and will not allow any of your Affiliates or any third party to):
- (a) copy, modify, adapt, translate, or otherwise create derivative works of the Product, Documentation, or any software, services, or other technology of third party vendor(s) or hosting provider(s) that we or our Affiliate engage;
- (b) disassemble, decompile or "unlock", decode or otherwise reverse translate or engineer, or attempt in any manner to reconstruct or discover the source code or underlying structure, ideas, or algorithms of the Product except as expressly permitted by law in effect in the jurisdiction in which you are located; © rent, lease, sell, distribute, pledge, assign, sublicense or otherwise transfer or encumber rights to the Product;
- (d) make the Product available on a timesharing or service bureau basis or otherwise allow any third party to use or access the Product;
- (e) remove or modify any proprietary notices, legends, or labels on the Product or Documentation;
- (f) use or access the Product in a manner that: (i) violates any Applicable Laws; (ii) violates the rights of any third party; (iii) purports to subject us or our Affiliates to any other obligations; (iv) could be fraudulent; or (v) is not permitted under this EULA;
- (g) use the Product to develop, test, support or market products that are competitive with and/or provide similar functionality to the Product; or

- (h) permit your Affiliates to access or use the Product unless specifically authorized elsewhere in this EULA or the Order.
- 1.2.5. Limitations on Evaluation or Trial Licenses. If the Product is licensed to you on an evaluation or trial basis, then you may use the Product only for such purposes until the earlier of: (a) the end of the evaluation period, if any, specified in the Order, this EULA or otherwise communicated by us to you at the time of delivery; or (b) the start date of a paid for license to the Product; or © termination in accordance with the terms of this EULA. You may not extend the evaluation period by uninstalling and re-installing the Product(s) or by any other means other than our written consent. You must not use the Product in a production environment. You will be required to pay for a license for the Product at our then applicable license price if you continue to use the Product, whether in a production or non-production environment, after the evaluation license expires or terminates, and the terms and conditions of the EULA in effect at that time will apply to your continued use of the Product. A Product licensed to you on an evaluation or trial basis may be subject to one or more usage limits specified in section 3 (Product Family Specific Terms), section 4 (Product Specific Terms), the Order or otherwise communicated at the time of delivery (including posting of such limits at the location where you download the Product for evaluation). We may, at our sole discretion, decide whether to offer any maintenance and support for the Product during the evaluation period, and to include any conditions or limits on such maintenance and support. You may not circumvent any technical limitations included in the Product licensed to you on an evaluation or trial basis.
- 1.2.6. Redistribution. If the Order or section 3 (Product Family Specific Terms) or section 4 (Product Specific Terms) grants you the express right to redistribute or offer access to all or a portion of the Product ("Redistributables"), then, in conjunction with any such grant, you must comply with any limitations or requirements specified in the Order, section 3 (Product Family Specific Terms) or section 4 (Product Specific Terms), as applicable, and you must distribute or offer access to the Redistributables subject to a license agreement or terms of use between you and each third party receiving or accessing the Redistributables ("your customer") that: (a) protects our interests consistent with the terms contained in this EULA, (b) prohibits your customer from any further distribution of the Redistributables (unless expressly permitted pursuant to section 3 (Product Family Specific Terms) or section 4 (Product Specific Terms)), © includes a limitation of damages clause that, to the maximum extent permitted by applicable law, disclaims on behalf of us, our Affiliates or our or their respective licensors, suppliers or Authorized Resellers, liability for any and all damages, whether direct, special, incidental or consequential damages, (d) contains terms substantially similar to those in subparts (a) through (g) of section 1.2.4 (Restrictions), section 1.5.1 (Export Compliance) and section 1.5.2 (U.S. Government Customers), and (e) includes a notice substantially similar to section 1.2.7 (Third Party Notices).
- 1.2.7. Third Party Notices. The Product may contain or be accompanied by certain third-party components which are subject to additional restrictions. These components, are identified in, and subject to, special license terms and conditions which, in the case of On-Premise Product(s), are set out in the "readme.txt" file, the "notices.txt" file, or the "Third Party Software" file accompanying the Product or portions thereof, and in the case of Hosted Services, are set out in the third-party license agreement or notices that comes with the third-party component or is otherwise provided on the web page on which such third-party component is made available ("Special Notices"). The Special Notices include important licensing and warranty information and disclaimers. Unless otherwise expressly stated for a given third-party component, all such third-party components may be used solely in connection with the use of the Product subject to and in accordance with the terms and conditions of this EULA and the Special Notices. In the event of conflict between the Special Notices and the other portions of this EULA, the Special Notices will take precedence (but solely with respect to the third-party component(s) to which the Special Notice relates).
- 1.2.8. Order of Precedence between EULA and Order. If there is any conflict between the terms and conditions in the Order and the terms and conditions of this EULA, or if the Order changes any of the terms of this EULA, the terms and conditions of the Order will apply, except if the Order is between you and an Authorized Reseller, or the Order is issued/generated by you. In the case where the Order is between you and an Authorized Reseller, the terms of the Order will apply subject to the following: (a) any terms and conditions in the Order imposing obligations on the Authorized Reseller that are in addition to or different from the obligations we have to you pursuant to this EULA will be born solely by the Authorized Reseller and our obligations to you and limits on our liability will be governed solely by the terms and conditions of this EULA and (b) any terms and conditions that conflict with or would otherwise alter any of the following under this EULA will have no effect unless expressly agreed to in a written instrument executed by us: our ownership rights, yours and our confidentiality obligations, your export compliance obligations, limitations on your rights as a U.S. Government customer (if applicable), our audit rights, restrictions on your right to assign, our publicity rights or governing law and jurisdiction. In cases where the Order is issued/generated by you, the terms and conditions of Section 1.19.2. of this EULA, governing a purchase order or other document you supply in connection with this EULA, shall apply to such Order.
- 1.2.9. Order of Precedence within EULA. If there is any conflict among the terms and conditions of this EULA, or if a section changes the terms of another section within this EULA, the order of precedence will be as follows: first, section 4 (Product Specific Terms) (if any); second, section 3 (Product Family Specific Terms) (if any); third, section 2.A (Terms for On-Premise Products) and/or section 2.B (Terms for Hosted Services), as applicable; and fourth and finally, section 1 (General Terms and Conditions).
- 1.3. License Types.
- 1.3.1. Overview of License Types. The license type for the Product will, unless otherwise specified in this EULA, be one of the following license types: perpetual, term or subscription. This will be confirmed in the Order or will be the default license type listed in section 3 (Product Family Specific Terms) or section 4 (Product Specific Terms).
- 1.3.2. Perpetual License Type. Your license to use the Product will continue in perpetuity unless earlier terminated in accordance with the terms of this EULA.
- 1.3.3. Term License Type. Your license to use the Product will continue until the expiration of the term identified in the Order unless earlier terminated in accordance with the terms of this EULA. If we continue to make the Product generally available to our customers, you may purchase a new term license for the Product from us or our Authorized Reseller.
- 1.3.4. Subscription License Type. Your license to use the Product will continue until the expiration of the subscription period identified in the Order unless earlier terminated in accordance with the terms of this EULA. The procedure for renewing your license to the Product is set out in section 3 (Product Family Specific Terms) or section 4 (Product Specific Terms). If you upgrade your subscription to the Product, the upgrade will take effect immediately and you will be charged and must pay the applicable fee, and the term of your then-current subscription period may be extended, as described at the time you upgrade. You may not downgrade a subscription to the Product.
- 1.4. Our Business Principles. We will apply the principles set out in our Code of Conduct and Business Ethics (published on our website at http://investors.progress.com/governance.cfm) in our performance under this EULA.

- 1.5. Export Compliance and U.S. Government Customers.
- 1.5.1. Export Compliance. Export laws and regulations of the United States and any other relevant local export laws and regulations apply to the Products. You agree that such export control laws, including, without limitation, the U.S. Export Administration Act and its associated regulations, govern your use of the Product (including technical data), and you agree to comply with all such export laws and regulations (including "deemed export" and "deemed re-export" regulations). You agree that no data, information and/or Product (or direct product thereof) will be exported, directly or indirectly, in violation of these laws, or will be used for any purpose prohibited by these laws including, without limitation, nuclear, chemical, or biological weapons proliferation, or development of missile technology.
- 1.5.2. U.S. Government Customers. If the Product is being acquired by or on behalf of the U.S. Government or by a U.S. Government prime contractor or subcontractor (at any tier), then the U.S. Government's rights in the Product will be only as set out herein. The Product and Documentation are "commercial items" as that term is defined at 48 C.F.R. 2.101, consisting of "commercial computer software" and "commercial software documentation" as such terms are used in 48 C.F.R. 12.212. Consistent with 48 C.F.R. 12.212 and 48 C.F.R. 227.7202-1 through 227.7202-4, all U.S. Government end users acquire the Product and such Documentation with only those rights set out herein.

1.6. IP Ownership and Feedback.

- 1.6.1. IP Ownership. The Product, Our Technology, Documentation, and all other current or future intellectual property developed by us or our Affiliates, and all worldwide Intellectual Property Rights in each of the foregoing and all Updates, upgrades, enhancements, new versions, releases, corrections, and other modifications thereto and derivative works thereof, are the exclusive property of us or our Affiliates or our or their licensors or suppliers. Except for the rights and licenses expressly granted herein, all such rights are reserved by us and our Affiliates and our or their licensors and suppliers. All title and Intellectual Property Rights in and to the content that may be accessed through use of the Product is the property of the respective content owner and may be protected by applicable copyright or other intellectual property laws and treaties. This EULA grants you no rights to use such content.
- 1.6.2. Feedback. If you provide us any ideas, thoughts, criticisms, suggested improvements or other feedback related to Our Technology (collectively "Feedback") you own the Feedback and you grant to us a worldwide, royalty-free, fully paid, perpetual, irrevocable license to use, reproduce, modify, translate, distribute, perform, display, import, sell, license, offer for sale, make, have made and otherwise exploit the Feedback in any form, media, or technology, whether now known or hereafter developed, and to allow others to do the same without restriction or obligation of any kind, on account of confidential information, intellectual property rights or otherwise, and may incorporate into our products or services any service, product, technology, enhancement, documentation or other development ("Improvement") incorporating or derived from any Feedback with no obligation to license or to make available the Improvement to you or any other person or entity. This is true whether you provide the Feedback through use of the Product or through any other method of communication with us, unless we have entered into a separate agreement with you that provides otherwise.

1.7. Maintenance.

- 1.7.1. Our Maintenance and Support Policies. If we offer and you purchase maintenance and support for the Product, then it will be provided in accordance with our then current maintenance and support policies for the applicable Product in effect at the time of purchase. You may access our maintenance and support policies by clicking on the applicable Product family link located at https://www.progress.com/support.
- 1.7.2. Maintenance and Support for Perpetual or Term License Types. For Perpetual and Term License Types, unless otherwise expressly stated by us in the Order, first year annual maintenance and support (if offered by us) is required for the Product and starts on the date the Product is delivered. Thereafter, you may choose to purchase annual maintenance and support (if offered by us). If you do not purchase renewal maintenance and support services for a Product, then you will not receive any maintenance and support services for that Product and will have no entitlement to any benefits of maintenance and support services including, bug fixes, patches, upgrades, enhancements, new releases or technical support. If you want to reinstate lapsed maintenance and support services on a Product, and we offer reinstatement to our customers, then you may re-instate maintenance and support services by paying the then-current fee, plus a reinstatement fee for the lapsed maintenance and support period in accordance with our maintenance and support reinstatement policies then in effect.
- 1.7.3. Maintenance and Support for Subscription License Type. If the license type for the Product licensed to you is the subscription license type, then maintenance and support (if offered by us) is included in the subscription fees for each subscription period.

1.8. Fees and Taxes.

1.8.1. Payment Terms and Taxes. All fees payable to us are payable in the currency specified in the Order, or if no currency is specified, in United States Dollars, are due within 30 days from the invoice date and, except as otherwise expressly specified herein, are non-cancellable and non-refundable. We may charge you interest at a rate of 1.5% per month (or the highest rate permitted by law, if less) on all overdue payments. You agree to pay any sales, value-added or other similar taxes imposed by applicable law that we must pay on such fees, except those based on our income. Invoices may be issued by our Affiliate. If you and we agree that you will pay by credit card, you will provide us with valid and updated credit card information and you authorize us to store such information and bill such credit card for all fees applicable: (a) at the time that you order the Product and (b) at the time of any renewal

or upgrade

- 1.8.2. Fees for Renewal Subscription Licenses. If the license type for the Product licensed to you is the Subscription License Type then each renewal subscription will be calculated at the then-current price offered for the Product at the time of renewal.
- 1.8.3. Fees for Renewal Maintenance Terms. If the license type for the Product licensed to you is a Perpetual license or Term license, then, unless otherwise specified in the Order or in section 3 (Product Family Specific Terms) or section 4 (Product-Specific Terms), the fee for an optional annual renewal maintenance and support term for the Product will be calculated based on the annual rate applicable for the initial maintenance and support term or immediately preceding renewal maintenance and support term, whichever is applicable, plus a rate increase, if applicable, calculated at the lesser of any standard price increase or CPI (or equivalent index) after applying any increases as a consequence of our Lifetime Support policy, if applicable.
- 1.8.4. Orders between You and Our Authorized Reseller. Notwithstanding the above terms of this section 1.8 (Fees and Taxes), if you purchased your license to the Product and/or maintenance and support from an Authorized Reseller, then the fees will be set out in the Order between you and the Authorized Reseller. The Authorized Reseller may be responsible for billing and/or

collecting payment from you and if so, the billing and collection terms agreed to between you and the Authorized Reseller may differ from the terms set out in this section 1.8 (Fees and Taxes).

- 1.8.5. No Reliance on Future Availability of any Product or Update. You agree that you have not relied on the future availability of any Product or Updates in your purchasing decision or in entering into the payment obligations in your Order.
- 1.9. Warranties.
- 1.9.1. Authority. Each party represents and warrants that it has the legal power and authority to enter into this EULA.
- 1.9.2. Product Compliance with Documentation. We warrant to you that, for six (6) months from delivery (in the case of an On-Premise Product) or for the duration of the license (in the case of a Hosted Service), the Product will comply with the applicable Documentation in all material respects. Your exclusive remedy, and our sole liability, with respect to any breach of this warranty will be for us to use commercially reasonable efforts to promptly correct the non-compliance (provided that you notify us in writing within the warranty period and allow us a reasonable cure period). If we, at our discretion, reasonably determine that correction is not economically or technically feasible, we may terminate your license to the Product and provide you a full refund of the fees paid to us with respect to the Product (in the case of an On-Premise Product) or a refund of the prepaid fees for the unused portion of the license period (in the case of a Hosted Service). Delivery of additional copies of, or Updates to, the Product will not restart or otherwise affect the warranty period.
- 1.9.3. Warranty Exclusions. The warranty specified in section 1.9.2 (Product Compliance with Documentation) does not cover any Product provided on an unpaid evaluation or trial basis, or defects to the Product due to accident, abuse, service, alteration, modification or improper installation or configuration by you, your Affiliates, your or their personnel or any third party not engaged by us.
- 1.9.4. Warranty Disclaimers. EXCEPT FOR THE WARRANTIES EXPRESSLY STATED IN THIS SECTION 1.9 OR THE ADDITIONAL WARRANTIES (IF ANY) EXPRESSLY STATED IN SECTION 3 (PRODUCT FAMILY SPECIFIC TERMS) OR SECTION 4 (PRODUCT SPECIFIC TERMS), THE PRODUCT, DOCUMENTATION AND OUR TECHNOLOGY ARE PROVIDED "AS IS", WITH ALL FAULTS, AND WE DISCLAIM ALL WARRANTIES, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE, NONINFRINGEMENT, AVAILABILITY, ERROR-FREE OR UNINTERRUPTED OPERATION, AND ANY WARRANTIES ARISING FROM COURSE OF DEALING, COURSE OF PERFORMANCE, OR USAGE OF TRADE. TO THE EXTENT THAT WE MAY NOT AS A MATTER OF APPLICABLE LAW DISCLAIM ANY IMPLIED WARRANTY, THE SCOPE AND DURATION OF SUCH WARRANTY WILL BE THE MINIMUM PERMITTED UNDER APPLICABLE LAW.
- 1.10. Indemnification.
- 1.10.1. Our Indemnification Obligation.
- 1.10.1.1. Intellectual Property Infringement. We will defend you, and your officers, directors, employees, and agents from and against any and all third party claims, lawsuits, and proceedings alleging that your use of the Product, in accordance with the terms and conditions of this EULA, constitutes a direct infringement or misappropriation of such third party's patent, copyright or trade secret rights (the "IP Claim"), and we will indemnify you for damages finally awarded against you by a court of competent jurisdiction with respect to the IP Claim.
- 1.10.1.2. Exceptions. We will not indemnify you to the extent that the alleged infringement or misappropriation results from (a) use of the Product in combination with any other software or item not supplied by us; (b) failure to promptly implement an Update provided by us pursuant to 1.10.1.3 (Our Options); © modification of the Product not made or provided by us; or (d) use of the Product in a manner not permitted by this EULA. We also will not indemnify you if we notify you of our decision to terminate this EULA, and the license to the Product granted hereunder, in accordance with section 1.10.1.3 (Our Options) and you have not ceased all use of the Product within thirty (30) days of such notification.
- 1.10.1.3. Our Options. If a final injunction is, or we reasonably believe that it could be, obtained against your use of the Product, or if in our opinion the Product is likely to become the subject of a successful claim of infringement, we may, at our option and expense, (a) replace or modify the Product so that it becomes non-infringing (provided that the functionality is substantially equivalent), (b) obtain for you a license to continue to use the Product, or © if neither (a) nor (b) are reasonably practicable, terminate this EULA on thirty (30) days' notice and, if the Product was licensed to you on a Perpetual License or Term License basis, refund to you the license fee paid to us for the Product less an amount for depreciation determined on a straight-line five year (or actual term if shorter) depreciation basis with a commencement date as of the date of delivery of the Product, or if the Product was licensed to you on a Subscription License basis, refund to you the unused portion of the fees paid in advance to us for the then-current subscription period for the Product. THE INDEMNIFICATION PROVISIONS SET OUT IN THIS SECTION 1.10.1 STATE OUR ENTIRE LIABILITY AND YOUR SOLE AND EXCLUSIVE REMEDY WITH RESPECT TO ANY INFRINGEMENT OR ALLEGED INFRINGEMENT BY US OF ANY INTELLECTUAL PROPERTY RIGHTS OR PROPRIETARY RIGHTS IN RESPECT OF THE PRODUCT OR ITS USE.
- $1.10.2.\ Your\ Indemnification\ Obligation.$
- 1.10.2.1. Indemnification for Third Party-Claims. To the extent permitted by applicable law, you will defend us and our Affiliates, and our and their respective officers, directors, employees, and agents from and against any and all third party claims, lawsuits, and proceedings that arise or result from (a) your breach of this EULA, (b) your use, distribution and/or licensing of the Redistributables, if applicable, except to the extent it arises from an IP Claim covered under section 1.10.1 above, or © your failure or alleged failure to comply with Applicable Laws or any violation of a third party's rights in connection with your use of the Product (each a "Third-Party Claim" and collectively "Third-Party Claims") and you will indemnify for damages finally awarded by a court of competent jurisdiction with respect to any Third-Party Claim.
- 1.10.3. Control of the Defense or Settlement. For any indemnification obligation covered in section 1.10.1, "Indemnifying Party" means us, "Indemnified Party" means you, and "Claim" means an IP Claim. For any indemnification obligation covered in section 1.10.2, "Indemnifying Party" means you, "Indemnified Party" means us, and "Claim" means a Third-Party Claim. The Indemnified Party must provide the Indemnifying Party with prompt written notice of a Claim; however, the Indemnified Party's failure to provide or delay in providing such notice will not relieve the Indemnifying Party of its obligations under this section except to the extent the Indemnifying Party is prejudiced by the Indemnified Party's failure or delay. The Indemnified Party will give the Indemnifying Party full control of the defense and settlement of the Claim as long as such settlement does not include a financial obligation on or admission of liability by the Indemnified Party. If the Indemnified Party does not do so, then the Indemnified Party waives the Indemnifying Party's indemnification obligations under section 1.10.1 or 1.10.2, as applicable. The Indemnified Party will reasonably cooperate in the defense of the Claim and may appear, at its own expense, through counsel reasonably

acceptable to the Indemnifying Party.

1.11. Confidentiality.

- 1.11.1. Confidentiality Obligations. Except as otherwise provided herein, each party agrees to retain in confidence all information and know-how transmitted or disclosed to the other that the disclosing party has identified as being proprietary and/or confidential or should reasonably be understood to be confidential given the nature of the information and the circumstances surrounding its disclosure, and agrees to make no use of such information and know-how except under the terms of this EULA. However, neither party will have an obligation to maintain the confidentiality of information that (a) it received rightfully from a third party without an obligation to maintain such information in confidence; (b) was known to the receiving party prior to its disclosure by the disclosing party; © is or becomes a matter of public knowledge through no fault of the receiving party; or (d) is independently developed by the receiving party without use of the confidential information of the disclosing party. Further, either party may disclose confidential information of the other party as required by governmental or judicial order, provided such party gives the other party prompt written notice prior to such disclosure (unless such prior notice is not permitted by applicable law) and complies with any protective order (or equivalent) imposed on such disclosure. You will treat any source code for the Product as our confidential information and will not disclose, disseminate or distribute such materials to any third party without our prior written permission. Each party's obligations under this section 1.11 will apply during the term of this EULA and for five (5) years following termination of this EULA, provided, however, that (i) obligations with respect to source code will survive forever and (ii) trade secrets will be maintained as such until they fall into the public domain
- 1.11.2. Product Benchmark Results. You acknowledge that any benchmark results pertaining to the Product are our confidential information and may not be disclosed or published without our prior written consent. This provision applies regardless of whether the benchmark tests are conducted by you or us.
- 1.11.3. Remedies for Breach of Confidentiality Obligations. Each party acknowledges that in the event of a breach or threat of breach of this section
- 1.11, money damages will not be adequate. Therefore, in addition to any other legal or equitable remedies, the non-breaching party will be entitled to seek injunctive or similar equitable relief against such breach or threat of breach without proof of actual injury and without posting of a bond.
- 1.12. Data Collection and Personal Data.
- 1.12.1. Data Collection through use of the Product. THE PRODUCT MAY INCLUDE FEATURE(S) THAT (A) GATHER PRODUCT ACTIVATION, USAGE AND/OR ENVIRONMENT INFORMATION, (B) IDENTIFY TRENDS AND/OR BUGS, © COLLECT USAGE STATISTICS, AND/OR (D) TRACK OTHER DATA RELATED TO YOUR USE OF THE PRODUCT, AS FURTHER DESCRIBED IN THE CURRENT VERSION OF OUR PRIVACY POLICY AVAILABLE AT https://www.progress.com/legal/privacy-policy. BY YOUR ACCEPTANCE OF THE TERMS OF THIS EULA AND/OR USE OF THE PRODUCT, YOU AUTHORIZE THE COLLECTION, USE AND DISCLOSURE OF THIS DATA FOR THE PURPOSES PROVIDED FOR IN THIS EULA AND/OR THE PRIVACY POLICY.
- 1.12.2. Additional Data Collection Terms. Depending on the Product licensed to you, this EULA may contain additional data collection terms in section 3 (Product Family Specific Terms) or section 4 (Product Specific Terms) and/or, if we are hosting the Product, in section 2.B (Terms for Hosted Services).
- 1.12.3. Your Personal Data. If you determine that you will be supplying us with your Personal Data (as defined in the Data Processing Addendum referenced below) for us to process on your behalf, in the provision of maintenance and support services or hosting services (if the Product licensed to you is a Hosted Service) or during the course of any audits we conduct pursuant to section 1.14 (Audit), you may submit a written request at privacy@progress.com for the mutual execution of a Data Processing Addendum substantially in the form we make available at https://www.progress.com/docs/default-source/progress-software/data-processing-a ddendum.pdf and we will enter into such Data Processing Addendum with you. To the extent there is any conflict between this EULA and such Data Processing Addendum, the Data Processing Addendum will prevail with respect to our handling and processing of your Personal Data.
- 1.13. Limitation of Liability and Disclaimer of Certain Types of Damages.
- 1.13.1. Limitation of Liability. EXCEPT FOR A PARTY'S INDEMNIFICATION OBLIGATIONS SET OUT IN THIS EULA OR A PARTY'S BREACH OF ITS CONFIDENTIALITY OBLIGATIONS PURSUANT TO SECTION 1.11 (CONFIDENTIALITY), OR YOUR MATERIAL VIOLATION OF OUR INTELLECTUAL PROPERTY RIGHTS OR OF THE LICENSE RESTRICTIONS SET OUT IN THIS EULA, TO THE EXTENT PERMITTED BY APPLICABLE LAW, IN NO EVENT WILL EITHER PARTY'S LIABILITY FOR ALL COSTS, DAMAGES, AND EXPENSES ARISING OUT OF OR RELATED TO THIS EULA WHETHER BASED UPON WARRANTY, CONTRACT, TORT (INCLUDING NEGLIGENCE), STRICT LIABILITY OR OTHERWISE AT LAW EXCEED, IN THE AGGREGATE, THE FEES PAID TO US FOR THE PRODUCT AND/OR SERVICE THAT IS THE SUBJECT OF THE CLAIM, PROVIDED, HOWEVER, THAT IF THE FEES PAID FOR SUCH PRODUCT AND/OR SERVICE ARE PAID ON A RECURRING BASIS, THEN THE NOT TO EXCEED LIMIT WILL BE THE FEES PAID TO US FOR THE PRODUCT AND/OR SERVICE DURING THE TWELVE (12) MONTH PERIOD IMMEDIATELY PRECEDING THE DATE THE CLAIM AROSE. OUR AFFILIATES AND LICENSORS, AND THE SUPPLIERS TO US, OUR AFFILIATES OR LICENSORS, WILL, TO THE EXTENT PERMITTED BY APPLICABLE LAW, HAVE NO LIABILITY TO YOU OR TO ANY OTHER PERSON OR ENTITY FOR DAMAGES, DIRECT OR OTHERWISE, ARISING OUT OF THIS EULA, INCLUDING, WITHOUT LIMITATION, DAMAGES IN CONNECTION WITH THE PERFORMANCE OR OPERATION OF OUR PRODUCTS OR OUR PERFORMANCE OF SERVICES.
- 1.13.2 Disclaimer of Certain Types of Damages. EXCEPT FOR A PARTY'S INDEMNIFICATION OBLIGATIONS SET OUT IN THIS EULA OR YOUR MATERIAL VIOLATION OF OUR INTELLECTUAL PROPERTY RIGHTS OR THE LICENSE RESTRICTIONS SET OUT IN THIS EULA, TO THE EXTENT PERMITTED BY APPLICABLE LAW, IN NO EVENT WILL EITHER PARTY, ITS AFFILIATES OR ITS LICENSORS OR THEIR RESPECTIVE SUPPLIERS BE LIABLE FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL, INCIDENTAL, PUNITIVE OR TORT DAMAGES ARISING IN CONNECTION WITH THIS EULA OR EITHER PARTY'S PERFORMANCE UNDER THIS EULA OR THE PERFORMANCE OF OUR PRODUCTS, OR FOR ANY DAMAGES RESULTING FROM LOSS OF USE, LOSS OF OPPORTUNITY, LOSS OF DATA, LOSS OF REVENUE, LOSS OF PROFITS, OR LOSS OF BUSINESS, EVEN IF THE PARTY, ITS AFFILIATES, ITS LICENSORS, OR ANY OF THEIR RESPECTIVE SUPPLIERS HAVE BEEN ADVISED OF THE POSSIBILITY OF THOSE DAMAGES.
- 1.14. Audit. We may install and use automated license tracking, management and/or enforcement solutions with the Product, which you may not disrupt or alter. You will maintain records in connection with this EULA and the use of the Product and any Updates and/or services provided hereunder. Such records will include at a minimum the number of licenses purchased and being used by you. At our expense and with reasonable written notice to you, we or a third party appointed by us may audit the

records, and if necessary and as applicable, the systems on which the Product or any Update is installed for the sole purpose of ensuring compliance with the terms of this EULA. We will have the right to conduct audits as necessary. These audits may be conducted on site at a location where you have installed the Product, remotely from our offices, or a combination of both, if applicable to the Product. On-site audits will be conducted during regular business hours, and neither on-site nor remote audits will interfere unreasonably with your business operations. You agree to share with us copies of all records referenced herein, as well as Product log files and other information reasonably requested by us promptly following such request, but in no event more than five (5) business days following receipt of our written request (or such longer period, if applicable, that we specify in the written request). We will treat all such information obtained or accessed by us during the audit as confidential information pursuant to section 1.11 (Confidentiality) for use by us only as necessary to ensure compliance with and enforcement of the terms of this EULA. If any audit reveals that you have underpaid license, maintenance and support or subscription fees, you will be invoiced for all such underpaid fees based on our list price in effect at the time the audit is completed. If the underpaid fees exceed five percent (5%) of the fees previously paid by you, then you will also pay our reasonable costs of conducting the audit and enforcement of this EULA.

1.15. Termination.

- 1.15.1. Termination for Breach. We may terminate this EULA by written notice at any time if you do not comply with any of your obligations under this EULA and fail to cure such failure to our satisfaction within thirty (30) days after such notice. This remedy will not be exclusive and will be in addition to any other remedies which we may have under this EULA or otherwise. 1.15.2. Effect of Termination. Upon expiration of your license term to the Product (if applicable) or earlier termination of this EULA, your license to access and/or use the Product and/or distribute the Redistributables (if applicable) will terminate. You must immediately cease use of the Product and destroy all copies of the Product in your possession (and required any Permitted Third Parties to do the same). Any licenses you have granted to the Redistributables in accordance with the terms and conditions of this EULA will, unless otherwise specified in section 3 (Product Family Specific Terms) or section 4 (Product Specific Terms), survive termination of this EULA.
- 1.15.3. Survival. Any provisions of this EULA containing licensing restrictions, warranties and warranty disclaimers, confidentiality obligations, limitations of liability and/or indemnity terms, audits rights, and any term of this EULA which, by its nature, is intended to survive termination or expiration, will remain in effect following any termination or expiration if this EULA, as will your obligation to pay any fees accrued and owing to us as of termination or expiration.
- 1.16. Assignment. You may not, without our prior written consent, assign or novate this EULA, any of your rights or obligations under this EULA, or the Products or any of our Confidential Information, in whole or in part, by operation of law, sale of assets, merger or otherwise, to any other party, including any parent, subsidiary or affiliated entity. Your Change of Control will constitute an assignment for purposes of the preceding sentence. A "Change of Control" will include, but not be limited to, any merger, consolidation, amalgamation, reorganization or sale, transfer or exchange of the capital stock or equity interests of you in a transaction or series of transactions which results in the holders of your capital stock or equity interests holding less than 50% of the outstanding capital stock or equity interests immediately following such transaction(s).
- 1.17. Choice of Law. This EULA is governed by the laws of the Commonwealth of Massachusetts, U.S.A., without regard to the conflict of laws principles thereof. If any dispute, controversy, or claim cannot be resolved by a good-faith discussion between the parties, then it will be submitted for resolution to a state or federal court in Boston, Massachusetts, USA, and the parties hereby irrevocably and unconditionally agree to submit to the exclusive jurisdiction and venue of such court. The Uniform Computer Information Transactions Act and the United Nations Convention on the International Sale of Goods will not apply to this EULA.
- 1.18. Publicity. You agree that we may, in our sole discretion, publicize your

use of the Product, and you license to us (and our Affiliates and necessary sublicensees) any intellectual property rights required to allow us (and our Affiliates and necessary sublicensees) to use your name, trade name(s), trademark(s), service mark(s), logo(s) and domain name(s) in connection with such publicity.

1.19. Miscellaneous.

- 1.19.1. Notices. Notices of termination, material breach, your insolvency or an indemnifiable claim ("Legal Notices") must be clearly identified as Legal Notices and sent via overnight courier or certified mail with proof of delivery to the following addresses: For us: 14 Oak Park Drive, Bedford, MA 01730, Attention: General Counsel. For you: your address set out in the Order. Legal Notices sent in accordance with the above will be effective upon the second business day after mailing. Either party may change its address for receipt of notices upon written notice to the other party.
- 1.19.2. Entire Agreement. This EULA, and any terms expressly incorporated herein by reference, will constitute the entire agreement between you and us with respect to the subject matter of this EULA and supersedes all prior and contemporaneous communications, oral or written, signed or unsigned, regarding such subject matter. Use of any purchase order or other document you supply in connection with this EULA will be for administrative convenience only and all terms and conditions stated therein will be void and of no effect. Except as otherwise expressly contemplated in this EULA, this EULA may not be modified or amended other than in writing signed by you and us.
- 1.19.3. Severability. If any provision of this EULA is terminated or held by a court of competent jurisdiction to be invalid, illegal, or unenforceable, the remainder of this EULA will remain in full force and effect.
- 1.19.4. Waiver. Failure or delay in exercising any right, power, privilege or remedy hereunder will not constitute a waiver thereof. A waiver of default will not operate as a waiver of any other default or of the same type of default on future occasions.
- 1.19.5. English Language. This EULA has been drawn up in English at the express wish of the parties. Le present contrat a ete redige en anglais a la demande expresse des parties.
- 1.19.6. Force Majeure. Neither you nor we will be liable for any delay or failure to take any action required under this EULA (except for payment) due to any cause beyond the reasonable control of you or us, as the case may be, including, but not limited to unavailability or shortages of labour, materials, or equipment, failure or delay in the delivery of vendors and suppliers and delays in transportation.
- 1.19.7. Our Use of Our Affiliates. We may, at our discretion, engage one or more of our Affiliates in the fulfilment of our obligations, including, our obligations for delivery of the Product to you and/or the provision of any maintenance and support services

2.A. TERMS FOR ON-PREMISE PRODUCTS

- 2.A.1. Delivery. Unless otherwise specified by us, On-Premise Product(s) will be provided to you via electronic delivery, and delivery is deemed complete when the On-Premise Product(s) is/are made available at the electronic software download site specified by us and you are e-mailed or otherwise provided with any necessary instructions, password and/or license keys required for you to be able to access, download and install the On-Premise Product(s). If we provide the On-Premise Product(s) on physical media, shipping terms will be FOB shipping point.
- 2.A.2. Updates. Each Update to an On-Premise Product replaces part or all of the On-Premise Product (or earlier Update) previously licensed to you ("Replaced Product") and will terminate such previously licensed Replaced Product to the extent replaced by the Update; provided, however, that you may continue to operate the Replaced Product for up to ninety (90) days from delivery of the Update to allow you to complete your implementation of the Update. You must cease all use of the Replaced Product at the end of the ninety (90) day period. Each Update will be subject to the terms and conditions of the license agreement accompanying the Update which must be accepted by you at the time you download or install the Update. If you do not agree to the license agreement accompanying the Update, do not download or install the Update.
- 2.A.3. Cloud Environment. You may upload the On-Premise Product(s) licensed to you pursuant to this EULA onto a cloud instance supplied by a third party, provided that the operation of the On-Premise Product(s) in the cloud instance complies with all license model restrictions and usage limitations applicable to the On-Premise Product(s). You may also allow the third party to upload, install, operate and/or use the On-Premise Products on the cloud instance, provided that the third party's access to and use of the On-Premise Products is solely for your benefit in accordance with the terms of this EULA. The third party will be considered a Permitted Third Party, and you will be responsible for the Permitted Third Party's compliance with this EULA in accordance with section 1.2.3 (Third Party Use).
- 2.B. TERMS FOR HOSTED SERVICES THIS SECTION IS NOT APPLICABLE

3. PRODUCT FAMILY SPECIFIC TERMS

This section specifies terms and conditions that are applicable to the following On-Premise Products, as made generally available by us to our customers: (1) all Progress OpenEdge products (excluding Progress OpenEdge Developer Kit), (2) all Progress DataDirect ODBC, JDBC, ADO.NET, SequeLink, OpenAccess, Hybrid Data Pipeline and Data Integration products; and (3) all Progress Corticon products. Default License Type for each of the above-referenced On-Premise Products: Perpetual

3.1. Product Family Definitions.

Any defined term used in this section 3 (Product Family Specific Terms) but not defined herein will have the meaning ascribed to it in section 1 (General Terms and Conditions) or section 2 (Terms for On-Premise Products).

- 3.1.1. "Client Device" means any input technology that allows you to access the Product, including but not limited to a workstation, a personal computer, a PDA device, a cellular phone, a tablet, a laptop or other device that is operated by an individual.
- 3.1.2. "Container" means a software package that relies on virtual isolation to deploy and run on the same Server multiple Product instances, components of the Product, or the applications utilizing the Product, while accessing a shared operating system kernel.
- 3.1.3. "Core" means a core processor of a CPU as allocated by you made up of an independent processor combined onto a single integrated circuit or silicon chip, in both a virtualized and/or non-virtualized environment, and regardless of whether used in a production or non-production (e.g., test, development) environment.
- 3.1.4. "CPU" means a central processing unit, also known as a processor or microprocessor. It can contain multiple cores in both virtualized and/or non-virtualized environments.
- 3.1.5. "Disaster" means any unplanned event or condition that renders you unable to use an application or database for its intended computer processing and related purposes.
- 3.1.6. "Named User" means a single individual, Non-Human Operated Device or Process.
- 3.1.7. "Non-Human Operated Device" means a device that is not operated by an individual including, but not limited to, a temperature device, a production line bar code scanner, or a tracking device.
- 3.1.8. "Platform" means a specific combination of the hardware and the operating system, a change to either would constitute a platform change.
- 3.1.9. "Process" means any automated process that is not initiated by a Client Device or a Non-Human Operated Device and includes, without limitation, automated controls and background jobs.
- 3.1.10. "Server" means a logical computer (can be a physical hardware or a virtual machine) with one or more CPUs on which the Product resides, along with the applications utilizing the Product, and which can be accessed by other computers. If the Product is installed on a Container, such Container is treated as a separate Server.
- 3.1.11. "Site" means a single building or campus of buildings sharing the same postal address.
- 3.1.12. "Use Case" means a description in the Order of one or more limitation(s) on the way the Product may be used by you. Such limitation(s) may include, but are not limited to, use of the Product only for a specific project, application, line of business, purpose of use or group of users.
- 3.2. License Grant. Subject to the terms and conditions contained in this EULA, including all Orders and License Addenda (as defined below), we grant to you a non-exclusive, non-transferable, limited, personal license (without the right to sublicense) to use the Product identified in the Order. We also grant you a non-exclusive, non-transferable, limited, personal license (without the right to sublicense) to use the Documentation. The license model applicable to the Product license granted to you will be identified in the Order and described in section 3.4 (Product Family License Models) or section 4.2 (Product-Specific License Models).
- 3.3. Product Family Applicable License Restrictions. The following license restrictions apply in addition to those set forth in section 1.2.3

(Restrictions):

- 3.3.1. Internal Use; No Affiliate Use. Your use of the Product and Documentation is limited to internal use within your company. The Product and Documentation may not be used by your Affiliates unless specifically authorized in the Order. Use will be in accordance with this EULA and any additional terms, if any, set out in any Order or additional agreement executed by us and you in connection with this EULA which specifically states the terms thereof will be in addition to or in lieu of any of the terms set out in this EULA (each a "License Addendum" and collectively, the "License Addenda").
- 3.3.2. No Use by Others. Except as otherwise expressly permitted in this EULA, you may not allow the Product to be accessed, used or possessed by another party. For these purposes, the term "use" will include, without limitation, direct or indirect use via thin-client or web-based remote access software which but for the use thereof would have required a copy of the Product to be installed or used locally by that user.
- 3.3.3. No Time Sharing or Similar Services. You will have no right to use the Product to provide outsourced services, or facility management services or to act as or operate a service bureau or provide information, data processing, subscription or hosting services for another party.
- 3.4. Product Family License Models. This section specifies license models that may be applicable to one or more Products. Note that not all license models are available for all Products.
- 3.4.1. CPU License. A CPU License grants you the right to run the Product, or an application which can access the Product, on a single CPU, on a single Server and on a single Platform, where such CPU contains only one Core. If the CPU contains more than one Core, then you must either purchase a Core License for the Product or, if no Core License model is available for the Product, then each Core in the CPU will count as one (1) CPU. Additional CPU License(s) are required for each CPU on a Server that runs the Product, including, without limitation, Servers configured for disaster recovery, load balancing, clustering, development, testing and reporting. A CPU License may not be transferred from one Server or Platform to another.
- 3.4.2. Named User/Seat License. A Named User or Seat License grants you the right to designate a Named User to access and use the Product or access and use an application which can access the Product. You must be able to identify and count each Named User. A Named User License is a multi-server license in that it is not limited by Core, CPU or Server count, but the Product must be used on a single Platform. A Named User may not be designated concurrently on different computers or devices or shared by multiple users. A Named User does not have to be logged on to the Product to be counted as a Named User. A Named User License designation may be transferred from one Named User to another provided that the original Named User no longer requires and is no longer permitted access to the Product. The foregoing transfer right will not affect the assignment prohibition set forth in section 1.16 (No Assignment). All Named Users must be bound to the terms and conditions of this EULA.
- 3.4.3. Server/Machine License. A Server or Machine License grants you the right to install and use the Product, or an application which can access the Product, on a single Server and on a single Platform. Additional Server or Machine License(s) are required for each Server that runs the Product including, without limitation, Servers configured for disaster recovery, load balancing, clustering, development, testing and reporting. A Server or Machine License may not be transferred from one Server or Platform to another.
- 3.4.4. Site License. A Site License grants you the right to install and use the Product, or an application which can access the Product, in any Server at a single Site and on any device (wherein the device can be a Client Device, a Non-Human Operated Device or a Process) operated by users who share the same principal place of business as the Site location, regardless of whether the Product, or an application which can access the Product is used in a virtualized and/or non-virtualized environment.
- 3.4.5. Application Specific License. If the Order lists a specific application authorized for use with the Product, or if you obtained the Product in conjunction with a software application provided by a third party, the Product is licensed for use only with that application. Use with any other application/product is strictly prohibited. You will also be subject to the license terms of the appropriate license model for the Product (e.g., CPU License, Core License, Named User/Seat License, etc.) designated in the Order. In no event do we make any representations or warranties, express, implied or arising by custom or trade usage, regarding a third party's application and assume no liability or responsibility for such application.
- 3.4.6. Developer License. A Developer License limits your access to and use of the Product to internal application development and support purposes only. If the license is for components of the Product only, then this right is limited to those components. You will also be subject to the license terms of the appropriate license model for the Product (e.g., CPU License, Core License, Server/Machine License, etc.) designated in the Order. A Developer License may not be transferred.
- 3.4.7. Disaster Recovery License. A Disaster Recovery License will only be used by you for the sole purpose of application recovery in the event a system fails or crashes or the Product or database files become corrupt. In that case, a Disaster Recovery License grants you the right to use the Product for a period not to exceed ninety (90) days ("D/R Deployment Period"). You will notify us in writing within five (5) business days of the commencement of the D/R Deployment Period. You will also be subject to the same quantity limits and license terms of the appropriate license model for the Product (e.g., CPU License, Core License, Server/Machine License, etc.) designated in the Order. A Disaster Recovery License may be transferred from one Server or CPU to another with prior notice to us so long as such change is permitted by the underlying license model. No Disaster Recovery License is required for a switch from a primary Server to a secondary Server so long as the primary Server is permanently disabled. However, a Disaster Recovery License is required for any other temporary reassignment between the primary Server and any other Server. The foregoing transfer right will not affect the assignment prohibition set forth in section 1.16 (No Assignment).
- 3.4.8. Evaluation License. An Evaluation License grants you the right to use the Product solely for evaluation, demonstration, prototyping, testing and/or proof of concept purposes in accordance with the terms set forth in section 1.2.5 (Limitations on Evaluation or Trial Licenses), and not in conjunction with the development or deployment of any software application. No commercial product development work is authorized. YOU UNDERSTAND AND ACKNOWLEDGE THAT THE PRODUCT LICENSED TO YOU PURSUANT TO AN EVALUATION LICENSE CONTAINS A DISABLING DEVICE THAT WILL AUTOMATICALLY DISABLE THE PRODUCT THIRTY (30) DAYS FROM INSTALLATION OR AT THE END OF SUCH LONGER OR SHORTER PERIOD OF TIME AS MAY BE SPECIFIED IN AN ORDER OR AGREED BY YOU AND US IN WRITING.
- 3.4.9. Testing and Staging License. A Testing and Staging License grants you the right to use the Product for internal quality assurance testing purposes only and you may not deploy the Product in a production environment. You will also be subject to the license terms of the appropriate license model for the Product (e.g., CPU License, Core License, Server/Machine License, etc.) designated in the Order. A Testing and Staging License may not be transferred.
- 3.5. General Usage Limitations

- 3.5.1. Templates, Samples and/or Accelerators.
- 3.5.1.1. No Warranties. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THIS EULA, INCLUDING WITHOUT LIMITATION ANYTHING TO THE CONTRARY IN SECTION 1.9 (WARRANTIES), ANY TEMPLATES, SAMPLES OR ACCELERATORS THAT ARE LOCATED IN THE PRODUCT AND DESCRIBED IN THE DOCUMENTATION AS A SAMPLE, TEMPLATE OR ACCELERATOR, ARE PROVIDED SOLELY ON AN "AS IS" BASIS WITHOUT WARRANTY OF ANY KIND. WE MAKE NO WARRANTIESWITH RESPECT TO SUCH TEMPLATES, SAMPLES AND/OR ACCELERATORS, EXPRESS, IMPLIED, OR ARISING BY CUSTOM OR TRADE USAGE, AND SPECIFICALLY MAKE NO WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE.
- 3.5.1.2. No Maintenance and Support. Any maintenance and support

services offered by us for the Product will not include maintenance and support

for the above-mentioned templates, samples or accelerators, as may be configured by you for your own implementation. You will be solely responsible for the use and configuration of said templates, samples and/or accelerators. 3.5.1.3. Ownership of Templates, Samples and/or Accelerators. We retain all ownership rights in and to the above-mentioned templates, samples and/or accelerators, and any modifications thereto or derivatives thereof, in accordance with section 1.6.1 (IP Ownership).

- 3.5.2. Optional Components. The Product licensed and delivered to you may include on the media or in the electronically delivered files, as applicable, components, features or other Products for which a separate license purchase and license key(s) or control code(s) are required. You are permitted to use only the Product(s) and/or components thereof for which you have ordered and received a valid license key or control code.
- 3.5.3. Use Case and/or User Group Restrictions. If the Order indicates that the Product may be used for specific Use Case(s) or by specific user group(s), the Product may only be used for the Use Case(s) and/or by the user group(s) listed in the Order.
- 3.5.4. Products Available via a Container Method. If we offer a download option for the Product in a container image then the Product will be provided in a container image that may also contain third party product(s) as identified in the Special Notices accompanying the Product, and subject to the additional licensing terms referenced in the Special Notices. Notwithstanding anything to the contrary in this EULA, we make no representations or warranties and assume no indemnification obligations in regard to such third party product(s), its/their operation or its/their security. User modifications to the version of the third party product(s) provided by us in the container image, even if permitted under the applicable license referenced in the Special Notices, may result in errors or instability in performance of the Product which are not covered by us under warranty or maintenance terms. You are responsible for obtaining, at your own expense, any required licenses from the supplier of the container technology to deploy the container image that contains the Product and the above-referenced third-party products and any such deployment of the Product must comply with the terms and conditions of this EULA.
- 3.5.5. Technical Preview Features. Product(s) may include features that are identified as a "Technical Preview", and if included, such features are documented in the Documentation such as the Product Releases Notes. We are not obligated to provide any maintenance, technical or other support for the Technical Preview features, but may, from time to time, provide technical support and/or Updates to the Technical Preview features at our sole discretion. All such Updates will be subject to the terms and conditions of this EULA, including, without limitation, the terms and conditions set forth in this section. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN IN SECTION 1.9 (WARRANTIES), TECHNICAL PREVIEW FEATURES, ANY UPDATES THERETO AND ANY RELATED TECHNICAL SUPPORT SERVICES ARE PROVIDED "AS IS" WITHOUT WARRANTY OF ANY KIND. WE MAKE NO WARRANTIES WITH RESPECT TO THE TECHNICAL PREVIEW FEATURES, ANY UPDATES THERETO, OR ANY RELATED TECHNICAL SUPPORT SERVICES, WHETHER EXPRESS, IMPLIED, OR ARISING BY CUSTOM OR TRADE USAGE, AND SPECIFICALLY MAKE NO WARRANTY OF MERCHANTABILITY, FITNESS FOR ANY PARTICULAR PURPOSE, TITLE OR NONINFRINGEMENT.

4. PRODUCT SPECIFIC TERMS

This section specifies terms and conditions that are applicable to Products included in the Progress DataDirect ODBC, JDBC, ADO.NET, SequeLink, OpenAccess, Hybrid Data Pipeline and Data Integration product lines.

- 4.1. Product Specific Definitions. Any defined term used in this section 4 (Product-Specific Terms) but not defined herein will have the meaning ascribed to it in section 1 (General Terms and Conditions), section 2 (Terms for On-Premise Products) or section 3 (Product Family Specific Terms).
- 4.1.1. "Custom Driver" means a data connectivity driver resulting from the use of the DataDirect OpenAccess Runtime Components in combination with the User Interface Provider.
- 4.1.2. "Progress OpenAccess Development Product(s)" means Product(s) (in object code form) described in the Order as an "OpenAccess SDK" (or similar label or identifier) and that consist of a software development toolkit (SDK) that may be used to develop a User Interface Provider.
- 4.1.3. "Progress OpenAccess Runtime Components" means the runtime software components (in object code form) that are included in the Progress OpenAccess Development Product(s).
- 4.1.4. "Progress OpenAccess Runtime Product(s)" means the Progress OpenAccess Runtime Components (in object code form) specified in the Order that may be used in connection with the deployment of the Custom Driver in an internal production environment.
- 4.1.5. "User Interface Provider" means computer program code written by or for you that provides the interface between the Progress OpenAccess Runtime Components and your data source.
- 4.2. Product Specific License Models. This section specifies license models that may be applicable to one or more Products in the Progress DataDirect ODBC, JDBC, ADO.NET, SequeLink, OpenAccess, Hybrid Data Pipeline and Data Integration product lines. Note that not all license models are available for all Products.
- 4.2.1. Authorized User License. If the Product is licensed on an Authorized User basis, then the Product may be used on a single computing machine which may be accessed by the number of Authorized Users specified in the Order. In no event shall the number of persons authorized to have access to the Product exceed the number of Authorized Users identified in the Order. The right to use the Product is not based on concurrent users, but instead on an actual "Authorized User" basis. You must, upon request by Progress, identify each Authorized User who will have access to the Product and provide that information to Progress.

- 4.2.2. Concurrent User License. If the Product is licensed on a Concurrent User basis, then the Product may be installed and used on multiple machines for access by multiple users provided that at any given time the maximum number of Concurrent Users accessing or using the Product may not exceed the number of Concurrent Users identified in the Order. A "Concurrent User" is a single individual making use of or accessing the Product under a Concurrent User License model and will be deemed a "Concurrent User" at all times while such use or access is occurring.
- 4.2.3. Connection Basis (for DataDirect Connect for ODBC, DataDirect for ODBC). If the Product licensed to you is a Progress DataDirect Product which incorporates the term "for ODBC" into the Product name or Product description or which is otherwise designated as being "for ODBC", and if the Product is licensed on a Connection basis, then a license fee must be paid for both the maximum number of simultaneous Connections utilized by the Server and the maximum number of CPUs utilized by the Server whether or not such access is simultaneous or consecutive. For the purposes of this provision only a "Connection" is defined as the driver state after a successful call to SQLConnect, SQLDriverConnect, or SQLBrowseConnect and before termination of this state, normally through a successful call to SQLDisconnect. Servers utilizing dual core processing (or multiple core processing) or other multi-processor technology shall have each core processor count as one (1) CPU. The Product may only be used by Server-based applications running on that Server accessed by other computers and may not be installed via file sharing on a local area network file system.
- 4.2.4. Connection Basis (for DataDirect Connect for JDBC, DataDirect for JDBC). If the Product licensed to you is a Progress DataDirect Product which incorporates the term "for JDBC" into the Product name or Product description or which is otherwise designated as being "for JDBC", and if the Product is licensed on a Connection basis, then a license fee must be paid for both the maximum number of simultaneous Connections utilized by the Server and the maximum number of CPUs utilized by the Server whether or not such access is simultaneous or consecutive. For the purposes of this provision only a "Connection" is defined as the driver state after a successful call to the DriverManager.getConnection method or the driver's Driver.connect, DataSource.getConnection ConnectionPoolDataSource.getPooledConnection, XADataSource.getXAConnection, or ManagedConnectionFactory.createManagedConnection methods and before termination of this state, normally through a successful call to the driver's Connection.close, PooledConnection.close or ManagedConnection.destroy methods.

Servers utilizing dual core processing (or multiple core processing) or other multi-processor technology shall have each core processor count as one (1) CPU. The Product may only be used by Server-based applications running on that Server accessed by other computers and may not be installed via file sharing on a local area network file system.

- 4.2.5. Connection Basis (for DataDirect Connect for ADO.NET only). If the Product is licensed on a Connection basis, then a license fee must be paid for both the maximum number of simultaneous Connections utilized by the Server and the maximum number of CPUs utilized by the Server whether or not such access is simultaneous or consecutive. For the purposes of this provision only a "Connection" is defined as the provider state after a successful call to the Open method in the System.Data.Common.DbConnection class or the DDTek.DB2.DB2Connection class or the DDTek.Oracle.Oracle.OracleConnection class or the DDTek.SQLServer.SQLServerConnection class or the DDTek.Sybase.SybaseConnection class and before termination of this state, normally through a successful call to the Close method in the System.Data.Common.DbConnection class or the DDTek.DB2.DB2Connection class or the DDTek.Oracle.Oracle.OracleConnection class or the DDTek.SQLServer.SQLServerConnection class or the DDTek.Sybase.SybaseConnection class. Servers utilizing dual core processing (or multiple core processing) or other multi-processor technology shall have each core processor count as one (1) CPU. The Product may only be used by Server-based applications running on that Server accessed by other computers and may not be installed via file sharing on a local area network file system.
- 4.2.6. Core License. If the Product is licensed on a Core basis, then you may run the Product on a single Core, on a single Server. The total number of Cores on the Server may not exceed the total number licensed to you. Additional Core License(s) are required for each Core of a Server on which the Product runs, including, without limitation, Servers configured for disaster recovery, load balancing, clustering, development, testing and reporting.
- 4.2.7. Virtualized Environment. The terms of this section apply only to Products licensed under a Core license model. Notwithstanding anything in the EULA to the contrary, you may install and use the Product(s) in a virtual machine environment, provided however, that you must not allow more than the number of applicable licensed Cores to be allocated to each Product via the virtual machine technology. The applicable licensed Cores will be the number of Cores that you are licensed to install and use each Product as specified in the Order. Any allocation of any portion of a Core's processing ability to the Product will constitute one Core toward the virtual machine environment licensed hereunder. You will be responsible for allocating the Cores to the Product(s) in a manner to ensure compliance with this restriction. Use of the Product(s) in any configuration or environment (including, without limitation, dynamic virtualization where additional Cores could be allocated to the Product(s) at peak load times) that at any time exceeds the restrictions set forth above is strictly prohibited, and requires the purchase of additional licenses.
- 4.2.8. z/OS Operating System Image License. If the Product is licensed to you on an z/OS Operating System Image basis, then a license fee must be paid for the number of z/OS Operating System Images where the Product is installed. You will identify the number of z/OS Operating System Images where the Product is installed and provide the information to us.
- 4.2.9. MVS or OS/390 Operating System Image License. If the Product is licensed to you on an MVS or OS/390 Operating System Image basis, then a license fee must be paid for the number of MVS or OS/390 Operating System Images, whichever is applicable, where the Product is installed. You will identify the number of MVS or OS/390 Operating System Images, whichever is applicable, where the Product is installed and provide the information to Progress.
- 4.2.10. Workstation License. If the Product is licensed on a Workstation basis, each Workstation using or accessing the Product must be licensed. A "Workstation" is defined as a computer that is operated, either attended or via remote access, by one person at a time, and cannot be used by more than one person, directly or indirectly, simultaneously. In no event will the number of workstations utilizing or accessing the Product exceed the number of Workstations identified in the Order.
- 4.2.11. Evaluation License Term. Notwithstanding anything to the contrary in section 3.4.8 (Evaluation License), the default license term for any evaluation license granted to you for the Product(s) pursuant to this EULA will be fifteen (15) days.
- 4.2.12. Term and Subscription Licenses. If the Order specifies that your license to the Product(s) is a term or subscription license, then, unless otherwise specified in the Order, your license to the Products will not auto renew and the term of this EULA and your license to the Product(s) hereunder will expire on the expiration date stated in the Order unless sooner terminated in accordance with the terms of this EULA. For the duration of the Term or Subscription license (as applicable) you will be further subject to the license terms of the appropriate license model for the Product (e.g., CPU, Core, Server/Machine, etc.) designated in the Order. Upon completion of the term or subscription (as applicable) the Product(s) may automatically timeout or cease to function. Upon completion of the term or subscription (as applicable), all of your license rights in and to the Products will terminate and you must satisfy the requirements in section 1.15.2 (Effect of Termination). 4.3. Product-Specific

Usage Limitations. 4.3.1. Runtime License Checker. The Products may include code to verify that you have not exceeded the license keys issued by us (the "Runtime License Checker") and in the event that the Runtime License Checker determines that you have exceeded your license the Products may cease to operate.

- 4.3.2. Operating Systems or Platforms. If the Order specifies that your license to the Product(s) is/are limited to a specific Operating System ("OS") or Platform, then the Product(s) is/are licensed with the additional restriction that it be used solely in conjunction with the specified OS or Platform. Use with any other OS or Platform is strictly prohibited. Without limiting the foregoing, if the Product is a Progress OpenAccess Development Product (as defined above), then the additional restriction specified in the preceding sentence applies to any part of the Progress OpenAccess Development Product, including, without limitation, the Progress OpenAccess Runtime Components (as defined above) contained therein.
- 4.3.3. Progress DataDirect and OpenAccess Products "for ODBC". If the Product licensed to you is a Progress DataDirect or OpenAccess Product which incorporates the term "for ODBC" into the Product name or Product description or which is otherwise designated as being "for ODBC", (each the "Licensed Product"), the following additional requirements will apply: we have licensed, and have rights to sub-license certain third party software that is part of the ODBC SDK components and which is usable for certain non-Windows platforms - Mac OS, OS/2, HP-UX, AIX, Linux, Solaris, and other non-Windows operating systems. This section sets forth the conditions that govern your permitted use of the various non-windows SDK components: you are not licensed to use the ODBC Core Components (hereinafter defined) other than with the Licensed Product, or in the case where the Licensed Product is Progress OpenAccess Development Product for ODBC or Progress OpenAccess Runtime Product for ODBC, other than with the Custom Driver (as defined above) created with the use of the Progress OpenAccess Development Product ("Permitted Driver"). Use of the ODBC Core Components with any ODBC drivers other than the Licensed Product, or in the case where the Licensed Product is Progress OpenAccess Development Product for ODBC or Progress OpenAccess Runtime Product for ODBC, other than the Permitted Driver, is expressly prohibited. You may not embed in any products, or otherwise sub-license the ODBC Core Components, without a separate license agreement. For the purposes hereof, the term "ODBC Core Components" means the ODBC Driver Manager and support files for non-Windows operating systems. A separate license must be obtained from us if you want to embed and sub-license the ODBC Header Files (hereinafter defined) with any software product. For the purposes hereof, the term "ODBC Header Files" means the text files for compilation and development of software.
- 4.3.4. Stylus Studio Products. If the Product licensed to you is a Stylus Studio product the following requirements apply:
- 4.3.4.1. If the Product is Stylus Studio, Stylus Studio XML Enterprise Suite, or Stylus Studio XML Professional Suite then the Product is licensed on either:
- (a) a named user basis meaning specific individuals are authorized to access the Product and the total number of named users may not exceed the total number licensed by you; or
- (b) a Concurrent User basis meaning the quantity of Concurrent Users (as defined above) identified in the Order for the Product must equal the maximum number of users concurrently using or accessing the Product at any given time in accordance with the Concurrent User license model described above. Stylus Studio, Stylus Studio XML Enterprise Suite and Stylus Studio XML

Professional Suite are subject to the additional restriction that it may be used by you solely for development and testing of software applications, and not in conjunction with the deployment of such software applications.

- 4.3.4.2. If the Product is Stylus Studio XML Home Edition, then the Product is licensed on a single named user basis for personal use only. The Stylus Studio XML Home Edition is subject to the additional restriction that it not be used in conjunction with any commercial or for-profit transactions. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN SECTION 1.9 (WARRANTIES), AND EXCEPT AS PROHIBITED BY APPLICABLE LAW, STYLUS STUDIO XML HOME EDITION IS PROVIDED "AS IS" WITHOUT WARRANTY OF ANY KIND. WE MAKE NO WARRANTIES WITH RESPECT TO THE STYLUS STUDIO XML HOME EDITION, EXPRESS, IMPLIED, OR ARISING BY CUSTOM OR TRADE USAGE, AND SPECIFICALLY MAKE NO WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE OR NONINFRINGEMENT. You have the sole responsibility for adequate protection and back-up of all data use in connection with the product.
- 4.3.4.3. Notwithstanding anything to the contrary in section 3.4.8 (Evaluation License), if the Product is an evaluation version of Stylus Studio, then the Product is licensed on a single named user basis and is subject to the additional restriction that it be used by you solely for evaluation of software applications, and not in conjunction with the development or deployment of such software applications. You must purchase a license for Stylus Studio for the development and testing of software applications. Evaluation includes the use of Products in performance benchmarking. We update the Product regularly and benchmarking data for the Product is subject to change. Benchmark tests on prior versions of the Product may yield results that are not reflective of the performance of the current version of the Product. The most current version of the Product is available at www.stylusstudio.com. YOU UNDERSTAND AND ACKNOWLEDGE THAT THE EVALUATION VERSION OF STYLUS STUDIO CONTAINS A DISABLING DEVICE THAT WILL AUTOMATICALLY DISABLE THE PRODUCT FIFTEEN (15) DAYS FROM INSTALLATION. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN SECTION 1.9 (WARRANTIES), THE EVALUATION VERSION OF STYLUS STUDIO IS PROVIDED "AS IS" WITHOUT WARRANTY OF ANY KIND. WE MAKE NO WARRANTIES WITH RESPECT TO THE EVALUATION VERSION OF STYLUS STUDIO, EXPRESS, IMPLIED, OR ARISING BY CUSTOM OR TRADE USAGE, AND SPECIFICALLY MAKE NO WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE OR NONINFRINGEMENT.
- 4.3.4.4. If you ordered or obtained the Product license from an Authorized Reseller, then, in addition to the terms and conditions of this EULA, your use of the Product may be subject to additional scope of license terms and conditions (if any) specified in the Order between you and the Authorized Reseller, including, without limitation, a restriction that you limit your use of the Product to use solely in conjunction with the Authorized Reseller's software product(s) licensed by the Authorized Reseller to you. In the event of conflict between the terms and conditions specified in the Order between you and the Authorized Reseller and the terms of this EULA, the order of precedence will be as set forth in section 1.2.8 (Order of Precedence between EULA and Order).
- 4.3.4.5. The Product may contain programs that perform automated collection of system data and transmit said data to equipment owned by us or under our control in order to (i) register your activation of the Product, (ii) perform automated updating services for the Product, and/or (iii) validate whether you are entitled to use the Product. System data collected through such programs may be used by us for the purpose of providing you with remote system services and/or for the purpose of enforcing the terms and conditions of this EULA.
- 4.3.4.6. Notwithstanding anything to the contrary in this EULA, maintenance and support for Stylus Studio products is only available online via the Stylus Studio Developer Network (SSDN).

- 4.3.5. Progress OpenAccess Development Product. If the Product licensed to you is a Progress OpenAccess Development Product the following additional requirements apply: Except as expressly permitted pursuant to this EULA, neither the Progress OpenAccess Development Product nor any part thereof may be (i) embedded in another software program or application or (ii) used in connection with the operation of a software program or application in a production environment. Any use by you of the Progress OpenAccess Runtime Components included in the Progress OpenAccess Development Product must be limited to use solely for the purpose of developing and testing the operation of the Custom Driver in a development environment, and under no circumstances may such Progress OpenAccess Runtime Components be used or deployed for productive use. You must purchase a separate deployment license for the Progress OpenAccess Runtime Product (as defined above) in order to use and deploy said Progress OpenAccess Runtime Components in a production environment.
- 4.3.6. Progress OpenAccess Runtime Product. If the Product licensed to you is a Progress OpenAccess Runtime Product the following additional requirements apply:
- 4.3.6.1. You are permitted to make and deploy copies of the Progress OpenAccess Runtime Product strictly on an internal deployment basis for use in conjunction with the User Interface Provider to create a Custom Driver where such Custom Driver is used solely by you in an internal testing and/or production environment. If the Progress OpenAccess Runtime Product licensed to you is restricted to use for a specific application in accordance with section 3.4.5 (Application Specific License), then your use of the Custom Driver and Progress OpenAccess Runtime Product is limited to use in connection with the designated application. You are further subject to the license terms of the appropriate license model for the Progress OpenAccess Runtime Product as identified in the Order and as described in this EULA.
- 4.3.6.2. NOTWITHSTANDING THE PROVISIONS OF SECTION 1.9 (WARRANTIES), YOU UNDERSTAND AND ACKNOWLEDGE THAT THE PROGRESS OPENACCESS RUNTIME COMPONENTS DEPLOYED BY YOU IN CONJUNCTION WITH THE CUSTOM DRIVER ARE PROVIDED "AS IS" WITHOUT WARRANTY OF ANY KIND, WHETHER EXPRESS, IMPLIED OR ARISING BY CUSTOM OR TRADE USAGE AND WE SPECIFICALLY MAKES NO WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE OR NONINFRINGEMENT.
- 4.3.7. Progress Hybrid Data Pipeline Product. If the Product licensed to you is the Progress Hybrid Data Pipeline Product the following additional requirements apply:
- 4.3.7.1. The Product consists of four main parts: (1) the Hybrid Data Pipeline Server, (2) the Hybrid Data Pipeline client libraries (e.g., ODBC and/or JDBC), (3) the Hybrid Data Pipeline data source connectors and (4) the Hybrid Data Pipeline on-premises connector. The Product is licensed under a Core license model; however, the Core license unit quantity limitation specified in the Order will apply only to the Hybrid Data Pipeline Server portion of the Product. You may install and use the Hybrid Data Pipeline client libraries and Hybrid Data Pipeline data source connectors designated in the Order and, subject to subsection 4.3.7.3 below, the Hybrid Data Pipeline on-premises connector on an unlimited number of Servers with any number of Cores.
- 4.3.7.2. The Order will designate each Hybrid Data Pipeline client library licensed to you (e.g., ODBC, JDBC and/or OData). The Order will also identify a Hybrid Data Pipeline data source connector for each data source supported by us and that you are authorized to use the Product to connect to. Each license to the Product includes Hybrid Data Pipeline data source connectors for up to three (3) supported data sources to be selected by you and identified in the Order. For each data source identified in the Order, you may utilize the applicable Hybrid Data Pipeline data source connector and, subject to subsection 4.3.7.3 below, the Hybrid Data Pipeline on-premises connector to access and transmit data to and from an unlimited number of database instances of that data source. You may purchase a license for additional Hybrid Data Pipeline data source connectors to access additional supported data sources at our then-current pricing.
- 4.3.7.3. The Hybrid Data Pipeline on-premises connector will be included in the license to the Progress Hybrid Data Pipeline Product unless expressly excluded in the applicable Order.
- 4.3.7.4. The Hybrid Data Pipeline data source connectors and the Hybrid Data Pipeline on-premises connector (if applicable in accordance with subsection 4.3.7.3 above) may only be used in conjunction with the operation of the Hybrid Data Pipeline Server.
- 4.3.7.5. The Product licensed and delivered to you may include on the media or in the electronically delivered files, as applicable, Hybrid Data Pipeline data source connectors enabling connectivity to data source(s) other than those identified in the applicable Order and/or the Hybrid Data Pipeline on-premises connector despite its express exclusion in the applicable Order. You may only use the Progress Hybrid Data Pipeline data source connectors to connect to the data sources that are listed in the applicable Order and may not use the Hybrid Data Pipeline on-premises connector if it has been expressly excluded in the applicable Order.

End User License Agreement

Rev. TMPLT06JAN2020DC&I06JAN2020

Proj - © 2000, Frank Warmerdam

Licensed under the MIT License, which is available in this document here.

Python - © 1991-1995 by Stichting Mathematisch Centrum, Amsterdam, The Netherlands. All Rights Reserved.

PYTHON SOFTWARE FOUNDATION LICENSE VERSION 2

- 1. This LICENSE AGREEMENT is between the Python Software Foundation ("PSF"), and the Individual or Organization ("Licensee") accessing and otherwise using this software ("Python") in source or binary form and its associated documentation.
- 2. Subject to the terms and conditions of this License Agreement, PSF hereby grants Licensee a nonexclusive, royalty-free, world-wide license to reproduce, analyze, test, perform and/or display publicly, prepare derivative works, distribute, and otherwise use Python alone or in any derivative version, provided, however, that PSF's License Agreement and PSF's notice of copyright, i.e., "Copyright © 2001, 2002, 2003, 2004, 2005, 2006 Python Software Foundation; All Rights Reserved" are retained in Python alone or in any derivative version prepared by Licensee.
- 3. In the event Licensee prepares a derivative work that is based on or incorporates Python or any part thereof, and wants to make the derivative work available to others as provided herein, then Licensee hereby agrees to include in any such work a brief summary of the changes made to Python.
- 4. PSF is making Python available to Licensee on an "AS IS" basis. PSF MAKES NO REPRESENTATIONS OR WARRANTIES.

EXPRESS OR IMPLIED. BY WAY OF EXAMPLE, BUT NOT LIMITATION, PSF MAKES NO AND DISCLAIMS ANY REPRESENTATION OR WARRANTY OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE OR THAT THE USE OF PYTHON WILL NOT INFRINGE ANY THIRD PARTY RIGHTS.

- 5. PSF SHALL NOT BE LIABLE TO LICENSEE OR ANY OTHER USERS OF PYTHON FOR ANY INCIDENTAL, SPECIAL, OR CONSEQUENTIAL DAMAGES OR LOSS AS A RESULT OF MODIFYING, DISTRIBUTING, OR OTHERWISE USING PYTHON, OR ANY DERIVATIVE THEREOF, EVEN IF ADVISED OF THE POSSIBILITY THEREOF.
- 6. This License Agreement will automatically terminate upon a material breach of its terms and conditions.
- 7. Nothing in this License Agreement shall be deemed to create any relationship of agency, partnership, or joint venture between PSF and Licensee. This License Agreement does not grant permission to use PSF trademarks or trade name in a trademark sense to endorse or promote products or services of Licensee, or any third party.
- $8. \ \ By copying$, installing or otherwise using Python, Licensee agrees to be bound by the terms and conditions of this License Agreement. BEOPEN.COM LICENSE AGREEMENT FOR PYTHON 2.0

BEOPEN PYTHON OPEN SOURCE LICENSE AGREEMENT VERSION 1

- 1. This LICENSE AGREEMENT is between BeOpen.com ("BeOpen"), having an office at 160 Saratoga Avenue, Santa Clara, CA 95051, and the Individual or Organization ("Licensee") accessing and otherwise using this software in source or binary form and its associated documentation ("the Software").
- 2. Subject to the terms and conditions of this BeOpen Python License Agreement, BeOpen hereby grants Licensee a non-exclusive, royalty-free, world-wide license to reproduce, analyze, test, perform and/or display publicly, prepare derivative works, distribute, and otherwise use the Software alone or in any derivative version, provided, however, that the BeOpen Python License is retained in the Software, alone or in any derivative version prepared by Licensee.
- 3. BeOpen is making the Software available to Licensee on an "AS IS" basis. BEOPEN MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED. BY WAY OF EXAMPLE, BUT NOT LIMITATION, BEOPEN MAKES NO AND DISCLAIMS ANY REPRESENTATION OR WARRANTY OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE OR THAT THE USE OF THE SOFTWARE WILL NOT INFRINGE ANY THIRD PARTY RIGHTS.
- 4. BEOPEN SHALL NOT BE LIABLE TO LICENSEE OR ANY OTHER USERS OF THE SOFTWARE FOR ANY INCIDENTAL, SPECIAL, OR CONSEQUENTIAL DAMAGES OR LOSS AS A RESULT OF USING, MODIFYING OR DISTRIBUTING THE SOFTWARE, OR ANY DERIVATIVE THEREOF, EVEN IF ADVISED OF THE POSSIBILITY THEREOF.
- 5. This License Agreement will automatically terminate upon a material breach of its terms and conditions.
- 6. This License Agreement shall be governed by and interpreted in all respects by the law of the State of California, excluding conflict of law provisions. Nothing in this License Agreement shall be deemed to create any relationship of agency, partnership, or joint venture between BeOpen and Licensee. This License Agreement does not grant permission to use BeOpen trademarks or trade names in a trademark sense to endorse or promote products or services of Licensee, or any third party. As an exception, the "BeOpen Python" logos available at http://www.pythonlabs.com/logos.html may be used according to the permissions granted on that web page.
- 7. By copying, installing or otherwise using the software, Licensee agrees to be bound by the terms and conditions of this License Agreement. CNRI OPEN SOURCE LICENSE AGREEMENT (for Python 1.6b1) IMPORTANT: PLEASE READ THE FOLLOWING AGREEMENT CAREFULLY.

BY CLICKING ON "ACCEPT" WHERE INDICATED BELOW, OR BY COPYING, INSTALLING OR OTHERWISE USING PYTHON 1.6, beta 1 SOFTWARE, YOU ARE DEEMED TO HAVE AGREED TO THE TERMS AND CONDITIONS OF THIS LICENSE AGREEMENT.

- 1. This LICENSE AGREEMENT is between the Corporation for National Research Initiatives, having an office at 1895 Preston White Drive, Reston, VA 20191 ("CNRI"), and the Individual or Organization ("Licensee") accessing and otherwise using Python 1.6, beta 1 software in source or binary form and its associated documentation, as released at the www.python.org Internet site on August 4, 2000 ("Python 1.6b1").
- 2. Subject to the terms and conditions of this License Agreement, CNRI hereby grants Licensee a non-exclusive, royalty-free, world-wide license to reproduce, analyze, test, perform and/or display publicly, prepare derivative works, distribute, and otherwise use Python 1.6b1 alone or in any derivative version, provided, however, that CNRIs License Agreement is retained in Python 1.6b1, alone or in any derivative version prepared by Licensee.

Alternately, in lieu of CNRIs License Agreement, Licensee may substitute the following text (omitting the quotes): "Python 1.6, beta 1, is made available subject to the terms and conditions in CNRIs License Agreement. This Agreement may be located on the Internet using the following unique, persistent identifier (known as a handle): 1895.22/1011. This Agreement may also be obtained from a proxy server on the Internet using the URL:http://hdl.handle.net/1895.22/1011".

- 3. In the event Licensee prepares a derivative work that is based on or incorporates Python 1.6b1 or any part thereof, and wants to make the derivative work available to the public as provided herein, then Licensee hereby agrees to indicate in any such work the nature of the modifications made to Python 1.6b1.
- 4. CNRI is making Python 1.6b1 available to Licensee on an "AS IS" basis. CNRI MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED. BY WAY OF EXAMPLE, BUT NOT LIMITATION, CNRI MAKES NO AND DISCLAIMS ANY REPRESENTATION OR WARRANTY OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE OR THAT THE USE OF PYTHON 1.6b1 WILL NOT INFRINGE ANY THIRD PARTY RIGHTS.
- 5. CNRI SHALL NOT BE LIABLE TO LICENSEE OR ANY OTHER USERS OF THE SOFTWARE FOR ANY INCIDENTAL, SPECIAL, OR CONSEQUENTIAL DAMAGES OR LOSS AS A RESULT OF USING, MODIFYING OR DISTRIBUTING PYTHON 1.6b1, OR ANY DERIVATIVE THEREOF, EVEN IF ADVISED OF THE POSSIBILITY THEREOF.
- 6. This License Agreement will automatically terminate upon a material breach of its terms and conditions.
- 7. This License Agreement shall be governed by and interpreted in all respects by the law of the State of Virginia, excluding conflict of law provisions. Nothing in this License Agreement shall be deemed to create any relationship of agency, partnership,

or joint venture between CNRI and Licensee. This License Agreement does not grant permission to use CNRI trademarks or trade name in a trademark sense to endorse or promote products or services of Licensee, or any third party.

8. By clicking on the "ACCEPT" button where indicated, or by copying, installing or otherwise using Python 1.6b1, Licensee agrees to be bound by the terms and conditions of this License Agreement. ACCEPT CWI LICENSE AGREEMENT FOR PYTHON 0.9.0 THROUGH 1.2

Copyright © 1991 - 1995, Stichting Mathematisch Centrum Amsterdam, The Netherlands. All rights reserved.

Permission to use, copy, modify, and distribute this software and its documentation for any purpose and without fee is hereby granted, provided that the above copyright notice appear in all copies and that both that copyright notice and this permission notice appear in supporting documentation, and that the name of Stichting Mathematisch Centrum or CWI not be used in advertising or publicity pertaining to distribution of the software without specific, written prior permission.

STICHTING MATHEMATISCH CENTRUM DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE, INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS, IN NO EVENT SHALL STICHTING MATHEMATISCH CENTRUM BE LIABLE FOR ANY SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

Qhull - © 1993-1999. The National Science and Technology Research Center for Computation and Visualization of Geometric Structures (The Geometry Center), University of Minnesota, 400 Lind Hall, 207 Church Street S.E., Minneapolis, MN 55455 USA. Email: qhull@geom.umn.edu.

This software includes Qhull from The Geometry Center. Qhull is copyrighted as noted above. Qhull is free software and may be obtained via anonymous ftp from geom.umn.edu. It may be freely copied, modified, and redistributed under the following conditions:

- 1. All copyright notices must remain intact in all files.
- 2. A copy of this text file must be distributed along with any copies of Qhull that you redistribute: this includes copes that you have modified, or copies of programs or other software products that include Qhull.
- 3. If you modify qhull, you must include a notice giving the name of the person performing the modification, the date of modification, and the reason for such modification
- 4. When distributing modified versions of Qhull, or other software products that include Qhull, you must provide notice that the original source code may be obtained as noted above.
- 5. There is no warranty or other guarantee of fitness for Qhull, it is provided solely "as is." Bug reports or fixes may be sent to qhull bug@geom.umn.edu; the authors may or may not act on them as they desire.

Regex - © 1992, 1993, 1994, 1997 Henry Spencer. All rights reserved.

Permission is granted to anyone to use this software for any purpose on any computer system, and to alter it and redistribute it, subject to the following restrictions:

- 1. The author is not responsible for the consequences of use of this software, no matter how awful, even if they arise from flaws in it.
- 2. The origin of this software must not be misrepresented, either by explicit claim or my omission. Since few users ever read sources, credits must appear in the documentation.
- 3. Altered versions must be plainly marked as such, and must not be misrepresented as being the original software. Since few users ever read sources, credits must appear in the documentation.
- 4. This notice may not be removed or altered.

Services functions for InnoSetup 5.x (services.iss) 1.1 - Luigi D. Sandon © 2006-2010 Luigi D. Sandon. All Rights Reserved.

Licensed under MPL 1.1, a copy of which is available at https://www.mozilla.org/en-US/MPL/1.1/.

ShapeLib - © 1999 Frank Warmerdam

Licensed under the MIT License, which is available in this document here.

SQLite software - Harris Geospatial Solutions, Inc. is not the owner.

szip - © 2001 Science & Technology Corporation @ UNM. All rights released. Copyright © 2003 Lowell H. Miles and Jack A. Venbrux.

Licensed to ICs, LLC, for distribution by the University of Illinois' National Center for Supercomputing Applications as a part of the HDF data storage and retrieval file format and software library products package. All Rights Reserved. Do not modify or use for other purposes.

Szip implements an extended Rice adaptive lossless compression algorithm for sample data. the primary algorithm was developed by R. F. Rice at Jet Propulsion Laboraty.

Szip embodies certain inventions patented by the National Aeronautics and Space Administration,. United States Patens Nos. 5,448,642, 5,687,255, and 5,822,457 have been licensed to ICs, LLC, for distribution with the HDF data storage and retrieval file format and software library products. All Rights Reserved.

Licensed under the revocable, royalty-free, nonexclusive sublicense to sue SZIP decompression software routines and underlying patents is hereby granted by ICs, LLC, to all users of and in conjunction with the HDF storage and retrieval file format and software library products.

TBB - © 2005-2016 Intel Corporation. All Rights Reserved.

Intel® Threading Building Blocks (TBB), Copyright © 2005-2020 Intel Corporation. All Rights Reserved.

Intel® Threading Building Blocks is used under a commercial (COM) license for use in Harris Geospatial Solutions, Inc. products and services only. All other use is prohibited.

Open Source Software (OSS) distributions of Intel® Threading Building Blocks are available from https://threadingbuildingblocks.org.

Tiff - © 1988-1997 Sam Leffler. © 1991-1997 Silicon Graphics, Inc.

Permission to use, copy, modify, distribute, and sell this software and its documentation for any purpose is hereby granted without fee, provided that (i) the above copyright notices and this permission notice appear in all copies of the software and related documentation, and (ii) the names of Sam Leffler and Silicon Graphics may not be used in any advertising or publicity relating to the software without the specific, prior written permission of Sam Leffler and Silicon Graphics.

THE SOFTWARE IS PROVIDED "AS IS" AND WITHOUT WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR OTHERWISE, INCLUDING WITHOUT LIMITATION, ANY WARRANTY OR MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE.

IN NO EVENT SHALL SAM LEFFLER OR SILICON GRAPHICS BE LIABLE FOR ANY SPECIAL, INCIDENTAL, INDIRECT OR CONSEQUENTIAL DAMAGES OF ANY KIND, OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER OR NOT ADVISED OF THE POSSIBILITY OF DAMAGE, AND ON ANY THEORY OF LIABILITY, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

Ultimate Grid - IDL software contains source code that is @ 1994, 1995 Dundas Software Ltd. All Rights Reserved.

wglext - © 2013-2017 The Khronos Group Inc.

Licensed under the MIT License, which is available in this document here.

Xdr - © 1997-1999 Sun Microsystems, Inc. 901 San Antonio Road, Palo Alto, California 94303 U.S.A.

Sun RPC is a product of Sun Microsystems, Inc. and is provided for unrestricted use provided that this legend is included on all tape media and as a part of the software program in whole or part. Users may copy or modify Sun RPC without charge, but are not authorized to license or distribute it to anyone else except as part of a product or program developed by the user.

SUN RPC IS PROVIDED AS IS WITH NO WARRANTIES OF ANY KIND INCLUDING THE WARRANTIES OF DESIGN, MERCHANTIBILITY AND FITNESS FOR A PARTICULAR PURPOSE, OR ARISING FROM A COURSE OF DEALING, USAGE OR TRADE PRACTICE.

Sun RPC is provided with no support and without any obligation on the part of Sun Microsystems, Inc. to assist in its use, correction, modification or enhancement.

SUN MICROSYSTEMS, INC. SHALL HAVE NO LIABILITY WITH RESPECT TO THE INFRINGEMENT OF COPYRIGHTS, TRADE SECRETS OR ANY PATENTS BY SUN RPC OR ANY PART THEREOF.

In no event will Sun Microsystems, Inc. be liable for any lost revenue or profits or other special, indirect and consequential damages, even if Sun has been advised of the possibility of such damages.

Xdt - © 1995, 1996 by Software Komponents International and Robert S. Werner. All Rights Reserved.

LICENSE FOR THE XDT LIBRARY

PREAMBLE

The Xdt library and all its supporting programs and documentation are:

Copyright © 1995, 1996 by Software Komponents International and Robert S. Werner. All Rights Reserved.

Xdt is not in the public domain. See the file COPYRIGHT for more information on the copyright.

Xdt is provided "as is" with no warranty of any kind. See the file NO WARRANTY for more information.

The Xdt library may only be used under the terms of this license. No fee is required to obtain a license of this version of the Xdt library. Registration is required. See the file REGISTRATION for details.

You'll find that this license gives you a lot of flexibility in the ways you use Xdt. It does not allow you, however, to sell Xdt or modified versions of Xdt, or to distribute Xdt as part of toolkits of your own.

This license below is a non-standard one, and is longer and more detailed than many. This is because I am trying to give the software developer as much freedom as possible to use Xdt, and at the same time protect some key commercial rights of my own. Indented paragraphs are the formal definitions, terms and conditions of this license. Text that is flush left is less formal commentary on those definitions, terms and conditions.

DEFINITIONS

A "licensed user" is anyone who has registered to use the Xdt library.

A "library" means a collection of software functions and/or data prepared so as to be conveniently linked with application programs (which use some of those functions and data) to form executables.

"Source code" for a work means the preferred form of the work for making modifications to it.

"Object code" is a compiled, machine-readable form of a work, which has not been linked with libraries or other object code to form an executable.

"Linking" is the process of combining modules of object code and libraries into a form that can be executed by a computer.

An "executable" is software that has been linked.

"Portions of the library" mean a subset of the software functions that

comprise the library which have been extracted from the library and are distributed directly as part of an application, and not as part of a library or as a library themselves.

TERMS AND CONDITIONS

1) Permission is granted to copy and distribute unmodified versions of the Xdt library for any non-commercial purpose. This distribution must be under terms identical to those set forth here.

Paragraph 1 of the license covers only distribution of the unmodified library: anyone may give an unmodified copy of this library to anyone who wants it, and anyone may also make the unmodified library available for anonymous ftp, for example. You can't charge money to make copies for people, and you can't distribute modified versions of the library.

"Unmodified" means that you have to include all files in the distribution, including those that contain this license. You don't have to be a licensed user to distribute unmodified copies, but note that you aren't allowed to actually use the library unless you are licensed.

2) Permission is granted to licensed users to use and modify the Xdt library, on a single CPU, for any purpose.

Paragraph 2 explains that licensed users can use and modify Xdt however they want. Note, though, that distribution of programs linked with Xdt or modified versions of Xdt is covered separately in the paragraphs below.

3) Permission is granted to licensed users to freely distribute executable programs that have been linked with the Xdt library or with modified versions of the library.

Paragraph 3 says that if licensed users link an application with this library, they can distribute the executable freely; there's no run-time royalty or any other charges.

4) Permission is granted to licensed users to distribute modified or unmodified portions of the Xdt library, in source code or object code form, as part of an application which is itself distributed in source or object code form, provided that the files LICENSE, COPYRIGHT, REGISTRATION and NO_WARRANTY from the Xdt library are included with the application distribution, and provided that any code derived from Xdt is clearly labeled as such. Documentation for the Xdt library, or for any portion of it, may not be distributed under the terms of this section.

Much useful software is distributed in source-code form. Paragraph 4 allows you to distribute applications that use Xdt in source code form without requiring all the users of the application to pay a license fee for Xdt: if you write an application that uses only some of modules from this library, or uses modified modules from the library, and you are distributing the application in source code form, then you can include the source code modules from Xdt or derived from Xdt with the application. You must include the files that explain this license if you do so. Note that you can only distribute portions of the library in this way--not the whole library--and that those portions must be part of an application--they may not form a library themselves. You may not distribute documentation for the portions of the library you distribute under the terms of this section--since your users need only to compile and run your application, there is no need for the Xdt programmer documentation. Distribution of the documentation might encourage modification or reuse of the distributed portions of Xdt, which is forbidden in Paragraph 5.

5) Permission is granted to use, without fee, portions of the Xdt library distributed with an application within that application only.

Paragraph 4 allows application developers to distribute portions of the Xdt library with their applications. Paragraph 5 specifies the terms under which those portions may be used by those who receive the distribution: anyone who receives portions of the Xdt library as part of an application may compile and run that application without becoming a licensed user. But no one may modify those portions, or reuse them in another application without becoming a licensed user.

6) Any permissions not explicitly granted above are reserved by Software Komponents International.

Paragraph 6 says that the paragraphs above specify the only ways that you are allowed to use, modify, and distribute the Xdt library. In particular, note that you are not permitted to distribute modified versions of Xdt without the prior written permission of Software Komponents International or myself (Robert S. Werner); this is to prevent a proliferation of versions and avoid confusion over which version is the "official" one. Also, while you are allowed to distribute portions of the library in or with your applications, you are not allowed to distribute portions of the library in your own libraries or toolkits, or in any way that would allow them to be reused, unless you obtain prior written permission to do so. For example, if you write an interface builder application, you may use the widgets from the Xdt library in your application's executable, but you may not include the source or object code for those widgets in any form that the users of your builder application can use in their applications. Instead, you must require your users to obtain a license for the Xdt library in order to use the Xdt widgets.

Xpm - © 1989-95 GROUPE BULL. Copyright © 1998 Arnaud LE HORS Copyright © 1996 Lorens Younes.

- © 1989-95 GROUPE BULL
- © 1998 Arnaud LE HORS
- © 1996 Lorens Younes

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED 'AS IS', WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL GROUPE BULL BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Except as contained in this notice, the name of GROUPE BULL shall not be used in advertising or otherwise to promote the sale,

use or other dealings in this Software without prior written authorization from GROUPE BULL.

Except as contained in this notice, the name of Arnaud LE HORS shall not be used in advertising or otherwise to promote the sale, use or other dealings in this Software without prior written authorization from Arnaud LE HORS.

Except as contained in this notice, the name of Lorens Younes shall not be used in advertising or otherwise to promote the sale, use or other dealings in this Software without prior written authorization from Lorens Younes.

Xprinter® is a registered trademark of Bristol Technologies, Inc. © 1990,1991,1992,1993,1994 Bristol Technology Inc.Property of Bristol Technologies Inc.All rights reserved.

Zlib

libHaru 2.1.0 - © 1999-2006 Takeshi Kanno © 2007-2008 Antony Dovgal

Tinyxml2 8.0.0

zlib 1.2.11 - Copyright © 1995-2013 Jean-loup Gailly and Mark Adler

zlib License Copyright © <year> <copyright holders>

This software is provided 'as-is', without any express or implied warranty. In no event will the authors be held liable for any damages arising from the use of this software.

Permission is granted to anyone to use this software for any purpose, including commercial applications, and to alter it and redistribute it freely, subject to the following restrictions:

- 1. The origin of this software must not be misrepresented; you must not claim that you wrote the original software. If you use this software in a product, an acknowledgment in the product documentation would be appreciated but is not required.
- 2. Altered source versions must be plainly marked as such, and must not be misrepresented as being the original software.
- 3. This notice may not be removed or altered from any source distribution.

 $\ensuremath{\text{@}}$ 1988-2021 Harris Geospatial Solutions, Inc. All Rights Reserved.

This information is not subject to the controls of the International Traffic in Arms Regulations (ITAR) or the Export Administration Regulations (EAR).

Licensing Information for Intel(R) Software Development Tools

The following list of Intel(R) Software Development Tools and associated licenses is for reference only. The controlling license for your Intel(R) Software Development Tools will be bundled with the distribution of the tool.

Additionally, third party software, even if included with the distribution of the Intel(R) Software Development Tools, may be governed by separate license terms, including without limitation, third party license terms, open source software notices and terms, and/or other Intel software license terms. These separate license terms solely govern your use of the third party software. Additional information on third party software can be found in the Section entitled 'Toolkits for oneAPI and Associated Third Party Program Files.'

Intel(R) Software Development Tools Licensed under the Intel End User License Agreement for Developer Tools (Version October 2021 or later)

- Installer Program
- Intel(R) Advisor
- Intel(R) C++ Compiler
- Intel(R) DPC++ Compatibility Tool
- Intel(R) Fortran Compiler
- Intel(R) Inspector
- Intel(R) oneAPI DPC++/C++ Compiler
- Intel(R) Trace Analyzer and Collector
- Intel(R) Vtune(TM) Profiler

Intel(R) Software Development Tools Licensed under the Intel Simplified Software License (Version August 2021 or later)

- Intel(R) Distribution for Python*
- Intel(R) Integrated Performance Primitives (Intel(R) IPP)
- Intel(R) Integrated Performance Primitives Cryptography (Intel(R) IPP Cryptography)
- Intel(R) oneAPI Math Kernel Library (oneMKL)
- Intel(R) MPI Library
- Intel(R) oneAPI Data Analytics Library (oneDAL)
- Intel(R) oneAPI Threading Building Blocks (oneTBB)

Intel(R) Software Development Tools Licensed under Open Source Licenses Apache License, Version 2.0

- Intel(R) Embree
- Intel(R) oneAPI Deep Neural Network Library (oneDNN)
- Intel(R) Open Image Denoise
- Intel(R) Open Volume Kernel Library (Intel(R) Open VKL)
- Intel(R) OSPRay
- Intel(R) OSPRay Studio
- Rendering Toolkit Utilities

Apache License, Version 2.0 with LLVM Exception

• Intel(R) oneAPI DPC++ Library (oneDPL)

3-Clause BSD

- Diagnostics Utility for Intel(R) oneAPI Toolkits
- Intel(R) Dev Utilities

The GNU General Public License v3.0

• Intel(R) Distribution for GDB

MIT License

• Intel(R) oneAPI Video Processing Library (oneVPL)

License Text Files (subject to change)

Intel End User License Agreement for Developer Tools (Version October 2021)

IMPORTANT NOTICE - PLEASE READ AND AGREE BEFORE DOWNLOADING, INSTALLING, COPYING OR USING

This Agreement is between you, or the company or other legal entity that you represent and warrant you have the legal authority to bind, (each, "You" or "Your") and Intel Corporation and its subsidiaries (collectively, "Intel") regarding Your use of the Materials. By downloading, installing, copying or using the Materials, You agree to be bound by the terms of this Agreement. If You do not agree to the terms of this Agreement, or do not have legal authority or required age to agree to them, do not download, install, copy or use the Materials.

1. LICENSE DEFINITIONS.

- A. "Cloud Provider" means a third party service provider offering a cloud-based platform, infrastructure, application or storage services, such as Microsoft Azure or Amazon Web Services, which You may only utilize to host the Materials subject to the restrictions set forth in Section 2.3 B.
- B. "Derivative Work" means a derivative work, as defined in 17 U.S.C. 101, of the Source Code.
- C. "Executable Code" means computer programming code in binary form suitable for machine execution by a processor without the intervening steps of interpretation or compilation.
- D. "Materials" mean the software, documentation, the software product serial number, and other collateral, including any updates, made available to You by Intel under this Agreement. Materials include Redistributables, Executable Code, Source Code, Sample Source Code, and Pre-Release Materials, but do not include Third Party Software.
- E. "Pre-Release Materials" mean the Materials, or portions of the Materials, that are identified (in the product release notes, on Intel's download website for the Materials or elsewhere) or labeled as pre-release, prototype, alpha or beta code and, as such, are deemed to be pre-release code (i) which may not be fully functional or tested and may contain bugs or errors; (ii) which Intel may substantially modify in its development of a production version; or (iii) for which Intel makes no assurances that it will ever develop or make a production version generally available. Pre-Release Materials are subject to the .terms of Section 3.2.
- F. "Reciprocal Open Source Software" means any software that is subject to a license which requires that (i) it must be distributed in source code form; (ii) it must be licensed under the same open source license terms; and (iii) its derivative works must be licensed under the same open source license terms. Examples of this type of license are the GNU General Public License or the Mozilla Public License.
- G. "Redistributables" mean the files (if any) listed in the "redist.txt," "redist-rt.txt" or similarly-named text files that may be included in the Materials. Redistributables include Sample Source Code.
- H. "Sample Source Code" means those portions of the Materials that are Source Code and are identified as sample code. Sample Source Code may not have been tested or validated by Intel and is provided purely as a programming example.
- I. "Source Code" means the software portion of the Materials provided in human readable format.
- J. "Third Party Software" mean the files (if any) listed in the "third-party-software.txt" or other similarly-named text file that may be included in the Materials for the applicable software. Third Party Software is subject to the terms of Section 2.2.
- K. "Your Product" means one or more applications, products or projects developed by or for You using the Materials.

2. LICENSE GRANTS.

- 2.1 License to the Materials. Subject to the terms and conditions of this Agreement, Intel grants You a non-exclusive, worldwide, non-assignable, non-sublicensable, limited right and license under its copyrights, to:
- A. reproduce internally a reasonable number of copies of the Materials for Your personal or business use;
- B. use the Materials solely for Your personal or business use to develop Your Product, in accordance with

the documentation included as part of the Materials;

- C. modify or create Derivative Works only of the Redistributables, or any portions, that are provided to You in Source Code;
- D. distribute (directly and through Your distributors, resellers, and other channel partners, if applicable), the Redistributables, including any modifications to or Derivative Works of the Redistributables or any portions made pursuant to Section 2.1.C subject to the following conditions:
- (1) Any distribution of the Redistributables must only be as part of Your Product which must add significant primary functionality different than that of the Redistributables themselves;
- (2) You must only distribute the Redistributables originally provided to You by Intel only in Executable Code subject to a license agreement that prohibits reverse engineering, decompiling or disassembling the Redistributables;
- (3) This distribution right includes a limited right to sublicense only the Intel copyrights in the Redistributables and only to the extent necessary to perform, display, and distribute the Redistributables (including Your modifications and Derivative Works of the Redistributables provided in Source Code) solely as incorporated in Your Product; and
- (4) You: (i) will be solely responsible to Your customers for any update, support obligation or other obligation or liability which may arise from the distribution of Your Product, (ii) will not make any statement that Your Product is "certified" or that its performance is guaranteed by Intel or its suppliers, (iii) will not use Intel's or its suppliers' names or trademarks to market Your Product, (iv) will comply with any additional restrictions which are included in the text files with the Redistributables and in Section 3 below, (v) will indemnify, hold harmless, and defend Intel and its suppliers from and against any claims or lawsuits, costs, damages, and expenses, including attorney's fees, that arise or result from (a) Your modifications or Derivative Works of the Materials or (b) Your distribution of Your Product.
- 2.2 Third Party Software. Third Party Software, even if included with the distribution of the Materials, may be governed by separate license terms, including without limitation, third party license terms, open source software notices and terms, and/or other Intel software license terms. These separate license terms solely govern Your use of the Third Party Software.

2.3 Third Party Use.

- A. If You are an entity, Your contractors may use the Materials under the license specified in Section 2, provided: (i) their use of the Materials is solely on behalf of and in support of Your business, (ii) they agree to the terms and conditions of this Agreement, and (iii) You are solely responsible for their use, misuse or disclosure of the Materials.
- B. You may utilize a Cloud Provider to host the Materials for You, provided: (i) the Cloud Provider may only host the Materials for Your exclusive use and may not use the Materials for any other purpose whatsoever, including the restriction set forth in Section 3.1(xi); (ii) the Cloud Provider's use of the Materials must be solely on behalf of and in support of Your Product, and (iii) You will indemnify, hold harmless, and defend Intel and its suppliers from and against any claims or lawsuits, costs, damages, and expenses, including attorney's fees, that arise or result from Your Cloud Provider's use, misuse or disclosure of the Materials.

3. LICENSE CONDITIONS.

- 3.1 Restrictions. Except as expressly provided in this Agreement, You may NOT: (i) use, reproduce, disclose, distribute, or publicly display the Materials; (ii) share, publish, rent or lease the Materials to any third party; (iii) assign this Agreement or transfer the Materials; (iv) modify, adapt, or translate the Materials in whole or in part; (v) reverse engineer, decompile, or disassemble the Materials, or otherwise attempt to derive the source code for the software; (vi) work around any technical limitations in the Materials; (vii) distribute, sublicense or transfer any Source Code, modifications or Derivative Works of any Source Code to any third party; (viii) remove, minimize, block or modify any notices of Intel or its suppliers in the Materials; (ix) include the Redistributables in malicious, deceptive, or unlawful programs or products or use the Materials in any way that is against the law; (x) modify, create a Derivative Work, link, or distribute the Materials so that any part of it becomes Reciprocal Open Source Software; (xi) use the Materials directly or indirectly for SaaS services or service bureau purposes (i.e., a service that allows use of or access to the Materials by a third party as part of that service, such as the salesforce.com service business model).
- 3.2 Pre-Release Materials. If You receive Pre-Release Materials, You may reproduce a reasonable number of copies and use the Pre-Release Materials for evaluation and testing purposes only. You may

- not (i) modify or incorporate the Pre-Release Materials into Your Product; (ii) continue to use the Pre-Release Materials once a commercial version is released; or (iii) disclose to any third party any benchmarks, performance results, or other information relating to the Pre-Release Materials. Intel may waive these restrictions in writing at its sole discretion; however, if You decide to use the Pre-Release Materials in Your Product (even with Intel's waiver), You acknowledge and agree that You are fully responsible for any and all issues that result from such use.
- 3.3 Safety-Critical, and Life-Saving Applications; Indemnity. The Materials may provide information relevant to safety-critical applications ("Safety-Critical Applications") to allow compliance with functional safety standards or requirements. You acknowledge and agree that safety is Your responsibility. To the extent You use the Materials to create, or as part of, products used in Safety-Critical Applications, it is Your responsibility to design, manage, and ensure that there are system-level safeguards to anticipate, monitor, and control system failures, and You agree that You are solely responsible for all applicable regulatory standards and safety-related requirements concerning Your use of the Materials in Safety Critical Applications.

Should You use the Materials for Safety-Critical Applications or in any type of a system or application in which the failure of the Materials could create a situation where personal injury or death may occur (e.g., medical systems, life-sustaining or life-saving systems) ("Life-Saving Applications"), You agree to indemnify, defend, and hold Intel and its representatives harmless against any claims or lawsuits, costs, damages, and expenses, including reasonable attorney fees, arising in any way out of Your use of the Materials in Safety-Critical Applications or Life-Saving Applications and claims of product liability, personal injury or death associated with those applications; even if such claims allege that Intel was negligent or strictly liable regarding the design or manufacture of the Materials or its failure to warn regarding the Materials.

3.4 Media Format Codecs and Digital Rights Management. You acknowledge and agree that Your use of the Materials or distribution of the Redistributables with Your Product as permitted by this Agreement may require You to procure license(s) from third parties that may hold intellectual property rights applicable to any media decoding, encoding or transcoding technology (e.g., the use of an audio or video codec) and/or digital rights management capabilities of the Materials, if any. Should any such additional licenses be required, You are solely responsible for obtaining any such licenses and agree to obtain any such licenses at Your own expense.

4. DATA COLLECTION AND PRIVACY.

- 4.1 Data Collection. The Materials may generate and collect anonymous data and/or provisioning data about the Materials and/or the development environment and transmit the data to Intel as a one-time event during installation. Optional data may also be collected by the Materials, however, You will be provided notice of the request to collect optional data and no optional data will be collected without Your consent. All data collection by Intel is performed pursuant to relevant privacy laws, including notice and consent requirements.
- 4.2 Intel's Privacy Notice. Intel is committed to respecting Your privacy. To learn more about Intel's privacy practices, please visit http://www.intel.com/privacy.
- 5. OWNERSHIP. Title to the Materials and all copies remain with Intel or its suppliers. The Materials are protected by intellectual property rights, including without limitation, United States copyright laws and international treaty provisions. You will not remove any copyright or other proprietary notices from the Materials. Except as expressly provided herein, no license or right is granted to You directly or by implication, inducement, estoppel or otherwise; specifically, Intel does not grant any express or implied right to You under Intel patents, copyrights, trademarks, or trade secrets.

6. NO WARRANTY AND NO SUPPORT.

- 6.1 No Warranty. Disclaimer. Intel disclaims all warranties of any kind and the terms and remedies provided in this Agreement are instead of any other warranty or condition, express, implied or statutory, including those regarding merchantability, fitness for any particular purpose, non-infringement or any warranty arising out of any course of dealing, usage of trade, proposal, specification or sample. Intel does not assume (and does not authorize any person to assume on its behalf) any liability.
- 6.2 No Support; Priority Support. Intel may make changes to the Materials, or to items referenced therein, at any time without notice, but is not obligated to support, update or provide training for the Materials under the terms of this Agreement. Intel offers free community and paid priority support options. More information on these support options can be found at: https://software.intel.com/content/www/us/en/develop/support/priority-support.html.

7. LIMITATION OF LIABILITY.

- 7.1 Intel will not be liable for any of the following losses or damages (whether such losses or damages were foreseen, foreseeable, known or otherwise): (i) loss of revenue; (ii) loss of actual or anticipated profits; (iii) loss of the use of money; (iv) loss of anticipated savings; (v) loss of business; (vi) loss of opportunity; (vii) loss of goodwill; (viii) loss of use of the Materials; (ix) loss of reputation; (x) loss of, damage to, or corruption of data; or (xi) any indirect, incidental, special or consequential loss of damage however caused (including loss or damage of the type specified in this Section 7).
- 7.2 Intel's total cumulative liability to You, including for direct damages for claims relating to this Agreement, and whether for breach of contract, negligence, or for any other reason, will not exceed \$100.
- 7.3 You acknowledge that the limitations of liability provided in this Section 7 are an essential part of this Agreement. You agree that the limitations of liability provided in this Agreement with respect to Intel will be conveyed to and made binding upon any customer of Yours that acquires the Redistributables.
- 8. USER SUBMISSIONS. Should you provide Intel with comments, modifications, corrections, enhancements or other input ("Feedback") related to the Materials, Intel will be free to use, disclose, reproduce, license or otherwise distribute or exploit the Feedback in its sole discretion without any obligations or restrictions of any kind, including without limitation, intellectual property rights or licensing obligations. If You wish to provide Intel with information that You intend to be treated as confidential information, Intel requires that such confidential information be provided pursuant to a non-disclosure agreement ("NDA"); please contact Your Intel representative to ensure the proper NDA is in place.
- 9. NON-DISCLOSURE. Information provided by Intel to You may include information marked as confidential. You must treat such information as confidential under the terms of the applicable NDA between Intel and You. If You have not entered into an NDA with Intel, You must not disclose, distribute or make use of any information marked as confidential, except as expressly authorized in writing by Intel. Intel retains all rights in and to its confidential information specifications, designs, engineering details, discoveries, inventions, patents, copyrights, trademarks, trade secrets, and other proprietary rights relating to the Materials. Any breach by You of the confidentiality obligations provided for in this Section 9 will cause irreparable injury to Intel for which money damages may be inadequate to compensate Intel for losses arising from such a breach. Intel may obtain equitable relief, including injunctive relief, if You breach or threaten to breach Your confidentiality obligations.
- 10. TERM AND TERMINATION. This Agreement becomes effective on the date You accept this Agreement and will continue until terminated as provided for in this Agreement. The term for any Pre-Release Materials terminates upon release of a commercial version. This Agreement will terminate if You are in breach of any of its terms and conditions. Upon termination, You will promptly destroy the Materials and all copies. In the event of termination of this Agreement, Your license to any Redistributables distributed by You in accordance with the terms and conditions of this Agreement, prior to the effective date of such termination, will survive any such termination of this Agreement. Sections 1, 2.1.D(4)(v), 2.2, 2.3.A(iii), 2.3.B(iii), 3.3, 5, 6, 7, 8, 9, 10 (with respect to these survival provisions in the last sentence), and 12 will survive expiration or termination of this Agreement.
- 11. U.S. GOVERNMENT RESTRICTED RIGHTS. The technical data and computer software covered by this license is a "Commercial Item," as such term is defined by the FAR 2.101 (48 C.F.R. 2.101) and is "commercial computer software" and "commercial computer software documentation" as specified under FAR 12.212 (48 C.F.R. 12.212) or DFARS 227.7202 (48 C.F.R. 227.7202), as applicable. This commercial computer software and related documentation is provided to end users for use by and on behalf of the U.S. Government with only those rights as are granted to all other end users pursuant to the terms and conditions of this Agreement.

12. GENERAL PROVISIONS.

12.1 ENTIRE AGREEMENT. This Agreement contains the complete and exclusive agreement and understanding between the parties concerning the subject matter of this Agreement, and supersedes all prior and contemporaneous proposals, agreements, understanding, negotiations, representations, warranties, conditions, and communications, oral or written, between the parties relating to the same subject matter. Each party acknowledges and agrees that in entering into this Agreement it has not relied on, and will not be entitled to rely on, any oral or written representations, warranties, conditions, understanding, or communications between the parties that are not expressly set forth in this Agreement. The express provisions of this Agreement control over any course of performance, course of dealing, or usage of the trade inconsistent with any of the provisions of this Agreement. The provisions of this Agreement will prevail notwithstanding any different, conflicting, or additional provisions that may

appear on any purchase order, acknowledgement, invoice, or other writing issued by either party in connection with this Agreement. No modification or amendment to this Agreement will be effective unless in writing and signed by authorized representatives of each party, and must specifically identify this Agreement by its title and version (e.g., "Intel End User License Agreement for Developer Tools (Version October 2021)"); except that Intel may make changes to this Agreement as it distributes new versions of the Materials. When changes are made, Intel will make a new version of the Agreement available on its website. If You received a copy of this Agreement translated into another language, the English language version of this Agreement will prevail in the event of any conflict between versions.

- 12.2 EXPORT. You acknowledge that the Materials and all related technical information are subject to export controls and you agree to comply with all laws and regulations of the United States and other applicable governments governing export, re-export, import, transfer, distribution, and use of the Materials. In particular, but without limitation, the Materials may not be exported or re-exported (i) into any U.S. embargoed countries or (ii) to any person or entity listed on a denial order published by the U.S. government or any other applicable governments. By using the Materials, You represent and warrant that You are not located in any such country or on any such list. You also agree that You will not use the Materials for, or sell or transfer them to a third party who is known or suspected to be involved in, any purposes prohibited by the U.S. government or other applicable governments, including, without limitation, the development, design, manufacture, or production of nuclear, missile, chemical or biological weapons.
- 12.3 GOVERNING LAW, JURISDICTION, AND VENUE. All disputes arising out of or related to this Agreement, whether based on contract, tort, or any other legal or equitable theory, will in all respects be governed by, and construed and interpreted under, the laws of the United States of America and the State of Delaware, without reference to conflict of laws principles. The parties agree that the United Nations Convention on Contracts for the International Sale of Goods (1980) is specifically excluded from and will not apply to this Agreement. All disputes arising out of or related to this Agreement, whether based on contract, tort, or any other legal or equitable theory, will be subject to the exclusive jurisdiction of the courts of the State of Delaware or of the Federal courts sitting in that State. Each party submits to the personal jurisdiction of those courts and waives all objections to that jurisdiction and venue for those disputes.
- 12.4 SEVERABILITY. The parties intend that if a court holds that any provision or part of this Agreement is invalid or unenforceable under applicable law, the court will modify the provision to the minimum extent necessary to make it valid and enforceable, or if it cannot be made valid and enforceable, the parties intend that the court will sever and delete the provision or part from this Agreement. Any change to or deletion of a provision or part of this Agreement under this Section will not affect the validity or enforceability of the remainder of this Agreement, which will continue in full force and effect.

Intel Simplified Software License (Version August 2021)

Use and Redistribution. You may use and redistribute the software (the "Software"), without modification, provided the following conditions are met:

- * Redistributions must reproduce the above copyright notice and the following terms of use in the Software and in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel nor the names of its suppliers may be used to endorse or promote products derived from this Software without specific prior written permission.
- * No reverse engineering, decompilation, or disassembly of this Software is permitted.

No other licenses. Except as provided in the preceding section, Intel grants no licenses or other rights by implication, estoppel or otherwise to, patent, copyright, trademark, trade name, service mark or other intellectual property licenses or rights of Intel.

Third party software. The Software may contain Third Party Software. "Third Party Software" is open source software, third party software, or other Intel software that may be identified in the Software itself or in the files (if any) listed in the "third-party-software.txt" or similarly named text file included with the Software. Third Party Software, even if included with the distribution of the Software, may be governed by separate license terms, including without limitation, open source software license terms, third party software license terms, and other Intel software license terms. Those separate license terms solely govern your use of the Third Party Software, and nothing in this license limits any rights under, or grants rights that supersede, the terms of the applicable license terms.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE DISCLAIMED. THIS SOFTWARE IS

NOT INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE PERSONAL INJURY OR DEATH AND YOU AGREE THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS, COSTS, DAMAGES, EXPENSES, AND ATTORNEYS FEES ARISING OUT OF ANY SUCH USE, EVEN IF ANY CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE SOFTWARE.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

No support. Intel may make changes to the Software, at any time without notice, and is not obligated to support, update or provide training for the Software.

Termination. Your right to use the Software is terminated in the event of your breach of this license.

Feedback. Should you provide Intel with comments, modifications, corrections, enhancements or other input ("Feedback") related to the Software, Intel will be free to use, disclose, reproduce, license or otherwise distribute or exploit the Feedback in its sole discretion without any obligations or restrictions of any kind, including without limitation, intellectual property rights or licensing obligations.

Compliance with laws. You agree to comply with all relevant laws and regulations governing your use, transfer, import or export (or prohibition thereof) of the Software.

Governing law. All disputes will be governed by the laws of the United States of America and the State of Delaware without reference to conflict of law principles and subject to the exclusive jurisdiction of the state or federal courts sitting in the State of Delaware, and each party agrees that it submits to the personal jurisdiction and venue of those courts and waives any objections. The United Nations Convention on Contracts for the International Sale of Goods (1980) is specifically excluded and will not apply to the Software.

Apache License

Version 2.0, January 2004

http://www.apache.org/licenses/

TERMS AND CONDITIONS FOR USE, REPRODUCTION, AND DISTRIBUTION

1. Definitions.

"License" shall mean the terms and conditions for use, reproduction, and distribution as defined by Sections 1 through 9 of this document.

"Licensor" shall mean the copyright owner or entity authorized by the copyright owner that is granting the License.

"Legal Entity" shall mean the union of the acting entity and all other entities that control, are controlled by, or are under common control with that entity. For the purposes of this definition, "control" means (i) the power, direct or indirect, to cause the direction or management of such entity, whether by contract or otherwise, or (ii) ownership of fifty percent (50%) or more of the outstanding shares, or (iii) beneficial ownership of such entity.

"You" (or "Your") shall mean an individual or Legal Entity exercising permissions granted by this License.

"Source" form shall mean the preferred form for making modifications, including but not limited to software source code, documentation source, and configuration files.

"Object" form shall mean any form resulting from mechanical transformation or translation of a Source form, including but not limited to compiled object code, generated documentation, and conversions to other media types.

"Work" shall mean the work of authorship, whether in Source or Object form, made available under the

License, as indicated by a copyright notice that is included in or attached to the work (an example is provided in the Appendix below).

"Derivative Works" shall mean any work, whether in Source or Object form, that is based on (or derived from) the Work and for which the editorial revisions, annotations, elaborations, or other modifications represent, as a whole, an original work of authorship. For the purposes of this License, Derivative Works shall not include works that remain separable from, or merely link (or bind by name) to the interfaces of, the Work and Derivative Works thereof.

"Contribution" shall mean any work of authorship, including the original version of the Work and any modifications or additions to that Work or Derivative Works thereof, that is intentionally submitted to Licensor for inclusion in the Work by the copyright owner or by an individual or Legal Entity authorized to submit on behalf of the copyright owner. For the purposes of this definition, "submitted" means any form of electronic, verbal, or written communication sent to the Licensor or its representatives, including but not limited to communication on electronic mailing lists, source code control systems, and issue tracking systems that are managed by, or on behalf of, the Licensor for the purpose of discussing and improving the Work, but excluding communication that is conspicuously marked or otherwise designated in writing by the copyright owner as "Not a Contribution."

"Contributor" shall mean Licensor and any individual or Legal Entity on behalf of whom a Contribution has been received by Licensor and subsequently incorporated within the Work.

- 2. Grant of Copyright License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare Derivative Works of, publicly display, publicly perform, sublicense, and distribute the Work and such Derivative Works in Source or Object form.
- 3. Grant of Patent License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable (except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by such Contributor that are necessarily infringed by their Contribution(s) alone or by combination of their Contribution(s) with the Work to which such Contribution(s) was submitted. If You institute patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Work or a Contribution incorporated within the Work constitutes direct or contributory patent infringement, then any patent licenses granted to You under this License for that Work shall terminate as of the date such litigation is filed.
- 4. Redistribution. You may reproduce and distribute copies of the Work or Derivative Works thereof in any medium, with or without modifications, and in Source or Object form, provided that You meet the following conditions:
- (a) You must give any other recipients of the Work or Derivative Works a copy of this License; and
- (b) You must cause any modified files to carry prominent notices stating that You changed the files; and
- (c) You must retain, in the Source form of any Derivative Works that You distribute, all copyright, patent, trademark, and attribution notices from the Source form of the Work, excluding those notices that do not pertain to any part of the Derivative Works; and
- (d) If the Work includes a "NOTICE" text file as part of its distribution, then any Derivative Works that You distribute must include a readable copy of the attribution notices contained within such NOTICE file, excluding those notices that do not pertain to any part of the Derivative Works, in at least one of the following places: within a NOTICE text file distributed as part of the Derivative Works; within the Source form or documentation, if provided along with the Derivative Works; or, within a display generated by the Derivative Works, if and wherever such third-party notices normally appear. The contents of the NOTICE file are for informational purposes only and do not modify the License. You may add Your own attribution notices within Derivative Works that You distribute, alongside or as an addendum to the NOTICE text from the Work, provided that such additional attribution notices cannot be construed as modifying the License.

You may add Your own copyright statement to Your modifications and may provide additional or different license terms and conditions for use, reproduction, or distribution of Your modifications, or for any such Derivative Works as a whole, provided Your use, reproduction, and distribution of the Work otherwise complies with the conditions stated in this License.

5. Submission of Contributions. Unless You explicitly state otherwise, any Contribution intentionally submitted for inclusion in the Work by You to the Licensor shall be under the terms and conditions of this

License, without any additional terms or conditions. Notwithstanding the above, nothing herein shall supersede or modify the terms of any separate license agreement you may have executed with Licensor regarding such Contributions.

- 6. Trademarks. This License does not grant permission to use the trade names, trademarks, service marks, or product names of the Licensor, except as required for reasonable and customary use in describing the origin of the Work and reproducing the content of the NOTICE file.
- 7. Disclaimer of Warranty. Unless required by applicable law or agreed to in writing, Licensor provides the Work (and each Contributor provides its Contributions) on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE. You are solely responsible for determining the appropriateness of using or redistributing the Work and assume any risks associated with Your exercise of permissions under this License.
- 8. Limitation of Liability. In no event and under no legal theory, whether in tort (including negligence), contract, or otherwise, unless required by applicable law (such as deliberate and grossly negligent acts) or agreed to in writing, shall any Contributor be liable to You for damages, including any direct, indirect, special, incidental, or consequential damages of any character arising as a result of this License or out of the use or inability to use the Work (including but not limited to damages for loss of goodwill, work stoppage, computer failure or malfunction, or any and all other commercial damages or losses), even if such Contributor has been advised of the possibility of such damages.
- 9. Accepting Warranty or Additional Liability. While redistributing the Work or Derivative Works thereof, You may choose to offer, and charge a fee for, acceptance of support, warranty, indemnity, or other liability obligations and/or rights consistent with this License. However, in accepting such obligations, You may act only on Your own behalf and on Your sole responsibility, not on behalf of any other Contributor, and only if You agree to indemnify, defend, and hold each Contributor harmless for any liability incurred by, or claims asserted against, such Contributor by reason of your accepting any such warranty or additional liability.

END OF TERMS AND CONDITIONS

APPENDIX: How to apply the Apache License to your work.

To apply the Apache License to your work, attach the following boilerplate notice, with the fields enclosed by brackets "[]" replaced with your own identifying information. (Don't include the brackets!) The text should be enclosed in the appropriate comment syntax for the file format. We also recommend that a file or class name and description of purpose be included on the same "printed page" as the copyright notice for easier identification within third-party archives. Copyright 2017, The TensorFlow Authors.

Licensed under the Apache License, Version 2.0 (the "License"); you may not use this file except in compliance with the License. You may obtain a copy of the License at

http://www.apache.org/licenses/LICENSE-2.0

Unless required by applicable law or agreed to in writing, software distributed under the License is distributed on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied. See the License for the specific language governing permissions and limitations under the License.

LLVM Exceptions to the Apache 2.0 License

As an exception, if, as a result of your compiling your source code, portions of this Software are embedded into an Object form of such source code, you may redistribute such embedded portions in such Object form without complying with the conditions of Sections 4(a), 4(b) and 4(d) of the License.

In addition, if you combine or link compiled forms of this Software with software that is licensed under the GPLv2 ("Combined Software") and if a court of competent jurisdiction determines that the patent provision (Section 3), the indemnity provision (Section 9) or other Section of the License conflicts with the conditions of the GPLv2, you may retroactively and prospectively choose to deem waived or otherwise exclude such Section(s) of the License, but only in their entirety and only with respect to the Combined Software.

3-Clause BSD license

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. Neither the names of Facebook, Deepmind Technologies, NYU, NEC Laboratories America and IDIAP Research Institute nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

GNU GENERAL PUBLIC LICENSE

Version 3, 29 June 2007

Copyright (C) 2007 Free Software Foundation, Inc. http://fsf.org/

Everyone is permitted to copy and distribute verbatim copies of this license document, but changing it is not allowed.

Preamble

The GNU General Public License is a free, copyleft license for software and other kinds of works.

The licenses for most software and other practical works are designed to take away your freedom to share and change the works. By contrast, the GNU General Public License is intended to guarantee your freedom to share and change all versions of a program--to make sure it remains free software for all its users. We, the Free Software Foundation, use the GNU General Public License for most of our software; it applies also to any other work released this way by its authors. You can apply it to your programs, too.

When we speak of free software, we are referring to freedom, not price. Our General Public Licenses are designed to make sure that you have the freedom to distribute copies of free software (and charge for them if you wish), that you receive source code or can get it if you want it, that you can change the software or use pieces of it in new free programs, and that you know you can do these things.

To protect your rights, we need to prevent others from denying you these rights or asking you to surrender the rights. Therefore, you have certain responsibilities if you distribute copies of the software, or if you modify it: responsibilities to respect the freedom of others.

For example, if you distribute copies of such a program, whether gratis or for a fee, you must pass on to the recipients the same freedoms that you received. You must make sure that they, too, receive or can get the source code. And you must show them these terms so they know their rights.

Developers that use the GNU GPL protect your rights with two steps: (1) assert copyright on the software, and (2) offer you this License giving you legal permission to copy, distribute and/or modify it.

For the developers' and authors' protection, the GPL clearly explains that there is no warranty for this free software. For both users' and authors' sake, the GPL requires that modified versions be marked as changed, so that their problems will not be attributed erroneously to authors of previous versions.

Some devices are designed to deny users access to install or run modified versions of the software inside

them, although the manufacturer can do so. This is fundamentally incompatible with the aim of protecting users' freedom to change the software. The systematic pattern of such abuse occurs in the area of products for individuals to use, which is precisely where it is most unacceptable. Therefore, we have designed this version of the GPL to prohibit the practice for those products. If such problems arise substantially in other domains, we stand ready to extend this provision to those domains in future versions of the GPL, as needed to protect the freedom of users.

Finally, every program is threatened constantly by software patents. States should not allow patents to restrict development and use of software on general-purpose computers, but in those that do, we wish to avoid the special danger that patents applied to a free program could make it effectively proprietary. To prevent this, the GPL assures that patents cannot be used to render the program non-free.

The precise terms and conditions for copying, distribution and modification follow.

TERMS AND CONDITIONS

0. Definitions.

"This License" refers to version 3 of the GNU General Public License.

"Copyright" also means copyright-like laws that apply to other kinds of works, such as semiconductor masks.

"The Program" refers to any copyrightable work licensed under this License. Each licensee is addressed as "you". "Licensees" and "recipients" may be individuals or organizations.

To "modify" a work means to copy from or adapt all or part of the work in a fashion requiring copyright permission, other than the making of an exact copy. The resulting work is called a "modified version" of the earlier work or a work "based on" the earlier work.

A "covered work" means either the unmodified Program or a work based on the Program.

To "propagate" a work means to do anything with it that, without permission, would make you directly or secondarily liable for infringement under applicable copyright law, except executing it on a computer or modifying a private copy. Propagation includes copying, distribution (with or without modification), making available to the public, and in some countries other activities as well.

To "convey" a work means any kind of propagation that enables other parties to make or receive copies. Mere interaction with a user through a computer network, with no transfer of a copy, is not conveying.

An interactive user interface displays "Appropriate Legal Notices" to the extent that it includes a convenient and prominently visible feature that (1) displays an appropriate copyright notice, and (2) tells the user that there is no warranty for the work (except to the extent that warranties are provided), that licensees may convey the work under this License, and how to view a copy of this License. If the interface presents a list of user commands or options, such as a menu, a prominent item in the list meets this criterion.

1. Source Code.

The "source code" for a work means the preferred form of the work for making modifications to it. "Object code" means any non-source form of a work.

A "Standard Interface" means an interface that either is an official standard defined by a recognized standards body, or, in the case of interfaces specified for a particular programming language, one that is widely used among developers working in that language.

The "System Libraries" of an executable work include anything, other than the work as a whole, that (a) is included in the normal form of packaging a Major Component, but which is not part of that Major Component, and (b) serves only to enable use of the work with that Major Component, or to implement a Standard Interface for which an implementation is available to the public in source code form. A "Major Component", in this context, means a major essential component (kernel, window system, and so on) of the specific operating system (if any) on which the executable work runs, or a compiler used to produce the work, or an object code interpreter used to run it.

The "Corresponding Source" for a work in object code form means all the source code needed to generate, install, and (for an executable work) run the object code and to modify the work, including scripts to control those activities. However, it does not include the work's System Libraries, or general-purpose tools or generally available free programs which are used unmodified in performing those

activities but which are not part of the work. For example, Corresponding Source includes interface definition files associated with source files for the work, and the source code for shared libraries and dynamically linked subprograms that the work is specifically designed to require, such as by intimate data communication or control flow between those subprograms and other parts of the work.

The Corresponding Source need not include anything that users can regenerate automatically from other parts of the Corresponding Source.

The Corresponding Source for a work in source code form is that same work.

2. Basic Permissions.

All rights granted under this License are granted for the term of copyright on the Program, and are irrevocable provided the stated conditions are met. This License explicitly affirms your unlimited permission to run the unmodified Program. The output from running a covered work is covered by this License only if the output, given its content, constitutes a covered work. This License acknowledges your rights of fair use or other equivalent, as provided by copyright law.

You may make, run and propagate covered works that you do not convey, without conditions so long as your license otherwise remains in force. You may convey covered works to others for the sole purpose of having them make modifications exclusively for you, or provide you with facilities for running those works, provided that you comply with the terms of this License in conveying all material for which you do not control copyright. Those thus making or running the covered works for you must do so exclusively on your behalf, under your direction and control, on terms that prohibit them from making any copies of your copyrighted material outside their relationship with you.

Conveying under any other circumstances is permitted solely under the conditions stated below. Sublicensing is not allowed; section 10 makes it unnecessary.

3. Protecting Users' Legal Rights From Anti-Circumvention Law.

No covered work shall be deemed part of an effective technological measure under any applicable law fulfilling obligations under article 11 of the WIPO copyright treaty adopted on 20 December 1996, or similar laws prohibiting or restricting circumvention of such measures.

When you convey a covered work, you waive any legal power to forbid circumvention of technological measures to the extent such circumvention is effected by exercising rights under this License with respect to the covered work, and you disclaim any intention to limit operation or modification of the work as a means of enforcing, against the work's users, your or third parties' legal rights to forbid circumvention of technological measures.

4. Conveying Verbatim Copies.

You may convey verbatim copies of the Program's source code as you receive it, in any medium, provided that you conspicuously and appropriately publish on each copy an appropriate copyright notice; keep intact all notices stating that this License and any non-permissive terms added in accord with section 7 apply to the code; keep intact all notices of the absence of any warranty; and give all recipients a copy of this License along with the Program.

You may charge any price or no price for each copy that you convey, and you may offer support or warranty protection for a fee.

5. Conveying Modified Source Versions.

You may convey a work based on the Program, or the modifications to produce it from the Program, in the form of source code under the terms of section 4, provided that you also meet all of these conditions:

- a) The work must carry prominent notices stating that you modified it, and giving a relevant date.
- b) The work must carry prominent notices stating that it is released under this License and any conditions added under section 7. This requirement modifies the requirement in section 4 to "keep intact all notices".
- c) You must license the entire work, as a whole, under this License to anyone who comes into possession of a copy. This License will therefore apply, along with any applicable section 7 additional terms, to the whole of the work, and all its parts, regardless of how they are packaged. This License gives no permission to license the work in any other way, but it does not invalidate such permission if you have separately received it.

d) If the work has interactive user interfaces, each must display Appropriate Legal Notices; however, if the Program has interactive interfaces that do not display Appropriate Legal Notices, your work need not make them do so.

A compilation of a covered work with other separate and independent works, which are not by their nature extensions of the covered work, and which are not combined with it such as to form a larger program, in or on a volume of a storage or distribution medium, is called an "aggregate" if the compilation and its resulting copyright are not used to limit the access or legal rights of the compilation's users beyond what the individual works permit. Inclusion of a covered work in an aggregate does not cause this License to apply to the other parts of the aggregate.

6. Conveying Non-Source Forms.

You may convey a covered work in object code form under the terms of sections 4 and 5, provided that you also convey the machine-readable Corresponding Source under the terms of this License, in one of these ways:

- a) Convey the object code in, or embodied in, a physical product (including a physical distribution medium), accompanied by the Corresponding Source fixed on a durable physical medium customarily used for software interchange.
- b) Convey the object code in, or embodied in, a physical product (including a physical distribution medium), accompanied by a written offer, valid for at least three years and valid for as long as you offer spare parts or customer support for that product model, to give anyone who possesses the object code either (1) a copy of the Corresponding Source for all the software in the product that is covered by this License, on a durable physical medium customarily used for software interchange, for a price no more than your reasonable cost of physically performing this conveying of source, or (2) access to copy the Corresponding Source from a network server at no charge.
- c) Convey individual copies of the object code with a copy of the written offer to provide the Corresponding Source. This alternative is allowed only occasionally and noncommercially, and only if you received the object code with such an offer, in accord with subsection 6b.
- d) Convey the object code by offering access from a designated place (gratis or for a charge), and offer equivalent access to the Corresponding Source in the same way through the same place at no further charge. You need not require recipients to copy the Corresponding Source along with the object code. If the place to copy the object code is a network server, the Corresponding Source may be on a different server (operated by you or a third party) that supports equivalent copying facilities, provided you maintain clear directions next to the object code saying where to find the Corresponding Source. Regardless of what server hosts the Corresponding Source, you remain obligated to ensure that it is available for as long as needed to satisfy these requirements.
- e) Convey the object code using peer-to-peer transmission, provided you inform other peers where the object code and Corresponding Source of the work are being offered to the general public at no charge under subsection 6d.

A separable portion of the object code, whose source code is excluded from the Corresponding Source as a System Library, need not be included in conveying the object code work.

A "User Product" is either (1) a "consumer product", which means any tangible personal property which is normally used for personal, family, or household purposes, or (2) anything designed or sold for incorporation into a dwelling. In determining whether a product is a consumer product, doubtful cases shall be resolved in favor of coverage. For a particular product received by a particular user, "normally used" refers to a typical or common use of that class of product, regardless of the status of the particular user or of the way in which the particular user actually uses, or expects or is expected to use, the product. A product is a consumer product regardless of whether the product has substantial commercial, industrial or non-consumer uses, unless such uses represent the only significant mode of use of the product.

"Installation Information" for a User Product means any methods, procedures, authorization keys, or other information required to install and execute modified versions of a covered work in that User Product from a modified version of its Corresponding Source. The information must suffice to ensure that the continued functioning of the modified object code is in no case prevented or interfered with solely because modification has been made.

If you convey an object code work under this section in, or with, or specifically for use in, a User Product, and the conveying occurs as part of a transaction in which the right of possession and use of the User

Product is transferred to the recipient in perpetuity or for a fixed term (regardless of how the transaction is characterized), the Corresponding Source conveyed under this section must be accompanied by the Installation Information. But this requirement does not apply if neither you nor any third party retains the ability to install modified object code on the User Product (for example, the work has been installed in ROM).

The requirement to provide Installation Information does not include a requirement to continue to provide support service, warranty, or updates for a work that has been modified or installed by the recipient, or for the User Product in which it has been modified or installed. Access to a network may be denied when the modification itself materially and adversely affects the operation of the network or violates the rules and protocols for communication across the network.

Corresponding Source conveyed, and Installation Information provided, in accord with this section must be in a format that is publicly documented (and with an implementation available to the public in source code form), and must require no special password or key for unpacking, reading or copying.

7. Additional Terms.

"Additional permissions" are terms that supplement the terms of this License by making exceptions from one or more of its conditions. Additional permissions that are applicable to the entire Program shall be treated as though they were included in this License, to the extent that they are valid under applicable law. If additional permissions apply only to part of the Program, that part may be used separately under those permissions, but the entire Program remains governed by this License without regard to the additional permissions.

When you convey a copy of a covered work, you may at your option remove any additional permissions from that copy, or from any part of it. (Additional permissions may be written to require their own removal in certain cases when you modify the work.) You may place additional permissions on material, added by you to a covered work, for which you have or can give appropriate copyright permission.

Notwithstanding any other provision of this License, for material you add to a covered work, you may (if authorized by the copyright holders of that material) supplement the terms of this License with terms:

- a) Disclaiming warranty or limiting liability differently from the terms of sections 15 and 16 of this License; or
- b) Requiring preservation of specified reasonable legal notices or author attributions in that material or in the Appropriate Legal Notices displayed by works containing it; or
- c) Prohibiting misrepresentation of the origin of that material, or requiring that modified versions of such material be marked in reasonable ways as different from the original version; or
- d) Limiting the use for publicity purposes of names of licensors or authors of the material; or
- e) Declining to grant rights under trademark law for use of some trade names, trademarks, or service marks; or
- f) Requiring indemnification of licensors and authors of that material by anyone who conveys the material (or modified versions of
- it) with contractual assumptions of liability to the recipient, for any liability that these contractual assumptions directly impose on those licensors and authors.

All other non-permissive additional terms are considered "further restrictions" within the meaning of section 10. If the Program as you received it, or any part of it, contains a notice stating that it is governed by this License along with a term that is a further restriction, you may remove that term. If a license document contains a further restriction but permits relicensing or conveying under this License, you may add to a covered work material governed by the terms of that license document, provided that the further restriction does not survive such relicensing or conveying.

If you add terms to a covered work in accord with this section, you must place, in the relevant source files, a statement of the additional terms that apply to those files, or a notice indicating where to find the applicable terms.

Additional terms, permissive or non-permissive, may be stated in the form of a separately written license, or stated as exceptions; the above requirements apply either way.

8. Termination.

You may not propagate or modify a covered work except as expressly provided under this License. Any attempt otherwise to propagate or modify it is void, and will automatically terminate your rights under this License (including any patent licenses granted under the third paragraph of section 11).

However, if you cease all violation of this License, then your license from a particular copyright holder is reinstated (a) provisionally, unless and until the copyright holder explicitly and finally terminates your license, and (b) permanently, if the copyright holder fails to notify you of the violation by some reasonable means prior to 60 days after the cessation.

Moreover, your license from a particular copyright holder is reinstated permanently if the copyright holder notifies you of the violation by some reasonable means, this is the first time you have received notice of violation of this License (for any work) from that copyright holder, and you cure the violation prior to 30 days after your receipt of the notice.

Termination of your rights under this section does not terminate the licenses of parties who have received copies or rights from you under this License. If your rights have been terminated and not permanently reinstated, you do not qualify to receive new licenses for the same material under section 10

9. Acceptance Not Required for Having Copies.

You are not required to accept this License in order to receive or run a copy of the Program. Ancillary propagation of a covered work occurring solely as a consequence of using peer-to-peer transmission to receive a copy likewise does not require acceptance. However, nothing other than this License grants you permission to propagate or modify any covered work. These actions infringe copyright if you do not accept this License. Therefore, by modifying or propagating a covered work, you indicate your acceptance of this License to do so.

10. Automatic Licensing of Downstream Recipients.

Each time you convey a covered work, the recipient automatically receives a license from the original licensors, to run, modify and propagate that work, subject to this License. You are not responsible for enforcing compliance by third parties with this License.

An "entity transaction" is a transaction transferring control of an organization, or substantially all assets of one, or subdividing an organization, or merging organizations. If propagation of a covered work results from an entity transaction, each party to that transaction who receives a copy of the work also receives whatever licenses to the work the party's predecessor in interest had or could give under the previous paragraph, plus a right to possession of the Corresponding Source of the work from the predecessor in interest, if the predecessor has it or can get it with reasonable efforts.

You may not impose any further restrictions on the exercise of the rights granted or affirmed under this License. For example, you may not impose a license fee, royalty, or other charge for exercise of rights granted under this License, and you may not initiate litigation (including a cross-claim or counterclaim in a lawsuit) alleging that any patent claim is infringed by making, using, selling, offering for sale, or importing the Program or any portion of it.

11. Patents.

A "contributor" is a copyright holder who authorizes use under this License of the Program or a work on which the Program is based. The work thus licensed is called the contributor's "contributor version".

A contributor's "essential patent claims" are all patent claims owned or controlled by the contributor, whether already acquired or hereafter acquired, that would be infringed by some manner, permitted by this License, of making, using, or selling its contributor version, but do not include claims that would be infringed only as a consequence of further modification of the contributor version. For purposes of this definition, "control" includes the right to grant patent sublicenses in a manner consistent with the requirements of this License.

Each contributor grants you a non-exclusive, worldwide, royalty-free patent license under the contributor's essential patent claims, to make, use, sell, offer for sale, import and otherwise run, modify and propagate the contents of its contributor version.

In the following three paragraphs, a "patent license" is any express agreement or commitment, however denominated, not to enforce a patent (such as an express permission to practice a patent or covenant not to sue for patent infringement). To "grant" such a patent license to a party means to make such an agreement or commitment not to enforce a patent against the party.

If you convey a covered work, knowingly relying on a patent license, and the Corresponding Source of the work is not available for anyone to copy, free of charge and under the terms of this License, through a publicly available network server or other readily accessible means, then you must either (1) cause the Corresponding Source to be so available, or (2) arrange to deprive yourself of the benefit of the patent license for this particular work, or (3) arrange, in a manner consistent with the requirements of this License, to extend the patent license to downstream recipients. "Knowingly relying" means you have actual knowledge that, but for the patent license, your conveying the covered work in a country, or your recipient's use of the covered work in a country, would infringe one or more identifiable patents in that country that you have reason to believe are valid.

If, pursuant to or in connection with a single transaction or arrangement, you convey, or propagate by procuring conveyance of, a covered work, and grant a patent license to some of the parties receiving the covered work authorizing them to use, propagate, modify or convey a specific copy of the covered work, then the patent license you grant is automatically extended to all recipients of the covered work and works based on it.

A patent license is "discriminatory" if it does not include within the scope of its coverage, prohibits the exercise of, or is conditioned on the non-exercise of one or more of the rights that are specifically granted under this License. You may not convey a covered work if you are a party to an arrangement with a third party that is in the business of distributing software, under which you make payment to the third party based on the extent of your activity of conveying the work, and under which the third party grants, to any of the parties who would receive the covered work from you, a discriminatory patent license (a) in connection with copies of the covered work conveyed by you (or copies made from those copies), or (b) primarily for and in connection with specific products or compilations that contain the covered work, unless you entered into that arrangement, or that patent license was granted, prior to 28 March 2007.

Nothing in this License shall be construed as excluding or limiting any implied license or other defenses to infringement that may otherwise be available to you under applicable patent law.

12. No Surrender of Others' Freedom.

If conditions are imposed on you (whether by court order, agreement or otherwise) that contradict the conditions of this License, they do not excuse you from the conditions of this License. If you cannot convey a covered work so as to satisfy simultaneously your obligations under this License and any other pertinent obligations, then as a consequence you may not convey it at all. For example, if you agree to terms that obligate you to collect a royalty for further conveying from those to whom you convey the Program, the only way you could satisfy both those terms and this License would be to refrain entirely from conveying the Program.

13. Use with the GNU Affero General Public License.

Notwithstanding any other provision of this License, you have permission to link or combine any covered work with a work licensed under version 3 of the GNU Affero General Public License into a single combined work, and to convey the resulting work. The terms of this License will continue to apply to the part which is the covered work, but the special requirements of the GNU Affero General Public License, section 13, concerning interaction through a network will apply to the combination as such.

14. Revised Versions of this License.

The Free Software Foundation may publish revised and/or new versions of the GNU General Public License from time to time. Such new versions will be similar in spirit to the present version, but may differ in detail to address new problems or concerns.

Each version is given a distinguishing version number. If the Program specifies that a certain numbered version of the GNU General Public License "or any later version" applies to it, you have the option of following the terms and conditions either of that numbered version or of any later version published by the Free Software Foundation. If the Program does not specify a version number of the GNU General Public License, you may choose any version ever published by the Free Software Foundation.

If the Program specifies that a proxy can decide which future versions of the GNU General Public License can be used, that proxy's public statement of acceptance of a version permanently authorizes you to choose that version for the Program.

Later license versions may give you additional or different permissions. However, no additional obligations are imposed on any author or copyright holder as a result of your choosing to follow a later version.

15. Disclaimer of Warranty.

THERE IS NO WARRANTY FOR THE PROGRAM, TO THE EXTENT PERMITTED BY APPLICABLE LAW. EXCEPT WHEN OTHERWISE STATED IN WRITING THE COPYRIGHT HOLDERS AND/OR OTHER PARTIES PROVIDE THE PROGRAM "AS IS" WITHOUT WARRANTY OF ANY KIND, EITHER EXPRESSED OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE. THE ENTIRE RISK AS TO THE QUALITY AND PERFORMANCE OF THE PROGRAM IS WITH YOU. SHOULD THE PROGRAM PROVE DEFECTIVE, YOU ASSUME THE COST OF ALL NECESSARY SERVICING, REPAIR OR CORRECTION.

16. Limitation of Liability.

IN NO EVENT UNLESS REQUIRED BY APPLICABLE LAW OR AGREED TO IN WRITING WILL ANY COPYRIGHT HOLDER, OR ANY OTHER PARTY WHO MODIFIES AND/OR CONVEYS THE PROGRAM AS PERMITTED ABOVE, BE LIABLE TO YOU FOR DAMAGES, INCLUDING ANY GENERAL, SPECIAL, INCIDENTAL OR CONSEQUENTIAL DAMAGES ARISING OUT OF THE USE OR INABILITY TO USE THE PROGRAM (INCLUDING BUT NOT LIMITED TO LOSS OF DATA OR DATA BEING RENDERED INACCURATE OR LOSSES SUSTAINED BY YOU OR THIRD PARTIES OR A FAILURE OF THE PROGRAM TO OPERATE WITH ANY OTHER PROGRAMS), EVEN IF SUCH HOLDER OR OTHER PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

17. Interpretation of Sections 15 and 16.

If the disclaimer of warranty and limitation of liability provided above cannot be given local legal effect according to their terms, reviewing courts shall apply local law that most closely approximates an absolute waiver of all civil liability in connection with the Program, unless a warranty or assumption of liability accompanies a copy of the Program in return for a fee.

END OF TERMS AND CONDITIONS

How to Apply These Terms to Your New Programs

If you develop a new program, and you want it to be of the greatest possible use to the public, the best way to achieve this is to make it free software which everyone can redistribute and change under these terms.

To do so, attach the following notices to the program. It is safest to attach them to the start of each source file to most effectively state the exclusion of warranty; and each file should have at least the "copyright" line and a pointer to where the full notice is found.

<one line to give the program's name and a brief idea of what it does.>
Copyright (C) <year> <name of author>

This program is free software: you can redistribute it and/or modify it under the terms of the GNU General Public License as published by the Free Software Foundation, either version 3 of the License, or (at your option) any later version.

This program is distributed in the hope that it will be useful, but WITHOUT ANY WARRANTY; without even the implied warranty of MERCHANTABILITY or FITNESS FOR A PARTICULAR PURPOSE. See the GNU General Public License for more details.

You should have received a copy of the GNU General Public License along with this program. If not, see http://www.qnu.org/licenses/.

Also add information on how to contact you by electronic and paper mail.

If the program does terminal interaction, make it output a short notice like this when it starts in an interactive mode:

This program comes with ABSOLUTELY NO WARRANTY; for details type `show w'.

This is free software, and you are welcome to redistribute it under certain conditions; type `show c' for details.

The hypothetical commands `show w' and `show c' should show the appropriate parts of the General Public License. Of course, your program's commands might be different; for a GUI interface, you would use an "about box".

You should also get your employer (if you work as a programmer) or school, if any, to sign a "copyright

disclaimer" for the program, if necessary. For more information on this, and how to apply and follow the GNU GPL, see http://www.gnu.org/licenses/.

The GNU General Public License does not permit incorporating your program into proprietary programs. If your program is a subroutine library, you may consider it more useful to permit linking proprietary applications with the library. If this is what you want to do, use the GNU Lesser General Public License instead of this License. But first, please read http://www.gnu.org/philosophy/why-not-lgpl.html.

MIT License

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Toolkits for oneAPI and Associated Third Party Program File Locations

<u>Note</u>: All paths are relative to the Intel(R) oneAPI Toolkits installation directory, which defaults to: $C:\Pr G = \mathbb{R}$

Intel(R) oneAPI Base Toolkit (Base Kit):

- $\bullet \ \, Intel(R) \ one API \ DPC++/C++ \ Compiler < install_dir>\\ \ compiler\\ \ latest\\ \ licensing\\ \ c\\ \ third-party-programs.txt$
- Intel(R) oneAPI DPC++ Library (oneDPL) <install_dir>\dpl\latest\licensing\dpcpp_library\third-party-programs.txt
- Intel(R) oneAPI Math Kernel Library (oneMKL) <install_dir>\mkl\latest\licensing\third-party-programs.txt
- Intel(R) oneAPI Threading Building Blocks (oneTBB) <install_dir>\tbb\latest\licensing\third-party-programs.txt
- Intel(R) oneAPI Data Analytics Library (oneDAL) <install_dir>\dal\latest\licensing\en\third-party-programs.txt
- Intel(R) oneAPI Deep Neural Network Library (oneDNN) <install_dir>\dnnl\latest\licensing\THIRD-PARTY-PROGRAMS
- $\bullet \ \, Intel(R) \ \, DPC++ \ \, Compatibility \ \, Tool < install_dir>\ \, dpcpp-ct\ \, licensing\ \, dpct\ \, third-party-programs.txt$
- $\bullet \ \ \, Intel(R) \ \, Distribution \ \, for \ \, GDB* < install_dir > \\ \ \, debugger \ \, latest \ \, licensing \ \, third-party-programs.txt$
- Intel(R) Integrated Performance Primitives (Intel(R) IPP) <install_dir>\ipp\latest\licensing\third-party-programs.txt
- Intel(R) Integrated Performance Primitives Cryptography (Intel(R) IPP Cryptography) <install_dir>\ippcp\latest\licensing\third-party-programs.txt
- Intel(R) Distribution for Python* <install_dir>\intelpython\latest\licensing\third-party-programs.txt
- Intel(R) Vtune(TM) Profiler < install dir>\vtune profiler\latest\licensing\third-party-programs.txt
- Intel(R) Advisor <install_dir>\advisor\latest\licensing\third-party-programs.txt
- Intel(R) Dev Utilities <install dir>\dev-utilities\latest\licensing\third-party-programs.txt
- Installer Program <install dir>\installer\third-party-programs.txt

Intel(R) oneAPI HPC Toolkit (HPC Kit):

- $\bullet \ \ Intel(R) \ Fortran\ Compiler < install_dir > \ \ \\ latest\ licensing \ fortran\ third-party-programs.txt$

- Intel(R) MPI Library <install dir>\mpi\latest\licensing\third-party-programs.txt
- Intel(R) Trace Analyzer and Collector <install dir>\itac\latest\licensing\third-party-programs.txt
- Intel(R) Inspector <install dir>\inspector\latest\licensing\third-party-programs.txt
- Intel(R) Dev Utilities <install dir>\dev-utilities\latest\licensing\third-party-programs.txt
- Installer Program <install dir>\installer\third-party-programs.txt

Intel(R) oneAPI IoT Toolkit (IoT Kit):

- Intel(R) C++ Compiler <install dir>\compiler\latest\licensing\c\third-party-programs.txt
- Intel(R) Inspector <install dir>\inspector\latest\licensing\third-party-programs.txt
- Intel(R) Dev Utilities < install dir>\dev-utilities\latest\licensing\third-party-programs.txt
- Installer Program <install dir>\installer\third-party-programs.txt

Intel(R) oneAPI Rendering Toolkit (Render Kit):

- Intel(R) Embree <install dir>\embree\latest\licensing\third-party-programs.txt
- Intel(R) OSPRay <install dir>\ospray\latest\licensing\third-party-programs.txt
- Intel(R) Open Image Denoise <install dir>\oidn\latest\licensing\third-party-programs.txt
- Intel(R) Open Volume Kernel Library <install_dir>\openvkl\latest\licensing\third-party-programs.txt
- Intel(R) OSPRay Studio <install_dir>\ospray_studio\latest\licensing\third-party-programs.txt
- Renderkit Utilities Package <install dir>\rkutil\latest\licensing\third-party-programs.txt
- Installer Program <install dir>\installer\third-party-programs.txt

This is the redist.txt file mentioned in the associated Intel End User License Agreement for Developer Tools (Version October 2020) ("Agreement"). All capitalized terms used but not defined in this file have the same meaning as specified in the Agreement. The Redistributables are listed below, and are subject to all of the terms and conditions of the Agreement.

Please note that Third Party Programs, if any, may also be redistributable. See the "third-party-programs.txt" or other similarly-named text file for more details."

Intel® oneAPI DPC++/C++ Compiler & Intel® C++ Compiler Classic 2023.1.0 <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libiomp5md.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libiomp5md.pdb <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/ libiompstubs5md.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ libiomp5md.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ libiomp5md.pdb <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ libiompstubs5md.dll <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libiomp5md.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32 win/libiomp5md db.dll <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32 win/libiompstubs5md.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64 win/libiomp5md.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/ libiomp5md_db.dll <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/ libiompstubs5md.lib <installdir>/compiler/2023.1.0/windows/bin/omptarget.rtl.opencl.dll <installdir>/compiler/2023.1.0/windows/bin/omptarget.rtl.level0.dll <installdir>/compiler/2023.1.0/windows/bin/omptarget.dll <installdir>/compiler/2023.1.0/windows/bin/libomp-fallback-cstring.spv <installdir>/compiler/2023.1.0/windows/lib/libomp-fallback-cstring.obj <installdir>/compiler/2023.1.0/windows/lib/libomptarget-opencl-optional.bc <installdir>/compiler/2023.1.0/windows/lib/libomptarget-opencl-required.bc <installdir>/compiler/2023.1.0/windows/lib/libomp-spirvdevicertl-optional.obj <installdir>/compiler/2023.1.0/windows/lib/libomp-spirvdevicertl-required.obj <installdir>/compiler/2023.1.0/windows/bin/omptarget.sycl.wrap.dll <installdir>/compiler/2023.1.0/windows/lib/libomp-device-svml.obj <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libbfp754.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libdecimal.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libicaio.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libipgo.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libirc.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libircmt.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libm.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libmatmul.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libmmd.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libmmdd.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libmmds.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libmmt.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/svml_disp.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/svml_dispmd.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/svml_dispmt.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libistrconv.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libbfp754.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libdecimal.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libicaio.lib

<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libipgo.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libirc.lib

```
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64 win/libircmt.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libm.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libmatmul.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libmmd.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libmmdd.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libmmds.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libmmt.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/svml_disp.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/svml_dispmd.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/svml_dispmt.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/libistrconv.lib
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libchkp.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libirngmd.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmd.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmd.pdb
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmdd.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmdd.pdb
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/svml_dispmd.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/svml_dispmd.pdb
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/cilkrts20.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/cilkrts20.pdb
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libchkp.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libirngmd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libmmd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libmmd.pdb
<installdir>/compiler/2023.1.0/windows/redist/intel64 win/compiler/libmmdd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64 win/compiler/libmmdd.pdb
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/
svml dispmd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/
svml dispmd.pdb
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/cilkrts20.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/cilkrts20.pdb
<installdir>/compiler/2023.1.0/windows/bin/pi_opencl.dll
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-cstring.obj
<installdir>/compiler/2023.1.0/windows/bin/xptifw.dll
<installdir>/compiler/2023.1.0/windows/lib/xpti.lib
<installdir>/compiler/2023.1.0/windows/lib/xptid.lib
<installdir>/compiler/2023.1.0/windows/lib/xptifw.lib
<installdir>/compiler/2023.1.0/windows/lib/libsycl-msvc-math.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-imf-fp64.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-imf.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-imf-fp64.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-imf.obj
<installdir>/compiler/2023.1.0/windows/lib/sycl-devicelib-host.lib
<installdir>/compiler/2023.1.0/windows/lib/libsycl-bfloat16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-bfloat16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-imf-bf16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-imf-bf16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-native-bfloat16.obj
<installdir>/compiler/2023.1.0/windows/lib/x64/cl.cfg
<installdir>/compiler/2023.1.0/windows/lib/OpenCL.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnz1.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryz1.o
<installdir>/compiler/2023.1.0/windows/lib/x64/task_executor64.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/task_executor64.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/libintelocl.so-gdb.py
<installdir>/compiler/2023.1.0/windows/lib/x64/intelocl64.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/intelocl64.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/common_clang64.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnz0.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnl9.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnh8.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfne9.rtl
```

```
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_z0.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_l9.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_h8.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_e9.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibrary.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibrarye9.o
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryh8.o
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryl9.o
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryz0.o
<installdir>/compiler/2023.1.0/windows/lib/clbltfnshared.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_z1.dll
<SystemFolder64>/OpenCL.dll
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/access_by.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/aligned.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/aligned_accessor.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/aligned_offset.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/aos1d_accessor.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/
aos1d_const_accessor.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/aos1d_container.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/asa1d_container.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/blocked.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/bounds.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/
buffer_offset_in_cachelines.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/common.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/
const_data_member_proxy.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/container.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/data_member_proxy.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/defaults.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/fixed.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/fixed_offset.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/isa.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/layout.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/linear_index.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/min_max_val.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/n_bounds.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/n_container.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/n_extent.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/n_index.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/no_offset.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/order.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/primitive.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/sdlt.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/simd_index.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/simd_traits.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/soa1d_container.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/
uniform_ref_over1d.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/
uniform_soa_over1d.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/
uniform_value_over1d.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/calloc.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/default.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/hybrid.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/malloc.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/
mem_align.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/
memory_block.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/
reserved_hugepage.h
```

```
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/
std_adapter.h
```

- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/allocator/
 transparent_hugepage.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/ aligned_access.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aligned_base.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aligned_stride.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 always_false_type.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_container_base.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_member.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_stride/exporter.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_stride/importer.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_stride/organization.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_stride/stride_offset_calculator.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_stride/stride_type_builder.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_struct/organization.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_struct/stride_offset_calculator.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos_by_struct/stride_type_builder.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/accessor.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/const_accessor.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/const_element.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/const_member.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/element.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/exporter.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/importer.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_stride/member.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_struct/accessor.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_struct/const_accessor.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_struct/const_element.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_struct/const_member.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_struct/element.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 aos1d_by_struct/member.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
 accessor.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
 const_accessor.h

```
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
const element.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
const_member_simdlane.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
const_stl_iterator.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
element.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
member_simdlane.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
organization.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/asa1d/
stl_iterator.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/capacity.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
check_for_primitive_declaration.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
compiler_defined.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
compound_index.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
const_bool.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
default_offset.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/deferred.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
disambiguator.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
emit_for_each.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
enable_if_type.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
find_largest_builtin.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
find_smallest_builtin.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
fixed_index.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
fixed_stride.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
immutable_value.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
index_list.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
indirect_assertion.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
int sequence.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
is_aligned.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
is_builtin.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
is_linear_compatible_index.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
is_same_type.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
is_simd_compatible_index.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
is_size_provider.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
logical and of.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
accessor.h
```

```
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
bounds_interface.h
```

- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 const_accessor.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 const_element.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 const_member.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 element.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 extent_interface.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 member.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 origin_type_builder.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/nd/
 organization_builder.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 operation.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 pass_through_land_rvalue_operators_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 pass_through_rvalue_operators_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 pointer or number.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 post_offset_calculator.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/ power_of_two.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 pre_offset_calculator.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 proxy_overload_assignment_operator_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/ proxy_overload_binary_operator_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 proxy_overload_unary_operator_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 rvalue_binary_operator_proxy_other_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 rvalue_binary_operator_proxy_proxy_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 rvalue_operator_helper.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 rvalue_unary_operator_helper_xmacro.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 simd_index_ref.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 simd_vector_register.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
 static_loop.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/ stride_list.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa/
 exporter.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa/ importer.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa/
 organization.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa/ stride_offset_calculator.h
- <installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa/ stride_type_builder.h

```
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
soa_per_row/exporter.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
```

soa_per_row/importer.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
soa_per_row/organization.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
soa_per_row/stride_offset_calculator.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
soa_per_row/stride_type_builder.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa/exporter.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa/importer.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa/organization.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa/stride_offset_calculator.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa/stride_type_builder.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa_per_row/exporter.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa_per_row/importer.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa_per_row/organization.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa_per_row/stride_offset_calculator.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
tiled_soa_per_row/stride_type_builder.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/tuple.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/
unqualified.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/ zero_offset.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/linear/
const_stl_iterator.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/linear/
element_mover.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/linear/
stl_iterator.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/scalar/
root_access.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/simd/ builtin_root_access.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/simd/
element.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/simd/
element_mover.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/simd/
indices.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/simd/
iterator_index.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/simd/
root_access.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/simd/
struct_root_access.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
accessor.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
const_accessor.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
const_element.h

<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/

```
const member.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
exporter.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
importer.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
member.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/soa1d/
organization.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const member.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const_scalar_accessor.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const_scalar_element.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const_scalar_member.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const_scalar_ref_accessor.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const_simd_accessor.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const_simd_element.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
const simd member.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
member.h
<installdir>/compiler/2023.1.0/windows/compiler/include/sdlt/internal/uniform/
scalar member.h
<installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/cilkrts.lib
<installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/cilkrts.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/intelocl64_emu.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/intelocl64_emu.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/task_executor64_emu.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/task_executor64_emu.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/cl.fpga_emu.cfg
<installdir>/compiler/2023.1.0/windows/bin/pi_level_zero.dll
<installdir>/compiler/2023.1.0/windows/bin/sycl6.dll
<installdir>/compiler/2023.1.0/windows/bin/sycl6d.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/task_executor32.lib
<installdir>/compiler/2023.1.0/windows/lib/x86/task_executor32.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/OpenCL.lib
<installdir>/compiler/2023.1.0/windows/lib/x86/libintelocl.so-qdb.py
<installdir>/compiler/2023.1.0/windows/lib/x86/intelocl32.lib
<installdir>/compiler/2023.1.0/windows/lib/x86/intelocl32.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/common_clang32.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/clbltfnx0.rtl
<installdir>/compiler/2023.1.0/windows/lib/x86/clbltfns9.rtl
<installdir>/compiler/2023.1.0/windows/lib/x86/clbltfnn8.rtl
<installdir>/compiler/2023.1.0/windows/lib/x86/clbltfng9.rtl
<installdir>/compiler/2023.1.0/windows/lib/x86/cl.cfg
<installdir>/compiler/2023.1.0/windows/lib/x86/__ocl_svml_x0.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/__ocl_svml_s9.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/__ocl_svml_n8.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/__ocl_svml_g9.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/cllibrary.rtl
<installdir>/compiler/2023.1.0/windows/lib/x86/cllibraryq9.o
<installdir>/compiler/2023.1.0/windows/lib/x86/cllibraryn8.o
<installdir>/compiler/2023.1.0/windows/lib/x86/cllibrarys9.o
<installdir>/compiler/2023.1.0/windows/lib/x86/cllibraryx0.o
<installdir>/compiler/2023.1.0/windows/lib/x86/clbltfnx1.rtl
```

<installdir>/compiler/2023.1.0/windows/lib/x86/cllibraryx1.o

```
<installdir>/compiler/2023.1.0/windows/lib/x86/intelocl32_emu.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/intelocl32_emu.lib
<installdir>/compiler/2023.1.0/windows/lib/x86/task_executor32_emu.dll
<installdir>/compiler/2023.1.0/windows/lib/x86/task_executor32_emu.lib
<installdir>/compiler/2023.1.0/windows/lib/x86/__ocl_svml_x1.dll
<SystemFolder32>/OpenCL.dll
```

This is the redist.txt file mentioned in the associated Intel End User License Agreement for Developer Tools (Version October 2020) ("Agreement"). All capitalized terms used but not defined in this file have the same meaning as specified in the Agreement. The Redistributables are listed below, and are subject to all of the terms and conditions of the Agreement.

Please note that Third Party Programs, if any, may also be redistributable. See the "third-party-programs.txt" or other similarly-named text file for more details."

Intel® Fortran Compiler & Intel® Fortran Compiler Classic 2023.1.0 <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libiomp5md.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libiomp5md.pdb <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/ libiompstubs5md.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ libiomp5md.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ libiomp5md.pdb <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ libiompstubs5md.dll <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32_win/libiomp5md.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32 win/libiomp5md db.dll <installdir>/compiler/2023.1.0/windows/compiler/lib/ia32 win/libiompstubs5md.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64 win/libiomp5md.lib <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/ libiomp5md_db.dll <installdir>/compiler/2023.1.0/windows/compiler/lib/intel64_win/ libiompstubs5md.lib <installdir>/compiler/2023.1.0/windows/bin/omptarget.rtl.opencl.dll <installdir>/compiler/2023.1.0/windows/bin/omptarget.rtl.level0.dll <installdir>/compiler/2023.1.0/windows/bin/omptarget.dll <installdir>/compiler/2023.1.0/windows/bin/libomp-fallback-cstring.spv <installdir>/compiler/2023.1.0/windows/lib/libomp-fallback-cstring.obj <installdir>/compiler/2023.1.0/windows/lib/libomptarget-opencl-optional.bc <installdir>/compiler/2023.1.0/windows/lib/libomptarget-opencl-required.bc <installdir>/compiler/2023.1.0/windows/lib/libomp-spirvdevicertl-optional.obj <installdir>/compiler/2023.1.0/windows/lib/libomp-spirvdevicertl-required.obj <installdir>/compiler/2023.1.0/windows/bin/omptarget.sycl.wrap.dll <installdir>/compiler/2023.1.0/windows/lib/libomp-device-svml.obj <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libchkp.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libirngmd.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmd.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmd.pdb <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmdd.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libmmdd.pdb <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/svml_dispmd.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/svml_dispmd.pdb <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/cilkrts20.dll <installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/cilkrts20.pdb <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libchkp.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libirngmd.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libmmd.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libmmd.pdb <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libmmdd.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libmmdd.pdb <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ svml_dispmd.dll <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ svml dispmd.pdb <installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/cilkrts20.dll

```
<installdir>/compiler/2023.1.0/windows/redist/intel64 win/compiler/cilkrts20.pdb
<installdir>/compiler/2023.1.0/windows/bin/pi_opencl.dll
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-cstring.obj
<installdir>/compiler/2023.1.0/windows/bin/xptifw.dll
<installdir>/compiler/2023.1.0/windows/lib/xpti.lib
<installdir>/compiler/2023.1.0/windows/lib/xptid.lib
<installdir>/compiler/2023.1.0/windows/lib/xptifw.lib
<installdir>/compiler/2023.1.0/windows/lib/libsycl-msvc-math.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-imf-fp64.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-imf.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-imf-fp64.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-imf.obj
<installdir>/compiler/2023.1.0/windows/lib/sycl-devicelib-host.lib
<installdir>/compiler/2023.1.0/windows/lib/libsycl-bfloat16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-bfloat16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-fallback-imf-bf16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-imf-bf16.obj
<installdir>/compiler/2023.1.0/windows/lib/libsycl-native-bfloat16.obj
<installdir>/compiler/2023.1.0/windows/lib/x64/cl.cfg
<installdir>/compiler/2023.1.0/windows/lib/OpenCL.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnz1.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryz1.o
<installdir>/compiler/2023.1.0/windows/lib/x64/task_executor64.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/task_executor64.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/libintelocl.so-qdb.py
<installdir>/compiler/2023.1.0/windows/lib/x64/intelocl64.lib
<installdir>/compiler/2023.1.0/windows/lib/x64/intelocl64.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/common clang64.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnz0.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnl9.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfnh8.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/clbltfne9.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_z0.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_l9.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_h8.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_e9.dll
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibrary.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibrarye9.o
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryh8.o
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryl9.o
<installdir>/compiler/2023.1.0/windows/lib/x64/cllibraryz0.o
<installdir>/compiler/2023.1.0/windows/lib/clbltfnshared.rtl
<installdir>/compiler/2023.1.0/windows/lib/x64/__ocl_svml_z1.dll
<SystemFolder64>/OpenCL.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/ifdlg100.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libicaf.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libifcoremd.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libifcoremdd.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libifcorert.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libifcorertd.dll
<installdir>/compiler/2023.1.0/windows/redist/ia32_win/compiler/libifportmd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/ifdlg100.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/libicaf.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/
libifcoremd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/
libifcoremdd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/
libifcorert.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/
libifcorertd.dll
<installdir>/compiler/2023.1.0/windows/redist/intel64_win/compiler/
libifportmd.dll
```

Intel Simplified Software License (Version October 2022)

Intel(R) MPI Library: Copyright (C) 2009 Intel Corporation

Use and Redistribution. You may use and redistribute the software, which is provided in binary form only, (the "Software"), without modification, provided the

following conditions are met:

- * Redistributions must reproduce the above copyright notice and these terms of
- in the Software and in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel nor the names of its suppliers may be used to endorse
- or promote products derived from this Software without specific prior written permission.
- * No reverse engineering, decompilation, or disassembly of the Software is permitted, nor any modification or alteration of the Software or its operation at any time, including during execution.

No other licenses. Except as provided in the preceding section, Intel grants no licenses or other rights by implication, estoppel or otherwise to, patent, copyright, trademark, trade name, service mark or other intellectual property licenses or rights of Intel.

Third party software. "Third Party Software" means the files (if any) listed in the "third-party-software.txt" or other similarly-named text file that may be included with the Software. Third Party Software, even if included with the distribution of the Software, may be governed by separate license terms, including

without limitation, third party license terms, open source software notices and terms, and/or other Intel software license terms. These separate license terms solely govern Your use of the Third Party Software.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE DISCLAIMED. THIS SOFTWARE IS NOT INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE PERSONAL INJURY OR DEATH AND YOU AGREE THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS, COSTS, DAMAGES, EXPENSES, AND ATTORNEYS' FEES ARISING OUT OF ANY SUCH USE, EVEN IF ANY CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE SOFTWARE.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

No support. Intel may make changes to the Software, at any time without notice, and is not obligated to support, update or provide training for the Software.

Termination. Your right to use the Software is terminated in the event of your breach of this license.

Feedback. Should you provide Intel with comments, modifications, corrections, enhancements or other input ("Feedback") related to the Software, Intel will be free to use, disclose, reproduce, license or otherwise distribute or exploit the Feedback in its sole discretion without any obligations or restrictions of any kind, including without limitation, intellectual property rights or licensing obligations.

Compliance with laws. You agree to comply with all relevant laws and regulations governing your use, transfer, import or export (or prohibition thereof) of the Software.

Governing law. All disputes will be governed by the laws of the United States of

America and the State of Delaware without reference to conflict of law principles

and subject to the exclusive jurisdiction of the state or federal courts sitting

in the State of Delaware, and each party agrees that it submits to the personal jurisdiction and venue of those courts and waives any objections. THE UNITED NATIONS CONVENTION ON CONTRACTS FOR THE INTERNATIONAL SALE OF GOODS (1980) IS SPECIFICALLY EXCLUDED AND WILL NOT APPLY TO THE SOFTWARE.

Intel(R) MPI Library 2021.9 Third Party Programs File

This file is the "third-party-programs.txt" file specified in the associated Intel end user license agreement for the Intel software you are licensing. Third party programs and their corresponding required notices and/or license terms are listed below.

.....

1. MPICH

Copyright Notice 1998--2020, Argonne National Laboratory

Mpich license

Permission is hereby granted to use, reproduce, prepare derivative works, and to redistribute to others. This software was authored by:

Mathematics and Computer Science Division Argonne National Laboratory, Argonne IL 60439

(and)

Department of Computer Science University of Illinois at Urbana-Champaign

GOVERNMENT LICENSE

Portions of this material resulted from work developed under a U.S. Government Contract and are subject to the following license: the Government is granted for itself and others acting on its behalf a paid-up, nonexclusive, irrevocable worldwide license in this computer software to reproduce, prepare derivative works, and perform publicly and display publicly.

DISCLAIMER

This computer code material was prepared, in part, as an account of work sponsored by an agency of the United States Government. Neither the United States, nor the University of Chicago, nor any of their employees, makes any warranty express or implied, or assumes any legal liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights.

2. Open MPI

- Copyright (c) 2004-2010 The Trustees of Indiana University and Indiana University Research and Technology Corporation. All rights reserved.
- Copyright (c) 2004-2017 The University of Tennessee and The University of Tennessee Research Foundation. All rights reserved.
- Copyright (c) 2004-2010 High Performance Computing Center Stuttgart, University of Stuttgart. All rights reserved.
- Copyright (c) 2004-2008 The Regents of the University of California. All rights reserved.
- Copyright (c) 2006-2018 Los Alamos National Security, LLC. All rights reserved.
- Copyright (c) 2006-2020 Cisco Systems, Inc. All rights reserved.
- Copyright (c) 2006-2010 Voltaire, Inc. All rights reserved.

```
Copyright (c) 2006-2017 Sandia National Laboratories. All rights reserved.
Copyright (c) 2006-2010 Sun Microsystems, Inc. All rights reserved.
                      Use is subject to license terms.
Copyright (c) 2006-2017 The University of Houston. All rights reserved.
Copyright (c) 2006-2009 Myricom, Inc. All rights reserved.
Copyright (c) 2007-2017 UT-Battelle, LLC. All rights reserved.
Copyright (c) 2007-2020 IBM Corporation. All rights reserved.
Copyright (c) 1998-2005 Forschungszentrum Juelich, Juelich Supercomputing
                      Centre, Federal Republic of Germany
Copyright (c) 2005-2008 ZIH, TU Dresden, Federal Republic of Germany
Copyright (c) 2007
                        Evergrid, Inc. All rights reserved.
                        Chelsio, Inc. All rights reserved.
Copyright (c) 2008
Copyright (c) 2008-2009 Institut National de Recherche en
                      Informatique. All rights reserved.
Copyright (c) 2007
                        Lawrence Livermore National Security, LLC.
                      All rights reserved.
Copyright (c) 2007-2017 Mellanox Technologies. All rights reserved.
Copyright (c) 2006-2010 QLogic Corporation. All rights reserved.
Copyright (c) 2008-2017 Oak Ridge National Labs. All rights reserved.
Copyright (c) 2006-2012 Oracle and/or its affiliates. All rights reserved.
Copyright (c) 2009-2015 Bull SAS. All rights reserved.
Copyright (c) 2010
                        ARM ltd. All rights reserved.
Copyright (c) 2016
                        ARM, Inc. All rights reserved.
Copyright (c) 2010-2011 Alex Brick . All rights reserved.
Copyright (c) 2012
                        The University of Wisconsin-La Crosse. All rights
                      reserved.
Copyright (c) 2013-2020 Intel, Inc. All rights reserved.
Copyright (c) 2011-2017 NVIDIA Corporation. All rights reserved.
                        Broadcom Limited. All rights reserved.
Copyright (c) 2016
Copyright (c) 2011-2017 Fujitsu Limited. All rights reserved.
Copyright (c) 2014-2015 Hewlett-Packard Development Company, LP.
                      rights reserved.
Copyright (c) 2013-2017 Research Organization for Information Science (RIST).
                      All rights reserved.
Copyright (c) 2017-2020 Amazon.com, Inc. or its affiliates. All Rights
                      reserved.
                        DataDirect Networks. All rights reserved.
Copyright (c) 2018
```

Copyright (c) 2001

\$COPYRIGHT\$

Copyright (c) 2020

Copyright (c) 2002

Additional copyrights may follow

\$HEADER\$

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

Copyright (c) 2018-2020 Triad National Security, LLC. All rights reserved.

Argonne National Laboratory

University of Chicago

Google, LLC. All rights reserved.

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer listed in this license in the documentation and/or other materials provided with the distribution.
- Neither the name of the copyright holders nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

The copyright holders provide no reassurances that the source code provided does not infringe any patent, copyright, or any other intellectual property rights of third parties. The copyright holders disclaim any liability to any recipient for claims brought against recipient by any third party for infringement of that parties intellectual property rights.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

-----[Copyright from inclusion of MPICH code]-----

The following is a notice of limited availability of the code, and disclaimer which must be included in the prologue of the code and in all source listings of the code.

Copyright Notice + 2002 University of Chicago

Permission is hereby granted to use, reproduce, prepare derivative works, and to redistribute to others. This software was authored by:

Mathematics and Computer Science Division Argonne National Laboratory, Argonne IL 60439

(and)

Department of Computer Science University of Illinois at Urbana-Champaign

GOVERNMENT LICENSE

Portions of this material resulted from work developed under a U.S. Government Contract and are subject to the following license: the Government is granted for itself and others acting on its behalf a paid-up, nonexclusive, irrevocable worldwide license in this computer software to reproduce, prepare derivative works, and perform publicly and display publicly.

DISCLAIMER

This computer code material was prepared, in part, as an account of work sponsored by an agency of the United States Government. Neither the United States, nor the University of Chicago, nor any of their employees, makes any warranty express or implied, or assumes any legal liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights.

3. hwloc

Copyright © 2004-2006 The Trustees of Indiana University and Indiana

University Research and Technology Corporation. All rights reserved.

Copyright © 2004-2005 The University of Tennessee and The University of Tennessee Research Foundation. All rights reserved.

Copyright © 2004-2005 High Performance Computing Center Stuttgart, University of Stuttgart. All rights reserved.

Copyright $\ensuremath{\text{@}}\xspace$ 2004-2005 The Regents of the University of California. All rights reserved.

Copyright © 2009 CNRS

Copyright © 2009-2016 Inria. All rights reserved.

Copyright © 2009-2015 Université Bordeaux

Copyright © 2009-2015 Cisco Systems, Inc. All rights reserved.

Copyright © 2009-2012 Oracle and/or its affiliates. All rights reserved.

Copyright © 2010 IBM

Copyright © 2010 Jirka Hladky

Copyright © 2012 Aleksej Saushev, The NetBSD Foundation

Copyright © 2012 Blue Brain Project, EPFL. All rights reserved.

Copyright © 2013-2014 University of Wisconsin-La Crosse. All rights reserved.

Copyright © 2015 Research Organization for Information Science and

Technology (RIST). All rights reserved.

Copyright © 2015-2016 Intel, Inc. All rights reserved.

See COPYING in top-level directory.

The 3-Clause BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- The name of the author may not be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE AUTHOR ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE AUTHOR BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

4. Libfabric and OpenFabrics Interfaces (OFI)

Copyright (c) 2015 Intel Corporation. All rights reserved. Copyright (c) 2015 Cisco Systems, Inc. All rights reserved.

The -2-Clause BSD license

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

5. Intel® Distribution for Python*

Intel Simplified Software License (Version August 2021)

Use and Redistribution. You may use and redistribute the software (the "Software"), without modification, provided the following conditions are met:

- * Redistributions must reproduce the above copyright notice and the following terms of use in the Software and in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel nor the names of its suppliers may be used to endorse or promote products derived from this Software without specific prior written permission.
- * No reverse engineering, decompilation, or disassembly of this Software is permitted.

No other licenses. Except as provided in the preceding section, Intel grants

licenses or other rights by implication, estoppel or otherwise to, patent, copyright, trademark, trade name, service mark or other intellectual property licenses or rights of Intel.

Third party software. The Software may contain Third Party Software. "Third Party Software" is open source software, third party software, or other Intel software that may be identified in the Software itself or in the files (if any)

listed in the "third-party-software.txt" or similarly named text file included with the Software. Third Party Software, even if included with the distribution

of the Software, may be governed by separate license terms, including without limitation, open source software license terms, third party software license terms, and other Intel software license terms. Those separate license terms solely govern your use of the Third Party Software, and nothing in this icense

limits any rights under, or grants rights that supersede, the terms of the applicable license terms.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE

DISCLAIMED. THIS SOFTWARE IS NOT INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE PERSONAL INJURY OR DEATH AND YOU AGREE THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS, COSTS, DAMAGES, EXPENSES, AND ATTORNEYS' FEES ARISING OUT OF ANY SUCH USE, EVEN IF ANY CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE SOFTWARE.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY

LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE

OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

No support. Intel may make changes to the Software, at any time without notice,

and is not obligated to support, update or provide training for the Software.

Termination. Your right to use the Software is terminated in the event of your breach of this license.

Feedback. Should you provide Intel with comments, modifications, corrections, enhancements or other input ("Feedback") related to the Software, Intel will

free to use, disclose, reproduce, license or otherwise distribute or exploit the

Feedback in its sole discretion without any obligations or restrictions of any kind, including without limitation, intellectual property rights or licensing obligations.

Compliance with laws. You agree to comply with all relevant laws and regulations

governing your use, transfer, import or export (or prohibition thereof) of the Software.

Governing law. All disputes will be governed by the laws of the United States of

America and the State of Delaware without reference to conflict of law principles and subject to the exclusive jurisdiction of the state or federal courts sitting in the State of Delaware, and each party agrees that it submits to the personal jurisdiction and venue of those courts and waives any objections. The United Nations Convention on Contracts for the International Sale of Goods (1980) is specifically excluded and will not apply to the Software.

6. uthash

Copyright (c) 2005-2014, Troy D. Hanson http://troydhanson.github.com/uthash/All rights reserved.

Troy Hanson BSD license

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

* Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED

TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

7. json-c

Copyright (c) 2009-2012 Eric Haszlakiewicz Copyright (c) 2004, 2005 Metaparadigm Pte Ltd

MIT license

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

8. zlib

Permissive software license

This software is provided 'as-is', without any express or implied warranty. In no event will the authors be held liable for any damages arising from the use of this software.

Permission is granted to anyone to use this software for any purpose, including commercial applications, and to alter it and redistribute it freely, subject to the following restrictions:

- 1. The origin of this software must not be misrepresented; you must not claim that you wrote the original software. If you use this software in a product, an acknowledgment in the product documentation would be appreciated but is not required.
- 2. Altered source versions must be plainly marked as such, and must not be misrepresented as being the original software.
- 3. This notice may not be removed or altered from any source distribution.

9. Intel(R) MPI Benchmarks Copyright (c) Intel Corporation.

OpenUCX/UCX				
			2014-2015	UT-Battelle, LLC. All rights reserved.
			2014-2020	Mellanox Technologies Ltd. All rights reserved.
		(C)	2014-2015	The University of Houston System. All rights
1	reserved.			
	Copyright	(C)		The University of Tennessee and The University
			01	Tennessee Research Foundation. All rights
	reserved.			
			2016-2020	ARM Ltd. All rights reserved.
	Copyright	(c)	2016	Los Alamos National Security, LLC. All rights
1	reserved.			
		(C)	2016-2020	Advanced Micro Devices, Inc. All rights
1	reserved.			
	Copyright			UChicago Argonne, LLC. All rights reserved.
			2018-2020	NVIDIA CORPORATION. All rights reserved.
	Copyright	(C)	2020	Huawei Technologies Co., Ltd. All rights
1	reserved.			
	Copyright	(C)	2016-2020	Stony Brook University. All rights reserved.
	Mellanox Hardware MultiCast			library
	Copyright	(C)	2014-2020	Mellanox Technologies Ltd. All rights reserved.
	Copyright	(c)	2018-2020	NVIDIA CORPORATION. All rights reserved.
	BSD 3-Clause "New" or "Revised" License			

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- Neither the name of the copyright holder nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

10. OpenPMIx

Most files in this release are marked with the copyrights of the organizations who have edited them. The copyrights below are in no particular order and generally reflect members of the Open MPI core team who have contributed code that may or may not have been ported to PMIx. Per the terms of that LICENSE, we include the list here. The copyrights for code used under license from other parties are included in the corresponding files.

```
Copyright (c) 2004-2010 The Trustees of Indiana University and Indiana University Research and Technology Corporation. All rights reserved.
```

- Copyright (c) 2004-2010 The University of Tennessee and The University of Tennessee Research Foundation. All rights reserved.
- Copyright (c) 2004-2010 High Performance Computing Center Stuttgart, University of Stuttgart. All rights reserved.
- Copyright (c) 2004-2008 The Regents of the University of California.

 All rights reserved.
- Copyright (c) 2006-2010 Los Alamos National Security, LLC. All rights reserved.
- Copyright (c) 2006-2010 Cisco Systems, Inc. All rights reserved.
- Copyright (c) 2006-2010 Voltaire, Inc. All rights reserved.
- Copyright (c) 2006-2011 Sandia National Laboratories. All rights reserved.
- Copyright (c) 2006-2010 Sun Microsystems, Inc. All rights reserved.

 Use is subject to license terms.
- Copyright (c) 2006-2010 The University of Houston. All rights reserved.
- Copyright (c) 2006-2009 Myricom, Inc. All rights reserved.
- Copyright (c) 2007-2008 UT-Battelle, LLC. All rights reserved.
- Copyright (c) 2007-2019 IBM Corporation. All rights reserved.
- Copyright (c) 1998-2005 Forschungszentrum Juelich, Juelich Supercomputing Centre, Federal Republic of Germany
- Copyright (c) 2005-2008 ZIH, TU Dresden, Federal Republic of Germany
- Copyright (c) 2007 Evergrid, Inc. All rights reserved.
- Copyright (c) 2008 Chelsio, Inc. All rights reserved.
- Copyright (c) 2008-2009 Institut National de Recherche en Informatique. All rights reserved.
- Copyright (c) 2007 Lawrence Livermore National Security, LLC.
 All rights reserved.
- Copyright (c) 2007-2019 Mellanox Technologies. All rights reserved.
- Copyright (c) 2006-2010 QLogic Corporation. All rights reserved.
- Copyright (c) 2008-2010 Oak Ridge National Labs. All rights reserved.
- Copyright (c) 2006-2010 Oracle and/or its affiliates. All rights reserved.
- Copyright (c) 2009 Bull SAS. All rights reserved.
- Copyright (c) 2010 ARM ltd. All rights reserved.
- Copyright (c) 2010-2011 Alex Brick <bri>ka@ccs.neu.edu>. All rights reserved.
- Copyright (c) 2012 The University of Wisconsin-La Crosse. All rights reserved.
- Copyright (c) 2013-2019 Intel, Inc. All rights reserved.
- Copyright (c) 2011-2014 NVIDIA Corporation. All rights reserved.
- Copyright (c) 2019 Amazon.com, Inc. or its affiliates. All Rights reserved.

\$COPYRIGHT\$

Additional copyrights may follow

\$HEADER\$

The following LICENSE pertains to both PMIx and any code ported from Open MPI.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer listed in this license in the documentation and/or other materials provided with the distribution.

- Neither the name of the copyright holders nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

The copyright holders provide no reassurances that the source code provided does not infringe any patent, copyright, or any other intellectual property rights of third parties. The copyright holders disclaim any liability to any recipient for claims brought against recipient by any third party for infringement of that parties intellectual property rights.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

- 11. libpciaccess
 - (C) Copyright IBM Corporation 2006, 2007
 - (C) Copyright Eric Anholt 2006
 - (C) Copyright Mark Kettenis 2011
 - (C) Copyright Robert Millan 2012

Copyright (c) 2007, 2008, 2009, 2011, 2012, 2013 Oracle and/or its affiliates. Copyright 2009, 2012 Red Hat, Inc.

All Rights Reserved.

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation on the rights to use, copy, modify, merge, publish, distribute, sub license, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice (including the next paragraph) shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NON-INFRINGEMENT. IN NO EVENT SHALL IBM AND/OR THEIR SUPPLIERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Copyright (c) 2008 Juan Romero Pardines

Copyright (c) 2008, 2011 Mark Kettenis

Copyright (c) 2009 Michael Lorenz

Copyright (c) 2009, 2012 Samuel Thibault

Permission to use, copy, modify, and distribute this software for any purpose with or without fee is hereby granted, provided that the above copyright notice and this permission notice appear in all copies.

THE SOFTWARE IS PROVIDED "AS IS" AND THE AUTHOR DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS. IN NO EVENT SHALL THE AUTHOR BE LIABLE FOR ANY SPECIAL, DIRECT, INDIRECT, OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

.....

Copyright (C) 2000 The XFree86 Project, Inc. All Rights Reserved.

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE XFREE86 PROJECT BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Except as contained in this notice, the name of the XFree86 Project shall not be used in advertising or otherwise to promote the sale, use or other dealings in this Software without prior written authorization from the XFree86 Project.

Copyright (c) 2007 Paulo R. Zanoni, Tiago Vignatti Copyright (c) 2009 Tiago Vignatti

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

12. Intel(R) oneAPI Level Zero (Level Zero)

MIT License

Copyright (C) 2019-2021 Intel Corporation

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

The following third party programs have their own third party programs. These additional third party program files are as follows:

1. Intel(R) MPI Benchmarks

<install_dir>/mpi/latest/benchmarks/imb/license/third-party-programs.txt

2. Intel(R) Distribution for Python*

<install_dir>/intelpython/latest/licensing/third-party-programs.txt

^{*} Other names and brands may be claimed as the property of others.

Intel Simplified Software License (Version October 2022)

Use and Redistribution. You may use and redistribute the software, which is provided in binary form only, (the "Software"), without modification, provided the following conditions are met:

- * Redistributions must reproduce the above copyright notice and these terms of use in the Software and in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel nor the names of its suppliers may be used to endorse or promote products derived from this Software without specific prior written permission.
- * No reverse engineering, decompilation, or disassembly of the Software is permitted, nor any modification or alteration of the Software or its operation at any time, including during execution.

No other licenses. Except as provided in the preceding section, Intel grants no licenses or other rights by implication, estoppel or otherwise to, patent, copyright, trademark, trade name, service mark or other intellectual property licenses or rights of Intel.

Third party software. "Third Party Software" means the files (if any) listed in the "third-party-software.txt" or other similarly-named text file that may be included with the Software. Third Party Software, even if included with the distribution of the Software, may be governed by separate license terms, including without limitation, third party license terms, open source software notices and terms, and/or other Intel software license terms. These separate license terms solely govern Your use of the Third Party Software.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE DISCLAIMED. THIS SOFTWARE IS NOT INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE PERSONAL INJURY OR DEATH AND YOU AGREE THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS, COSTS, DAMAGES, EXPENSES, AND ATTORNEYS' FEES ARISING OUT OF ANY SUCH USE, EVEN IF ANY CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE SOFTWARE.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

No support. Intel may make changes to the Software, at any time without notice, and is not obligated to support, update or provide training for the Software.

Termination. Your right to use the Software is terminated in the event of your breach of this license.

Feedback. Should you provide Intel with comments, modifications, corrections, enhancements or other input ("Feedback") related to the Software, Intel will be free to use, disclose, reproduce, license or otherwise distribute or exploit the Feedback in its sole discretion without any obligations or restrictions of any

kind, including without limitation, intellectual property rights or licensing obligations.

Compliance with laws. You agree to comply with all relevant laws and regulations governing your use, transfer, import or export (or prohibition thereof) of the Software.

Governing law. All disputes will be governed by the laws of the United States of America and the State of Delaware without reference to conflict of law principles and subject to the exclusive jurisdiction of the state or federal courts sitting in the State of Delaware, and each party agrees that it submits to the personal jurisdiction and venue of those courts and waives any objections. THE UNITED NATIONS CONVENTION ON CONTRACTS FOR THE INTERNATIONAL SALE OF GOODS (1980) IS SPECIFICALLY EXCLUDED AND WILL NOT APPLY TO THE SOFTWARE.

Intel(R) oneAPI Math Kernel Library (oneMKL) Third Party Programs File

This file is the "third-party-programs.txt" file specified in the associated Intel end user license agreement for the Intel software you are licensing.

Third party programs and their corresponding required notices and/or license terms are listed below.

1. Netlib BLACS - Basic Linear Algebra Communication Subprograms:

Copyright (c) 1992-2013 The University of Tennessee and The

University of Tennessee Research Foundation. All rights reserved.

Copyright (c) 2000-2013 The University of California Berkeley. All rights reserved.

Copyright (c) 2006-2013 The University of Colorado Denver. All rights reserved.

Netlib LAPACK:

Copyright (c) 1992-2013 The University of Tennessee and The

University of Tennessee Research Foundation. All rights reserved.

Copyright (c) 2000-2013 The University of California Berkeley. All rights reserved.

Copyright (c) 2006-2013 The University of Colorado Denver. All rights reserved.

Netlib LAPACK95:

Copyright (c) 1992-2013 The University of Tennessee and The

University of Tennessee Research Foundation. All rights reserved.

Copyright (c) 2000-2013 The University of California Berkeley. All rights reserved.

Copyright (c) 2006-2013 The University of Colorado Denver. All rights reserved.

Netlib ScaLAPACK:

Copyright (c) 1992-2013 The University of Tennessee and The

University of Tennessee Research Foundation. All rights reserved.

Copyright (c) 2000-2013 The University of California Berkeley. All rights reserved.

Copyright (c) 2006-2013 The University of Colorado Denver. All rights reserved.

Modified BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer listed in this license in the documentation and/or other materials provided with the distribution.
- Neither the name of the copyright holders nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

The copyright holders provide no reassurances that the source code provided does not infringe any patent, copyright, or any other intellectual property rights of third parties. The copyright holders disclaim any liability to any recipient for claims brought against recipient by any third party for infringement of that parties intellectual property rights.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE

DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INCIDENTAL,

SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT(INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

2. Mersenne Twister with improved initialization:

Copyright (C) 1997 - 2002, Makoto Matsumoto and Takuji Nishimura, All rights reserved.

Modified BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. The names of its contributors may not be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INCIDENTAL,

SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

SFMT:

Copyright (c) 2006,2007-2014 Mutsuo Saito, Makoto Matsumoto and Hiroshima University. All rights reserved.

Modified BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * Neither the name of the Hiroshima University nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED

WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT(INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

4. Sobol sequence generator:

Copyright (c) 2008, Frances Y. Kuo and Stephen Joe, All rights reserved.

Modified BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * Neither the names of the copyright holders nor the names of the University of New South Wales and the University of Waikato and its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDERS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES(INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT(INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

5. The FEAST Eigenvalue Solver:

Copyright (c) 2009-2012, The Regents of the University of Massachusetts, Amherst. Developed by E. Polizzi. All rights reserved.

Modified BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. Neither the name of the University nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE AUTHOR ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF

MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE AUTHOR BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT(INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

6. xbyak:

Copyright (c) 2007 MITSUNARI Shigeo All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.

Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

Neither the name of the copyright owner nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

ソースコード形式かバイナリ形式か、変更するかしないかを問わず、以下の条件を満た

す場合に限り、再頒布および使用が許可されます。

ソースコードを再頒布する場合、上記の著作権表示、本条件一覧、および下記免責条項 を含めること。

バイナリ形式で再頒布する場合、頒布物に付属のドキュメント等の資料に、上記の著作 権表示、本条件一覧、および下記免責条項を含めること。 書面による特別の許可なしに、本ソフトウェアから派生した製品の宣伝または販売促進

に、著作権者の名前またはコントリビューターの名前を使用してはならない。 本ソフトウェアは、著作権者およびコントリビューターによって「現状のまま」提供されており、明示黙示を問わず、商業的な使用可能性、および特定の目的に対する適合性 に関する暗黙の保証も含め、またそれに限定されない、いかなる保証もありません。 著作権者もコントリビューターも、事由のいかんを問わず、 損害発生の原因いかんを 問わず、かつ責任の根拠が契約であるか厳格責任であるか(過失その他の)不法行為で あるかを問わず、仮にそのような損害が発生する可能性を知らされていたとしても、 本ソフトウェアの使用によって発生した(代替品または代用サービスの調達、使用の 喪失、データの喪失、利益の喪失、業務の中断も含め、またそれに限定されない)直接 損害、間接損害、偶発的な損害、特別損害、懲罰的損害、または結果損害について、 一切責任を負わないものとします。

7. Intel(R) Instrumentation and Tracing Technology API: Copyright(c) 2005-2014, Intel Corporation, All rights reserved.

memkind

Copyright (C) 2014-2020 Intel Corporation

Modified BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, his list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel Corporation nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

8. Intel Open Source Technology Center Safe String Library: strcat s.c

October 2008, Bo Berry

Copyright (c) 2008-2011 by Cisco Systems, Inc. All rights

reserved.

level-zero:

Copyright (c) 2019 Intel Corporation

MIT License

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

9. Khronos Group - OpenCL: Copyright (c) 2008 - 2013 The Khronos Group Inc.

Apache License
Version 2.0, January 2004
http://www.apache.org/licenses/

TERMS AND CONDITIONS FOR USE, REPRODUCTION, AND DISTRIBUTION

1. Definitions.

"License" shall mean the terms and conditions for use, reproduction, and distribution as defined by Sections 1 through 9 of this document.

"Licensor" shall mean the copyright owner or entity authorized by the copyright owner that is granting the License.

"Legal Entity" shall mean the union of the acting entity and all other entities that control, are controlled by, or are under common control with that entity. For the purposes of this definition, "control" means (i) the power, direct or indirect, to cause the direction or management of such entity, whether by contract or otherwise, or (ii) ownership of fifty percent (50%) or more of the outstanding shares, or (iii) beneficial ownership of such entity.

"You" (or "Your") shall mean an individual or Legal Entity exercising permissions granted by this License.

"Source" form shall mean the preferred form for making modifications, including but not limited to software source code, documentation source, and configuration files.

"Object" form shall mean any form resulting from mechanical transformation or translation of a Source form, including but not limited to compiled object code, generated documentation, and conversions to other media types.

"Work" shall mean the work of authorship, whether in Source or Object form, made available under the License, as indicated by a copyright notice that is included in or attached to the work(an example is provided in the Appendix below).

"Derivative Works" shall mean any work, whether in Source or Object form, that is based on (or derived from) the Work and for which the editorial revisions, annotations, elaborations, or other modifications represent, as a whole, an original work of authorship. For the purposes of this License, Derivative Works shall not include works that remain separable from, or merely link (or bind by name) to the interfaces of, the Work and Derivative Works thereof.

"Contribution" shall mean any work of authorship, including the original version of the Work and any modifications or additions to that Work or Derivative Works thereof, that is intentionally submitted to Licensor for inclusion in the Work by the copyright owner or by an individual or Legal Entity authorized to submit on behalf of the copyright owner. For the purposes of this definition, "submitted" means any form of electronic, verbal, or written communication sent to the Licensor or its representatives, including but not limited to

communication on electronic mailing lists, source code control systems, and issue tracking systems that are managed by, or on behalf of, the Licensor for the purpose of discussing and improving the Work, but excluding communication that is conspicuously marked or otherwise designated in writing by the copyright owner as "Not a Contribution."

"Contributor" shall mean Licensor and any individual or Legal Entity on behalf of whom a Contribution has been received by Licensor and subsequently incorporated within the Work.

2. Grant of Copyright License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare Derivative Works of, publicly display, publicly perform, sublicense, and distribute the Work and such Derivative Works in Source or Object form.

3. Grant of Patent License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable(except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by such Contributor that are necessarily infringed by their

Contribution(s) alone or by combination of their Contribution(s) with the Work to which such Contribution(s) was submitted. If You institute patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Work or a Contribution incorporated within the Work constitutes direct or contributory patent infringement, then any patent licenses granted to You under this License for that Work shall terminate as of the date such litigation is filed.

- 4. Redistribution. You may reproduce and distribute copies of the Work or Derivative Works thereof in any medium, with or without modifications, and in Source or Object form, provided that You meet the following conditions:
 - (a) You must give any other recipients of the Work or Derivative Works a copy of this License; and
- (b) You must cause any modified files to carry prominent notices stating that You changed the files; and
- (c) You must retain, in the Source form of any Derivative Works that You distribute, all copyright, patent, trademark, and attribution notices from the Source form of the Work, excluding those notices that do not pertain to any part of the Derivative Works; and
- (d) If the Work includes a "NOTICE" text file as part of its distribution, then any Derivative Works that You distribute must include a readable copy of the attribution notices contained within such NOTICE file, excluding those notices that do not pertain to any part of the Derivative Works, in at least one of the following places: within a NOTICE text file distributed as part of the Derivative Works; within the Source form or documentation, if provided along with the Derivative Works; or, within a display generated by the Derivative Works, if and wherever such third-party notices normally appear. The contents of the NOTICE file are for informational purposes only and do not modify the License. You may add Your own attribution notices within Derivative Works that You distribute, alongside or as an addendum to the NOTICE text from the Work, provided that such additional attribution notices cannot be construed as modifying the License.

You may add Your own copyright statement to Your modifications and may provide additional or different license terms and conditions for use, reproduction, or distribution of Your modifications, or for any such Derivative Works as a whole, provided Your use, reproduction, and distribution of the Work otherwise complies with the conditions stated in this License.

5. Submission of Contributions. Unless You explicitly state otherwise, any Contribution intentionally submitted for inclusion in the Work by You to the Licensor shall be under the terms and conditions of this License, without any additional terms or conditions.

Notwithstanding the above, nothing herein shall supersede or modify the terms of any separate license agreement you may have executed with Licensor regarding such Contributions.

6. Trademarks. This License does not grant permission to use the trade names, trademarks, service marks, or product names of the Licensor, except as required for reasonable and customary use in describing the origin of the Work and reproducing the content of the NOTICE file.

- 7. Disclaimer of Warranty. Unless required by applicable law or agreed to in writing, Licensor provides the Work (and each Contributor provides its Contributions) on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE. You are solely responsible for determining the appropriateness of using or redistributing the Work and assume any risks associated with Your exercise of permissions under this License.
- 8. Limitation of Liability. In no event and under no legal theory, whether in tort (including negligence), contract, or otherwise, unless required by applicable law (such as deliberate and grossly negligent acts) or agreed to in writing, shall any Contributor be liable to You for damages, including any direct, indirect, special, incidental, or consequential damages of any character arising as a result of this License or out of the use or inability to use the Work (including but not limited to damages for loss of goodwill, work stoppage, computer failure or malfunction, or any and all other commercial damages or losses), even if such Contributor has been advised of the possibility of such damages.
- 9. Accepting Warranty or Additional Liability. While redistributing the Work or Derivative Works thereof, You may choose to offer, and charge a fee for, acceptance of support, warranty, indemnity, or other liability obligations and/or rights consistent with this License. However, in accepting such obligations, You may act only on Your own behalf and on Your sole responsibility, not on behalf of any other Contributor, and only if You agree to indemnify, defend, and hold each Contributor harmless for any liability incurred by, or claims asserted against, such Contributor by reason of your accepting any such warranty or additional liability.

END OF TERMS AND CONDITIONS

APPENDIX: How to apply the Apache License to your work.

To apply the Apache License to your work, attach the following boilerplate notice, with the fields enclosed by brackets "[]" replaced with your own identifying information. (Don't include the brackets!) The text should be enclosed in the appropriate comment syntax for the file format. We also recommend that a file or class name and description of purpose be included on the same "printed page" as the copyright notice for easier identification within third-party archives.

Copyright [yyyy] [name of copyright owner]

Licensed under the Apache License, Version 2.0 (the "License"); you may not use this file except in compliance with the License.

You may obtain a copy of the License at

http://www.apache.org/licenses/LICENSE-2.0

Unless required by applicable law or agreed to in writing, software distributed under the License is distributed on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied.

See the License for the specific language governing permissions and limitations under the License.

10. Intel(R) oneAPI Threading Building Blocks:
Copyright 2005-2019 Intel Corporation. All Rights Reserved.

Intel(R) Integrated Performance Primitives Library Copyright 2020 Intel Corporation. All Rights Reserved.

Intel(R) OpenMP* Runtime:
Copyright 1985-2019 Intel Corporation. All Rights Reserved.

The source code contained or described herein and all documents related to the source code ("Material") are owned by Intel Corporation or its suppliers or licensors. Title to the Material remains with Intel Corporation or its suppliers and licensors. The Material is protected by worldwide copyright laws and treaty provisions. No part of the Material may be used, copied, reproduced, modified, published, uploaded, posted, transmitted, distributed, or disclosed in any way without Intel's prior express written permission.

No license under any patent, copyright, trade secret or other intellectual property right is granted to or conferred upon you by disclosure or delivery of the Materials, either expressly, by implication, inducement, estoppel or otherwise. Any license under such intellectual property rights must be express and approved by Intel in writing.

Portions of this software are protected under the following patents:

- U.S. Patent 5,812,852
- U.S. Patent 6,792,599
- U.S. Patent 7,069,556
- U.S. Patent 7,328,433
- U.S. Patent 7,500,242

Intel Simplified Software License (Version April 2018)

Copyright (c) 2018 Intel Corporation.

Use and Redistribution. You may use and redistribute the software (the "Software"), without modification, provided the following conditions are met:

- * Redistributions must reproduce the above copyright notice and the following terms of use in the Software and in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel nor the names of its suppliers may be used to endorse or promote products derived from this Software without specific prior written permission.
- * No reverse engineering, decompilation, or disassembly of this Software is permitted.

Limited patent license. Intel grants you a world-wide, royalty-free, non-exclusive license under patents it now or hereafter owns or controls to make, have made, use, import, offer to sell and sell ("Utilize") this Software, but solely to the extent that any such patent is necessary to Utilize the Software alone. The patent license shall not apply to any combinations which include this software. No hardware per se is licensed hereunder.

Third party and other Intel programs. "Third Party Programs" are the files listed in the "third-party-programs.txt" text file that is included with the Software and may include Intel programs under separate license terms. Third Party Programs, even if included with the distribution of the Materials, are governed by separate license terms and those license terms solely govern your use of those programs.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE DISCLAIMED. THIS SOFTWARE IS NOT INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE PERSONAL INJURY OR DEATH AND YOU AGREE

THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS, COSTS, DAMAGES, EXPENSES, AND ATTORNEYS' FEES ARISING OUT OF ANY SUCH USE, EVEN IF ANY CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE MATERIALS.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE. YOU AGREE TO INDEMNIFY AND HOLD INTEL HARMLESS AGAINST ANY CLAIMS AND EXPENSES RESULTING FROM YOUR USE OR UNAUTHORIZED USE OF THE SOFTWARE.

No support. Intel may make changes to the Software, at any time without notice, and is not obligated to support, update or provide training for the Software.

Termination. Intel may terminate your right to use the Software in the event of your breach of this Agreement and you fail to cure the breach within a reasonable period of time.

Feedback. Should you provide Intel with comments, modifications, corrections, enhancements or other input ("Feedback") related to the Software Intel will be free to use, disclose, reproduce, license or otherwise distribute or exploit the Feedback in its sole discretion without any obligations or restrictions of any kind, including without limitation, intellectual property rights or licensing obligations.

Compliance with laws. You agree to comply with all relevant laws and regulations governing your use, transfer, import or export (or prohibition thereof) of the Software.

Governing law. All disputes will be governed by the laws of the United States of America and the State of Delaware without reference to conflict of law principles and subject to the exclusive jurisdiction of the state or federal courts sitting in the State of Delaware, and each party agrees that it submits to the personal jurisdiction and venue of those courts and waives any objections. The United Nations Convention on Contracts for the International Sale of Goods (1980) is specifically excluded and will not apply to the Software.

*Other names and brands may be claimed as the property of others.

11. Microsoft HPC Pack Software Development Kit (SDK)

Micrososft Software License

MICROSOFT SOFTWARE LICENSE TERMS

MICROSOFT HPC PACK SOFTWARE DEVELOPMENT KIT (SDK)

Last updated: August 2018

These license terms are an agreement between Microsoft Corporation (or based on where you live, one of its affiliates) and you. Please read them. They apply to the software named above, which includes the media on which you received it, if any. The terms also apply to any Microsoft

updates,

supplements,

Internet-based services, and

support services

for this software, unless other terms accompany those items. If so, those terms apply.

By using the software, you accept these terms. If you do not accept them, do not

use the software.

If you comply with these license terms, you have the perpetual rights below. INSTALLATION AND USE RIGHTS. You may install and use any number of copies of the software on your devices to design, develop and test your programs. Subject to your compliance with the terms below, you are permitted to distribute the software, in object code form only, in programs you develop and you may permit distributors of your programs to copy and distribute the software as part of those programs.

Distribution Requirements. For any software you distribute in your programs, you must

add significant primary functionality to it in your programs;

for any Distributable Code having a filename extension of .lib, distribute only the results of running such Distributable Code through a linker with your program;

require distributors and external end users to agree to terms that protect it at least as much as this agreement;

display your valid copyright notice on your programs; and

indemnify, defend, and hold harmless Microsoft from any claims, including attorneys' fees, related to the distribution or use of your programs.

ii. Distribution Restrictions. You may not

alter any copyright, trademark or patent notice in the software;

use Microsoft's trademarks in your programs' names or in a way that suggests your programs come from or are endorsed by Microsoft;

distribute software to run on a platform other than the Windows platform; include software in malicious, deceptive or unlawful programs; or modify or distribute the software so that any part of it becomes subject to

modify or distribute the software so that any part of it becomes subject to an Excluded License. An Excluded License is one that requires, as a condition of use, modification or distribution, that

the code be disclosed or distributed in source code form; or others have the right to modify it.

2. SCOPE OF LICENSE. The software is licensed, not sold. This agreement only gives you some rights to use the software. Microsoft reserves all other rights. Unless applicable law gives you more rights despite this limitation, you may use the software only as expressly permitted in this agreement. In doing so, you must comply with any technical limitations in the software that only allow you to use it in certain ways. You may not

work around any technical limitations in the software;

reverse engineer, decompile or disassemble the software, except and only to the extent that applicable law expressly permits, despite this limitation; publish the software for others to copy;

rent, lease or lend the software;

transfer the software or this agreement to any third party; or use the software for commercial software hosting services.

- 3. DOCUMENTATION. Any person that has valid access to your computer or internal network may copy and use the documentation for your internal, reference purposes.
- 4. EXPORT RESTRICTIONS. You must comply with all domestic and international export laws and regulations that apply to the software, which include restrictions on destinations, end users, and end use. For further information on export restrictions, visit http://aka.ms/exporting.
- 5. SUPPORT SERVICES. Microsoft is not obligated under this agreement to provide any support services for the software. Any support provided is "as is", "with all faults", and without warranty of any kind.
- 6. UPDATES. The software may periodically check for updates, and download and install them for you. You may obtain updates only from Microsoft or authorized sources. Microsoft may need to update your system to provide you with updates. You agree to receive these automatic updates without any additional notice. Updates may not include or support all existing software features, services, or peripheral devices.
- 7. BINDING ARBITRATION AND CLASS ACTION WAIVER. This Section applies if you live in (or, if a business, your principal place of business is in) the United States. If you and Microsoft have a dispute, you and Microsoft agree to try for 60 days to resolve it informally. If you and Microsoft can't, you and Microsoft agree to binding individual arbitration before the American Arbitration

Association under the Federal Arbitration Act ("FAA"), and not to sue in court in front of a judge or jury. Instead, a neutral arbitrator will decide. Class action lawsuits, class-wide arbitrations, private attorney-general actions, and any other proceeding where someone acts in a representative capacity are not allowed; nor is combining individual proceedings without the consent of all parties. The complete Arbitration Agreement contains more terms and is at http://aka.ms/arb-agreement-1. You and Microsoft agree to these terms.

8. ENTIRE AGREEMENT. This agreement, and any other terms Microsoft may provide

for supplements, updates, or third-party applications, is the entire agreement

- for the software. 9. APPLICABLE LAW AND PLACE TO RESOLVE DISPUTES. If you acquired the software in the United States or Canada, the laws of the state or province where you live (or, if a business, where your principal place of business is located) govern the interpretation of this agreement, claims for its breach, and all other claims (including consumer protection, unfair competition, and tort claims), regardless of conflict of laws principles, except that the FAA governs everything related to arbitration. If you acquired the software in any other country, its laws apply, except that the FAA governs everything related to arbitration. If U.S. federal jurisdiction exists, you and Microsoft consent to exclusive jurisdiction and venue in the federal court in King County, Washington for all disputes heard in court (excluding arbitration). If not, you and Microsoft consent to exclusive jurisdiction and venue in the Superior Court of King County, Washington for all disputes heard in court (excluding arbitration). 10. CONSUMER RIGHTS; REGIONAL VARIATIONS. This agreement describes certain legal rights. You may have other rights, including consumer rights, under the laws of your state or country. Separate and apart from your relationship with Microsoft, you may also have rights with respect to the party from which you acquired the software. This agreement does not change those other rights if the laws of your state or country do not permit it to do so. For example, if you acquired the software in one of the below regions, or mandatory country law applies, then the following provisions apply to you:
- Australia. You have statutory guarantees under the Australian Consumer Law and nothing in this agreement is intended to affect those rights.
- Canada. If you acquired this software in Canada, you may stop receiving updates by turning off the automatic update feature, disconnecting your device from the Internet (if and when you re-connect to the Internet, however, the software will resume checking for and installing updates), or uninstalling the software. The product documentation, if any, may also specify how to turn off updates for your specific device or software.

Germany and Austria.

Warranty. The properly licensed software will perform substantially as described in any Microsoft materials that accompany the software. However, Microsoft gives no contractual guarantee in relation to the licensed software.

no contractual guarantee in relation to the licensed software. Limitation of Liability. In case of intentional conduct, gross negligence, claims based on the Product Liability Act, as well as, in case of death or personal or physical injury, Microsoft is liable according to the statutory law. Subject to the foregoing clause ii., Microsoft will only be liable for slight negligence if Microsoft is in breach of such material contractual obligations, the fulfillment of which facilitate the due performance of this agreement, the breach of which would endanger the purpose of this agreement and the compliance with which a party may constantly trust in (so-called "cardinal obligations"). In other cases of slight negligence, Microsoft will not be liable for slight negligence.

- 11. DISCLAIMER OF WARRANTY. THE SOFTWARE IS LICENSED "AS IS." YOU BEAR THE RISK OF USING IT. MICROSOFT GIVES NO EXPRESS WARRANTIES, GUARANTEES, OR CONDITIONS. TO THE EXTENT PERMITTED UNDER APPLICABLE LAWS, MICROSOFT EXCLUDES ALL IMPLIED WARRANTIES, INCLUDING MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT.
- 12. LIMITATION ON AND EXCLUSION OF DAMAGES. IF YOU HAVE ANY BASIS FOR RECOVERING DAMAGES DESPITE THE PRECEDING DISCLAIMER OF WARRANTY, YOU CAN RECOVER FROM MICROSOFT AND ITS SUPPLIERS ONLY DIRECT DAMAGES UP TO U.S. \$5.00. YOU CANNOT RECOVER ANY OTHER DAMAGES, INCLUDING CONSEQUENTIAL, LOST PROFITS, SPECIAL, INDIRECT, OR INCIDENTAL DAMAGES.
- This limitation applies to (a) anything related to the software, services,

content (including code) on third party Internet sites, or third party applications; and (b) claims for breach of contract, warranty, guarantee, or condition; strict liability, negligence, or other tort; or any other claim; in each case to the extent permitted by applicable law.

It also applies even if Microsoft knew or should have known about the possibility of the damages. The above limitation or exclusion may not apply to you because your state, province, or country may not allow the exclusion or limitation of incidental, consequential, or other damages.

Please note: As this software is distributed in Canada, some of the clauses in this agreement are provided below in French.

Remarque: Ce logiciel étant distribué au Canada, certaines des clauses dans ce contrat sont fournies ci-dessous en français.

EXONÉRATION DE GARANTIE. Le logiciel visé par une licence est offert « tel quel ». Toute utilisation de ce logiciel est à votre seule risque et péril. Microsoft n'accorde aucune autre garantie expresse. Vous pouvez bénéficier de droits additionnels en vertu du droit local sur la protection des consommateurs, que ce contrat ne peut modifier. La ou elles sont permises par le droit locale, les garanties implicites de qualité marchande, d'adéquation à un usage particulier et d'absence de contrefaçon sont exclues.

LIMITATION DES DOMMAGES-ÎNTÉRÊTS ET EXCLUSION DE RESPONSABILITÉ POUR LES DOMMAGES. Vous pouvez obtenir de Microsoft et de ses fournisseurs une indemnisation en cas de dommages directs uniquement à hauteur de 5,00 \$ US. Vous ne pouvez prétendre à aucune indemnisation pour les autres dommages, y compris les dommages spéciaux, indirects ou accessoires et pertes de bénéfices. Cette limitation concerne:

- tout ce qui est relié au logiciel, aux services ou au contenu (y compris le code) figurant sur des sites Internet tiers ou dans des programmes tiers; et
- les réclamations au titre de violation de contrat ou de garantie, ou au titre de responsabilité stricte, de négligence ou d'une autre faute dans la limite autorisée par la loi en viqueur.

Elle s'applique également, même si Microsoft connaissait ou devrait connaître l'éventualité d'un tel dommage. Si votre pays n'autorise pas l'exclusion ou la limitation de responsabilité pour les dommages indirects, accessoires ou de quelque nature que ce soit, il se peut que la limitation ou l'exclusion cidessus ne s'appliquera pas à votre égard.

EFFET JURIDIQUE. Le présent contrat décrit certains droits juridiques. Vous pourriez avoir d'autres droits prévus par les lois de votre pays. Le présent contrat ne modifie pas les droits que vous confèrent les lois de votre pays si celles-ci ne le permettent pas.

*Other names and brands may be claimed as the property of others.

12. Netlib XBLAS - Extra Precise Basic Linear Algebra Subroutines:
Copyright (c) 2008-2009 The University of California Berkeley. All rights reserved.

\$COPYRIGHT\$

Additional copyrights may follow

\$HEADER\$

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer listed

in this license in the documentation and/or other materials provided with the distribution.

- Neither the name of the copyright holders nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

The following third party programs have their own third party program files. These additional third party program files are as follows:

- 1. List of third party for Intel(R) oneAPI Math Kernel Library Benchmarks is available in the third-party-programs-benchmarks.txt.
- 2. List of third party for Intel Open Source Technology Center Safe String Library is available in the third-party-programs-safestring.txt.
- 2. List of third party for Intel(R) oneAPI Threading Building Blocks for Linux is available in <install-dir>/tbb/<version>/licensing/third-party-programs.txt.
- 3. List of third party for Intel(R) oneAPI Threading Building Blocks for Windows is available in <install-dir>\tbb\<version>\licensing\third-party-programs.txt.
- 4. List of third party for Intel(R) OpenMP* Runtime is available in the third-party-programs-openmp.txt file.
- 5. List of third party for Intel(R) Integrated Performance Primitives Library is available in the third-party-programs-ipp.txt file.

^{*}Other names and brands may be claimed as the property of others.

Intel(R) oneAPI Math Kernel Library (oneMKL) Benchmarks Third Party Programs
File

This file is the "third-party-programs.txt" file specified in the associated Intel end user license agreement for the Intel software you are licensing.

Third party programs and their corresponding required notices and/or license terms are listed below.

.....

HPCG: High Performance Conjugate Gradient Benchmark:
 HPCG - 3.1 - March 28, 2019
 Michael A. Heroux
 Scalable Algorithms Group, Center for Computing Research
 Sandia National Laboratories, Albuquerque, NM
 Piotr Luszczek
 Jack Dongarra
 University of Tennessee, Knoxville
 Innovative Computing Laboratory
 (C) Copyright 2013-2019 All Rights Reserved

Modified BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions, and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. The name of the University, the name of the Laboratory, or the names of its contributors may not be used to endorse or promote products derived from this software without specific written permission.
- -- Disclaimer:

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE UNIVERSITY OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT(INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

2. HPL - High-Performance Linpack Benchmark: HPL - 2.3 - December 2, 2018 Antoine P. Petitet University of Tennessee, Knoxville Innovative Computing Laboratory (C) Copyright 2000-2008 All Rights Reserved

HPL License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions, and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. All advertising materials mentioning features or use of this software must display the following acknowledgement:

This product includes software developed at the University of Tennessee, Knoxville, Innovative Computing Laboratory.

4. The name of the University, the name of the Laboratory, or the names of its contributors may not be used to endorse or promote products derived from this software without specific written permission.

-- Disclaimer:

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE UNIVERSITY OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT(INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

^{*}Other names and brands may be claimed as the property of others.

Intel(R) Integrated Performance Primitives Third Party Programs File

This file is the "third-party-programs-ipp.txt" file specified in the associated Intel end user license

agreement for the Intel software you are licensing.

Third party programs and their corresponding required notices and/or license terms are listed below.

1. Intel® Threading Building Blocks (Intel® TBB)

Copyright 2018 Intel Corporation

Intel Simplified Software License

Use and Redistribution. You may use and redistribute the software (the "Software"), without modification, provided the following conditions are met:

- * Redistributions must reproduce the above copyright notice and the following terms of use in the Software and in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel nor the names of its suppliers may be used to endorse
- or promote products derived from this Software without specific prior written permission.
- * No reverse engineering, decompilation, or disassembly of this Software is permitted.

Limited patent license. Intel grants you a world-wide, royalty-free, non-exclusive

license under patents it now or hereafter owns or controls to make, have made, use,

import, offer to sell and sell ("Utilize") this Software, but solely to the extent

that any such patent is necessary to Utilize the Software alone. The patent license

shall not apply to any combinations which include this software. No hardware per se is licensed hereunder.

Third party and other Intel programs. "Third Party Programs" are the files listed

in the "third-party-programs.txt" text file that is included with the Software and

may include Intel programs under separate license terms. Third Party Programs, even if included with the distribution of the Materials, are governed by separate

license terms and those license terms solely govern your use of those programs.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES,

INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS

FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE DISCLAIMED. THIS SOFTWARE IS NOT

INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE

PERSONAL INJURY OR DEATH AND YOU AGREE THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS,

COSTS, DAMAGES, EXPENSES, AND ATTORNEYS' FEES ARISING OUT OF ANY SUCH USE, EVEN

IF ANY

CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE MATERIALS.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT,

INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO,

PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS

INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT

LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE

OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE. YOU AGREE TO INDEMNIFY

AND HOLD INTEL HARMLESS AGAINST ANY CLAIMS AND EXPENSES RESULTING FROM YOUR USE OR

UNAUTHORIZED USE OF THE SOFTWARE.

No support. Intel may make changes to the Software, at any time without notice, and is not

obligated to support, update or provide training for the Software.

Termination. Intel may terminate your right to use the Software in the event of your breach

of this Agreement and you fail to cure the breach within a reasonable period of time.

Feedback. Should you provide Intel with comments, modifications, corrections, enhancements

or other input ("Feedback") related to the Software Intel will be free to use, disclose,

reproduce, license or otherwise distribute or exploit the Feedback in its sole discretion

without any obligations or restrictions of any kind, including without limitation,

intellectual property rights or licensing obligations.

Compliance with laws. You agree to comply with all relevant laws and regulations governing

your use, transfer, import or export (or prohibition thereof) of the Software.

Governing law. All disputes will be governed by the laws of the United States of America and

the State of Delaware without reference to conflict of law principles and subject to the

exclusive jurisdiction of the state or federal courts sitting in the State of Delaware, and

each party agrees that it submits to the personal jurisdiction and venue of those courts and $% \left(1\right) =\left(1\right) +\left(1\right$

waives any objections. The United Nations Convention on Contracts for the International $\ensuremath{\mathsf{I}}$

Sale of Goods (1980) is specifically excluded and will not apply to the Software.

The following third party programs have their own third party program files. These additional

third party program files are as follows:

1. List of third party for Intel(R) oneAPI Threading Building Blocks for Linux is available in <install-dir>/tbb/<version>/licensing/third-party-programs.txt.

2. List of third party for Intel(R) oneAPI Threading Building Blocks for Windows is available in <install-dir>\tbb\<version>\licensing\third-party-programs.txt.

Other names and brands may be claimed as the property of others.

This file is the "third-party-programs.txt" file specified in the associated Intel end user

license agreement for the Intel software you are licensing. The third party programs and

their corresponding required notices and/or license terms are listed below.

1. The BGET Memory Allocator

This program is in the public domain.

2. Instrumentation and Tracing Technology (ITT) Notify User API Copyright (c) 2005-2014 Intel Corporation. All rights reserved.

The 3-Clause BSD License

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * Neither the name of Intel Corporation nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

3. LLVM OpenMP

Apache License v2.0 with LLVM Exceptions

Apache License
Version 2.0, January 2004
http://www.apache.org/licenses/

TERMS AND CONDITIONS FOR USE, REPRODUCTION, AND DISTRIBUTION

Definitions.

"License" shall mean the terms and conditions for use, reproduction, and distribution as defined by Sections 1 through 9 of this document.

"Licensor" shall mean the copyright owner or entity authorized by the copyright owner that is granting the License.

"Legal Entity" shall mean the union of the acting entity and all other entities that control, are controlled by, or are under common control with that entity. For the purposes of this definition, "control" means (i) the power, direct or indirect, to cause the direction or management of such entity, whether by contract or

otherwise, or (ii) ownership of fifty percent (50%) or more of the outstanding shares, or (iii) beneficial ownership of such entity.

"You" (or "Your") shall mean an individual or Legal Entity exercising permissions granted by this License.

"Source" form shall mean the preferred form for making modifications, including but not limited to software source code, documentation source, and configuration files.

"Object" form shall mean any form resulting from mechanical transformation or translation of a Source form, including but not limited to compiled object code, generated documentation, and conversions to other media types.

"Work" shall mean the work of authorship, whether in Source or Object form, made available under the License, as indicated by a copyright notice that is included in or attached to the work (an example is provided in the Appendix below).

"Derivative Works" shall mean any work, whether in Source or Object form, that is based on (or derived from) the Work and for which the editorial revisions, annotations, elaborations, or other modifications represent, as a whole, an original work of authorship. For the purposes of this License, Derivative Works shall not include works that remain separable from, or merely link (or bind by name) to the interfaces of, the Work and Derivative Works thereof.

"Contribution" shall mean any work of authorship, including the original version of the Work and any modifications or additions to that Work or Derivative Works thereof, that is intentionally submitted to Licensor for inclusion in the Work by the copyright owner or by an individual or Legal Entity authorized to submit on behalf of the copyright owner. For the purposes of this definition, "submitted" means any form of electronic, verbal, or written communication sent to the Licensor or its representatives, including but not limited to communication on electronic mailing lists, source code control systems, and issue tracking systems that are managed by, or on behalf of, the Licensor for the purpose of discussing and improving the Work, but excluding communication that is conspicuously marked or otherwise designated in writing by the copyright owner as "Not a Contribution."

"Contributor" shall mean Licensor and any individual or Legal Entity on behalf of whom a Contribution has been received by Licensor and subsequently incorporated within the Work.

- 2. Grant of Copyright License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare Derivative Works of, publicly display, publicly perform, sublicense, and distribute the Work and such Derivative Works in Source or Object form.
- 3. Grant of Patent License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable (except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by such Contributor that are necessarily infringed by their Contribution(s) alone or by combination of their Contribution(s) with the Work to which such Contribution(s) was submitted. If You institute patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Work

or a Contribution incorporated within the Work constitutes direct or contributory patent infringement, then any patent licenses granted to You under this License for that Work shall terminate as of the date such litigation is filed.

- 4. Redistribution. You may reproduce and distribute copies of the Work or Derivative Works thereof in any medium, with or without modifications, and in Source or Object form, provided that You meet the following conditions:
 - (a) You must give any other recipients of the Work or Derivative Works a copy of this License; and
 - (b) You must cause any modified files to carry prominent notices stating that You changed the files; and
 - (c) You must retain, in the Source form of any Derivative Works that You distribute, all copyright, patent, trademark, and attribution notices from the Source form of the Work, excluding those notices that do not pertain to any part of the Derivative Works; and
 - (d) If the Work includes a "NOTICE" text file as part of its distribution, then any Derivative Works that You distribute must include a readable copy of the attribution notices contained within such NOTICE file, excluding those notices that do not pertain to any part of the Derivative Works, in at least one of the following places: within a NOTICE text file distributed as part of the Derivative Works; within the Source form or documentation, if provided along with the Derivative Works; or, within a display generated by the Derivative Works, if and wherever such third-party notices normally appear. The contents of the NOTICE file are for informational purposes only and do not modify the License. You may add Your own attribution notices within Derivative Works that You distribute, alongside or as an addendum to the NOTICE text from the Work, provided that such additional attribution notices cannot be construed as modifying the License.

You may add Your own copyright statement to Your modifications and may provide additional or different license terms and conditions for use, reproduction, or distribution of Your modifications, or for any such Derivative Works as a whole, provided Your use, reproduction, and distribution of the Work otherwise complies with the conditions stated in this License.

- 5. Submission of Contributions. Unless You explicitly state otherwise, any Contribution intentionally submitted for inclusion in the Work by You to the Licensor shall be under the terms and conditions of this License, without any additional terms or conditions. Notwithstanding the above, nothing herein shall supersede or modify the terms of any separate license agreement you may have executed with Licensor regarding such Contributions.
- 6. Trademarks. This License does not grant permission to use the trade names, trademarks, service marks, or product names of the Licensor, except as required for reasonable and customary use in describing the origin of the Work and reproducing the content of the NOTICE file.
- 7. Disclaimer of Warranty. Unless required by applicable law or agreed to in writing, Licensor provides the Work (and each Contributor provides its Contributions) on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions

of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE. You are solely responsible for determining the appropriateness of using or redistributing the Work and assume any risks associated with Your exercise of permissions under this License.

- 8. Limitation of Liability. In no event and under no legal theory, whether in tort (including negligence), contract, or otherwise, unless required by applicable law (such as deliberate and grossly negligent acts) or agreed to in writing, shall any Contributor be liable to You for damages, including any direct, indirect, special, incidental, or consequential damages of any character arising as a result of this License or out of the use or inability to use the Work (including but not limited to damages for loss of goodwill, work stoppage, computer failure or malfunction, or any and all other commercial damages or losses), even if such Contributor has been advised of the possibility of such damages.
- 9. Accepting Warranty or Additional Liability. While redistributing the Work or Derivative Works thereof, You may choose to offer, and charge a fee for, acceptance of support, warranty, indemnity, or other liability obligations and/or rights consistent with this License. However, in accepting such obligations, You may act only on Your own behalf and on Your sole responsibility, not on behalf of any other Contributor, and only if You agree to indemnify, defend, and hold each Contributor harmless for any liability incurred by, or claims asserted against, such Contributor by reason of your accepting any such warranty or additional liability.

END OF TERMS AND CONDITIONS

APPENDIX: How to apply the Apache License to your work.

To apply the Apache License to your work, attach the following boilerplate notice, with the fields enclosed by brackets "[]" replaced with your own identifying information. (Don't include the brackets!) The text should be enclosed in the appropriate comment syntax for the file format. We also recommend that a file or class name and description of purpose be included on the same "printed page" as the copyright notice for easier identification within third-party archives.

Copyright [yyyy] [name of copyright owner]

Licensed under the Apache License, Version 2.0 (the "License"); you may not use this file except in compliance with the License. You may obtain a copy of the License at

http://www.apache.org/licenses/LICENSE-2.0

Unless required by applicable law or agreed to in writing, software distributed under the License is distributed on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied. See the License for the specific language governing permissions and limitations under the License.

--- LLVM Exceptions to the Apache 2.0 License ----

As an exception, if, as a result of your compiling your source code, nortions

of this Software are embedded into an Object form of such source code, you may redistribute such embedded portions in such Object form without complying

with the conditions of Sections 4(a), 4(b) and 4(d) of the License.

In addition, if you combine or link compiled forms of this Software with software that is licensed under the GPLv2 ("Combined Software") and if a court of competent jurisdiction determines that the patent provision (Section

3), the indemnity provision (Section 9) or other Section of the License conflicts with the conditions of the GPLv2, you may retroactively and prospectively choose to deem waived or otherwise exclude such Section(s) of the License, but only in their entirety and only with respect to the Combined

Software.

Software from third parties included in the LLVM Project:

The LLVM Draiget contains third party software which is under different

The LLVM Project contains third party software which is under different license

terms. All such code will be identified clearly using at least one of two mechanisms:

- 1) It will be in a separate directory tree with its own `LICENSE.txt` or `LICENSE` file at the top containing the specific license and restrictions which apply to that software, or
- It will contain specific license and restriction terms at the top of every file.

Legacy LLVM License (https://llvm.org/docs/DeveloperPolicy.html#legacy):

University of Illinois/NCSA Open Source License

Copyright (c) 2003-2019 University of Illinois at Urbana-Champaign. All rights reserved.

Developed by:

LLVM Team

University of Illinois at Urbana-Champaign

http://llvm.org

Permission is hereby granted, free of charge, to any person obtaining a copy of

this software and associated documentation files (the "Software"), to deal with

the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do

so, subject to the following conditions:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimers.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimers in the documentation and/or other materials provided with the distribution.
- * Neither the names of the LLVM Team, University of Illinois at

Urbana-Champaign, nor the names of its contributors may be used to endorse or promote products derived from this Software without specific prior written permission.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS

FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE CONTRIBUTORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS WITH THE

SOFTWARE.

Copyright (c) 1997-2019 Intel Corporation

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Intel Corporation

Software Grant License Agreement ("Agreement")

Except for the license granted herein to you, Intel Corporation ("Intel") reserves

all right, title, and interest in and to the Software (defined below).

Definition

"Software" means the code and documentation as well as any original work of authorship, including any modifications or additions to an existing work, that

is intentionally submitted by Intel to llvm.org (http://llvm.org) ("LLVM") for

inclusion in, or documentation of, any of the products owned or managed by ${\sf LLVM}$

(the "Work"). For the purposes of this definition, "submitted" means any form of

electronic, verbal, or written communication sent to LLVM or its representatives, including but not limited to communication on electronic mailing lists, source code control systems, and issue tracking systems that are

managed by, or on behalf of, LLVM for the purpose of discussing and improving

the Work, but excluding communication that is conspicuously marked otherwise.

- 1. Grant of Copyright License. Subject to the terms and conditions of this Agreement, Intel hereby grants to you and to recipients of the Software distributed by LLVM a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare derivative works of, publicly display, publicly perform, sublicense, and distribute the Software and such derivative works.
- 2. Grant of Patent License. Subject to the terms and conditions of this Agreement, Intel hereby grants you and to recipients of the Software distributed by LLVM a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable (except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by Intel that are necessarily infringed by Intel's Software alone or by combination of the Software with the Work to which such Software was submitted. If any entity institutes patent litigation against Intel or any other entity (including a cross-claim or counterclaim in a lawsuit) alleging that Intel's Software, or the Work to which Intel has contributed constitutes direct or contributory patent infringement, then any patent licenses granted to that entity under this Agreement for the Software or Work shall terminate as of the date such litigation is filed.

Unless required by applicable law or agreed to in writing, the software is provided on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE.

4. Portable Hardware Locality (hwloc)

New BSD license

Copyright © 2004-2006 The Trustees of Indiana University and Indiana University Research and Technology Corporation. All rights reserved.

Copyright © 2004-2005 The University of Tennessee and The University of Tennessee Research Foundation. All rights reserved.

Copyright © 2004-2005 High Performance Computing Center Stuttgart, University of Stuttgart. All rights reserved.

Copyright © 2004-2005 The Regents of the University of California. All rights reserved.

Copyright © 2009 CNRS

Copyright 🖟 2009-2016 Inria. All rights reserved.

Copyright 🛭 2009-2015 Universite Bordeaux

Copyright @ 2009-2015 Cisco Systems, Inc. All rights reserved.

Copyright 🖟 2009-2012 Oracle and/or its affiliates. All rights reserved.

Copyright **©** 2010 IBM

Copyright © 2010 Jirka Hladky

Copyright © 2012 Aleksej Saushev, The NetBSD Foundation

Copyright © 2012 Blue Brain Project, EPFL. All rights reserved.

Copyright @ 2013-2014 University of Wisconsin-La Crosse. All rights reserved.

Copyright 🛭 2015 Research Organization for Information Science and

Technology (RIST). All rights reserved.

Copyright © 2015-2016 Intel, Inc. All rights reserved.

See COPYING in top-level directory.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

1. Redistributions of source code must retain the above copyright

- notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. The name of the author may not be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE AUTHOR ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE AUTHOR BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

5. Safe C Library

Copyright (c) 2008-2013 by Cisco Systems, Inc Copyright (c) 2017-2018 by Reini Urban All rights reserved.

MIT License

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

6. Threading Building Blocks

Intel Simplified Software License (Version April 2018)

Copyright (c) 2018 Intel Corporation.

Use and Redistribution. You may use and redistribute the software (the "Software"), without modification, provided the following conditions are met:

* Redistributions must reproduce the above copyright notice and the following terms of use in the Software and in the documentation and/or other materials

provided with the distribution.

* Neither the name of Intel nor the names of its suppliers may be used to endorse or promote products derived from this Software without specific prior

written permission.

* No reverse engineering, decompilation, or disassembly of this Software is permitted.

Limited patent license. Intel grants you a world-wide, royalty-free, non-exclusive license under patents it now or hereafter owns or controls to make, have made, use, import, offer to sell and sell ("Utilize") this Software.

but solely to the extent that any such patent is necessary to Utilize the Software alone. The patent license shall not apply to any combinations which include this software. No hardware per se is licensed hereunder.

Third party and other Intel programs. "Third Party Programs" are the files listed in the "third-party-programs.txt" text file that is included with the Software and may include Intel programs under separate license terms. Third Party Programs, even if included with the distribution of the Materials, are governed by separate license terms and those license terms solely govern your use of those programs.

DISCLAIMER. THIS SOFTWARE IS PROVIDED "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT ARE DISCLAIMED. THIS SOFTWARE IS NOT INTENDED FOR USE IN SYSTEMS OR APPLICATIONS WHERE FAILURE OF THE SOFTWARE MAY CAUSE PERSONAL INJURY OR DEATH AND YOU AGREE

THAT YOU ARE FULLY RESPONSIBLE FOR ANY CLAIMS, COSTS, DAMAGES, EXPENSES, AND ATTORNEYS' FEES ARISING OUT OF ANY SUCH USE, EVEN IF ANY CLAIM ALLEGES THAT INTEL WAS NEGLIGENT REGARDING THE DESIGN OR MANUFACTURE OF THE MATERIALS.

LIMITATION OF LIABILITY. IN NO EVENT WILL INTEL BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING,

BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF

LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE

OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE. YOU AGREE TO INDEMNIFY AND HOLD TNTFI

HARMLESS AGAINST ANY CLAIMS AND EXPENSES RESULTING FROM YOUR USE OR UNAUTHORIZED

USE OF THE SOFTWARE.

No support. Intel may make changes to the Software, at any time without notice,

and is not obligated to support, update or provide training for the Software.

Termination. Intel may terminate your right to use the Software in the event of

your breach of this Agreement and you fail to cure the breach within a reasonable period of time.

This file is the "third-party-programs.txt" for Intel Open Source Technology Center Safe String Library. This library is based from the Cisco CSafelib. The copyright and license terms are listed below.

MIT License

Copyright (c) 2014-2018 Intel Corporation

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Copyright (C) 2012, 2013 Cisco Systems All rights reserved.

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

JSON-Fortran: A Fortran 2008 JSON API https://github.com/jacobwilliams/json-fortran

Copyright (c) 2014-2019, Jacob Williams All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * The names of its contributors may not be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

>

>
 Original FSON License:

> Copyright (c) 2012 Joseph A. Levin

> Permission is hereby granted, free of charge, to any person obtaining a copy of this

- > software and associated documentation files (the "Software"), to deal in the Software
- > without restriction, including without limitation the rights to use, copy, modify, merge,
- > publish, distribute, sublicense, and/or sell copies of the Software, and to permit
- > persons to whom the Software is furnished to do so, subject to the following conditions:
- > The above copyright notice and this permission notice shall be included in all copies or
- > substantial portions of the Software.
- > THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED,
- > INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR
- > PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE
- > LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT
- > OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER
- > DEALINGS IN THE SOFTWARE.

> > Copyright 2024 Mathis Rosenhauer, Moritz Hanke, Joerg Behrens, Luis Kornblueh All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS `AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

BSD-2-Clause Copyright (c) <year> <owner>.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Copyright & License Notice

Copyright 1995-2013, Regents of the University of Minnesota

Licensed under the Apache License, Version 2.0 (the "License"); you may not use this file except in compliance with the License. You may obtain a copy of the License at

http://www.apache.org/licenses/LICENSE-2.0

Unless required by applicable law or agreed to in writing, software distributed under the License is distributed on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied. See the License for the specific language governing permissions and limitations under the License.

Apache License Version 2.0, January 2004 http://www.apache.org/licenses/

TERMS AND CONDITIONS FOR USE, REPRODUCTION, AND DISTRIBUTION

1. Definitions.

"License" shall mean the terms and conditions for use, reproduction, and distribution as defined by Sections 1 through 9 of this document.

"Licensor" shall mean the copyright owner or entity authorized by the copyright owner that is granting the License.

"Legal Entity" shall mean the union of the acting entity and all other entities that control, are controlled by, or are under common control with that entity. For the purposes of this definition, "control" means (i) the power, direct or indirect, to cause the direction or management of such entity, whether by contract or otherwise, or (ii) ownership of fifty percent (50%) or more of the outstanding shares, or (iii) beneficial ownership of such entity.

"You" (or "Your") shall mean an individual or Legal Entity exercising permissions granted by this License.

"Source" form shall mean the preferred form for making modifications, including but not limited to software source code, documentation source, and configuration files.

"Object" form shall mean any form resulting from mechanical transformation or translation of a Source form, including but not limited to compiled object code, generated documentation, and conversions to other media types.

"Work" shall mean the work of authorship, whether in Source or Object form, made available under the License, as indicated by a copyright notice that is included in or attached to the work (an example is provided in the Appendix below).

"Derivative Works" shall mean any work, whether in Source or Object form, that is based on (or derived from) the Work and for which the editorial revisions, annotations, elaborations, or other modifications represent, as a whole, an original work of authorship. For the purposes of this License, Derivative Works shall not include works that remain separable from, or merely link (or bind by name) to the interfaces of, the Work and Derivative Works thereof.

"Contribution" shall mean any work of authorship, including the original version of the Work and any modifications or additions to that Work or Derivative Works thereof, that is intentionally submitted to Licensor for inclusion in the Work by the copyright owner or by an individual or Legal Entity authorized to submit on behalf of the copyright owner. For the purposes of this definition, "submitted" means any form of electronic, verbal, or written communication sent to the Licensor or its representatives, including but not limited to communication on electronic mailing lists, source code control systems, and issue tracking systems that are managed by, or on behalf of, the Licensor for the purpose of discussing and improving the Work, but excluding communication that is conspicuously marked or otherwise designated in writing by the copyright owner as "Not a Contribution."

"Contributor" shall mean Licensor and any individual or Legal Entity on behalf of whom a Contribution has been received by Licensor and subsequently incorporated within the Work.

- 2. Grant of Copyright License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare Derivative Works of, publicly display, publicly perform, sublicense, and distribute the Work and such Derivative Works in Source or Object form.
- 3. Grant of Patent License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable (except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by such Contributor that are necessarily infringed by their Contribution(s) alone or by combination of their Contribution(s) with the Work to which such Contribution(s) was submitted. If You institute patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Work or a Contribution incorporated within the Work constitutes direct or contributory patent infringement, then any patent licenses granted to You under this License for that Work shall terminate as of the date such litigation is filed.
- 4. Redistribution. You may reproduce and distribute copies of the Work or Derivative Works thereof in any medium, with or without modifications, and in Source or Object form, provided that You meet the following conditions:
 - (a) You must give any other recipients of the Work or Derivative Works a copy of this License; and
 - (b) You must cause any modified files to carry prominent notices stating that You changed the files; and
 - (c) You must retain, in the Source form of any Derivative Works that You distribute, all copyright, patent, trademark, and attribution notices from the Source form of the Work, excluding those notices that do not pertain to any part of the Derivative Works; and
 - (d) If the Work includes a "NOTICE" text file as part of its distribution, then any Derivative Works that You distribute must include a readable copy of the attribution notices contained within such NOTICE file, excluding those notices that do not pertain to any part of the Derivative Works, in at least one of the following places: within a NOTICE text file distributed as part of the Derivative Works; within the Source form or documentation, if provided along with the Derivative Works; or, within a display generated by the Derivative Works, if and wherever such third-party notices normally appear. The contents of the NOTICE file are for informational purposes only and do not modify the License. You may add Your own attribution notices within Derivative Works that You distribute, alongside or as an addendum to the NOTICE text from the Work, provided that such additional attribution notices cannot be construed as modifying the License.

You may add Your own copyright statement to Your modifications and may provide additional or different license terms and conditions for use, reproduction, or distribution of Your modifications, or for any such Derivative Works as a whole, provided Your use, reproduction, and distribution of the Work otherwise complies with

the conditions stated in this License.

- 5. Submission of Contributions. Unless You explicitly state otherwise, any Contribution intentionally submitted for inclusion in the Work by You to the Licensor shall be under the terms and conditions of this License, without any additional terms or conditions. Notwithstanding the above, nothing herein shall supersede or modify the terms of any separate license agreement you may have executed with Licensor regarding such Contributions.
- 6. Trademarks. This License does not grant permission to use the trade names, trademarks, service marks, or product names of the Licensor, except as required for reasonable and customary use in describing the origin of the Work and reproducing the content of the NOTICE file.
- 7. Disclaimer of Warranty. Unless required by applicable law or agreed to in writing, Licensor provides the Work (and each Contributor provides its Contributions) on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE. You are solely responsible for determining the appropriateness of using or redistributing the Work and assume any risks associated with Your exercise of permissions under this License.
- 8. Limitation of Liability. In no event and under no legal theory, whether in tort (including negligence), contract, or otherwise, unless required by applicable law (such as deliberate and grossly negligent acts) or agreed to in writing, shall any Contributor be liable to You for damages, including any direct, indirect, special, incidental, or consequential damages of any character arising as a result of this License or out of the use or inability to use the Work (including but not limited to damages for loss of goodwill, work stoppage, computer failure or malfunction, or any and all other commercial damages or losses), even if such Contributor has been advised of the possibility of such damages.
- 9. Accepting Warranty or Additional Liability. While redistributing the Work or Derivative Works thereof, You may choose to offer, and charge a fee for, acceptance of support, warranty, indemnity, or other liability obligations and/or rights consistent with this License. However, in accepting such obligations, You may act only on Your own behalf and on Your sole responsibility, not on behalf of any other Contributor, and only if You agree to indemnify, defend, and hold each Contributor harmless for any liability incurred by, or claims asserted against, such Contributor by reason of your accepting any such warranty or additional liability.

END OF TERMS AND CONDITIONS

APPENDIX: How to apply the Apache License to your work.

To apply the Apache License to your work, attach the following boilerplate notice, with the fields enclosed by brackets "[]" replaced with your own identifying information. (Don't include the brackets!) The text should be enclosed in the appropriate comment syntax for the file format. We also recommend that a file or class name and description of purpose be included on the same "printed page" as the copyright notice for easier identification within third-party archives.

Copyright [yyyy] [name of copyright owner]

Licensed under the Apache License, Version 2.0 (the "License");

you may not use this file except in compliance with the License. You may obtain a copy of the License at

http://www.apache.org/licenses/LICENSE-2.0

Unless required by applicable law or agreed to in writing, software distributed under the License is distributed on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied. See the License for the specific language governing permissions and limitations under the License.

MICROSOFT SOFTWARE LICENSE TERMS

MICROSOFT VISUAL STUDIO ENTERPRISE 2019, VISUAL STUDIO PROFESSIONAL 2019, VISUAL STUDIO TEST PROFESSIONAL 2019 AND TRIAL EDITION

These license terms are an agreement between you and Microsoft Corporation (or based on where you live, one of its affiliates). They apply to the software named above. The terms also apply to any Microsoft services and updates for the software, except to the extent those have different terms.

BY USING THE SOFTWARE, YOU ACCEPT THESE TERMS. IF YOU DO NOT ACCEPT THEM, DO NOT USE THE SOFTWARE. INSTEAD, RETURN IT TO THE RETAILER FOR A REFUND OR CREDIT. If you cannot obtain a refund there, contact Microsoft about Microsoft's refund policies. See www.microsoft.com/worldwide. In the United States and Canada, call (800) MICROSOFT or see aka.ms/nareturns.

TRIAL EDITION USE RIGHTS. If you have not acquired a valid full-use license, the software is a trial edition, and this Section applies to your use of the trial edition.

- **A. GENERAL.** You may use any number of copies of the trial edition on your devices. You may only use the trial edition for internal evaluation purposes, and only during the trial period. You may not distribute or deploy any applications you make with the trial edition to a production environment. You may run load tests of up to 250 virtual users during the trial period.
- **B. TRIAL PERIOD AND CONVERSION.** The trial period lasts for 30 days after you install the trial edition, plus any permitted extension period. After the expiration of the trial period, the trial edition will stop running. You may extend the trial period an additional 90 days if you sign in to the software. You may not be able to access data used with the trial edition after it stops running. You may convert your trial rights at any time to the full-use rights described below by acquiring a valid full-use license.
- C. DISCLAIMER OF WARRANTY. THE TRIAL EDITION IS LICENSED "AS-IS". YOU BEAR THE RISK OF USING IT. MICROSOFT GIVES NO EXPRESS WARRANTIES, GUARANTEES OR CONDITIONS. TO THE EXTENT PERMITTED UNDER YOUR LOCAL LAWS, MICROSOFT EXCLUDES THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NON-INFRINGEMENT.
 - FOR AUSTRALIA YOU HAVE STATUTORY GUARANTEES UNDER THE AUSTRALIAN CONSUMER LAW AND NOTHING IN THESE TERMS IS INTENDED TO AFFECT THOSE RIGHTS.
- **D. SUPPORT.** Because the trial edition is "as is", we may not provide support services for it.
- E. LIMITATIONS ON DAMAGES. YOU CAN RECOVER FROM MICROSOFT AND ITS SUPPLIERS ONLY DIRECT DAMAGES UP TO U.S. \$5.00. YOU CANNOT RECOVER ANY OTHER DAMAGES, INCLUDING CONSEQUENTIAL, LOST PROFITS, SPECIAL, INDIRECT OR INCIDENTAL DAMAGES.

This limitation applies to (a) anything related to the trial version, services, content (including code) on third party Internet sites, or third party programs; and (b) claims for breach of contract, breach of warranty, guarantee or condition, strict liability, negligence, or other tort to the extent permitted by applicable law.

It also applies even if Microsoft knew or should have known about the possibility of the damages. The above limitation or exclusion may not apply to you because your country may not allow the exclusion or limitation of incidental, consequential or other damages.

FULL-USE LICENSE TERMS FOR THE SOFTWARE: When you acquire a valid license and either enter a product key or sign in to the software, the terms below apply. You may not share your product key or access credentials.

1. OVERVIEW.

- **a. Software.** The software includes development tools, applications, and documentation.
- **b. License Model.** The software is licensed on a per user basis.

2. USE RIGHTS.

- **a. General.** One user may use copies of the software on your devices to develop and test applications. This includes using copies of the software on your own internal servers that remain fully dedicated to your own use. You may not, however, separate the components of the software (except as otherwise stated in this agreement) and run those in a production environment, or on third party devices, or for any purpose other than developing and testing your applications. Running the software on Microsoft Azure may require separate online usage fees.
- **b. Workloads.** These license terms apply to your use of the workloads made available to you within the software, except to the extent a workload or a workload component comes with different license terms and support policies.
- c. Backup copy. You may make one backup copy of the software, for reinstalling the software.
- **d. Online Services in the Software.** Some features of the software make use of online services to provide you with information about updates to the software or extensions, or to enable you to retrieve content, collaborate with others, or otherwise supplement your development experience. As used throughout this agreement, the term "software" includes these online service features.
- **e. Demo Use**. The use rights permitted above include using the software to demonstrate your applications.

3. TERMS FOR SPECIFIC COMPONENTS.

- a. Utilities. The software contains items on the Utilities List at https://aka.ms/vs/16/utilities. You may copy and install those items onto your devices to debug and deploy your applications and databases you developed with the software. The Utilities are designed for temporary use. Microsoft may not be able to patch or update Utilities separately from the rest of the software. Some Utilities by their nature may make it possible for others to access the devices on which the Utilities are installed. You should delete all Utilities you have installed after you finish debugging or deploying your applications and databases. Microsoft is not responsible for any third party use or access of devices, or of the applications or databases on devices, on which Utilities have been installed.
- **b. Build Devices and Visual Studio Build Tools.** You may copy and install files from the software or from Visual Studio Build Tools onto your build devices, including physical devices and virtual machines or containers on those machines, whether on-premises or remote machines that are owned by you, hosted on Microsoft Azure for you, or dedicated solely to your use (collectively, "Build Devices"). You and others in your organization may use these files on your Build Devices solely to compile, build, and verify applications developed by using the software, or run quality or performance tests of those applications as part of the build process.
- **c. Font Components.** While the software is running, you may use its fonts to display and print content. You may only: (i) embed fonts in content as permitted by the embedding restrictions in the fonts; and (ii) temporarily download them to a printer or other output device to help print content.

d. Licenses for Other Components.

Microsoft Platforms. The software may include components from Microsoft Windows,
Microsoft Windows Server, Microsoft SQL Server, Microsoft Exchange, Microsoft Office, or
Microsoft SharePoint. These components are governed by separate agreements and their own
product support policies, as described in the Microsoft "Licenses" folder accompanying the
software, except that, if separate license terms for those components are included in the
associated installation directly, those license terms control.

- **Third Party Components.** The software may include third party components with separate legal notices or governed by other agreements, as may be described in the ThirdPartyNotices file(s) accompanying the software.
- **e. Package Managers.** The software includes package managers, like NuGet, that give you the option to download other Microsoft and third party software packages to use with your applications. Those packages are under their own licenses, and not these license terms. Microsoft does not distribute, license or provide any warranties for any of the third party packages.
- **4. DISTRIBUTABLE CODE.** The software contains code that you are permitted to distribute in applications you develop as described in this Section. For purposes of this Section 4, the term "distribution" also means deployment of your applications for third parties to access over the Internet.
 - a. Right to Use and Distribute. The code and text files listed below are "Distributable Code".
 - **Distributable List.** You may copy and distribute the object code form of code listed on the Distributable List located at https://aka.ms/vs/16/redistribution.
 - **Sample Code, Templates, and Styles.** You may copy, modify, and distribute the source and object code form of code marked as "sample", "template", "simple styles", and "sketch styles".
 - **Third Party_Distribution.** You may permit distributors of your applications to copy and distribute the Distributable Code as part of those applications.
 - b. Distribution Requirements. For any Distributable Code you distribute, you must:
 - add significant primary functionality to it in your applications;
 - require distributors and external end users to agree to terms that protect the Distributable
 Code at least as much as this agreement; and
 - indemnify, defend, and hold harmless Microsoft from any claims, including attorneys' fees, related to the distribution or use of your applications, except to the extent that any claim is based solely on the Distributable Code.
 - c. Distribution Restrictions. You may not:
 - use Microsoft's trademarks in your applications' names or in a way that suggests your applications come from or are endorsed by Microsoft; or
 - modify or distribute the source code of any Distributable Code so that any part of it becomes subject to an Excluded License. An Excluded License is one that requires, as a condition of use, modification or distribution of code, that (i) it be disclosed or distributed in source code form; or (ii) others have the right to modify it.

5. DEVELOPING EXTENSIONS.

- **a. Limits on Extensions.** You may not develop or enable others to develop extensions for the software (or any other component of the Visual Studio family of products) which circumvent the technical limitations implemented in the software. If Microsoft technically limits or disables extensibility for the software, you may not extend the software by, among other things, loading or injecting into the software any non-Microsoft add-ins, macros, or packages; modifying the software registry settings; or adding features or functionality equivalent to that found in the Visual Studio family of products.
- **b. No Degrading the Software**. If you develop an extension for the software (or any other component of the Visual Studio family of products), you must test the installation, uninstallation, and operation of your extension to ensure that such processes do not disable any features or adversely affect the functionality of the software (or such component) or of any previous version or edition of thereof.

6. DATA.

a. Data Collection. The software may collect information about you and your use of the software, and send that to Microsoft. Microsoft may use this information to provide services and improve our products and services. You may opt out of many of these scenarios, but not all, as described in the software documentation. There are also some features in the software that may enable

you and Microsoft to collect data from users of your applications. If you use these features, you must comply with applicable law, including providing appropriate notices to users of your applications together with Microsoft's privacy statement. Our privacy statement is located at https://go.microsoft.com/fwlink/?LinkID=824704. You can learn more about data collection and its use from the software documentation and our privacy statement. Your use of the software operates as your consent to these practices.

- **b. Processing of Personal Data.** To the extent Microsoft is a processor or subprocessor of personal data in connection with the software, Microsoft makes the commitments in the European Union General Data Protection Regulation Terms of the Online Services Terms to all customers effective May 25, 2018, at https://docs.microsoft.com/en-us/legal/gdpr.
- 7. SCOPE OF LICENSE. The software is licensed, not sold. These license terms only give you some rights to use the software. Microsoft reserves all other rights. Unless applicable law gives you more rights despite this limitation, you may use the software only as expressly permitted in these license terms. In doing so, you must comply with any technical limitations in the software that only allow you to use it in certain ways. In addition, you may not:
 - · work around any technical limitations in the software;
 - reverse engineer, decompile or disassemble the software, or otherwise attempt to derive the source code for the software, except and to the extent required by third party licensing terms governing use of certain open source components that may be included in the software;
 - remove, minimize, block, or modify any notices of Microsoft or its suppliers in the software;
 - use the software in any way that is against the law;
 - · share, publish, rent, or lease the software; or
 - provide the software as a stand-alone offering or combine it with any of your applications for others to use.
- **8. NOT FOR RESALE SOFTWARE.** You may not sell the software if it is marked as "NFR" or "Not for Resale".
- **9. PREVIOUS VERSIONS OR OTHER EDITIONS.** These license terms do not supersede your right to use validly licensed previous versions or other editions of the software. You may use the software and those previous versions or other editions of the software concurrently.
- **10.PROOF OF LICENSE.** If you acquired the software on a disc or other media, your proof of license is the Microsoft certificate of authenticity label, the accompanying product key, and your receipt. If you purchased an online copy of the software, your proof of license is the Microsoft product key you received with your purchase and your receipt and/or being able to access the software service through your Microsoft account. To identify genuine Microsoft software, see www.howtotell.com.
- **11.TRANSFER TO A THIRD PARTY.** If you are a valid licensee of the software, you may transfer it and this agreement directly to another party. Before the transfer, that party must agree that these license terms apply to the transfer and use of the software. The transfer must include the software, this agreement, the genuine Microsoft product key, and (if applicable) the Proof of License label. The transferor must uninstall all copies of the software after transferring it from the device. The transferor may not retain any copies of the genuine Microsoft product key to be transferred, and may only retain copies of the software if otherwise licensed to do so. If you have acquired a non-perpetual license to use the software or if the software is marked Not for Resale you may not transfer the software or the software license agreement to another party.
- **12.EXPORT RESTRICTIONS.** You must comply with all domestic and international export laws and regulations that apply to the software, which include restrictions on destinations, end users, and end use. For further information on export restrictions, visit www.microsoft.com/exporting.
- **13.SUPPORT.** Microsoft provides support for the software as described at https://support.microsoft.com.
- **14.ENTIRE AGREEMENT.** These license terms (including the warranty below), and the terms for supplements, updates, Internet-based services, and support services, are the entire agreement for the software and support services.
- 15.APPLICABLE LAW. If you acquired the software in the United States, Washington State law applies

to interpretation of and claims for breach of this agreement, and the laws of the state where you live apply to all other claims. If you acquire the software in any other country, its laws apply.

- **16.CONSUMER RIGHTS; REGIONAL VARIATIONS.** These license terms describe certain legal rights. You may have other rights, including consumer rights, under the laws of your state or country. You may also have rights with respect to the party from which you acquired the software. This agreement does not change those other rights if the laws of your state or country do not permit it to do so. For example, if you acquired the software in one of the below regions, or if mandatory country law applies, then the following provisions apply to you:
 - **a) Australia.** References to "Limited Warranty" are references to the express warranty provided by Microsoft. This warranty is given in addition to other rights and remedies you may have under law, including your rights and remedies in accordance with the statutory guarantees in the Australian Consumer Law.

In this section, "goods" refers to the software for which Microsoft provides the express warranty. Our goods come with guarantees that cannot be excluded under the Australian Consumer Law. You are entitled to a replacement or refund for a major failure and compensation for any other reasonably foreseeable loss or damage. You are also entitled to have the goods repaired or replaced if the goods fail to be of acceptable quality and the failure does not amount to a major failure.

- **b) Canada.** You may stop receiving updates on your device by turning off Internet access. If and when you re-connect to the Internet, the software will resume checking for and installing updates.
- c) Germany and Austria.
 - **(i) Warranty**. The properly licensed software will perform substantially as described in any Microsoft materials that accompany the software. However, Microsoft gives no contractual guarantee in relation to the software.
 - (ii) Limitation of Liability. In case of intentional conduct, gross negligence, claims based on the Product Liability Act, as well as, in the case of death or personal or physical injury, Microsoft is liable according to the statutory law.

Subject to the preceding sentence (ii), Microsoft will only be liable for slight negligence if Microsoft is in breach of such material contractual obligations, the fulfillment of which facilitate the due performance of this agreement, the breach of which would endanger the purpose of this agreement and the compliance with which a party may constantly trust in (so-called "cardinal obligations"). In other cases of slight negligence, Microsoft will not be liable for slight negligence.

LIMITED WARRANTY

Microsoft warrants that properly licensed software will perform substantially as described in any Microsoft materials that accompany the software. This limited warranty does not cover problems that you cause, that arise when you fail to follow instructions, or that are caused by events beyond Microsoft's reasonable control. The limited warranty starts when the first user acquires the software, and lasts for one year. Any supplements, updates, or replacement software that you may receive from Microsoft during that year are also covered, but only for the remainder of that one-year period or for 30 days, whichever is longer. Transferring the software will not extend the limited warranty.

Microsoft gives no other express warranties, guarantees, or conditions. Microsoft excludes all implied warranties and conditions, including those of merchantability, fitness for a particular purpose, and non-infringement. If your local law does not allow the exclusion of implied warranties, then any implied warranties, guarantees, or conditions last only during the term of the limited warranty and are limited as much as your local law allows. If your local law requires a longer limited warranty term, despite this agreement, then that longer term will apply, but you can recover only the remedies this agreement allows.

If Microsoft breaches its limited warranty, it will, at its election, either: (i) repair or replace the software at no charge, or (ii) accept return of the software (or at its election the Microsoft branded device on which the software was preinstalled) for a refund of the amount paid, if any. **These are your only remedies for breach of warranty.** This limited warranty gives you specific legal rights, and you may also have other rights which vary from state to state or country to country.

Except for any repair, replacement, or refund Microsoft may provide, you may not recover under this limited warranty, under any other part of this agreement, or under any theory, any damages or other remedy, including lost profits or direct, consequential, special, indirect, or incidental damages. The damage exclusions and remedy limitations in this agreement apply even if repair, replacement or a refund does not fully compensate you for any losses, if Microsoft knew or should have known about the possibility of the damages, or if the remedy fails of its essential purpose. Some states and countries do not allow the exclusion or limitation of incidental, consequential, or other damages, so those limitations or exclusions may not apply to you. If your local law allows you to recover damages from Microsoft even though this agreement does not, you cannot recover more than you paid for the software (or up to \$50 USD if you acquired the software for no charge).

Warranty Procedures

For service or a refund, you must provide a copy of your proof of purchase and comply with Microsoft's return policies, which might require you to uninstall the software and return it to Microsoft or return the software with the entire Microsoft branded device on which the software is installed; the certificate of authenticity label including the product key (if provided with your device) must remain affixed.

- 1. <u>United States and Canada</u>. For limited warranty service or information about how to obtain a refund for software acquired in the United States or Canada, contact Microsoft via telephone at (800) MICROSOFT; via mail at Microsoft Customer Service and Support, One Microsoft Way, Redmond, WA 98052-6399; or visit (aka.ms/nareturns).
- 2. <u>Europe, Middle East, and Africa</u>. If you acquired the software in Europe, the Middle East, or Africa, Microsoft Ireland Operations Limited makes the limited warranty. To make a claim under the limited warranty, you must contact either Microsoft Ireland Operations Limited, Customer Care Centre, Atrium Building Block B, Carmanhall Road, Sandyford Industrial Estate, Dublin 18, Ireland, or the Microsoft affiliate serving your country (aka.ms/msoffices).
- 3. <u>Australia</u>. If you acquired the software in Australia, contact Microsoft to make a claim at 13 20 58; or Microsoft Pty Ltd, 1 Epping Road, North Ryde NSW 2113 Australia.
- 4. <u>Other countries</u>. If you acquired the software in another country, contact the Microsoft affiliate serving your country (aka.ms/msoffices).

EULAID: VS 2019 ENU.1033

Minpack Copyright Notice (1999) University of Chicago. All rights reserved

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. The end-user documentation included with the redistribution, if any, must include the following acknowledgment:

"This product includes software developed by the University of Chicago, as Operator of Argonne National Laboratory.

Alternately, this acknowledgment may appear in the software itself, if and wherever such third-party acknowledgments normally appear.

- 4. WARRANTY DISCLAIMER. THE SOFTWARE IS SUPPLIED "AS IS" WITHOUT WARRANTY OF ANY KIND. THE COPYRIGHT HOLDER, THE UNITED STATES, THE UNITED STATES DEPARTMENT OF ENERGY, AND THEIR EMPLOYEES: (1) DISCLAIM ANY WARRANTIES, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE OR NON-INFRINGEMENT, (2) DO NOT ASSUME ANY LEGAL LIABILITY OR RESPONSIBILITY FOR THE ACCURACY, COMPLETENESS, OR USEFULNESS OF THE SOFTWARE, (3) DO NOT REPRESENT THAT USE OF THE SOFTWARE WOULD NOT INFRINGE PRIVATELY OWNED RIGHTS, (4) DO NOT WARRANT THAT THE SOFTWARE WILL FUNCTION UNINTERRUPTED, THAT IT IS ERROR-FREE OR THAT ANY ERRORS WILL BE CORRECTED.
- 5. LIMITATION OF LIABILITY. IN NO EVENT WILL THE COPYRIGHT HOLDER, THE UNITED STATES, THE UNITED STATES DEPARTMENT OF ENERGY, OR THEIR EMPLOYEES: BE LIABLE FOR ANY INDIRECT, INCIDENTAL, CONSEQUENTIAL, SPECIAL OR PUNITIVE DAMAGES OF ANY KIND OR NATURE, INCLUDING BUT NOT LIMITED TO LOSS OF PROFITS OR LOSS OF DATA, FOR ANY REASON WHATSOEVER, WHETHER SUCH LIABILITY IS ASSERTED ON THE BASIS OF CONTRACT, TORT (INCLUDING NEGLIGENCE OR STRICT LIABILITY), OR OTHERWISE, EVEN IF ANY OF SAID PARTIES HAS BEEN WARNED OF THE POSSIBILITY OF SUCH LOSS OR DAMAGES.

BSD 2-Clause License

Copyright (c) 2019-2020, Volker Jacht All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Copyright 1991-2022 CERFACS, CNRS, ENS Lyon, INP Toulouse, Inria, Mumps Technologies, University of Bordeaux.

This version of MUMPS is provided to you free of charge. It is released under the CeCILL-C license (see doc/CeCILL-C_V1-en.txt, doc/CeCILL-C_V1-fr.txt, and https://cecill.info/licences/Licence_CeCILL-C_V1-en.html), except for variants of AMD ordering and xMUMPS_TRUNCATED_RRQR derived from the LAPACK package distributed under BSD 3-clause license (see headers of ana_orderings.F and lr_core.F), and except for the external and optional ordering PORD provided in a separate directory PORD (see PORD/README for License information).

You can acknowledge (using references [1] and [2]) the contribution of this package in any scientific publication dependent upon the use of the package. Please use reasonable endeavours to notify the authors of the package of this publication.

- [1] P. R. Amestoy, I. S. Duff, J. Koster and J.-Y. L'Excellent, A fully asynchronous multifrontal solver using distributed dynamic scheduling, SIAM Journal on Matrix Analysis and Applications, Vol 23, No 1, pp 15-41 (2001).
- [2] P. R. Amestoy, A. Buttari, J.-Y. L'Excellent and T. Mary, Performance and scalability of the block low-rank multifrontal factorization on multicore architectures, ACM Transactions on Mathematical Software, Vol 45, Issue 1, pp 2:1-2:26 (2019)

As a counterpart to the access to the source code and rights to copy, modify and redistribute granted by the license, users are provided only with a limited warranty and the software's author, the holder of the economic rights, and the successive licensors have only limited liability.

In this respect, the user's attention is drawn to the risks associated with loading, using, modifying and/or developing or reproducing the software by the user in light of its specific status of free software, that may mean that it is complicated to manipulate, and that also therefore means that it is reserved for developers and experienced professionals having in-depth computer knowledge. Users are therefore encouraged to load and test the software's suitability as regards their requirements in conditions enabling the security of their systems and/or data to be ensured and, more generally, to use and operate it in the same conditions as regards security.

The fact that you are presently reading this means that you have had knowledge of the CeCILL-C license and that you accept its terms.

CeCILL-C FREE SOFTWARE LICENSE AGREEMENT

Notice

This Agreement is a Free Software license agreement that is the result of discussions between its authors in order to ensure compliance with the two main principles guiding its drafting:

- * firstly, compliance with the principles governing the distribution of Free Software: access to source code, broad rights granted to users,
- * secondly, the election of a governing law, French law, with which it is conformant, both as regards the law of torts and intellectual property law, and the protection that it offers to both authors and holders of the economic rights over software.

The authors of the CeCILL-C (for Ce[a] C[nrs] I[nria] L[ogiciel] L[ibre]) license are:

Commissariat à l'Energie Atomique - CEA, a public scientific, technical and industrial research establishment, having its principal place of business at 25 rue Leblanc, immeuble Le Ponant D, 75015 Paris, France.

Centre National de la Recherche Scientifique - CNRS, a public scientific and technological establishment, having its principal place of business at 3 rue Michel-Ange, 75794 Paris cedex 16, France.

Institut National de Recherche en Informatique et en Automatique - INRIA, a public scientific and technological establishment, having its principal place of business at Domaine de Voluceau, Rocquencourt, BP 105, 78153 Le Chesnay cedex, France.

Preamble

The purpose of this Free Software license agreement is to grant users the right to modify and re-use the software governed by this license.

The exercising of this right is conditional upon the obligation to make available to the community the modifications made to the source code of the software so as to contribute to its evolution.

In consideration of access to the source code and the rights to copy, modify and redistribute granted by the license, users are provided only with a limited warranty and the software's author, the holder of the economic rights, and the successive licensors only have limited liability.

In this respect, the risks associated with loading, using, modifying and/or developing or reproducing the software by the user are brought to the user's attention, given its Free Software status, which may make it complicated to use, with the result that its use is reserved for developers and experienced professionals having in-depth computer knowledge. Users are therefore encouraged to load and test the suitability of the software as regards their requirements in conditions enabling the security of their systems and/or data to be ensured and, more generally, to use and operate it in the same conditions of security. This Agreement may be freely reproduced and published, provided it is not altered, and that no provisions are either added or removed herefrom.

This Agreement may apply to any or all software for which the holder of the economic rights decides to submit the use thereof to its provisions.

Article 1 - DEFINITIONS

For the purpose of this Agreement, when the following expressions commence with a capital letter, they shall have the following meaning:

Agreement: means this license agreement, and its possible subsequent versions and annexes.

Software: means the software in its Object Code and/or Source Code form and, where applicable, its documentation, "as is" when the Licensee accepts the Agreement.

Initial Software: means the Software in its Source Code and possibly its Object Code form and, where applicable, its documentation, "as is" when it is first distributed under the terms and conditions of the Agreement.

Modified Software: means the Software modified by at least one Integrated Contribution.

Source Code: means all the Software's instructions and program lines to which access is required so as to modify the Software.

Object Code: means the binary files originating from the compilation of the Source Code.

Holder: means the holder(s) of the economic rights over the Initial Software.

Licensee: means the Software user(s) having accepted the Agreement.

Contributor: means a Licensee having made at least one Integrated Contribution.

Licensor: means the Holder, or any other individual or legal entity, who distributes the Software under the Agreement.

Integrated Contribution: means any or all modifications, corrections, translations, adaptations and/or new functions integrated into the Source Code by any or all Contributors.

Related Module: means a set of sources files including their documentation that, without modification to the Source Code, enables supplementary functions or services in addition to those offered by the Software.

Derivative Software: means any combination of the Software, modified or not, and of a Related Module.

Parties: mean both the Licensee and the Licensor.

These expressions may be used both in singular and plural form.

Article 2 - PURPOSE

The purpose of the Agreement is the grant by the Licensor to the Licensee of a non-exclusive, transferable and worldwide license for the Software as set forth in Article 5 hereinafter for the whole term of the protection granted by the rights over said Software.

Article 3 - ACCEPTANCE

- 3.1 The Licensee shall be deemed as having accepted the terms and conditions of this Agreement upon the occurrence of the first of the following events:
 - * (i) loading the Software by any or all means, notably, by downloading from a remote server, or by loading from a physical medium;
 - * (ii) the first time the Licensee exercises any of the rights granted hereunder.
- 3.2 One copy of the Agreement, containing a notice relating to the characteristics of the Software, to the limited warranty, and to the fact that its use is restricted to experienced users has been provided to the Licensee prior to its acceptance as set forth in Article 3.1 hereinabove, and the Licensee hereby acknowledges that it has read and understood it.

Article 4 - EFFECTIVE DATE AND TERM

4.1 EFFECTIVE DATE

The Agreement shall become effective on the date when it is accepted by the Licensee as set forth in Article 3.1.

4.2 TERM

The Agreement shall remain in force for the entire legal term of protection of the economic rights over the Software.

Article 5 - SCOPE OF RIGHTS GRANTED

The Licensor hereby grants to the Licensee, who accepts, the following rights over the Software for any or all use, and for the term of the Agreement, on the basis of the terms and conditions set forth hereinafter.

Besides, if the Licensor owns or comes to own one or more patents protecting all or part of the functions of the Software or of its components, the Licensor undertakes not to enforce the rights granted by these patents against successive Licensees using, exploiting or modifying the Software. If these patents are transferred, the Licensor undertakes to have the transferees subscribe to the obligations set forth in this paragraph.

5.1 RIGHT OF USE

The Licensee is authorized to use the Software, without any limitation as to its fields of application, with it being hereinafter specified that this comprises:

- 1. permanent or temporary reproduction of all or part of the Software by any or all means and in any or all form.
- 2. loading, displaying, running, or storing the Software on any or all medium.
- 3. entitlement to observe, study or test its operation so as to determine the ideas and principles behind any or all constituent elements of said Software. This shall apply when the Licensee

carries out any or all loading, displaying, running, transmission or storage operation as regards the Software, that it is entitled to carry out hereunder.

5.2 RIGHT OF MODIFICATION

The right of modification includes the right to translate, adapt, arrange, or make any or all modifications to the Software, and the right to reproduce the resulting software. It includes, in particular, the right to create a Derivative Software.

The Licensee is authorized to make any or all modification to the Software provided that it includes an explicit notice that it is the author of said modification and indicates the date of the creation thereof.

5.3 RIGHT OF DISTRIBUTION

In particular, the right of distribution includes the right to publish, transmit and communicate the Software to the general public on any or all medium, and by any or all means, and the right to market, either in consideration of a fee, or free of charge, one or more copies of the Software by any means.

The Licensee is further authorized to distribute copies of the modified or unmodified Software to third parties according to the terms and conditions set forth hereinafter.

5.3.1 DISTRIBUTION OF SOFTWARE WITHOUT MODIFICATION

The Licensee is authorized to distribute true copies of the Software in Source Code or Object Code form, provided that said distribution complies with all the provisions of the Agreement and is accompanied by:

- 1. a copy of the Agreement,
- a notice relating to the limitation of both the Licensor's warranty and liability as set forth in Articles 8 and 9,

and that, in the event that only the Object Code of the Software is redistributed, the Licensee allows effective access to the full Source Code of the Software at a minimum during the entire period of its distribution of the Software, it being understood that the additional cost of acquiring the Source Code shall not exceed the cost of transferring the data.

5.3.2 DISTRIBUTION OF MODIFIED SOFTWARE

When the Licensee makes an Integrated Contribution to the Software, the terms and conditions for the distribution of the resulting Modified Software become subject to all the provisions of this Agreement.

The Licensee is authorized to distribute the Modified Software, in source code or object code form, provided that said distribution complies with all the provisions of the Agreement and is accompanied by:

- 1. a copy of the Agreement,
- 2. a notice relating to the limitation of both the Licensor's warranty and liability as set forth in Articles 8 and 9,

and that, in the event that only the object code of the Modified Software is redistributed, the Licensee allows effective access to the full source code of the Modified Software at a minimum during the entire period of its distribution of the Modified Software, it being understood that the additional cost of acquiring the source code shall not exceed the cost of transferring the data.

5.3.3 DISTRIBUTION OF DERIVATIVE SOFTWARE

When the Licensee creates Derivative Software, this Derivative Software may be distributed under a license agreement other than this Agreement, subject to compliance with the requirement to include a notice concerning the rights over the Software as defined in Article 6.4. In the event the creation of the Derivative Software required modification of the Source Code, the Licensee undertakes that:

- 1. the resulting Modified Software will be governed by this Agreement,
- 2. the Integrated Contributions in the resulting Modified Software will be clearly identified and documented,
- 3. the Licensee will allow effective access to the source code of the Modified Software, at a minimum during the entire period of distribution of the Derivative Software, such that such modifications may be carried over in a subsequent version of the Software; it being understood that the additional cost of purchasing the source code of the Modified Software shall not exceed the cost of transferring the data.

5.3.4 COMPATIBILITY WITH THE CECTLL LICENSE

When a Modified Software contains an Integrated Contribution subject to the CeCILL license agreement, or when a Derivative Software contains a Related Module subject to the CeCILL license agreement, the provisions set forth in the third item of Article 6.4 are optional.

Article 6 - INTELLECTUAL PROPERTY

6.1 OVER THE INITIAL SOFTWARE

The Holder owns the economic rights over the Initial Software. Any or all use of the Initial Software is subject to compliance with the terms and conditions under which the Holder has elected to distribute its work and no one shall be entitled to modify the terms and conditions for the distribution of said Initial Software.

The Holder undertakes that the Initial Software will remain ruled at least by this Agreement, for the duration set forth in Article 4.2.

6.2 OVER THE INTEGRATED CONTRIBUTIONS

The Licensee who develops an Integrated Contribution is the owner of the intellectual property rights over this Contribution as defined by applicable law.

6.3 OVER THE RELATED MODULES

The Licensee who develops a Related Module is the owner of the intellectual property rights over this Related Module as defined by applicable law and is free to choose the type of agreement that shall

govern its distribution under the conditions defined in Article 5.3.3.

6.4 NOTICE OF RIGHTS

The Licensee expressly undertakes:

- 1. not to remove, or modify, in any manner, the intellectual property notices attached to the Software;
- 2. to reproduce said notices, in an identical manner, in the copies of the Software modified or not;
- 3. to ensure that use of the Software, its intellectual property notices and the fact that it is governed by the Agreement is indicated in a text that is easily accessible, specifically from the interface of any Derivative Software.

The Licensee undertakes not to directly or indirectly infringe the intellectual property rights of the Holder and/or Contributors on the Software and to take, where applicable, vis-à-vis its staff, any and all measures required to ensure respect of said intellectual property rights of the Holder and/or Contributors.

Article 7 - RELATED SERVICES

7.1 Under no circumstances shall the Agreement oblige the Licensor to provide technical assistance or maintenance services for the Software.

However, the Licensor is entitled to offer this type of services. The terms and conditions of such technical assistance, and/or such maintenance, shall be set forth in a separate instrument. Only the Licensor offering said maintenance and/or technical assistance services shall incur liability therefor.

7.2 Similarly, any Licensor is entitled to offer to its licensees, under its sole responsibility, a warranty, that shall only be binding upon itself, for the redistribution of the Software and/or the Modified Software, under terms and conditions that it is free to decide. Said warranty, and the financial terms and conditions of its application, shall be subject of a separate instrument executed between the Licensor and the Licensee.

Article 8 - LIABILITY

- 8.1 Subject to the provisions of Article 8.2, the Licensee shall be entitled to claim compensation for any direct loss it may have suffered from the Software as a result of a fault on the part of the relevant Licensor, subject to providing evidence thereof.
- 8.2 The Licensor's liability is limited to the commitments made under this Agreement and shall not be incurred as a result of in particular: (i) loss due the Licensee's total or partial failure to fulfill its obligations, (ii) direct or consequential loss that is suffered by the Licensee due to the use or performance of the Software, and (iii) more generally, any consequential loss. In particular the Parties expressly agree that any or all pecuniary or business loss (i.e. loss of data, loss of profits, operating loss, loss of customers or orders, opportunity cost, any disturbance to business activities) or any or all legal proceedings instituted against the Licensee by a third party, shall constitute consequential loss and shall not provide entitlement to any or all compensation from the Licensor.

Article 9 - WARRANTY

9.1 The Licensee acknowledges that the scientific and technical state-of-the-art when the Software was distributed did not enable all possible uses to be tested and verified, nor for the presence of possible defects to be detected. In this respect, the Licensee's attention has been drawn to the risks associated with loading, using, modifying and/or developing and reproducing the Software which are reserved for experienced users.

The Licensee shall be responsible for verifying, by any or all means, the suitability of the product for its requirements, its good working order, and for ensuring that it shall not cause damage to either persons or properties.

- 9.2 The Licensor hereby represents, in good faith, that it is entitled to grant all the rights over the Software (including in particular the rights set forth in Article 5).
- 9.3 The Licensee acknowledges that the Software is supplied "as is" by the Licensor without any other express or tacit warranty, other than that provided for in Article 9.2 and, in particular, without any warranty as to its commercial value, its secured, safe, innovative or relevant nature.

Specifically, the Licensor does not warrant that the Software is free from any error, that it will operate without interruption, that it will be compatible with the Licensee's own equipment and software configuration, nor that it will meet the Licensee's requirements.

9.4 The Licensor does not either expressly or tacitly warrant that the Software does not infringe any third party intellectual property right relating to a patent, software or any other property right. Therefore, the Licensor disclaims any and all liability towards the Licensee arising out of any or all proceedings for infringement that may be instituted in respect of the use, modification and redistribution of the Software. Nevertheless, should such proceedings be instituted against the Licensee, the Licensor shall provide it with technical and legal assistance for its defense. Such technical and legal assistance shall be decided on a case-by-case basis between the relevant Licensor and the Licensee pursuant to a memorandum of understanding. The Licensor disclaims any and all liability as regards the Licensee's use of the name of the Software. No warranty is given as regards the existence of prior rights over the name of the Software or as regards the existence of a trademark.

Article 10 - TERMINATION

- 10.1 In the event of a breach by the Licensee of its obligations hereunder, the Licensor may automatically terminate this Agreement thirty (30) days after notice has been sent to the Licensee and has remained ineffective.
- 10.2 A Licensee whose Agreement is terminated shall no longer be authorized to use, modify or distribute the Software. However, any licenses that it may have granted prior to termination of the Agreement shall remain valid subject to their having been granted in compliance with the terms and conditions hereof.

11.1 EXCUSABLE EVENTS

Neither Party shall be liable for any or all delay, or failure to perform the Agreement, that may be attributable to an event of force majeure, an act of God or an outside cause, such as defective functioning or interruptions of the electricity or telecommunications networks, network paralysis following a virus attack, intervention by government authorities, natural disasters, water damage, earthquakes, fire, explosions, strikes and labor unrest, war, etc.

- 11.2 Any failure by either Party, on one or more occasions, to invoke one or more of the provisions hereof, shall under no circumstances be interpreted as being a waiver by the interested Party of its right to invoke said provision(s) subsequently.
- 11.3 The Agreement cancels and replaces any or all previous agreements, whether written or oral, between the Parties and having the same purpose, and constitutes the entirety of the agreement between said Parties concerning said purpose. No supplement or modification to the terms and conditions hereof shall be effective as between the Parties unless it is made in writing and signed by their duly authorized representatives.
- 11.4 In the event that one or more of the provisions hereof were to conflict with a current or future applicable act or legislative text, said act or legislative text shall prevail, and the Parties shall make the necessary amendments so as to comply with said act or legislative text. All other provisions shall remain effective. Similarly, invalidity of a provision of the Agreement, for any reason whatsoever, shall not cause the Agreement as a whole to be invalid.

11.5 LANGUAGE

The Agreement is drafted in both French and English and both versions are deemed authentic.

Article 12 - NEW VERSIONS OF THE AGREEMENT

- 12.1 Any person is authorized to duplicate and distribute copies of this Agreement.
- 12.2 So as to ensure coherence, the wording of this Agreement is protected and may only be modified by the authors of the License, who reserve the right to periodically publish updates or new versions of the Agreement, each with a separate number. These subsequent versions may address new issues encountered by Free Software.
- 12.3 Any Software distributed under a given version of the Agreement may only be subsequently distributed under the same version of the Agreement or a subsequent version.

Article 13 - GOVERNING LAW AND JURISDICTION

- 13.1 The Agreement is governed by French law. The Parties agree to endeavor to seek an amicable solution to any disagreements or disputes that may arise during the performance of the Agreement.
- 13.2 Failing an amicable solution within two (2) months as from their occurrence, and unless emergency proceedings are necessary, the

Version 1.0 dated 2006-09-05.

Copyright (c) 2005-2023, NumPy Developers. All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * Neither the name of the NumPy Developers nor the names of any contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Copyright (c) 2011-2014, The OpenBLAS Project All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. Neither the name of the OpenBLAS project nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL

DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

The MIT License (MIT)

Copyright (c) 2014-2017 Matthias C. M. Troffaes Copyright (c) 2012-2014 Antoine Pitrou and contributors

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Copyright (c) 1991-2021, UChicago Argonne, LLC and the PETSc Development Team All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT HOLDER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

This license DOES NOT apply to any software that may be obtained via the --download-package option of the PETSc configuration. Each of those packages is covered by its own license(s).

Copyright (c) 2012 University of North Carolina at Chapel Hill Luke Campagnola ('luke.campagnola@%s.com' % 'gmail')

The MIT License

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

GNU LESSER GENERAL PUBLIC LICENSE Version 3, 29 June 2007

Copyright (C) 2007 Free Software Foundation, Inc. http://fsf.org/ Everyone is permitted to copy and distribute verbatim copies of this license document, but changing it is not allowed.

This version of the GNU Lesser General Public License incorporates the terms and conditions of version 3 of the GNU General Public License, supplemented by the additional permissions listed below.

O. Additional Definitions.

As used herein, "this License" refers to version 3 of the GNU Lesser General Public License, and the "GNU GPL" refers to version 3 of the GNU General Public License.

"The Library" refers to a covered work governed by this License, other than an Application or a Combined Work as defined below.

An "Application" is any work that makes use of an interface provided by the Library, but which is not otherwise based on the Library. Defining a subclass of a class defined by the Library is deemed a mode of using an interface provided by the Library.

A "Combined Work" is a work produced by combining or linking an Application with the Library. The particular version of the Library with which the Combined Work was made is also called the "Linked Version".

The "Minimal Corresponding Source" for a Combined Work means the Corresponding Source for the Combined Work, excluding any source code for portions of the Combined Work that, considered in isolation, are based on the Application, and not on the Linked Version.

The "Corresponding Application Code" for a Combined Work means the object code and/or source code for the Application, including any data and utility programs needed for reproducing the Combined Work from the Application, but excluding the System Libraries of the Combined Work.

1. Exception to Section 3 of the GNU GPL.

You may convey a covered work under sections 3 and 4 of this License without being bound by section 3 of the GNU GPL.

2. Conveying Modified Versions.

If you modify a copy of the Library, and, in your modifications, a facility refers to a function or data to be supplied by an Application that uses the facility (other than as an argument passed when the facility is invoked), then you may convey a copy of the modified version:

- a) under this License, provided that you make a good faith effort to ensure that, in the event an Application does not supply the function or data, the facility still operates, and performs whatever part of its purpose remains meaningful, or
- b) under the GNU GPL, with none of the additional permissions of this License applicable to that copy.
- 3. Object Code Incorporating Material from Library Header Files.

The object code form of an Application may incorporate material from a header file that is part of the Library. You may convey such object code under terms of your choice, provided that, if the incorporated material is not limited to numerical parameters, data structure layouts and accessors, or small macros, inline functions and templates (ten or fewer lines in length), you do both of the following:

- a) Give prominent notice with each copy of the object code that the Library is used in it and that the Library and its use are covered by this License.
- b) Accompany the object code with a copy of the GNU GPL and this license document.
- 4. Combined Works.

You may convey a Combined Work under terms of your choice that, taken together, effectively do not restrict modification of the portions of the Library contained in the Combined Work and reverse engineering for debugging such modifications, if you also do each of the following:

- a) Give prominent notice with each copy of the Combined Work that the Library is used in it and that the Library and its use are covered by this License.
- b) Accompany the Combined Work with a copy of the GNU GPL and this license document.
- c) For a Combined Work that displays copyright notices during execution, include the copyright notice for the Library among these notices, as well as a reference directing the user to the copies of the GNU GPL and this license document.
- d) Do one of the following:
 - 0) Convey the Minimal Corresponding Source under the terms of this License, and the Corresponding Application Code in a form suitable for, and under terms that permit, the user to recombine or relink the Application with a modified version of the Linked Version to produce a modified Combined Work, in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.
 - 1) Use a suitable shared library mechanism for linking with the Library. A suitable mechanism is one that (a) uses at run time a copy of the Library already present on the user's computer system, and (b) will operate properly with a modified version of the Library that is interface-compatible with the Linked Version.
- e) Provide Installation Information, but only if you would otherwise be required to provide such information under section 6 of the GNU GPL, and only to the extent that such information is necessary to install and execute a modified version of the Combined Work produced by recombining or relinking the Application with a modified version of the Linked Version. (If you use option 4d0, the Installation Information must accompany the Minimal Corresponding Source and Corresponding Application Code. If you use option 4d1, you must provide the Installation Information in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.)
- 5. Combined Libraries.

You may place library facilities that are a work based on the Library side by side in a single library together with other library facilities that are not Applications and are not covered by this License, and convey such a combined library under terms of your choice, if you do both of the following:

- a) Accompany the combined library with a copy of the same work based on the Library, uncombined with any other library facilities, conveyed under the terms of this License.
- b) Give prominent notice with the combined library that part of it is a work based on the Library, and explaining where to find the accompanying uncombined form of the same work.
- 6. Revised Versions of the GNU Lesser General Public License.

The Free Software Foundation may publish revised and/or new versions of the GNU Lesser General Public License from time to time. Such new versions will be similar in spirit to the present version, but may differ in detail to address new problems or concerns.

Each version is given a distinguishing version number. If the Library as you received it specifies that a certain numbered version of the GNU Lesser General Public License "or any later version" applies to it, you have the option of following the terms and conditions either of that published version or of any later version published by the Free Software Foundation. If the Library as you received it does not specify a version number of the GNU Lesser General Public License, you may choose any version of the GNU Lesser General Public License ever published by the Free Software Foundation.

If the Library as you received it specifies that a proxy can decide whether future versions of the GNU Lesser General Public License shall apply, that proxy's public statement of acceptance of any version is permanent authorization for you to choose that version for the Library.

BEOPEN.COM LICENSE AGREEMENT FOR PYTHON 2.0

BEOPEN PYTHON OPEN SOURCE LICENSE AGREEMENT VERSION 1

1. This LICENSE AGREEMENT is between BeOpen.com ("BeOpen"), having an office at 160 Saratoga Avenue, Santa Clara, CA 95051, and the Individual or Organization

("Licensee") accessing and otherwise using this software in source or binary form and its associated documentation ("the Software").

- 2. Subject to the terms and conditions of this BeOpen Python License Agreement, BeOpen hereby grants Licensee a non-exclusive, royalty-free, world-wide license
- to reproduce, analyze, test, perform and/or display publicly, prepare derivative
 - works, distribute, and otherwise use the Software alone or in any derivative version, provided, however, that the BeOpen Python License is retained in the Software, alone or in any derivative version prepared by Licensee.
- 3. BeOpen is making the Software available to Licensee on an "AS IS" basis.
 BEOPEN MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED. BY WAY OF
 EXAMPLE, BUT NOT LIMITATION, BEOPEN MAKES NO AND DISCLAIMS ANY REPRESENTATION
 OR
 - WARRANTY OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE OR THAT THE USE OF THE SOFTWARE WILL NOT INFRINGE ANY THIRD PARTY RIGHTS.
- 4. BEOPEN SHALL NOT BE LIABLE TO LICENSEE OR ANY OTHER USERS OF THE SOFTWARE FOR ANY INCIDENTAL, SPECIAL, OR CONSEQUENTIAL DAMAGES OR LOSS AS A RESULT OF USING.

MODIFYING OR DISTRIBUTING THE SOFTWARE, OR ANY DERIVATIVE THEREOF, EVEN IF ADVISED OF THE POSSIBILITY THEREOF.

- 5. This License Agreement will automatically terminate upon a material breach of its terms and conditions.
- 6. This License Agreement shall be governed by and interpreted in all respects by the law of the State of California, excluding conflict of law provisions. Nothing in this License Agreement shall be deemed to create any relationship of
- agency, partnership, or joint venture between BeOpen and Licensee. This License
- Agreement does not grant permission to use BeOpen trademarks or trade names in a
- trademark sense to endorse or promote products or services of Licensee, or any
 - third party. As an exception, the "BeOpen Python" logos available at http://www.pythonlabs.com/logos.html may be used according to the permissions granted on that web page.
- 7. By copying, installing or otherwise using the software, Licensee agrees to be bound by the terms and conditions of this License Agreement.

CNRI LICENSE AGREEMENT FOR PYTHON 1.6.1

- 1. This LICENSE AGREEMENT is between the Corporation for National Research Initiatives, having an office at 1895 Preston White Drive, Reston, VA 20191 ("CNRI"), and the Individual or Organization ("Licensee") accessing and otherwise using Python 1.6.1 software in source or binary form and its associated documentation.
- 2. Subject to the terms and conditions of this License Agreement, CNRI hereby grants Licensee a nonexclusive, royalty-free, world-wide license to reproduce,

analyze, test, perform and/or display publicly, prepare derivative works, distribute, and otherwise use Python 1.6.1 alone or in any derivative

provided, however, that CNRI's License Agreement and CNRI's notice of copyright, i.e., "Copyright © 1995-2001 Corporation for National Research Initiatives;

Rights Reserved" are retained in Python 1.6.1 alone or in any derivative version

prepared by Licensee. Alternately, in lieu of CNRI's License Agreement, Licensee may substitute the following text (omitting the quotes): "Python 1.6.1

is made available subject to the terms and conditions in CNRI's License Agreement. This Agreement together with Python 1.6.1 may be located on the internet using the following unique, persistent identifier (known as a handle):

1895.22/1013. This Agreement may also be obtained from a proxy server on the internet using the following URL: http://hdl.handle.net/1895.22/1013."

3. In the event Licensee prepares a derivative work that is based on or incorporates Python 1.6.1 or any part thereof, and wants to make the derivative

work available to others as provided herein, then Licensee hereby agrees to include in any such work a brief summary of the changes made to Python 1.6.1.

4. CNRI is making Python 1.6.1 available to Licensee on an "AS IS" basis. MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED. BY WAY OF EXAMPLE,

BUT NOT LIMITATION, CNRI MAKES NO AND DISCLAIMS ANY REPRESENTATION OR WARRANTY

OF MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE OR THAT THE USE OF PYTHON 1.6.1 WILL NOT INFRINGE ANY THIRD PARTY RIGHTS.

- 5. CNRI SHALL NOT BE LIABLE TO LICENSEE OR ANY OTHER USERS OF PYTHON 1.6.1 FOR ANY INCIDENTAL, SPECIAL, OR CONSEQUENTIAL DAMAGES OR LOSS AS A RESULT OF MODIFYING, DISTRIBUTING, OR OTHERWISE USING PYTHON 1.6.1, OR ANY DERIVATIVE THEREOF, EVEN IF ADVISED OF THE POSSIBILITY THEREOF.
- 6. This License Agreement will automatically terminate upon a material breach of its terms and conditions.
- 7. This License Agreement shall be governed by the federal intellectual property law of the United States, including without limitation the federal copyright law, and, to the extent such U.S. federal law does not apply, by the law of

Commonwealth of Virginia, excluding Virginia's conflict of law provisions. Notwithstanding the foregoing, with regard to derivative works based on

1.6.1 that incorporate non-separable material that was previously distributed under the GNU General Public License (GPL), the law of the Commonwealth of Virginia shall govern this License Agreement only as to issues arising under

with respect to Paragraphs 4, 5, and 7 of this License Agreement. Nothing in this License Agreement shall be deemed to create any relationship of agency, partnership, or joint venture between CNRI and Licensee. This License

does not grant permission to use CNRI trademarks or trade name in a trademark sense to endorse or promote products or services of Licensee, or any third party.

8. By clicking on the "ACCEPT" button where indicated, or by copying, installing or otherwise using Python 1.6.1, Licensee agrees to be bound by the terms and conditions of this License Agreement.

CWI LICENSE AGREEMENT FOR PYTHON 0.9.0 THROUGH 1.2

Copyright © 1991 - 1995, Stichting Mathematisch Centrum Amsterdam, The Netherlands. All rights reserved.

Permission to use, copy, modify, and distribute this software and its documentation for any purpose and without fee is hereby granted, provided that the above copyright notice appear in all copies and that both that copyright notice and this permission notice appear in supporting documentation, and that the name of Stichting Mathematisch Centrum or CWI not be used in advertising or publicity pertaining to distribution of the software without specific, written prior permission.

STICHTING MATHEMATISCH CENTRUM DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE, INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS, IN NO EVENT SHALL STICHTING MATHEMATISCH CENTRUM BE LIABLE FOR ANY SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

ZERO-CLAUSE BSD LICENSE FOR CODE IN THE PYTHON 3.11.8 DOCUMENTATION

Permission to use, copy, modify, and/or distribute this software for any purpose with or without fee is hereby granted.

THE SOFTWARE IS PROVIDED "AS IS" AND THE AUTHOR DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS. IN NO EVENT SHALL THE AUTHOR BE LIABLE FOR ANY SPECIAL, DIRECT, INDIRECT, OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

Licenses and Acknowledgements for Incorporated Software

This section is an incomplete, but growing list of licenses and acknowledgements for third-party software incorporated in the Python distribution.

Mersenne Twister

The _random module includes code based on a download from http://www.math.sci.hiroshima-u.ac.jp/~m-mat/MT/MT2002/emt19937ar.html. The following are the verbatim comments from the original code:

A C-program for MT19937, with initialization improved 2002/1/26. Coded by Takuji Nishimura and Makoto Matsumoto.

Before using, initialize the state by using init_genrand(seed) or init_by_array(init_key, key_length).

Copyright (C) 1997 - 2002, Makoto Matsumoto and Takuji Nishimura, All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

3. The names of its contributors may not be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS
"AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT
LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR
A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR
CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL,
EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO,
PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR
PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF
LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING
NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS
SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Any feedback is very welcome.

http://www.math.sci.hiroshima-u.ac.jp/~m-mat/MT/emt.html
email: m-mat @ math.sci.hiroshima-u.ac.jp (remove space)

Sockets

The socket module uses the functions, getaddrinfo(), and getnameinfo(), which are coded in separate source files from the WIDE Project, https://www.wide.ad.jp/.

Copyright (C) 1995, 1996, 1997, and 1998 WIDE Project. All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- 3. Neither the name of the project nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE PROJECT AND CONTRIBUTORS `AS IS' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE PROJECT OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Asynchronous socket services

The asynchat and asyncore modules contain the following notice:

Copyright 1996 by Sam Rushing

All Rights Reserved

Permission to use, copy, modify, and distribute this software and its documentation for any purpose and without fee is hereby

granted, provided that the above copyright notice appear in all copies and that both that copyright notice and this permission notice appear in supporting documentation, and that the name of Sam Rushing not be used in advertising or publicity pertaining to distribution of the software without specific, written prior permission.

SAM RUSHING DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE, INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS, IN NO EVENT SHALL SAM RUSHING BE LIABLE FOR ANY SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

Cookie management

The http.cookies module contains the following notice:

Copyright 2000 by Timothy O'Malley <timo@alum.mit.edu>

All Rights Reserved

Permission to use, copy, modify, and distribute this software and its documentation for any purpose and without fee is hereby granted, provided that the above copyright notice appear in all copies and that both that copyright notice and this permission notice appear in supporting documentation, and that the name of Timothy O'Malley not be used in advertising or publicity pertaining to distribution of the software without specific, written prior permission.

Timothy O'Malley DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE, INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS, IN NO EVENT SHALL Timothy O'Malley BE LIABLE FOR ANY SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

Execution tracing

The trace module contains the following notice:

portions copyright 2001, Autonomous Zones Industries, Inc., all rights... err... reserved and offered to the public under the terms of the

Python 2.2 license.

Author: Zooko O'Whielacronx

http://zooko.com/ mailto:zooko@zooko.com

Copyright 2000, Mojam Media, Inc., all rights reserved.

Author: Skip Montanaro

Copyright 1999, Bioreason, Inc., all rights reserved.

Author: Andrew Dalke

Copyright 1995-1997, Automatrix, Inc., all rights reserved.

Author: Skip Montanaro

Copyright 1991-1995, Stichting Mathematisch Centrum, all rights reserved.

Permission to use, copy, modify, and distribute this Python software and its associated documentation for any purpose without fee is hereby granted, provided that the above copyright notice appears in all copies, and that both that copyright notice and this permission notice appear in supporting documentation, and that the name of neither Automatrix, Bioreason or Mojam Media be used in advertising or publicity pertaining to distribution of the software without specific, written prior permission.

UUencode and UUdecode functions

The uu module contains the following notice:

Copyright 1994 by Lance Ellinghouse Cathedral City, California Republic, United States of America. All Rights Reserved

Permission to use, copy, modify, and distribute this software and its documentation for any purpose and without fee is hereby granted, provided that the above copyright notice appear in all copies and that both that copyright notice and this permission notice appear in supporting documentation, and that the name of Lance Ellinghouse not be used in advertising or publicity pertaining to distribution of the software without specific, written prior permission.

LANCE ELLINGHOUSE DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE, INCLUDING ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS, IN NO EVENT SHALL LANCE ELLINGHOUSE CENTRUM BE LIABLE FOR ANY SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

Modified by Jack Jansen, CWI, July 1995:

- Use binascii module to do the actual line-by-line conversion between ascii and binary. This results in a 1000-fold speedup. The C version is still 5 times faster, though.
- Arguments more compliant with Python standard

XML Remote Procedure Calls

The xmlrpc.client module contains the following notice:

The XML-RPC client interface is

Copyright (c) 1999-2002 by Secret Labs AB Copyright (c) 1999-2002 by Fredrik Lundh

By obtaining, using, and/or copying this software and/or its associated documentation, you agree that you have read, understood, and will comply with the following terms and conditions:

Permission to use, copy, modify, and distribute this software and its associated documentation for any purpose and without fee is hereby granted, provided that the above copyright notice appears in all copies, and that both that copyright notice and this permission notice appear in supporting documentation, and that the name of Secret Labs AB or the author not be used in advertising or publicity pertaining to distribution of the software without specific, written prior permission.

SECRET LABS AB AND THE AUTHOR DISCLAIMS ALL WARRANTIES WITH REGARD TO THIS SOFTWARE, INCLUDING ALL IMPLIED WARRANTIES OF MERCHANT-ABILITY AND FITNESS. IN NO EVENT SHALL SECRET LABS AB OR THE AUTHOR BE LIABLE FOR ANY SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES WHATSOEVER RESULTING FROM LOSS OF USE, DATA OR PROFITS, WHETHER IN AN ACTION OF CONTRACT, NEGLIGENCE OR OTHER TORTIOUS

ACTION, ARISING OUT OF OR IN CONNECTION WITH THE USE OR PERFORMANCE OF THIS SOFTWARE.

test_epoll

The test.test_epoll module contains the following notice:

Copyright (c) 2001-2006 Twisted Matrix Laboratories.

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Select kqueue

The select module contains the following notice for the kqueue interface:

Copyright (c) 2000 Doug White, 2006 James Knight, 2007 Christian Heimes All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

THIS SOFTWARE IS PROVIDED BY THE AUTHOR AND CONTRIBUTORS ``AS IS'' AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE AUTHOR OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

SipHash24

The file Python/pyhash.c contains Marek Majkowski' implementation of Dan Bernstein's SipHash24 algorithm. It contains the following note:

<MIT License>

Copyright (c) 2013 Marek Majkowski <marek@popcount.org>

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.
</MIT License>

Original location:

https://github.com/majek/csiphash/

Solution inspired by code from:

Samuel Neves (supercop/crypto_auth/siphash24/little)
djb (supercop/crypto_auth/siphash24/little2)
Jean-Philippe Aumasson (https://131002.net/siphash/siphash24.c)

strtod and dtoa

The file Python/dtoa.c, which supplies C functions dtoa and strtod for conversion of C doubles to and from strings, is derived from the file of the same name by David M. Gay, currently available from https://web.archive.org/web/20220517033456/http://www.netlib.org/fp/dtoa.c. The original file, as retrieved on March 16, 2009, contains the following copyright and licensing notice:

* THIS SOFTWARE IS BEING PROVIDED "AS IS", WITHOUT ANY EXPRESS OR IMPLIED
* WARRANTY. IN PARTICULAR, NEITHER THE AUTHOR NOR LUCENT MAKES ANY
* REPRESENTATION OR WARRANTY OF ANY KIND CONCERNING THE MERCHANTABILITY
* OF THIS SOFTWARE OR ITS FITNESS FOR ANY PARTICULAR PURPOSE.
*

OpenSSL

The modules hashlib, posix, ssl, crypt use the OpenSSL library for added performance if made available by the operating system. Additionally, the Windows and macOS installers for Python may include a copy of the OpenSSL libraries, so we include a copy of the OpenSSL license here. For the OpenSSL 3.0 release, and later releases derived from that, the Apache License v2 applies:

Apache License Version 2.0, January 2004 https://www.apache.org/licenses/

TERMS AND CONDITIONS FOR USE, REPRODUCTION, AND DISTRIBUTION

1. Definitions.

"License" shall mean the terms and conditions for use, reproduction,

and distribution as defined by Sections 1 through 9 of this document.

"Licensor" shall mean the copyright owner or entity authorized by the copyright owner that is granting the License.

"Legal Entity" shall mean the union of the acting entity and all other entities that control, are controlled by, or are under common control with that entity. For the purposes of this definition, "control" means (i) the power, direct or indirect, to cause the direction or management of such entity, whether by contract or otherwise, or (ii) ownership of fifty percent (50%) or more of the outstanding shares, or (iii) beneficial ownership of such entity.

"You" (or "Your") shall mean an individual or Legal Entity exercising permissions granted by this License.

"Source" form shall mean the preferred form for making modifications, including but not limited to software source code, documentation source, and configuration files.

"Object" form shall mean any form resulting from mechanical transformation or translation of a Source form, including but not limited to compiled object code, generated documentation, and conversions to other media types.

"Work" shall mean the work of authorship, whether in Source or Object form, made available under the License, as indicated by a copyright notice that is included in or attached to the work (an example is provided in the Appendix below).

"Derivative Works" shall mean any work, whether in Source or Object form, that is based on (or derived from) the Work and for which the editorial revisions, annotations, elaborations, or other modifications represent, as a whole, an original work of authorship. For the purposes of this License, Derivative Works shall not include works that remain separable from, or merely link (or bind by name) to the interfaces of, the Work and Derivative Works thereof.

"Contribution" shall mean any work of authorship, including the original version of the Work and any modifications or additions to that Work or Derivative Works thereof, that is intentionally submitted to Licensor for inclusion in the Work by the copyright owner or by an individual or Legal Entity authorized to submit on behalf of the copyright owner. For the purposes of this definition, "submitted" means any form of electronic, verbal, or written communication sent to the Licensor or its representatives, including but not limited to communication on electronic mailing lists, source code control systems, and issue tracking systems that are managed by, or on behalf of, the Licensor for the purpose of discussing and improving the Work, but excluding communication that is conspicuously marked or otherwise designated in writing by the copyright owner as "Not a Contribution."

"Contributor" shall mean Licensor and any individual or Legal Entity on behalf of whom a Contribution has been received by Licensor and subsequently incorporated within the Work.

- 2. Grant of Copyright License. Subject to the terms and conditions of this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable copyright license to reproduce, prepare Derivative Works of, publicly display, publicly perform, sublicense, and distribute the Work and such Derivative Works in Source or Object form.
- 3. Grant of Patent License. Subject to the terms and conditions of

this License, each Contributor hereby grants to You a perpetual, worldwide, non-exclusive, no-charge, royalty-free, irrevocable (except as stated in this section) patent license to make, have made, use, offer to sell, sell, import, and otherwise transfer the Work, where such license applies only to those patent claims licensable by such Contributor that are necessarily infringed by their Contribution(s) alone or by combination of their Contribution(s) with the Work to which such Contribution(s) was submitted. If You institute patent litigation against any entity (including a cross-claim or counterclaim in a lawsuit) alleging that the Work or a Contribution incorporated within the Work constitutes direct or contributory patent infringement, then any patent licenses granted to You under this License for that Work shall terminate as of the date such litigation is filed.

- 4. Redistribution. You may reproduce and distribute copies of the Work or Derivative Works thereof in any medium, with or without modifications, and in Source or Object form, provided that You meet the following conditions:
 - (a) You must give any other recipients of the Work or Derivative Works a copy of this License; and
 - (b) You must cause any modified files to carry prominent notices stating that You changed the files; and
 - (c) You must retain, in the Source form of any Derivative Works that You distribute, all copyright, patent, trademark, and attribution notices from the Source form of the Work, excluding those notices that do not pertain to any part of the Derivative Works; and
 - (d) If the Work includes a "NOTICE" text file as part of its distribution, then any Derivative Works that You distribute must include a readable copy of the attribution notices contained within such NOTICE file, excluding those notices that do not pertain to any part of the Derivative Works, in at least one of the following places: within a NOTICE text file distributed as part of the Derivative Works; within the Source form or documentation, if provided along with the Derivative Works; or, within a display generated by the Derivative Works, if and wherever such third-party notices normally appear. The contents of the NOTICE file are for informational purposes only and do not modify the License. You may add Your own attribution notices within Derivative Works that You distribute, alongside or as an addendum to the NOTICE text from the Work, provided that such additional attribution notices cannot be construed as modifying the License.

You may add Your own copyright statement to Your modifications and may provide additional or different license terms and conditions for use, reproduction, or distribution of Your modifications, or for any such Derivative Works as a whole, provided Your use, reproduction, and distribution of the Work otherwise complies with the conditions stated in this License.

5. Submission of Contributions. Unless You explicitly state otherwise, any Contribution intentionally submitted for inclusion in the Work by You to the Licensor shall be under the terms and conditions of this License, without any additional terms or conditions. Notwithstanding the above, nothing herein shall supersede or modify the terms of any separate license agreement you may have executed with Licensor regarding such Contributions.

- 6. Trademarks. This License does not grant permission to use the trade names, trademarks, service marks, or product names of the Licensor, except as required for reasonable and customary use in describing the origin of the Work and reproducing the content of the NOTICE file.
- 7. Disclaimer of Warranty. Unless required by applicable law or agreed to in writing, Licensor provides the Work (and each Contributor provides its Contributions) on an "AS IS" BASIS, WITHOUT WARRANTIES OR CONDITIONS OF ANY KIND, either express or implied, including, without limitation, any warranties or conditions of TITLE, NON-INFRINGEMENT, MERCHANTABILITY, or FITNESS FOR A PARTICULAR PURPOSE. You are solely responsible for determining the appropriateness of using or redistributing the Work and assume any risks associated with Your exercise of permissions under this License.
- 8. Limitation of Liability. In no event and under no legal theory, whether in tort (including negligence), contract, or otherwise, unless required by applicable law (such as deliberate and grossly negligent acts) or agreed to in writing, shall any Contributor be liable to You for damages, including any direct, indirect, special, incidental, or consequential damages of any character arising as a result of this License or out of the use or inability to use the Work (including but not limited to damages for loss of goodwill, work stoppage, computer failure or malfunction, or any and all other commercial damages or losses), even if such Contributor has been advised of the possibility of such damages.
- 9. Accepting Warranty or Additional Liability. While redistributing the Work or Derivative Works thereof, You may choose to offer, and charge a fee for, acceptance of support, warranty, indemnity, or other liability obligations and/or rights consistent with this License. However, in accepting such obligations, You may act only on Your own behalf and on Your sole responsibility, not on behalf of any other Contributor, and only if You agree to indemnify, defend, and hold each Contributor harmless for any liability incurred by, or claims asserted against, such Contributor by reason of your accepting any such warranty or additional liability.

END OF TERMS AND CONDITIONS

expat

The pyexpat extension is built using an included copy of the expat sources unless the build is configured --with-system-expat:

Copyright (c) 1998, 1999, 2000 Thai Open Source Software Center Ltd and Clark Cooper

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT,

TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

libffi

The _ctypes extension is built using an included copy of the libffi sources unless the build is configured --with-system-libffi:

Copyright (c) 1996-2008 Red Hat, Inc and others.

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the ``Software''), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED `AS IS'', WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

7lib

The zlib extension is built using an included copy of the zlib sources if the zlib version found on the system is too old to be used for the build:

Copyright (C) 1995-2011 Jean-loup Gailly and Mark Adler

This software is provided 'as-is', without any express or implied warranty. In no event will the authors be held liable for any damages arising from the use of this software.

Permission is granted to anyone to use this software for any purpose, including commercial applications, and to alter it and redistribute it freely, subject to the following restrictions:

- 1. The origin of this software must not be misrepresented; you must not claim that you wrote the original software. If you use this software in a product, an acknowledgment in the product documentation would be appreciated but is not required.
- 2. Altered source versions must be plainly marked as such, and must not be misrepresented as being the original software.
- 3. This notice may not be removed or altered from any source distribution.

Jean-loup Gailly Mark Adler jloup@gzip.org madler@alumni.caltech.edu

cfuhash

The implementation of the hash table used by the tracemalloc is based on the cfuhash project:

Copyright (c) 2005 Don Owens

All rights reserved.

This code is released under the BSD license:

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * Neither the name of the author nor the names of its contributors may be used to endorse or promote products derived from this software without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

libmpdec

The _decimal module is built using an included copy of the libmpdec library unless the build is configured --with-system-libmpdec:

Copyright (c) 2008-2020 Stefan Krah. All rights reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- 1. Redistributions of source code must retain the above copyright notice, this list of conditions and the following disclaimer.
- 2. Redistributions in binary form must reproduce the above copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.

THIS SOFTWARE IS PROVIDED BY THE AUTHOR AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE AUTHOR OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

W3C C14N test suite

The C14N 2.0 test suite in the test package (Lib/test/xmltestdata/c14n-20/) was retrieved from the W3C website at https://www.w3.org/TR/xml-c14n2-testcases/ and is distributed under the 3-clause BSD license:

Copyright (c) 2013 W3C(R) (MIT, ERCIM, Keio, Beihang), All Rights Reserved.

Redistribution and use in source and binary forms, with or without modification, are permitted provided that the following conditions are met:

- * Redistributions of works must retain the original copyright notice, this list of conditions and the following disclaimer.
- * Redistributions in binary form must reproduce the original copyright notice, this list of conditions and the following disclaimer in the documentation and/or other materials provided with the distribution.
- * Neither the name of the W3C nor the names of its contributors may be used to endorse or promote products derived from this work without specific prior written permission.

THIS SOFTWARE IS PROVIDED BY THE COPYRIGHT HOLDERS AND CONTRIBUTORS "AS IS" AND ANY EXPRESS OR IMPLIED WARRANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE ARE DISCLAIMED. IN NO EVENT SHALL THE COPYRIGHT OWNER OR CONTRIBUTORS BE LIABLE FOR ANY DIRECT, INDIRECT, INCIDENTAL, SPECIAL, EXEMPLARY, OR CONSEQUENTIAL DAMAGES (INCLUDING, BUT NOT LIMITED TO, PROCUREMENT OF SUBSTITUTE GOODS OR SERVICES; LOSS OF USE, DATA, OR PROFITS; OR BUSINESS INTERRUPTION) HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY, WHETHER IN CONTRACT, STRICT LIABILITY, OR TORT (INCLUDING NEGLIGENCE OR OTHERWISE) ARISING IN ANY WAY OUT OF THE USE OF THIS SOFTWARE, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGE.

Audioop

The audioop module uses the code base in g771.c file of the SoX project. https://sourceforge.net/projects/sox/files/sox/12.17.7/sox-12.17.7.tar.gz

This source code is a product of Sun Microsystems, Inc. and is provided for unrestricted use. Users may copy or modify this source code without charge.

SUN SOURCE CODE IS PROVIDED AS IS WITH NO WARRANTIES OF ANY KIND INCLUDING THE WARRANTIES OF DESIGN, MERCHANTIBILITY AND FITNESS FOR A PARTICULAR PURPOSE, OR ARISING FROM A COURSE OF DEALING, USAGE OR TRADE PRACTICE.

Sun source code is provided with no support and without any obligation on the part of Sun Microsystems, Inc. to assist in its use, correction, modification or enhancement.

SUN MICROSYSTEMS, INC. SHALL HAVE NO LIABILITY WITH RESPECT TO THE INFRINGEMENT OF COPYRIGHTS, TRADE SECRETS OR ANY PATENTS BY THIS SOFTWARE OR ANY PART THEREOF.

In no event will Sun Microsystems, Inc. be liable for any lost revenue or profits or other special, indirect and consequential damages, even if Sun has been advised of the possibility of such damages.

Sun Microsystems, Inc. 2550 Garcia Avenue Mountain View, California 94043 asyncio

Parts of the asyncio module are incorporated from uvloop 0.16, which is distributed under the MIT license:

Copyright (c) 2015-2021 MagicStack Inc. http://magic.io

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Qt

http://doc.qt.io/qt-5

version: 5.12

GNU LESSER GENERAL PUBLIC LICENSE Version 3, 29 June 2007

Qt is available under different licensing options designed to accommodate the needs of our various users:

Qt licensed under commercial licenses is appropriate for development of proprietary/commercial software where you do not want to share any source code with third parties or otherwise cannot comply with the terms of the GNU LGPL version 3. Qt licensed under the GNU Lesser General Public License (LGPL) version 3 is appropriate for the development of Qt applications provided you can comply with the terms and conditions of the GNU LGPL version 3 (or GNU GPL version 3). Qt contains also third-party code that is licensed under specific open-source licenses from the original authors.

Note: For open-source licensed Qt, some specific parts (modules) are not available under the GNU LGPL version 3, but under the GNU General Public License (GPL) instead. See the list of Qt modules for details. For commercial licensees, all modules are available under a single, commercial Qt license.

Qt documentation is licensed under the terms of the GNU Free Documentation License (FDL) version 1.3, as published by the Free Software Foundation. Alternatively, you may use the documentation in accordance with the terms contained in a written agreement between you and The Qt Company.

See http://qt.io/licensing/ for an overview of Qt licensing.

GNU LESSER GENERAL PUBLIC LICENSE Version 3, 29 June 2007

Copyright (C) 2007 Free Software Foundation, Inc. http://fsf.org/ Everyone is permitted to copy and distribute verbatim copies of this license document, but changing it is not allowed.

This version of the GNU Lesser General Public License incorporates the terms and conditions of version 3 of the GNU General Public License, supplemented by the additional permissions listed below.

0. Additional Definitions.

As used herein, "this License" refers to version 3 of the GNU Lesser General Public License, and the "GNU GPL" refers to version 3 of the GNU General Public License.

"The Library" refers to a covered work governed by this License, other than an Application or a Combined Work as defined below.

An "Application" is any work that makes use of an interface provided by the Library, but which is not otherwise based on the Library. Defining a subclass of a class defined by the Library is deemed a mode of using an interface provided by the Library.

A "Combined Work" is a work produced by combining or linking an Application with the Library. The particular version of the Library with which the Combined Work was made is also called the "Linked Version".

The "Minimal Corresponding Source" for a Combined Work means the Corresponding Source for the Combined Work, excluding any source code for portions of the Combined Work that, considered in isolation, are based on the Application, and not on the Linked Version.

The "Corresponding Application Code" for a Combined Work means the object code and/or source code for the Application, including any data and utility programs needed for reproducing the Combined Work from the Application, but excluding the System Libraries of the Combined Work.

1. Exception to Section 3 of the GNU GPL.

You may convey a covered work under sections 3 and 4 of this License without being bound by section 3 of the GNU GPL.

2. Conveying Modified Versions.

If you modify a copy of the Library, and, in your modifications, a facility refers to a function or data to be supplied by an Application that uses the facility (other than as an argument passed when the facility is invoked), then you may convey a copy of the modified version:

- a. under this License, provided that you make a good faith effort to ensure that, in the event an Application does not supply the function or data, the facility still operates, and performs whatever part of its purpose remains meaningful, or
- b. under the GNU GPL, with none of the additional permissions of this License applicable to that copy.
- 3. Object Code Incorporating Material from Library Header Files.

The object code form of an Application may incorporate material from a header file that is part of the Library. You may convey such object code under terms of your choice, provided that, if the incorporated material is not limited to numerical parameters, data structure layouts and accessors, or small macros, inline functions and templates (ten or fewer lines in length), you do both of the following:

- a. Give prominent notice with each copy of the object code that the Library is used in it and that the Library and its use are covered by this License.
- b. Accompany the object code with a copy of the GNU GPL and this license document.
- 4. Combined Works.

You may convey a Combined Work under terms of your choice that, taken together, effectively do not restrict modification of the portions of the Library contained in the Combined Work and reverse engineering for debugging such modifications, if you also do each of the following:

- a. Give prominent notice with each copy of the Combined Work that the Library is used in it and that the Library and its use are covered by this License.
- b. Accompany the Combined Work with a copy of the GNU GPL and this license document.
- c. For a Combined Work that displays copyright notices during execution, include the copyright notice for the Library among these notices, as well as a reference directing the user to the copies of the GNU GPL and this license document.
- d. Do one of the following:
 - On Convey the Minimal Corresponding Source under the terms of this License, and the Corresponding Application Code in a form suitable for, and under terms that permit, the user to recombine or relink the Application with a modified version of the Linked Version to produce a modified Combined Work, in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.
 - 1. Use a suitable shared library mechanism for linking with the Library. A suitable mechanism is one that (a) uses at run time a copy of the Library already present on the user's computer system, and (b) will operate properly with a modified version of the Library that is interface-compatible with the Linked Version.
- e. Provide Installation Information, but only if you would otherwise be required to provide such information under section 6 of the GNU GPL, and only to the extent that such information is necessary to install and execute a modified version of the Combined Work produced by recombining or relinking the Application with a modified version of the Linked Version. (If you use option 4d0, the Installation Information must accompany the Minimal Corresponding Source and Corresponding Application Code. If you use option 4d1, you must provide the Installation Information in the manner specified by section 6 of the GNU GPL for conveying Corresponding Source.)

5. Combined Libraries.

You may place library facilities that are a work based on the Library side by side in a single library together with other library facilities that are not Applications and are not covered by this License, and convey such a combined library under terms of your choice, if you do both of the following:

- a. Accompany the combined library with a copy of the same work based on the Library, uncombined with any other library facilities, conveyed under the terms of this License.
- b. Give prominent notice with the combined library that part of it is a work based on the Library, and explaining where to find the accompanying uncombined form of the same work.
- 6. Revised Versions of the GNU Lesser General Public License.

The Free Software Foundation may publish revised and/or new versions of the GNU Lesser General Public License from time to time. Such new versions will be similar in spirit to the present version, but may differ in detail to address new problems or concerns.

Each version is given a distinguishing version number. If the Library as you received it specifies that a certain numbered version of the GNU Lesser General Public License "or any later version" applies to it, you have the option of following the terms and conditions either of that published version or of any later version published by the Free Software Foundation. If the Library as you received it does not specify a version number of the GNU Lesser General Public License, you may choose any version of the GNU Lesser General Public License ever published by the Free Software Foundation.

If the Library as you received it specifies that a proxy can decide whether future versions of the GNU Lesser General Public License shall apply, that proxy's public statement of acceptance of any version is permanent authorization for you to choose that version for the Library.

Licensed Code in Qt

http://doc.qt.io/qt-5/licenses-used-in-qt.html

Qt Core

Software and Link	License
<u>Data Compression Library (zlib)</u>	ZLib license
Easing Equations by Robert Penner	BSD 3-clause "New" or "Revised" License
Efficient Binary-Decimal and Decimal-Binary Conversion Routines	BSD 3-clause "New" or "Revised" License
for IEEE Doubles	
FreeBSD strtoll and strtoull	BSD 3-clause "New" or "Revised" License
MD4	Public Domain
<u>MD5</u>	Public Domain
PCRE2	BSD 3-clause "New" or "Revised" License
QEventDispatcher on macOS	BSD 3-clause "New" or "Revised" License
Secure Hash Algorithm SHA-1	Public Domain
Secure Hash Algorithm SHA-3 - Keccak	Creative Commons Zero v1.0 Universal
Secure Hash Algorithm SHA-3 - brg_endian	BSD 3-clause "New" or "Revised" License
Secure Hash Algorithms SHA-384 and SHA-512	BSD 3-clause "New" or "Revised" License
Text Codec: EUC-JP	BSD 2-clause "Simplified" License
Text Codec: EUC-KR	BSD 2-clause "Simplified" License
<u>Text Codec: GBK</u>	BSD 2-clause "Simplified" License
Text Codec: ISO 2022-JP (JIS)	BSD 2-clause "Simplified" License
Text Codec: Shift-JIS	BSD 2-clause "Simplified" License
Text Codec: TSCII	BSD 2-clause "Simplified" License
Text Codecs: Big5, Big5-HKSCS	BSD 2-clause "Simplified" License
The Public Suffix List	Mozilla Public License 2.0
Unicode Character Database (UCD)	Unicode License Agreement - Data Files and Software (2016)
Unicode Common Locale Data Repository (CLDR)	Unicode License Agreement - Data Files and Software (2016)
<u>forkfd</u>	MIT License

Qt GUI

Software and Link	License
ANGLE Library	BSD 3-clause "New" or "Revised" License
ANGLE: Array Bounds Clamper for WebKit	BSD 2-clause "Simplified" License
ANGLE: Murmurhash	Public Domain
ANGLE: Systeminfo	BSD 2-clause "Simplified" License
ANGLE: trace_event	BSD 3-clause "New" or "Revised" License
Anti-aliasing rasterizer from FreeType 2	Freetype Project License or GNU General Public License v2.0 only
Cocoa Platform Plugin	BSD 3-clause "New" or "Revised" License
Freetype 2	Freetype Project License or GNU General Public License v2.0 only
Freetype 2 - Bitmap Distribution Format (BDF)	MIT License

CII	n	n	\sim	rt
נוכ	IJ	u	u	,,,

Freetype 2 - Portable Compiled Format (PCF) support	MIT License
Freetype 2 - zlib	Zlib License
<u>HarfBuzz</u>	MIT License
<u>HarfBuzz-NG</u>	MIT License
IAccessible2 IDL Specification	BSD 3-clause "New" or "Revised" License
<u>LibJPEG-turbo</u>	Independent JPEG Group License
<u>LibPNG</u>	Libpng License
Native Style for Android	Apache License 2.0
OpenGL ES 2 Headers	MIT License
OpenGL Headers	MIT License
<u>Pixman</u>	MIT License
Smooth Scaling Algorithm	BSD 2-clause "Simplified" License and Imlib2 License
<u>Vulkan API Registry</u>	MIT License
Wintab API	Public Domain
<u>XCB</u>	MIT License
sRGB color profile icc file	International Color Consortium License
xkbcommon	MIT Licenses (with no-advertisement clause)

Qt Location

Software and Link	License
<u>Boost</u>	Boost Software License 1.0
CSS Color Parser	MIT License
Clip2Tri Polygon Triangulation Library	MIT License
Clipper Polygon Clipping Library	Boost Software License 1.0
<u>Earcut</u>	ISC License
Earcut Polygon Triangulation Library	ISC License
Mapbox GL Native	BSD 2-clause "Simplified" License
Poly2Tri Polygon Triangulation Library	BSD 3-clause "New" or "Revised" License
<u>RapidJSON</u>	MIT License
<u>Vector Tile Library</u>	ISC License
Wagyu Geometry Processing Library	MIT License
cURL Parse Date	MIT License
geojson-cpp	ISC License
geojson-vt-cpp	ISC License
geometry.hpp	ISC License
kdbush.hpp	ISC License
<u>libc++</u>	MIT and UIUC (BSD-like)
<u>polylabel</u>	ISC License
<u>protozero</u>	BSD 2-clause "Simplified" License
shelf-pack-cpp	ISC License
supercluster.hpp	ISC License
tao_tuple	MIT License
unique_resource	Boost Software License 1.0
<u>variant</u>	BSD 3-clause "New" or "Revised" License

Qt SVG

Software and Link License

XSVG MIT "Old Style" License

Qt XML Patterns

Software and Link License

XML Schema W3C Software Notice and Document License (2015-05-13)

The MIT License (MIT)

Copyright (c) 2014-2021 Anthon van der Neut, Ruamel byba

Permission is hereby granted, free of charge, to any person obtaining a copy of this software and associated documentation files (the "Software"), to deal in the Software without restriction, including without limitation the rights to use, copy, modify, merge, publish, distribute, sublicense, and/or sell copies of the Software, and to permit persons to whom the Software is furnished to do so, subject to the following conditions:

The above copyright notice and this permission notice shall be included in all copies or substantial portions of the Software.

THE SOFTWARE IS PROVIDED "AS IS", WITHOUT WARRANTY OF ANY KIND, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT. IN NO EVENT SHALL THE AUTHORS OR COPYRIGHT HOLDERS BE LIABLE FOR ANY CLAIM, DAMAGES OR OTHER LIABILITY, WHETHER IN AN ACTION OF CONTRACT, TORT OR OTHERWISE, ARISING FROM, OUT OF OR IN CONNECTION WITH THE SOFTWARE OR THE USE OR OTHER DEALINGS IN THE SOFTWARE.

Creative Commons Legal Code

CCO 1.0 Universal

CREATIVE COMMONS CORPORATION IS NOT A LAW FIRM AND DOES NOT PROVIDE LEGAL SERVICES. DISTRIBUTION OF THIS DOCUMENT DOES NOT CREATE AN ATTORNEY-CLIENT RELATIONSHIP. CREATIVE COMMONS PROVIDES THIS INFORMATION ON AN "AS-IS" BASIS. CREATIVE COMMONS MAKES NO WARRANTIES REGARDING THE USE OF THIS DOCUMENT OR THE INFORMATION OR WORKS PROVIDED HEREUNDER, AND DISCLAIMS LIABILITY FOR DAMAGES RESULTING FROM THE USE OF THIS DOCUMENT OR THE INFORMATION OR WORKS PROVIDED HEREUNDER.

Statement of Purpose

The laws of most jurisdictions throughout the world automatically confer exclusive Copyright and Related Rights (defined below) upon the creator and subsequent owner(s) (each and all, an "owner") of an original work of authorship and/or a database (each, a "Work").

Certain owners wish to permanently relinquish those rights to a Work for the purpose of contributing to a commons of creative, cultural and scientific works ("Commons") that the public can reliably and without fear of later claims of infringement build upon, modify, incorporate in other works, reuse and redistribute as freely as possible in any form whatsoever and for any purposes, including without limitation commercial purposes. These owners may contribute to the Commons to promote the ideal of a free culture and the further production of creative, cultural and scientific works, or to gain reputation or greater distribution for their Work in part through the use and efforts of others.

For these and/or other purposes and motivations, and without any expectation of additional consideration or compensation, the person associating CCO with a Work (the "Affirmer"), to the extent that he or she is an owner of Copyright and Related Rights in the Work, voluntarily elects to apply CCO to the Work and publicly distribute the Work under its terms, with knowledge of his or her Copyright and Related Rights in the Work and the meaning and intended legal effect of CCO on those rights.

- 1. Copyright and Related Rights. A Work made available under CCO may be protected by copyright and related or neighboring rights ("Copyright and Related Rights"). Copyright and Related Rights include, but are not limited to, the following:
 - i. the right to reproduce, adapt, distribute, perform, display, communicate, and translate a Work;
 - ii. moral rights retained by the original author(s) and/or performer(s);
- iii. publicity and privacy rights pertaining to a person's image or likeness depicted in a Work;
 - iv. rights protecting against unfair competition in regards to a Work, subject to the limitations in paragraph 4(a), below;
 - v. rights protecting the extraction, dissemination, use and reuse of data in a Work;
 - vi. database rights (such as those arising under Directive 96/9/EC of the European Parliament and of the Council of 11 March 1996 on the legal protection of databases, and under any national implementation thereof, including any amended or successor version of such directive); and
- vii. other similar, equivalent or corresponding rights throughout the world based on applicable law or treaty, and any national implementations thereof.
- 2. Waiver. To the greatest extent permitted by, but not in contravention of, applicable law, Affirmer hereby overtly, fully, permanently,

irrevocably and unconditionally waives, abandons, and surrenders all of Affirmer's Copyright and Related Rights and associated claims and causes of action, whether now known or unknown (including existing as well as future claims and causes of action), in the Work (i) in all territories worldwide, (ii) for the maximum duration provided by applicable law or treaty (including future time extensions), (iii) in any current or future medium and for any number of copies, and (iv) for any purpose whatsoever, including without limitation commercial, advertising or promotional purposes (the "Waiver"). Affirmer makes the Waiver for the benefit of each member of the public at large and to the detriment of Affirmer's heirs and successors, fully intending that such Waiver shall not be subject to revocation, rescission, cancellation, termination, or any other legal or equitable action to disrupt the quiet enjoyment of the Work by the public as contemplated by Affirmer's express Statement of Purpose.

3. Public License Fallback. Should any part of the Waiver for any reason be judged legally invalid or ineffective under applicable law, then the Waiver shall be preserved to the maximum extent permitted taking into account Affirmer's express Statement of Purpose. In addition, to the extent the Waiver is so judged Affirmer hereby grants to each affected person a royalty-free, non transferable, non sublicensable, non exclusive, irrevocable and unconditional license to exercise Affirmer's Copyright and Related Rights in the Work (i) in all territories worldwide, (ii) for the maximum duration provided by applicable law or treaty (including future time extensions), (iii) in any current or future medium and for any number of copies, and (iv) for any purpose whatsoever, including without limitation commercial, advertising or promotional purposes (the "License"). The License shall be deemed effective as of the date CCO was applied by Affirmer to the Work. Should any part of the License for any reason be judged legally invalid or ineffective under applicable law, such partial invalidity or ineffectiveness shall not invalidate the remainder of the License, and in such case Affirmer hereby affirms that he or she will not (i) exercise any of his or her remaining Copyright and Related Rights in the Work or (ii) assert any associated claims and causes of action with respect to the Work, in either case contrary to Affirmer's express Statement of Purpose.

4. Limitations and Disclaimers.

- a. No trademark or patent rights held by Affirmer are waived, abandoned, surrendered, licensed or otherwise affected by this document.
- b. Affirmer offers the Work as-is and makes no representations or warranties of any kind concerning the Work, express, implied, statutory or otherwise, including without limitation warranties of title, merchantability, fitness for a particular purpose, non infringement, or the absence of latent or other defects, accuracy, or the present or absence of errors, whether or not discoverable, all to the greatest extent permissible under applicable law.
- c. Affirmer disclaims responsibility for clearing rights of other persons that may apply to the Work or any use thereof, including without limitation any person's Copyright and Related Rights in the Work. Further, Affirmer disclaims responsibility for obtaining any necessary consents, permissions or other rights required for any use of the Work.
- d. Affirmer understands and acknowledges that Creative Commons is not a party to this document and has no duty or obligation with respect to this CCO or use of the Work.

ZLIB DATA COMPRESSION LIBRARY

zlib 1.2.11 is a general purpose data compression library. All the code is thread safe. The data format used by the zlib library is described by RFCs (Request for Comments) 1950 to 1952 in the files http://tools.ietf.org/html/rfc1950 (zlib format), rfc1951 (deflate format) and rfc1952 (gzip format).

All functions of the compression library are documented in the file zlib.h (volunteer to write man pages welcome, contact zlib@gzip.org). A usage example of the library is given in the file test/example.c which also tests that the library is working correctly. Another example is given in the file test/minigzip.c. The compression library itself is composed of all source files in the root directory.

To compile all files and run the test program, follow the instructions given at the top of Makefile.in. In short "./configure; make test", and if that goes well, "make install" should work for most flavors of Unix. For Windows, use one of the special makefiles in win32/ or contrib/vstudio/ . For VMS, use make_vms.com.

Questions about zlib should be sent to <zlib@gzip.org>, or to Gilles Vollant <info@winimage.com> for the Windows DLL version. The zlib home page is http://zlib.net/ . Before reporting a problem, please check this site to verify that you have the latest version of zlib; otherwise get the latest version and check whether the problem still exists or not.

PLEASE read the zlib FAQ http://zlib.net/zlib_faq.html before asking for help.

Mark Nelson <markn@ieee.org> wrote an article about zlib for the Jan. 1997 issue of Dr. Dobb's Journal; a copy of the article is available at http://marknelson.us/1997/01/01/zlib-engine/.

The changes made in version 1.2.11 are documented in the file ChangeLog.

Unsupported third party contributions are provided in directory contrib/ .

zlib is available in Java using the java.util.zip package, documented at http://java.sun.com/developer/technicalArticles/Programming/compression/ .

A Perl interface to zlib written by Paul Marquess <pmqs@cpan.org> is available at CPAN (Comprehensive Perl Archive Network) sites, including http://search.cpan.org/~pmqs/IO-Compress-Zlib/ .

A Python interface to zlib written by A.M. Kuchling <amk@amk.ca> is available in Python 1.5 and later versions, see http://docs.python.org/library/zlib.html .

zlib is built into tcl: http://wiki.tcl.tk/4610 .

An experimental package to read and write files in .zip format, written on top of zlib by Gilles Vollant <info@winimage.com>, is available in the contrib/minizip directory of zlib.

Notes for some targets:

- For Windows DLL versions, please see win32/DLL_FAQ.txt
- For 64-bit Irix, deflate.c must be compiled without any optimization. With -0, one libpng test fails. The test works in 32 bit mode (with the -n32 compiler flag). The compiler bug has been reported to SGI.
- zlib doesn't work with gcc 2.6.3 on a DEC 3000/300LX under OSF/1 2.1 it works

when compiled with cc.

- On Digital Unix 4.0D (formely OSF/1) on AlphaServer, the cc option -std1 is necessary to get gzprintf working correctly. This is done by configure.
- zlib doesn't work on HP-UX 9.05 with some versions of /bin/cc. It works with other compilers. Use "make test" to check your compiler.
- gzdopen is not supported on RISCOS or BEOS.
- For PalmOs, see http://palmzlib.sourceforge.net/

Acknowledgments:

The deflate format used by zlib was defined by Phil Katz. The deflate and zlib specifications were written by L. Peter Deutsch. Thanks to all the people who reported problems and suggested various improvements in zlib; they are too numerous to cite here.

Copyright notice:

(C) 1995-2017 Jean-loup Gailly and Mark Adler

This software is provided 'as-is', without any express or implied warranty. In no event will the authors be held liable for any damages arising from the use of this software.

Permission is granted to anyone to use this software for any purpose, including commercial applications, and to alter it and redistribute it freely, subject to the following restrictions:

- 1. The origin of this software must not be misrepresented; you must not claim that you wrote the original software. If you use this software in a product, an acknowledgment in the product documentation would be appreciated but is not required.
- 2. Altered source versions must be plainly marked as such, and must not be misrepresented as being the original software.
- 3. This notice may not be removed or altered from any source distribution.

Jean-loup Gailly Mark Adler jloup@gzip.org madler@alumni.caltech.edu

If you use the zlib library in a product, we would appreciate *not* receiving lengthy legal documents to sign. The sources are provided for free but without warranty of any kind. The library has been entirely written by Jean-loup Gailly and Mark Adler; it does not include third-party code.

If you redistribute modified sources, we would appreciate that you include in the file ChangeLog history information documenting your changes. Please read the FAQ for more information on the distribution of modified source versions. zlib.h -- interface of the 'zlib' general purpose compression library version 1.3.1, January 22nd, 2024

Copyright (C) 1995-2024 Jean-loup Gailly and Mark Adler

This software is provided 'as-is', without any express or implied warranty. In no event will the authors be held liable for any damages arising from the use of this software.

Permission is granted to anyone to use this software for any purpose, including commercial applications, and to alter it and redistribute it freely, subject to the following restrictions:

- 1. The origin of this software must not be misrepresented; you must not claim that you wrote the original software. If you use this software in a product, an acknowledgment in the product documentation would be appreciated but is not required.
- 2. Altered source versions must be plainly marked as such, and must not be misrepresented as being the original software.
- 3. This notice may not be removed or altered from any source distribution.

Jean-loup Gailly Mark Adler jloup@gzip.org madler@alumni.caltech.edu